Exploring the interconnections between climate change-gender in adaptation and mitigation contexts

Dissertation

zur Erlangung des Grades Doktorin der Agrarwissenschaften (Dr. agr.) der Agrar-, Ernährungs- und Ingenieurwissenschaftlichen Fakultät der Rheinischen Friedrich-Wilhelms-Universität Bonn

von

Marina Kovaleva

aus Bischkek, Kirgisistan

Bonn 2025

Referent: Prof. Dr. (mult.) Dr. h.c. (mult.) Walter Leal

Korreferent: Prof. Dr. Christian Borgemeister

Tag der mündlichen Prüfung: 28.05.2024

Angefertigt mit Genehmigung der Agrar-, Ernährungs- und Ingenieurwissenschaftlichen Fakultät der Universität Bonn

Abstract

Global human-driven climate change and its consequences are continuously transforming natural and socio-economic systems by altering the environment and posing threats to people's health, well-being and livelihood stability. The population's characteristics including geographical distribution, education, poverty and gender predefine to varying degrees individuals' capacity and ability to overcome climate change challenges. In the light of the recent developments at the international arena, gender has been mentioned more often in the climate change-related context, compared to other characteristics. Interconnections between climate change and gender are characterized by complex synergies and dynamics, continuous development, and ongoing mutual impacts. The integration of these interconnections is among the factors that are expected to influence the development and implementation of effective global-scale adaptation and mitigation solutions.

This PhD work explores some key aspects of the climate change-gender interconnections and sheds light on their interactions in a variety of contexts in different geographical regions including Central Asia (CA), Africa, and the Pacific. The work is structured around three main domains of climate change-gender

interconnections: vulnerability, leadership/empowerment, and benefits, which are subject to thematic and geographical contexts, as well as level of application: regional, national or local.

Chapter 1 provides an introduction to the topic including the objectives and scope of this work and discusses gaps in the climate change-gender interconnections landscape. Chapter 2 explores the overall developments of inclusion of gender/women's issues in scientific contributions in climate change research. It also outlines the main themes that can be referred to the vulnerability, leadership/empowerment, and/or benefits domain(s) of climate change-gender interconnections. Chapter 3 examines the leadership/empowerment domain by analysing advances made towards attaining gender-balanced leadership/empowerment in climate change adaptation and mitigation in 12 African countries. Chapter 4 investigates the progress in addressing gender/women considerations by climate change-related areas in CA countries. Chapter 5 discusses climate change-gender interconnections by reviewing a feminisation process, its drivers, positive and negative outcomes, and associated challenges within the broader agricultural context in different geographical regions including Latin America and CA. Chapters 6, 7 and 8 explore the domains of climate change-gender interconnections, in the context of additional issue areas such as the partly successful implementation of the Sustainable Development Goals (SDGs), extreme events and mental health, and climate change-induced migration.

The findings obtained during this PhD study reflect a new perspective under a unified umbrella of climate change-gender interconnections framework. The work provides innovative insights on the progress made towards taking into consideration men's and women's differences in handling climate change issues, and the potential added value that an improved understanding of these interconnections may bring from a climate change perspective, at meta level.

Zusammenfassung

Der globale, vom Menschen verursachte Klimawandel und seine Folgen verändern kontinuierlich die natürlichen und sozioökonomischen Systeme, indem sie die Umwelt verändern und die Gesundheit, das Wohlergehen und die Stabilität der Lebensgrundlage der Menschen gefährden. Die Merkmale der Bevölkerung wie geografische Verteilung, Bildung, Armut und Geschlecht bestimmen in unterschiedlichem Maße die Fähigkeit der Menschen, die Herausforderungen des Klimawandels zu bewältigen. Angesichts der jüngsten Entwicklungen auf internationaler Ebene wird ,Geschlecht' im Zusammenhang mit dem Klimawandel im Vergleich zu anderen Merkmalen häufiger erwähnt. Die Verflechtungen zwischen Klimawandel und ,Geschlecht' sind durch komplexe Synergien und Dynamiken, eine kontinuierliche Entwicklung und anhaltende gegenseitige Auswirkungen gekennzeichnet. Die Integration dieser Verflechtungen gehört zu den Faktoren, von denen erwartet wird, dass sie die Entwicklung und Umsetzung wirksamer globaler Lösungen zur Anpassung an und Eindämmung des Klimawandels beeinflussen werden.

Diese Studie untersucht einige Schlüsselaspekte der Klimawandel-Geschlecht Verflechtungen und beleuchtet deren Wechselwirkungen in einer Vielzahl von Kontexten in unterschiedlichen geografischen Regionen, einschließlich Zentralasien, Afrika und den Pazifikraum. Die Forschungsarbeit wurde um drei zentrale Themenbereiche dieser Verflechtungen herum strukturiert: i) Anfälligkeit, ii) Führung/Ermächtigung und iii) Vorteile, die dem thematischen und geografischen Kontext, sowie der Anwendungsebene: regional, national oder lokal unterliegen.

Eine Einführung in das Thema, einschließlich der Ziele und des Umfangs dieser Arbeit, sowie der Forschungstand im Themenbereich Klimawandel und Geschlecht, wird in Kapitel 1 gegeben. In Kapitel 2 werden die allgemeinen Entwicklungen bei der Einbeziehung der Geschlechter/Frauenaspekten in wissenschaftliche Beiträge zur Klimawandelforschung untersucht. Außerdem werden die Hauptthemen skizziert, die sich auf Anfälligkeit, Führung/Ermächtigung und/oder Vorteile Bereiche der Klimawandel-Geschlecht Verflechtungen beziehen lassen. Kapitel 3 analysiert die Fortschritte, die auf dem Weg zu einer geschlechtergerechten Führung/ Ermächtigung bei der Anpassung an den Klimawandel und seiner Eindämmung in 12 afrikanischen Ländern erzielt wurden. In Kapitel 4 wird untersucht, wie Genderbzw. Frauenaspekte in den mit dem Klimawandel verbundenen Bereichen in den Ländern Zentralasiens bereits berücksichtigt wurden. In Kapitel 5 werden die Klimawandel-Geschlecht Verflechtungen erörtert. Es untersucht ein Feminisierungsprozess, seine Triebkräfte, positive und negative Ergebnisse und die damit verbundenen Herausforderungen im breiteren landwirtschaftlichen Kontext die in verschiedenen geografischen Regionen, einschließlich Zentralasien und Lateinamerika, untersucht werden. In den Kapiteln 6, 7 und 8 werden verschiedene Thementeilbereiche Klimawandel-Geschlecht weitere der Verflechtungen untersucht, und zwar im Zusammenhang mit der teilweise erfolgreichen Umsetzung der Ziele für nachhaltige Entwicklung (SDGs), mit extremen Wetterereignissen und psychischer Gesundheit sowie mit klimawandelbedingter Migration.

Die im Rahmen dieser Doktorandenstudie erzielten Ergebnisse reflektieren eine neue Perspektive eines einheitlichen Rahmens für die Klimawandel-Geschlecht Verflechtungen. Die Arbeit bietet innovative Einblicke in die Fortschritte bei der Berücksichtigung der Unterschiede zwischen Männern und Frauen im Umgang mit Fragen des Klimawandels. Es zeigt auch den potenziellen Mehrwert auf, den ein besseres Verständnis dieser Zusammenhänge aus der Perspektive des Klimawandels auf Metaebene bringen kann.

Table of Contents¹

2. Gender issues within climate change research: a bibliometric analysis
2.2 Introduction
2.3 Methods
2.3.1 Data selection
2.3.2 Data analysis
2.4 Results and discussion
2.4.1 Total published works
2.4.1.1 Literature retrieved from WoS
2.4.1.2 Literature retrieved from Google Scholar
2.4.2 Co-occurrence analysis
2.4.3 Bibliographic coupling analysis
2.4.4 Co-citation analysis
2.4.5 Authors
2.4.6 Co-authorship analysis
2.4.6.1 Authors
2.4.6.2 Countries
2.4.6.3 Institutions
2.4.6.4 Contributions of international organizations
2.4 Limitations
2.5 Conclusions
5. Feminization of Agriculture and Its Impact on Sustainable Development
5.2 Introduction
5.3 Drivers of the Phenomenon
5.3.1 Male Outmigration from Rural Areas
5.3.2 Economic Crisis and Climate Change
5.3.3 Diseases and Armed Conflicts
5.3.4 Opportunities in the Agri-business Industry
5.3.5 Different Methods of Statistical Data Collection
5.4 Consequences of the Agricultural Feminization
5.4.1 Women's Empowerment
5.4.2 Exacerbation of Gender Gaps in Wealth and Work Burden
5.4.3 Low Productivity and Access to Land

 $^{^{\}rm 1}$ The following chapters and subchapters are not available for republication due to closed access in the published journals

5.4.4 Impact on Sustainable Development
5.5 Latin America
5.6 Central Asia
5.7 Conclusions

8. Assessing causes and implications of climate-induced migration in Kenya and Ethiopia	
8.2 Introduction: climate change in Eastern Africa	•••
8.3 Climate change and migration in Kenya and Ethiopia	
8.4 Methods	••••
8.5 Findings from the analysis and discussion	•••
8.5.1 Descriptive statistics of the survey respondents and responses	
8.6 Conclusions	•••

Table of Contents

Abstract
Zusammenfassung
List of Tables
List of Figures
List of Abbreviations
1. Introduction
1.1 Background
1.1.1Climate change impacts
1.1.2Responding to climate change17
1.1.3 Climate change and sustainable development interconnection
1.2 Filing the research gap: Climate change and gender landscape
1.3 Scope of the dissertation
1.3.1 Aim and objectives
1.3.2 Structure of the dissertation
2. Gender issues within climate change research: a bibliometric analysis
2.1 Abstract
3. Understanding Needs and Potentials for Gender-Balanced Empowerment and Leadership in Climate Change Adaptation and Mitigation in Africa 24
31 Abstract 24
3.2 Introduction: Climate Change–Gender Interconnections 24
3.3 Climate Change-Gender Interconnections in Leadership/Empowerment
3.4 Gender Leadership/Empowerment in Climate Change in the African Context 32
3.5 Materials and Methods
3.5.1 Bibliometric Analysis
3.5.2 Online Questionnaire
3.6 Results and Discussion
3.6.1 Bibliometric Analysis
3.6.2 Online Questionnaire
3.6.3 Final Remarks
3.7 Conclusions
4. Central Asia: Exploring Insights on Gender Considerations in Climate Change 52
4.1 Abstract
4.2 Introduction: Gender in the Climate Change Context
4.3 Central Asia: Climate Change, Gender/Women, and International Donors54

	4.3.1 Climate change	54
	4.3.2 Gender/Women	55
	4.3.3 Donors	57
	4.4 Methods	58
	4.4.1 Bibliometric Analysis	58
	4.4.2 Expert-Driven Assessment Approach	59
	4.4.3 Development Projects Analysis	60
,	4.5 Results	60
	4.5.1 Bibliometric Analysis	60
	4.5.2 On-Line Survey	61
	4.5.2 Development Projects Analysis	65
,	4.6 Discussion	66
	4.6.1 Bibliometric Analysis	66
	4.6.2 On-Line Survey	67
	4.6.3 Development Projects Analysis	68
	4.6.4 Implications and Recommendations	68
	4.7 Conclusions	69
5.	Feminization of Agriculture and Its Impact on Sustainable Development	71
	5.1 Abstract	71
:	5.2 Introduction	71
6.	Promoting gender equality across the sustainable development goals	72
	6.1 Abstract	72
	6.2 Introducing SDG5–gender equality	72
	6.3 Research on gender equality and the SDGs: assessing the relations	74
	6.4 Methodology	79
	6.5 Results and discussion	80
	6.5.1 Gender equality and the SDGs	80
	6.5.2 Case Studies	81
	6.5.3 The proposed framework for assessing gender equality impact across the S	DGs
		82
	6.6 Conclusions	85
7.	Climate change, extreme events and mental health in the Pacific region	88
,	7.1 Abstract	88
,	7.2 Introduction	88
	7.2.1 Introducing climate change, extreme weather events and health	88
	7.2.2 Impacts of climate change and extreme events in the Pacific Island countrie	es90

7.3 Methods	94
7.4 Results and discussion	96
7.4.1 Bibliometric Analysis	96
7.4.2 Case studies: reviewing the impacts of climate change and extreme weather a climate events on the life-support systems, livelihoods, well-being and mental heal in the Pacific Islands Countries	and .th 97
7.4.3 Overall Discussion	99
7.5 Conclusions	102
8. Assessing causes and implications of climate-induced migration in Kenya and	
Ethiopia	103
8.1 Abstract	103
9. Synthesis of results	104
10. Outlook for future research	107
11. References	108
12. Appendices	170
Appendix I	170
Appendix II	171
Appendix III	181
Appendix IV	184
Appendix V	202
Acknowledgments	203

The shorter version of the dissertation has been approved for publication by the dissertation supervisors:

Prof. Dr. (mult.) Dr. h.c. (mult.) Walter Leal

Prof. Dr. Christian Borgemeister

List of Tables

Table 3.1 Climate change-gender interconnections. Source: author	.27
Table 3.2 Selection criteria. Source: author	.34
Table 3.3 Barriers to women's representation in climate-related leadership	
positions. Source: author	.44
Table 3.4 Outcomes of the change in women's and men's roles in climate action.	
Source: author	.46
Table 4.1 Central Asia countries' characteristics. Source: Authors compilation	
based on data from the World Bank, UNDP, national official websites, and	
statistical offices.	.55
Table 4.2 Total net ODA disbursements received by the CA countries over the	
period 2016–2020, USD million. Source: The authors' calculations based on the	
Query Wizard for International Development Statistics (QWIDS)	.57
Table 4.3 Number and size of the ADB and World Bank commitments by country	y;
USD billions. Source: The authors, based on data from the Asian Development	-
Bank and the World Bank 2023	.58
Table 4.4 Selection criteria. Source: author	.59
Table 4.5 Socio-demographic characteristics of the on-line survey respondents.	
Source: author	.62
Table 6.1 SDG5 Targets and Indicators from United Nations (2021). Source: tear	n
of authors	.73
Table 7.1 Examples of extreme weather events in the Pacific islands (2003-2020)
Source: team of authors supported by the literature	.93
Table 12.1. List of Abbreviations of the UNFCCC bodies (based on Gender	
Climate Tracker). Source: author1	70
Table 12.2 Main causes of rural men's and women's vulnerability to climate	
change. Source: Authors' compilation from the survey responses1	71
Table 12.3 Development projects supported by international organizations/donors	s.
Source: author	74
Table 12.4 International case studies illustrating worldwide initiatives that address	SS
gender equality among the SDGs. Source: team of authors1	81
Table 12.5 SDG Matrix. Source: team of authors1	84

List of Figures

Figure 1.1 Domains of climate change-gender interconnections framework at meta Figure 3.1 Female/male ratio of the number of the representatives on UNFCCC boards and bodies between the years 2013 and 2020, except TEC; the data have been available since 2017, FWG and KCI—since 2019, PAICC—since 2020 (based on Gender Climate Tracker); full body names are listed in Appendix I. Figure 3.2 Percentage of Female Chair, Co-Chair or Vice Chair out of the total number of female representatives on UNFCCC boards/bodies between the years 2013 and 2020, except TEC; the data have been available since 2017, FWG and KCI-since 2019, PAICC-since 2020 (based on Gender Climate Tracker); full Figure 3.3 (a) Co-occurrence map of keywords (Scopus/WoS dataset); (b) Cooccurrence map of keywords (GS dataset); (c) Co-occurrence map of terms (Scopus/WoS dataset); (d) Co-occurrence map of terms (GS dataset). Created with Figure 3.4 Countries of the respondents participated in the online questionnaire Figure 3.6 Characteristics of women's leadership in climate change. Source: author. Figure 3.7 Change in women's and men's roles in climate action. Source: author, 42 Figure 3.8 Areas where women's empowerment has most increased. Source: Figure 3.9 The extent of women's engagement in climate-change-related processes. Figure 3.10 Progress in women's influence on climate-change-related decisions and improvements in their ability to overcome the impacts of climate change. Source: Figure 3.11 Efforts to advance gender-equal participation and representation in Figure 3.12 Time needed to achieve gender-equal participation and representation in leadership positions in climate action at the national level. Source: author.......48 Figure 4.1 Co-occurrence map of terms (merged Scopus/GS dataset). Created with Figure 4.2 Main impacts of climate change on the rural population. Source: author. Figure 4.3 How men and women are affected in the key sectors. Source: author...63 Figure 4.5 Importance of including more rural women in decision-making Figure 4.6 Rural women's input in decision-making processes at the town/village Figure 4.7 Change in women's role in climate change adaptation and mitigation

Figure 6.1 Gender Inequality Index, by developing region, 2020, modified from
UNDP (2020) Source: team of authors
Figure 6.2 Life-course Gender Gap, 2019, modified from UNDP (2020). Source:
team of authors77
Figure 6.3 Proposed framework for considering gender impact across all the SDGs.
Source: team of authors
Figure 6.4 Percentage of gender inequality impact on SDG goals, according to the
authors' proposed methodology. Source: team of authors
Figure 7.1 Some of the impacts of climate change on mental health. Source: team
of authors
Figure 7.2 A conceptualisation of the pathways by which climate change will affect
mental health in the Pacific Islands Countries (PICs) and the significant anticipated
impacts throughout the region. Source: team of authors
Figure 7.3 Co-occurrence map of mental health and climate change-related terms.
Source: team of authors

List of Abbreviations

AAP	Africa Adaptation Programme
AC	Adaptation Committee
ADB	Asian Development Bank
ASAL	Arid and Semi-Arid Land
AWGGCC	African Working Group on Gender and Climate Change
CA	Central Asia
CAP	Central Asia Program
CAREC	The Regional Environmental Centre for Central Asia
CBD	Convention on Biological Diversity
CC	Compliance Committee
CDM	Clean Development Mechanism
	Central and Eastern Europe and the Commonwealth of
CEE/CIS	Independent States
CEDAW	Convention on the Elimination of All Forms of Discrimination
CLDIIII	against Women
CEO	Chief Executive Officer
CGE	Consultative Group of Experts
COP	Conference of the Parties
COVID-19	Coronavirus disease 2019
CRC-PEG	Courant Research Centre - Poverty Equity and Growth
CRM/TASP	Climate Risk Management Technical Assistance Project
CTCN	Climate Technology Centre and Network
DRR	Disaster Risk Reduction
DIAUD	Disability Inclusive and Accessible Urban Development
FCLAC	Economic Commission for Latin America and the Caribbean
EEA	European Environment Agency
ESCAP	Economic and Social Commission for Asia and the Pacific
FSD	Education for Sustainable Development
FU	Furopean Union
EWEs	Extreme Weather and Climate Events
FAO	Food and Agriculture Organisation
FSM	Federated States of Micronesia
FWG	Facilitative Working Group
GAP	Gender Action Plan
GCF	Green Climate Fund
GDP	Gross Domestic Product
GHG	Greenhouse gas
GS	Google Scholar
	International Assessment of Agricultural Knowledge Science
IAASID	and Tachpology for Davalopment
IDM	Individual Deprivation Measure
IDMC	International Displacement Monitoring Contra
IEDDI	International Displacement Monitoring Centre
	International Institute for Systematic Development
	International I abour Organisation
IOM	International Crassization for Migration
	International Organization for Migration
	Intergovernmental Faller on Childle Challge
JISC IDC	Joint Implementation Supervisory Committee
JAC	Joint Research Centre Vatomica Committae of Exports on the Imports of the
NU	Implementation of Despanse Measure
LAC	Implementation of Kesponse Measure
LAU	Laun America and Caribbean

LEG	Least Developed Countries Expert Group	
MDD	Major Depressive Disorder	
NAP	National Adaptation Plan	
NAPA	National Adaptation Programme of Action	
NDC	Nationally Determined Contribution	
ND-GAIN	Notre Dame Global Adaptation Initiative	
NDMO	National Disaster Management Office	
NWO	Dutch Research Council	
OCHA	Office for the Coordination of Humanitarian Affairs	
ODA	Official Development Assistance	
OECD	Organisation for Economic Co-operation and Development	
0100	Office of the United Nations High Commissioner for Human	
OHCHR	Rights	
OSII	Open Society Justice Initiative	
PAIC	Paris Agreement Implementation and Compliance Committee	
PCCB	Paris Committee on Canacity-Building	
	Pacific Island Countries	
DTSD	Post traumatic distress syndrome	
	Overy Wizerd for International Davelonment Statistics	
QWIDS	Standing Committee on Einenee	
SCP	Standing Committee on Finance	
SCF	Sustainable consumption and production	
SDC	Sustainable development	
SDGS	Sustainable Development Goals	
Slua	Swedish International Development Co-operation Agency	
SLK		
STEM	Science, Technology, Engineering, and Mathematics	
TEC	Technology Executive Committee	
TROSA	Trans-boundary Rivers of South Asia	
UN	United Nations	
UNCLOS	United Nations Convention on the Law of the Sea	
UNCTAD	United Nations Conference on Trade and Development	
UN DESA	United Nations Department of Economic and Social Affairs	
UNDP	United Nations Development Programme	
UNECA	United Nations Economic Commission for Africa	
UNECE	United Nations Economic Commission for Europe	
UNEP-FI	United Nations Environment Programme Finance Initiative	
UNESCAP	United Nations Economic and Social Commission for Asia and	
	the Pacific	
UNESCO	United Nations Educational, Scientific and Cultural	
UNLDCO	Organization	
UNFCCC	United Nations Framework Convention on Climate Change	
UNHCR	United Nations High Commissioner for Refugees	
UNICEF	United Nations International Children's Emergency Fund	
UNIDO	United Nations Industrial Development Organisation	
UNIFEM	United Nations Development Fund for Women	
UNODC	United Nations Office on Drugs and Crime	
UNSDG	United Nations Sustainable Development group	
WB	World Bank	
WDS	World Development Sources	
WEDO	Women's Environment and Development Organization	
WHO	World Health Organization	
WMO	World Meteorological Organization	
WoS	Web of Science	
WUAs	Water User Associations	

1. Introduction

1.1 Background

1.1.1 Climate change impacts

Human-driven climate change has become an undeniable reality and is one of the greatest challenges faced by humanity. This global phenomenon continuously transforms natural and socio-economic systems by altering the environment and threatening the populations' health, well-being and livelihood stability. On the other hand, the projected growth of the global population that is expected to reach 9.7 billion in 2050 (UN Department of Economic and Social Affairs, 2022), fossil fuelbased economic and production activities, land and energy use patterns not only amplify existing multiple climate threats, but also will incrementally create more complex and cascading risks increasing related losses and damages (IPCC, 2021; IPCC, 2023). In 2023, various regions of the world experienced numerous catastrophes from tornados, destructive wildfires to wide-ranging impacts of cyclones with were followed by floods that caused preliminarily around USD 200 billion in damages (Aon plc, 2023). Over the last two decades, climate changerelated meteorological, hydrological and climatological extremes have led to total global economic losses exceeding USD 2 trillion (Eckstein et al., 2021; WMO, 2021a; 2021b). Furthermore, climate change increases the volatility of the global economy by reducing its output at between 11% and 14% by 2050 compared to the 'without climate change' scenario (i.e., 0°C temperature change) (Swiss Re Institute, 2021). No industry or economic sector have zero contribution to climate change or remains 'immune' to its impacts, though some are significantly more sensitive to them compared to others. For instance, agriculture is responsible for 15-29% of global greenhouse gas (GHG) emissions (Magazzino et al., 2023; Ivanovich et al., 2023), but it also experiences reduction in crop yields mainly in mid- and low latitude regions (IPCC, 2023). Furthermore, the sector faces an increasing risk, particularly of food, water- and vector-borne diseases, as well as decreasing availability of water supply for irrigation needs, and an acceleration of biodiversity loss (Abbass et al., 2022). Studies estimate that the total economic climate change-related losses in agriculture could reach 0.3% of the global gross domestic product (GDP) per annum by the end of the century (Stevanović et al., 2016).

The energy sector, another major contributor to climate change, experiences disruptions in fuel supply and energy production at an unprecedented pace and scale. Massive power outages as a result of weather extremes affect millions of people worldwide (e.g., Xi, 2016; WMO, 2022). Heatwaves, droughts and water stress are undermining the resilience of current and future energy infrastructure and jeopardizing supplies of renewable energy (WMO, 2022). Economic losses due to natural hazards (e.g., floods, droughts, heatwaves) reach approximately USD 0.6 billion per year in the European energy sector only. They are projected to rise by 394% and 860% by the 2020s and 2050s, respectively (Forzieri et al., 2018).

The health sector contributes significantly less to climate change compared to agriculture and energy. It is accounting for about 5% of global GHG emissions (Watts et al., 2019) but is also significantly affected by the climate change impacts. The sector's capacity and infrastructure are under constant pressure due to the growing and amplified health risks associated with climate change including mental health issues, morbidity and mortality rates (Rocque et al., 2021). It is estimated that

climate change will lead to an additional 250,000 deaths annually between 2030 and 2050 due to malnutrition, malaria, diarrhea and heat stress (WHO, 2023). Over two decades, about 9 000 natural disasters affected more than 4 billion people and were responsible for more than 730 000 fatalities (EM-DAT, CRED, 2023). Furthermore, affected populations can experience heat exhaustion, physiological heat strain, heatrelated and respiratory illnesses, longer-term impacts on mental health including psychological distress and anxiety about the future and severe illnesses requiring hospital care (Flouris et al., 2018; Palinkas and Wong, 2020; Ebi et al., 2021). Climate change also influences economies by triggering people's displacement. Extreme weather events, decreasing water resources, loss of arable agricultural land, droughts and other critical changes in environmental conditions force individuals to migrate in an attempt to reduce vulnerability and risks to their livelihoods (McLeman, 2017). Additional driver for displacement is a 'threat multiplier' (Goodman and Baudu, 2023) that amplifies other factors that threaten security, such as poverty, tensions and conflicts over depleting natural resources, loss of income source, etc. According to the projections, in the worst-case scenario, over 1 billion people could be forcibly displaced by 2050 due to climate change and natural disaster events (Institute for Economics and Peace, 2020).

1.1.2 Responding to climate change

Mitigation and adaptation are the efforts to address the unprecedented rates of global climate change and its associated risks that pose a direct threat to human health and well-being. The global scale of the problem calls for increased international cooperation and nations' commitment to advance these efforts, including adequate and sustained funding, enhancement of the scientific basis and technological development (Ford and King, 2015; Midgley and Methner, 2016; IPCC, 2023). According to estimates, achieving a low-carbon transition by 2050 requires about USD 2.4 trillion of annual average energy investments, which represent around 2.5% of the global GDP (IPCC, 2018). Adaptation costs are rising due to the growing level of associated climate change risks. Recent projections indicate that adaptation finance for developing countries will require between USD 215 billion and 387 billion per annum by 2030, which is equivalent to between 0.6% and 1.0% of their combined GDP (United Nations Environment Programme, 2023). It is worth mentioning that there is no single amount reflecting the adaptation finance needed across developed countries. Several studies conducted in Europe showed that the required investments for the 1.5°C scenario, are estimated at approximately USD 44 billion per annum, while for the 2°C scenario are at USD 88-132 billion, and for the 3-4°C scenario are at USD 192-220 billion (Feyen et al., 2020; Watkiss et al., 2018). Mitigation and adaptation actions stimulate technological innovations. Companies are introducing new materials (e.g., carbon fibre), climate-smart appliances, noncarbon and carbon capture and storage technologies as well as innovations that allow farmers to adopt alternative types of farming, and new stress-tolerant plant and animal breeds (Wilson et al., 2018; Sovacool et al., 2019; Loboguerrero et al., 2019; Özatağan, and Ayalp, 2021). In addition to intensive financing and technological development, effective long-term planning and implementation of mitigation and adaptation require a flexible, multi-sectoral and inclusive approach. Failure to develop this type of approaches, without including individual (e.g., age, gender, education and disabilities, personal past experience), group (e.g., religion, ethnicity), or livelihood (e.g., capabilities, material and social resources) differences may worsen the conditions that mitigation and adaptation strategies were initially addressing (Schipper, 2020; Meijer et al., 2019). The importance of inclusiveness of climate action has been captured in a broad range of reports, policies and initiatives including the United Nations Sustainable Development Goals (UN SDGs) (Gupta and Vegelin, 2016).

1.1.3 Climate change and sustainable development interconnection

The Sustainable Development Goals (SDGs) initiative along with the Paris Agreement represent major global multilateral efforts to mitigate and adapt to climate change impacts and promote sustainable development (Dagnachew et al., 2021). A growing body of evidence demonstrates that climate change impacts can hinder the achievement of most of the SDGs (e.g., Fleurbaey et al., 2014; von Stechow et al., 2016; IPCC, 2018; Fuso Nerini et al., 2019). This may become a constraint for simultaneous transition to a low-carbon society (IPCC, 2018; Gomez-Echeverri, 2018; Moreno et al., 2023). The strong interlinkage between climate change and SDGs creates both synergies and trade-offs that may either maximise or impede successful associated decision-making and co-benefits across multiple targets (Thornton and Comberti, 2017; Fuso Nerini et al., 2019; Dagnachew et al., 2021; Moreno et al. 2023). One of the examples is climate change-induced migration that was addressed by the Paris Agreement only in the context of migrants' vulnerability, while the SDGs recognize migration as one of the powerful drivers of sustainable development without mentioning how it might be altered by climate change (United Nations, 2015c). Nevertheless, climate action and vulnerability reduction are among the key factors in achieving a progress on SDGs, particularly in improving access to health services, ensuring food and water security, and reducing poverty and inequality. This, in turn, will increase population resilience against the climate change impacts (Pelling and Garschagen, 2019; Schipper, 2020; Fuso Nerini et al., 2019; Dzebo and Shawoo, 2023). However, none of these activities will be fully effective without consideration of population characteristics.

1.2 Filing the research gap: Climate change and gender landscape

The successful development and implementation of adaptation and mitigation strategies require, among others, a deeper understanding of the roles played by population's socio-economic and demographic characteristics in the climate change contexts and consequently, their inclusion in the respective processes. All these characteristics, including population age structure, size, spatial distribution, mortality and natality rates, education, poverty, life expectancy, and gender, predefine, to various degrees, a persons' capacity and ability to overcome climate change impacts.

Compared to other characteristics, gender is discussed more frequently in the climate change-related context, especially after the international recognition of gender non-neutrality of climate change impacts (UN Secretary-General, 2008; Pham and Saner, 2021). The term 'gender' is defined as relationships between men and women, boys and girls, with the determined behaviour, roles, expectations, rules, and practices (FAO, 1997; March et al., 1999; UNESCO, 2014). It is widely perceived as a core principle that "often governs the processes of production and reproduction, consumption and distribution" (FAO, 1997) shaping access to power, control over natural and financial resources, vulnerability, resilience and engagement in decision-making processes (UN Women, 2001; Sida, 2015). Understanding and addressing the disproportionate climate change effects experienced by women and men has

become one of the central points for vulnerability reduction and building resilience. The late 1990s were marked as a pivotal point, when gender/women issues began to be actively included into the climate change agenda. Among the notable milestones are:

- 1995 the Beijing Declaration and Platform for Action that outlined a greater impact women face from environmental degradation, pollution, and resource depletion (United Nations, 1995).
- 2002 the call of the Commission on the Status of Women to "mainstream a gender perspective into ongoing research and encourage the application of the research results in policies and programmes on the impacts and causes of climate change" (UN Commission on the Status of Women, 2002).
- 2008 the acknowledgement of climate change as a non-gender-neutral phenomenon by the United Nations Secretary-General (UN Secretary-General, 2008).
- ⁻ 2017 the United Nations Framework Convention on Climate Change (UNFCCC) Gender Action Plan that set out objectives and activities to "advance knowledge and understanding of gender-responsive climate action and its coherent mainstreaming in the implementation of the UNFCCC ... as well as women's full, equal and meaningful participation in its process" (UNFCCC, 2017).

Previous research works have discussed the linkages between gender differences in environmental attitudes and support for environmental policies (e.g., Greenbaum, 1995; Davidson and Freudenburg, 1996; Mohai, 1997) and women roles and their vulnerability in natural disasters in comparison to men (Enarson and Morrow 1998; Fothergill, 1999). In the early 2000s, the thematic focus shifted to climate change, ranging from its impacts on exacerbating gender inequalities and gender-based violence (e.g., Brody et al., 2008; Adzawla et al., 2019; Desai and Mandal, 2021), to the recent studies on the effects of the COVID-19 pandemic on already vulnerable to climate change gender groups (e.g., Belsey-Priebe et al., 2021; Nyahunda et al., 2021; Sifa et al., 2021). However, despite an apparent interconnection between these studies, it often remains difficult to draw a holistic perspective on climate changegender interconnections due to thematic, geographical, and scaled scattering of the obtained findings. The works mainly focus on an interlinkage between gender/women and a specific activity, process or sector affected by climate change in a particular geographical region. For instance, Lama et al. (2021) discuss the complexity and nonlinearity of the interrelation between gender, migration, and climate change. The authors defined gender as an organising principle, and climate change as a risk modifier rather than an additional risk. Rahman and Khatun (2019) investigated the linkages between men's and women's vulnerability and water quality, salinity intrusion, land, socio-economic and demographic conditions that are affected by extreme weather events (e.g., cyclonic storms) in Bangladesh. Kironde et al. (2022) examined the insights and outcomes of gender inclusion in climate change and water policies, demonstrating existing gaps and challenges for the implementation in Tanzania. Gartaula et al. (2020) highlighted the interrelation between emissions reduction, female labour participation in the agricultural sector, and adoption of more effective rice production technologies in India. The study noticed that the potential benefits vary depending on the extent of women's engagement in the production and the agro-climatic conditions of the regions.

The climate change-gender interconnections contribute to a better understanding of their common ground, identification of critical points, trade-offs, and failures to

develop effective adaptation and mitigation solutions. The global nature of the climate change problem requires solutions that can have an effect on a similar scale. Consequently, being considered as one of the key elements of adaptation and mitigation, climate change-gender interconnections are needed to be explored by taking into account a holistic perspective.

1.3 Scope of the dissertation

1.3.1 Aim and objectives

Interconnections between climate change and gender are characterized by complex synergies and dynamics, continuous development, and ongoing mutual impacts. This PhD work aimed to investigate some key aspects of the climate change-gender interconnections and shed some light on their interactions in a variety of contexts in different geographical regions. The work was built around three main domains of climate change-gender interconnections: vulnerability, empowerment, and benefits, which are subject to thematic and geographical, regional, national or local level context (Figure 1.1). Vulnerability here is considered as a factor that has been amplified by gender imbalances in decision-making power, land ownership rights, access to natural and financial resources, information and knowledge, and genderblindness/neutrality policies of climate change and action. Leadership/Empowerment is characterised by male domination, women's underrepresentation and their limited decision-making power from local to international levels. Empowerment provides women with opportunities to increase their social, economic, and political capacities and contribute to adaptation and mitigation strategies and related decision-making processes. Benefits are reflected in improved effectiveness of adaptation and mitigation solutions as result of gender/women considerations integration.





To attain the main aim, this work focused on the following specific objectives that explored:

• how climate change research addresses gender and women issues;

- how changes concerning/progress on disbalances in gender empowerment and leadership are reflected in climate change adaptation and mitigation in African countries;
- how gender and women considerations and needs are perceived in the climate change context in the Central Asian countries.

Additionally, the work looked at

- how changes in gender balance of those who have been engaged in agriculture affected the sector;
- where stands gender in the climate change-induced extreme events and mental health context in the Pacific region;
- how gender as a cross-cutting issue interact with the SDGs towards their implementation and attainment of SDGs; and
- how gender issues are perceived within the context of climate changeinduced migration in Kenya and Ethiopia.

These are the aspects that could be considered as one or more sub-domains of the climate change-gender interconnections. Figure 1.2 shows the research implementation roadmap for this PhD research. The work deploys a variety of methods, which include literature reviews, content analysis, bibliometric analysis and on-line surveys. For the purposes of this research, in some cases the terms 'gender' and 'women' terms were used interchangeably, and 'gender groups' were referred to men and women groups. The following approach could be influenced by the specific social and cultural environments in the investigated countries, as well as by the size and scope of the impacts that women and men groups make in the climate change adaptation and mitigation context at the global level.





1.3.2 Structure of the dissertation

The structure of this dissertation follows the objectives set out above. Chapter 2 explores the overall developments of inclusion of gender/women issues in scientific contributions in climate change research and the main themes that can be referred to the vulnerability, leadership/empowerment, and/or benefits domain(s) of climate change-gender interconnections. Chapter 3 discusses the leadership/empowerment domain based on analysis of the advances made towards attaining gender-balanced leadership/empowerment in climate change adaptation and mitigation in 12 African countries. Chapter 4 investigates the progress made on including gender/women considerations by climate change-related in CA countries. Chapter 5 reviews climate change-gender interconnections in terms of feminization, its drivers, positive and negative outcomes, and associated challenges within the broader agricultural context

in different geographical regions including Latin America and CA. Chapters 6, 7 and 8 explore several additional domains of climate change-gender interconnections in the context of additional issue areas such as the successful SDGs implementation, extreme events and mental health, and climate change-induced migration. The last chapters synthesize the obtained findings, discuss their implications and provide an outlook for future research.

2. Gender issues within climate change research: a bibliometric analysis

This chapter was published as Kovaleva, M. et al. (2021) *Climate and Development*, DOI: 10.1080/17565529.2021.1980365

2.1 Abstract

The last decades are recognized as a period when a gender-sensitive perspective being broadly acknowledged by the majority of actors involved in the field of climate change. Existing gender inequalities exacerbated by climate change urge the inclusion of a differentiated female-male approach into research, decision-making processes, adaptation, and mitigation. This paper quantitatively reviews scientific contributions in the gender subfield within climate change research published between 1996-2020 by adapting a bibliometric analysis approach. The literature was extracted from Web of Science, Google Scholar, and World Bank's repositories. The results of the co-occurrence, bibliographic coupling, co-citation, and co-authorship analyses were mapped by using the VOS viewer software. The selection criteria were developed to maximize the thematic coverage of the publications that could be referred to the gender subfield within climate change research. The findings help better understand the overall developments in the gender subfield in terms of discussed topics, authors and institutional collaborations, contributions of international organizations in the capacity of authors, and an amount of grey literature. The review might represent an interest for further studies in consideration of new research topics, geographical focus, compositions of co-authors teams, and be communicated in debates on gender within the climate change context.

3. Understanding Needs and Potentials for Gender-Balanced Empowerment and Leadership in Climate Change Adaptation and Mitigation in Africa

This chapter was published as Kovaleva, M. et al. (2022) Sustainability, DOI: 10.3390/su14159410

3.1 Abstract

The past years were marked by the COVID-19 pandemic, economic downfall, the 5th anniversary of the Paris Climate Agreement, and the end of the African Women's Decade. According to the latest projections, African countries will continue to face increasing inequalities, as well as risks to human health, water and food security, due to climate change. African countries are also struggling to reduce gender-related power imbalances in adaptation and mitigation that magnify existing vulnerabilities, particularly those of women. Therefore, any advances made in this narrative are significant. This paper investigates the needs and potential for gender-balanced leadership/empowerment in adaptation and mitigation based on climate change experts' views on the advances made in Africa. This is complemented by a bibliometric analysis of the literature published on the topic between the years 2015 and 2022. The study suggests that although women's influence on climate change related decisions is growing, a series of barriers need to be overcome, among which are lack of knowledge and political will. The COVID-19 pandemic is seen as having both positive and negative potentials for gender-balanced leadership/empowerment. The findings provide a premise for identifying possible directions of further actions towards gender-balanced leadership/empowerment in climate change in African countries.

3.2 Introduction: Climate Change–Gender Interconnections

The continuous climate crisis poses a threat to the already fragile ecosystems, livelihoods, health, food and water security, increasing poverty and marginalization. Climate change and its related issues have been investigated in various contexts in relation to a wide range of factors. One such factor is the inclusion of a gender variable. The United Nations acknowledged climate change as a non-gender-neutral phenomenon (UN Secretary-General, 2008) and the United Nations Framework Convention on Climate Change (UNFCCC) called "... to advance knowledge and understanding of gender-responsive climate actions . . . as well as women's full, equal and meaningful participation in its process . . . " (UNFCCC, 2017). Inequalities between gender groups have economic, political and social implications regarding how and whether individuals are affected by and respond to climate change (Björnberg and Hansson, 2013; Rohr, 2007; IPCC, 2014b). Therefore, it is vital to understand climate change-gender interconnections, which are characterised by their complex synergies and dynamics, continuous development, and ongoing mutual impacts across all levels and dimensions. The majority of the studies are very often evidence-based and focus on the interconnection within a specific thematic and geographical context, at the regional, national, or community level. For instance, Mensah et al. (2022) investigated gender differences in adaptation strategies and climate change perception among smallholder farmers in a semi-arid region of Africa. Bunce and Ford (2015) discussed the connections between gender and adaptation, resilience, and vulnerability research. The authors pointed out that the investigations are strongly focused on women's experiences (Bunce and Ford, 2015). von Lander Svendsen et al. (2022) investigated the mutual impact of gender and climate policies, including equality in respective decision-making processes, in the Nordic countries.

Pearse (2016) analysed gender and climate change relations in a broader context. The study was focused on five dimensions: (i) vulnerability and climate change impacts; (ii) adaptations in various contexts; (iii) responsibility for greenhouse gas emissions; (iv) inequalities in climate governance; (v) knowledge and social action on climate change. The author also indicated a key role of gender in social transformation in connection to climate change (Pearse, 2016). Brody et al. (2008) extended the scope by linking the climate change–gender interconnections to (i) health; (ii) agriculture; (iii) water; (iv) wage labour; (v) disasters and their aftermath; (vi) migration; (vii) conflict (Brody et al., 2008). In 2016, Global Gender and Climate Alliance (GGCA) report illustrated how gender differences shape vulnerability to climate change and decisions on its adaptation, including policy issues and health impacts (Sellers, 2016).

In the broader scope, i.e., at the meta-level, climate change–gender interconnections can be clustered into three thematic domains: vulnerability, benefits and leadership/empowerment.

Climate change **vulnerability** is subject to gender differences. Women face climate change threats and struggle to overcome its challenges to a greater extent, as they have fewer means and adaptive capacities than men (e.g., Lambrou and Piana, 2006; Alston, 2013; Reggers, 2019; FAO and ARC, 2021; Eskenazi et al., 2020). Moreover, their vulnerabilities are amplified by gender imbalances in decision-making power, land ownership rights, and access to natural and financial resources, information and knowledge (Khandekar et al., 2019; Habtezion, 2013; Yadav and Lal, 2018; Kabir et al., 2016). Very often, governments delay in introducing the gender variable to adaptation and mitigation processes and national climate policies (Khandekar et al., 2019; Nhamo, 2014; Least Developed Countries Expert Group, 2015; Ford et al., 2015). Gender-blind or gender-neutral measures and policies create additional barriers to reducing and easing women's vulnerability (e.g., Habtezion, 2016; Davis et al., 2015).

On the other hand, attaining a gender balance might provide a range of climate change-related **benefits**. The effectiveness of adaptation and mitigation solutions can be significantly improved as a result of gender inclusion and the more active involvement of women, who are perceived as powerful agents of change to address climate change at scale (Glemarec et al., 2016; Dankelman, 2010; Glazebrook, 2011; Osman-Elasha, 2013; Aguilar et al., 2015; IUCN, 2015; Cook et al., 2019). Furthermore, more gender-balanced representation in decision-making positively affects the integration of climate-smart agricultural technologies, the ratification of climate treaties, the building-up of community resilience and improvements in adaptive capacities (e.g., Habtezion, 2016; Andrijevic et al., 2020; FAO, 2011; Carvajal-Escobar et al., 2020). According to estimations, the 20–30% increase in productivity achieved by women farmers would reduce the number of undernourished people by 100–150 million worldwide (FAO, 2011).

Leadership, empowerment, and their associated processes are crucial in confronting and overcoming global challenges such as climate change and its impacts (Parker et al., 2015; United Nations, 2015a). Similar to other processes, leadership/empowerment in climate change is characterised by male domination, and

women have limited decision-making power from local to international levels (FAO, 2020d). Empowerment aims to reduce power imbalances and provides women with opportunities to increase their social, economic and political capacities and make strategic choices that, in turn, contribute to adaptation strategies and related decision-making processes (UNDP, 2007; Chen and Tanaka, 2014; Mekonnen, 2022; Habib et al., 2022).

Table 3.1 provides examples of specific climate change–gender interconnections for each domain: vulnerability, benefits or leadership/empowerment. Given the enormity and diversity of the works, particularly at the local level, the list of included studies is far from being exhaustive.

Domains	Interconnections (Selected)	Description	Reference
Vulnerability	Natural disasters and	Natural disasters: extreme weather events and their subsequent	IPCC (2014b), Neumayer and
	extreme weather	impacts might disproportionately affect different gender groups,	Plümper (2007), Soroptimist
	events	increasing existing inequalities, and undermining their water and	International of the Americas
		food security. The events also narrow the gender gap in life	(2011), Racioppi and
		expectancy.	Kajagopalan (2016), FAO $(2017a)$ Detail at al. (2020)
	Climate change	Climate in dward diamle convert is covered by an in dividually includivy	(2017a), Patel et al. (2020)
	Climate change-	Climate-induced displacement is caused by an individual's inability	CARE (2020), Sams (2019), Biggoud et al. (2018)
	displacement/	of climate change as a result of inequalities or a lack of resources. It	Rigaud et al. (2018)
	migration	is triggered by direct physical harm from extreme weather events or	
	migration	slow-onset impacts, indirect consequences on food insecurity and	
		conflict over natural resources and land rights. Women represent	
		more than half of displaced people. Very often, they face an	
		increased risk of domestic or sexual violence and deterioration in	
		their physical, emotional and mental health, and have less access to	
		relief resources.	
	Unequal adaptive	Gender groups have unequal adaptive capacities due to a lack of or	IPCC (2014b), Khandekar et
	capacities	limited decision-making powers, assigned roles and tasks, labour	al. (2019), FAO (2017a), Patel
		division, gender-biased legislation, limited access to and availability	et al. (2020)
		of technology, economic capital and productive resources, and gaps	
		in literacy level, health and nutritional status.	
	Limited or no access	Gender inequalities in access to education and information are	Habtezion (2013), Yadav and
	to information,	among the key causes of vulnerability to climate change that	Lal (2018), Kabir et al. (2016)
	education, knowledge	decrease an individual's ability to build up resilience to its impacts.	

Table 3.1 Climate change–gender interconnections. Source: author.

	Limited or no access	Socio-cultural gender roles and norms that determine access to	Khandekar et al. (2019),
	to resources (e.g.,	resources and (land) ownership rights influence and shape an	Habtezion (2013), Sen Roy
	natural, financial,	individual's vulnerability. Very often, women have limited or no	(2018a), Mnimbo et al. (2016)
	etc.), ownership	access/rights.	
	rights		
	Gender-blind climate	Gender-blind climate finance might reinforce climate change	Habtezion (2016), Davis et al.
	finance/policies/inter	vulnerabilities. Very often, funding institutions lack a gender	(2015), GEF Independent
	ventions	perspective in their policy frameworks. Prior to 2015, projects that	Evaluation Office (2017),
		address climate change and women's rights received 0.01% of	Schalatek (2018)
		worldwide funding support. The efficacy and effectiveness of	
		mitigation and adaptation interventions might be significantly	
		reduced due to the exclusion of gender issues from climate-related	
		projects and policies, although, in recent years, the situation has been	
		changing.	
Benefits	Agents of change	Women are powerful agents of change to address climate change at	Habtezion (2016), Glemarec
		scale.	et al. (2016)
	Gendered-balanced	Women's increased participation in decision-making processes at	Yadav and Lal (2018),
	participation and	various levels of governance makes considerable contributions to	Habtezion (2016), Leisher et
	engagement	natural resource management, biodiversity conservation efforts,	al. (2016), Loarne-Lemaire et
		ratification of international environmental treaties, building	al. (2021)
		community resilience and responding to climate-related disasters.	
		Their expertise and experience contribute to the reduction in adverse	
		impacts and accelerate the development of technological innovations	
		to address climate change.	
	Climate resilience	A reduction of gender imbalances is one of the most effective	Glemarec et al. (2016),
	and adaptive capacity	mechanisms in the development of climate resilience and adaptive	Andrijevic et al. (2020), Doss
		capacity, which, in turn, improves food security.	(2018)

	Paris Agreement	Support of gender equality and women's rights is one of the most	Glemarec et al. (2016)
		powerful ways to implement the Paris Agreement.	
Leadership/	Equal representation	Equal gender representation in administrative and political bodies	Loarne-Lemaire et al. (2021),
Empowerment	on boards/	engaged in climate change contributes to effective climate change	Magnusdottir and Kronsel
	administrative and policies. However, sometimes, having a critical number of women		(2015)
political bodies		on these bodies does not directly lead to the development of gender-	
		sensitive climate change policies and/or the inclusion of gender	
		differences in climate issues.	
	Underrepresentation	There is a gender gap in representation (women are	FAO (2020d),
		underrepresented) in climate-change-related decision-making	UNFCCC (2021)
		processes in climate governance, including in international	
		delegations on	
		UNFCCC bodies.	
	Adaptation strategies	Adaptation capacities to climate change are positively influenced by	Mekonnen (2022), Habib et
	and decision-making	women's empowerment and decision-making power.	al. (2022)

The current work focuses only on climate change–gender interconnections in the leadership/empowerment domain in a specific geographical region. More precisely, the paper investigates the needs and potentials for gender-balanced leadership/empowerment in climate change adaptation and mitigation, based on the views and perspectives of climate change experts regarding the advances made in a number of African countries over the last five years. The study is complemented by a bibliometric analysis (i.e. keywords and terms co-occurrence analysis) of the literature on women's leadership and empowerment in climate change in the region published between the years 2015 and 2022.

The continuously increasing role of African nations in the global climate change negotiations, a growing action on the national level, and the transformation of women's role as an agent of change lead to the need to ensure that gender imbalances in leadership and empowerment will be thoroughly addressed and not neglected along the way. Therefore, it is essential to demonstrate the implications of the steps that have been already taken towards attaining equality, to outline additional potential benefits and reductions in risk, in addition to simple improvements in gender-disaggregated statistical data.

The work contributes to the current literature by providing initial insights on the outcomes of the advances made in recent years. The findings can form a basis to show the importance of the measures that could be taken towards more gender-balanced leadership and empowerment in the climate change adaptation and mitigation context.

The following two sections discuss the aspects of climate change–gender leadership/empowerment interconnections and the characteristics of gender leadership/ empowerment in climate change in Africa. The fourth section describes the methods used. The penultimate section presents the obtained findings and their implications for gender-balanced leadership and empowerment in climate change. The paper concludes with a summary of the findings and the study limitations.

3.3 Climate Change-Gender Interconnections in Leadership/Empowerment

Climate change leadership and empowerment with reference to gender issues at international, national or sectoral levels vary depending on a range of factors, among which are economic development, political will, strong social rules, and beliefs. Gender imbalances in this context are perceived as one of the hindrances to more effectively addressing climate change stressors and impacts. Therefore, recent decades have been marked by initiatives to attain gender parity in climate leadership roles in governance bodies and boards across countries, sectors and levels of implementation.

The intermediate analysis of the progress made by the United Nations Framework Convention on Climate Change (UNFCCC) demonstrated that equal or female-dominated representation was only reached in the Adaptation Committee (AC) in 2018–2020, Consultative Group of Experts (CGE) in 2013 and 2016, and Paris Committee on Capacity-Building (PCCB) in 2017–2020. Figure 3.1 shows the female/male ratio of the representatives on UNFCCC boards and bodies during the period 2013–2020. Values below 1 signify the prevalence of male representatives on a board or body in a particular year. Figure 3.2 presents the share of women who occupied the leadership position of Chair, Co-Chair or Vice Chair on these boards and bodies from 2013 to 2020. The percentage value is calculated based on the total number of female representatives on a board or body in that year. The results indicate that even the bodies/boards with a prevalence of female representatives are often

male-chaired, such as the Consultative Group of Experts (CGE) in 2013 and 2016 (Figures 3.1 and 3.2). Both figures exemplify the challenging nature and complexity of the processes to achieve a significant reduction of gender gaps and attain gender parity in leadership positions in climate change related bodies. Thus, it can be assumed that similar processes at lower (e.g., regional, national and community) levels might be characterised by even larger barriers, hindrances, and complexity.



Figure 3.1 Female/male ratio of the number of the representatives on UNFCCC boards and bodies between the years 2013 and 2020, except TEC; the data have been available since 2017, FWG and KCI—since 2019, PAICC—since 2020 (based on Gender Climate Tracker); full body names are listed in Appendix I. Source: author.



Figure 3.2 Percentage of Female Chair, Co-Chair or Vice Chair out of the total number of female representatives on UNFCCC boards/bodies between the years 2013 and 2020, except TEC; the data have been available since 2017, FWG and KCI—since 2019, PAICC—since 2020 (based on Gender Climate Tracker); full body names are listed in Appendix I. Source: author.

At the national level, according to the National Adaptation Plan (NAP) Global Network Synthesis Report, only a few nations place women as stakeholders or as agents of change rather than victims and indicate their empowerment as a priority in adaptation measures (Dazé and Dekens, 2018). Holvoet and Inberg (2014) suggested that having a woman lead the governmental agency responsible for drafting a National Adaptation Programme of Action (NAPA) does not always actively contribute to gender inclusion, women's participation, or expertise on gender in the programme or its related decision-making processes and evaluation. However, Mavisakalyan and Tarverdi (2019) suggested that women's representation in national parliaments significantly contributes to the acceptance of more rigorous climate change policies across countries. Such contrary findings might exemplify not only the topic's complexity but also the still low number of studies, and inconsistencies in research methodology or estimation approaches. Nevertheless, these factors do not oppose the need to reduce gender imbalances in climate change leadership and empowerment. Similarly to the situation at the international level, these processes are hindered by a range of economic, political, and cultural aspects, among which are societal resistance, lack of education, insufficient knowledge, lack of motivation or confidence, mainly among women (Wen et al., 2013; Pham and Brockhaus, 2015). Numerous case studies demonstrate diverse solutions and approaches to overcoming some of these barriers. For instance, in India, local women leaders created their own groups to adapt to the effects of climate change, increase resilience, and receive access to international sources of income and capacitybuilding (Christoff et al., 2019). The BSR's Business Action for Women offered the cluster "Empowering Women to Lead through Climate Resilience", which supports women's empowerment in climate-resilient agriculture in the private sector by promoting gender-sensitive climate resilience methods and strategies (Harris and Abbott, 2018).

In the business sector, the gender discrepancies in leadership are about two times larger than they are in labour-force participation (Woetzel et al., 2015). As of January 2022, only 6.6% of CEO positions in the companies listed in the SandP 500 list are occupied by women (Catalyst, 2022). In 2021, 31% of senior leadership roles were held by women globally and 90% of firms had at least one woman in senior management (Grant Thornton, 2021).

In the United States, the energy sector is mainly characterised by the overrepresentation of male leaders (Pearl-Martinez and Stephens, 2016), although female leadership plays a significant role in advancing energy system transformations (Allen et al., 2019). Increasing investments in renewable energy, and companies' more effective commitment to climate change management, including the implementation of climate policies and carbon performance, have been linked with gender diversity in companies' leadership (Pearl-Martinez and Stephens, 2016; Allen et al., 2019; Jizi et al., 2021) and particularly with having a critical number of women in top management teams (Nuber and Velte, 2021; Caby et al., 2022).

3.4 Gender Leadership/Empowerment in Climate Change in the African Context

Characteristics of gender leadership/empowerment in climate change in the region have been continuously changing as a result of the efforts to reduce the imbalances and transform of women's position. Nevertheless, the role of women in African countries remains dependent on geographical location and their belonging to a specific religious or ethnic group (Punnett and Clarke, 2017). These factors create further challenges and complexities in terms of gender dynamics when shaping leadership/empowerment in climate change at every level. According to the Global Gender Gap Index, the gender-based gap that needs to be closed in the Political Empowerment dimension in African countries ranges between 44.7% (Rwanda), 50.7% (Mozambique), 95.3% (Nigeria), and 99.9% (Yemen) (World Economic Forum, 2021). The studies also demonstrate men's overrepresentation in decision-making processes in the sectors sensitive to climate change (Dekens and Dazé, 2019).

Addressing gender gaps in leadership and empowerment significantly contributes to reducing vulnerability and building resilience while the continent faces high exposure to climate change and its impacts, with low adaptive capacities (Niang et al., 2014). National governments, with the support of international organisations, introduced and implemented a set of programs (e.g., Africa Adaptation Programme (AAP)) aimed at women's empowerment, more active engagement in decision-making processes and capacity-building. The ratification of key international documents, such as the Convention on the Elimination of All Forms of Discrimination Against Women and the African Union's Protocol on the Rights of Women in Africa, paves the road for more women to enter the labour market and be promoted to managers and leaders (Punnett and Clarke, 2017). The McKinsey Global Institute study found that if North Africa, the Middle East and sub-Saharan African countries reached their fastest-moving neighbour gender parity performance, they would contribute about USD 0.9 trillion to the global gross domestic product (GDP) in 2025 (McKinsey, 2015).

In 2013–2014, the African Working Group on Gender and Climate Change (AWGGCC) and Women and Gender Programme on Climate Change were established to support the engagement of African countries in global and regional climate change-gender processes, as well as the participation of women negotiators in climate action (AWGGCC, 2017). At the national level, 85% of the submitted Nationally Determined Contribution (NDC) reports referenced gender (Remteng et al., 2021). Furthermore, countries supported by the international community introduced a range of initiatives with the inclusion of the gender variable. For instance, in Niger, the gender dimension was included in the Climate Risk Management Technical Assistance Project (CRM/TASP). A gender intervention approach, "Transformative Adaptation Prioritizing the Adaptation Needs of Women in the AAP", focused on climate change adaptation, was presented in Nigeria. In Kenya, steps were taken to mainstream gender into the country's climate change impacts through their empowerment (Least Developed Countries Expert Group, 2015).

Despite the growing trend of this type of action, the majority of studies at present focus on or refer to one of these two issues: (i) gender imbalance, its causes, and associated impacts, (e.g., Holvoet and Inberg, 2014; Makinaa and Moyob, 2016; Angula et al., 2021; Patnaik, 2021); (ii) actions taken to reduce these, (e.g., Babugura, 2010; Wagner et al., 2015; Grillos, 2018; Nyahunda, 2021).

3.5 Materials and Methods

The study used two methods: a bibliometric analysis and an online questionnaire.

3.5.1 Bibliometric Analysis

Bibliometric analysis has attracted growing attention among scholars as a tool to evaluate the relevant scientific literature and identify publication trends and research elements. This analysis was performed to outline prominent themes in the climate change–gender domain in leadership/empowerment in selected African countries. The datasets used included peer-reviewed and grey literature, retrieved from the Web of Science (WoS) core collection, Scopus, and Google Scholar (GS) databases. These three databases represent the largest collections of scientific and scholarly publications published worldwide. The downloaded datasets included such variables as publication title, keywords, abstract, authors' name, year of publication, and journal name. However, only the first three variables were used for analysis and the rest were used for cross-checking and the elimination of duplicates.

Table 3.2 the search data range was limited to the years 2015–2022 and specific geographical locations. The datasets were retrieved in February 2022. Only publications in English were selected.

Criteria	Description	
Data range	2015–2022	
Language	English	
Type of publication	All types available in the collections	
Databases	Scopus, Web of Science, Google Scholar	
Geographical focus	Selected African countries	

Table 3.2 Selection criteria. Source: author.

The search query returned publications where the defined key terms appeared in titles and abstracts or as keywords. The main key terms were 'climate change', terms associated with 'leadership' and 'empowerment', and names of the online questionnaire respondents' countries. It is worth mentioning that, in this article, the terms 'gender' and 'women' were used as synonyms.

The set of the Google Scholar publications was retrieved using the Publish or Perish software program (Harzing, 2007) based on the appearance of indicated keywords similar to those used for the Scopus and WoS search. The full search strings included the following key terms, attributes, and Boolean operators:

Web of Science:

TS = ((wom*n OR gender) AND ("climat* chang*" OR "climat* adapt*" OR "climat* mitigat*") AND (leader* OR empower* OR "decision-making" OR negotiat*) AND (afric* OR tanzan* OR niger* OR kenya OR cameroon OR gambia OR eswatini OR congo OR uganda OR ethiopia OR tunis*)) Data range: 1 January 2015 to 26 January 2022.

Scopus:

TITLE-ABS-KEY ((wom*n OR gender) AND ("climat* chang*" OR "climat* adapt*" OR "climat* mitigat*") AND (leader* OR empower* OR "decision-making" OR negotiat*) AND (afric* OR tanzan* OR niger* OR kenya OR cameroon OR gambia OR eswatini OR congo OR uganda OR ethiopia OR tunis*)) AND (PUBYEAR > 2014)

Google Scholar:

Keywords: (women OR gender) AND ("climate change" OR "climat* adaptation" OR "climate mitigation") AND (leader OR empowerment OR "decision-making" OR negotiation) AND (africa OR tanzania OR niger* OR kenya OR cameroon OR gambia OR eswatini OR congo OR uganda OR ethiopia OR tunisia) Years: 2015 to 2022; Other options: exclude citations; exclude patents.

Both the joined Scopus/WoS and Google Scholar (GS) datasets were used as the input data for co-occurrence analysis. This was based on keywords and terms that appeared in the titles and abstracts of the publications included in each dataset (van Eck and Waltman, 2014). Titles, keywords, and abstracts were considered to effectively describe and reflect papers' contents (Springer, 2020). This type of

analysis is a favourable tool to identify prominent topic clusters. The VOSviewer software (van Eck and Waltman, 2010) was used to conduct the analysis and visualise its results in the form of network maps. Each node in a network represents a term or keyword. The size of the node indicates the occurrence of the term/keyword; the larger the node, the higher the occurrence of the term/keyword. All nodes are divided into thematic clusters, which are differentiated by colours.

3.5.2 Online Questionnaire

The online questionnaire was developed to collect the opinions of experts who are engaged in climate-change-related field(s) in African countries on the advances made to attain gender-balanced leadership/empowerment in climate change adaptation and mitigation.

The set of 40 questions was characterised by several different types of question, including closed and open-ended questions. The Likert scale was used to assess respondents' agreement/disagreement with a particular situation, state, or achievement. A relatively large number of open questions was included to provide respondents with more opportunities to share their experiences and comments, rather than limiting their responses to already provided specific answers. This questionnaire structure was chosen to reduce authors' biases, which might affect the range and direction of close answers. Consequently, it was expected that this may result in a lower-than-usual number of questionnaire respondents. The responses were collected anonymously, and respondents were asked for their consent. The questions referred to the period of the last five years.

Thematically, the questions were divided into several blocks: (i) questions on respondents' characteristics (e.g., country, experience and engagement in climate change related field(s), etc.), (ii) questions focused on women's/men's situation regarding climate change in their country, (iii) questions focused on advances made in the field over the last five years, and (iv) questions asking the respondents to share their comments on possible solutions.

The initial set of questions was reviewed by external experts from related fields. The latest version of the questionnaire was adjusted based on their comments and recommendations. The MS Excel software was used to conduct the statistical analysis and visualise the obtained findings. The invitation to participate in the questionnaire was distributed via a number of thematic expert networks and mailing lists. The voluntary nature of participation reduced the number of respondents to those who expressed their interest in sharing their opinion and views on the matter based on their experience. The participants were not necessarily gender experts, nor did they all participate in gender mainstreaming activities. The type of expertise and geographical diversity of the respondents were not predefined, and the instrument was open to experts engaged in climate change related fields from any African country.

3.6 Results and Discussion

3.6.1 Bibliometric Analysis

The results of the bibliometric analysis based on the joined (Scopus/WoS) dataset of 137 publications and the GS dataset of 995 publications outlined the following characteristics of the overall thematic structure of the climate change–gender domain in leadership/empowerment in the selected African countries. The large difference in

the number of retrieved publications demonstrated a significant prevalence of the grey literature. This leads to the conclusion that topics relevant to this domain are of more interest among civil society and international development organisations compared to the scientific community. It might also imply that the respective literature mainly discusses the applied rather than the theoretical aspects.

The co-occurrence analysis demonstrated that, thematically, the domain is divided into five major clusters based on keywords (Figure 3.3(a,b)), and into four (Scopus/WoS dataset) and six (GS dataset) clusters based on terms (Figure 3.3(c,d)). The maps of terms are more comprehensive and complement the co-occurrence maps of keywords. By analysing all clusters, the thematic structure of the domain could be outlined as follows. The key topics discussed that pertained to climate changegender in leadership/empowerment were those related to agriculture, food security, vulnerability, adaptation mechanisms, resilience, gender gaps and differences. The prominence of 'agriculture' and related issues is apparent, due to the multiple threats that climate change poses to the sector, including threats to wellbeing, livelihood and food security (Campbell et al., 2016; Mbow et al., 2019; Zakari et al., 2022; Pickson and Boateng, 2022). Furthermore, the rural population, which comprises, on average, about 52% of the total population in Africa (TheGlobalEconomy.com, 2022) and mainly relies on agricultural activities, is becoming more vulnerable, particularly women who are poorer and have less access to, e.g., financial and natural resources, knowledge and information, limited or no land ownership rights, and lower adaptive and coping capacities (Mekonnen, 2022; Adzawla et al., 2020; Afriyie et al., 2018). On the other hand, the effectiveness of adaptation mechanisms depends not only on their gender sensitivity but also on women's inclusion in decision-making processes and their presence in leadership positions on relevant boards and bodies, due to the significant positive contributions that women make to climate change solutions (e.g., Least Developed Countries Expert Group, 2015; Cook et al., 2019; Mavisakalyan and Tarverdi, 2016; Grillos, 2018; Huyer, 2016).






Figure 3.3 (a) Co-occurrence map of keywords (Scopus/WoS dataset); (b) Co-occurrence map of keywords (GS dataset); (c) Co-occurrence map of terms (Scopus/WoS dataset); (d) Co-occurrence map of terms (GS dataset). Created with VOSviewer. Source: author.

In terms of geographical locations, only certain countries (e.g., Ethiopia, Kenya, Cameroon) were included in the clusters. These results might imply that the climate change–gender in leadership/empowerment theme received unequal attention among researchers across different countries in the African region. Furthermore, it could also be assumed that the number of relevant development programs and initiatives implemented by civil society and international organizations varies geographically, since these activities are usually accompanied by a set of publications (e.g., reports, briefs, letters, etc.) and are added at least to the GS database.

It is worth mentioning that 'COVID-19' and the associated terms/keywords were not included in any cluster, meaning that the number of works in this domain with a connection to the COVID-19 pandemic is still significantly low, even though there are works on climate change–gender without connection to leadership/empowerment (e.g., Akrofi et al., 2021; Nyahunda et al., 2021; Belsey-Priebe et al., 2021).

3.6.2 Online Questionnaire

3.6.2.1 Respondents' Characteristics

Thirty experts from South Africa, Tunisia, Uganda, Nigeria, Niger, Cameroon, DR Congo, Kenya, Gambia, Tanzania, Ethiopia, and Swaziland participated in the online questionnaire (Figure 3.4).



Figure 3.4 Countries of the respondents participated in the online questionnaire (created with Datawrapper). Source: author.

The respondents were representatives of academia, business, international, governmental, and non-profit organisations, working in one or more of the following fields: climate change assessment, climate change impacts, climate projections, climate change and conservation, climate change mitigation, climate change adaptation and resilience, climate education and climate change governance. Their international, regional, national or community working experience in the field(s) ranged from less than five years (7% of the respondents) to more than five (50% of the respondents), ten, and more (43% of the respondents) years.

3.6.2.2 Women's Leadership/Empowerment in Climate Change

The dynamics between the economic sectors and climate change is characterised, among others, by the gendered support received by individuals engaged in the sector. The respondents were asked to compare and indicate sectors in which women are better-supported than men and vice versa. Based on their assessment, women are supported best in agriculture, followed by water and waste management (Figure 3.5). The attention paid to the agricultural sector is explained by its large share in total employment, i.e., 43.8% as of 2020 (OECD, 2021). Being on the frontline of the sector, women's participation in the labour force varies, on average, between 40 and 43% depending on the country, crop, and type of activity (Doss, 2014; Palacios-Lopez et al., 2017). Agriculture is characterised by gender imbalances in decision-making power and leadership, as well as by gaps in adaptive capacities to climate change (Mnimbo et al., 2016).



Figure 3.5 Sectors with better support. Source: author.

It is worth mentioning that the respondents named renewable energy, transportation, oil, gas, and mining as the sectors in which men receive better support than women (Figure 3.5). This opinion might be expected regarding oil, gas and mining, since these are often considered male-dominated industries (Catalyst, 2016; Kansake et al., 2021). However, renewable energy has often been positioned as an industry that provides a substantial opportunity for women's empowerment through their inclusion in decision-making processes and the energy-value chain because the continent has significant potential for clean energy (UN Women et al., 2015). Since the countries are challenging to design and implement gender-sensitive climate programmes and policies with reference to the energy sector (UN Women et al., 2015), it might be assumed that the support provided for women in renewables has low visibility or is still not sufficient enough to be noted by the outer audience.

When defining the characteristics of women's leadership in climate change, the majority of respondents indicated factors such as willingness to reduce climate change impact and the vulnerability of different gender groups, the effective, practical implementation of gender-related climate decisions, sustainable economic growth and development (Figure 3.6). These responses come along with studies that posture women leaders as contributors to more rigorous climate action and resilience, successful adaptation and mitigation strategies, and innovative and sustainable solutions (Women4ClimateAction, 2019; Ndiritu et al., 2014; Bob and Babugura, 2014). They also take the lead in advocating for behavioral changes to reduce climate change impacts and improve the wellbeing of disadvantaged and marginalized groups (Women4ClimateAction, 2019; Ndiritu et al., 2014; Bob and Babugura, 2014).



Figure 3.6 Characteristics of women's leadership in climate change. Source: author.

It is worth mentioning that, despite the widespread notion of male domination in decision-making power and their overrepresentation in leadership positions (FAO and ARC, 2021; UNFCCC, 2019; 2021), only a few respondents saw strong resistance from male counterparts as a characteristic of women's leadership. Several assumptions could be made based on these results. Firstly, if this type of resistance does exist, it is hardly identified or is not visible, particularly to those whose climate-change-related working area is not directly interconnected with 'gender/women' issues. Secondly, the significance of other factors affecting women's leadership in climate change is so large that it eliminates even the need for such resistance to develop. Among these factors are the economic and social gender inequalities that exist in societies that lock women's leadership potential away. The majority of these inequalities refer to legal rights, access to education, resources, power, opportunities to build social capital, etc. (e.g., Women4ClimateAction, 2019; UNDP, 2021).

When assessing the change in women's and men's roles in climate action over the investigated period (Figure 3.7), the respondents indicated a range from no degree at all to a very significant degree for both roles. Women's roles have changed to a certain degree, while men's roles have changed to small, significant and very significant degrees. These changes might reflect the measures taken regarding the countries' international and national gender-related obligations in climate change. It is worth mentioning that even if the assessment based on the respondents' experience and observations is subjective, in some cases, this might paint a better picture of the actual and visible outcomes of the progress that has been made on this matter than available, statistical, gender-disaggregated data.





The change in women's roles is associated among others with advances in their empowerment and engagement in climate change related processes.

When identifying the areas in which women have become more empowered, the majority of respondents named climate change adaptation, resilience, mitigation, impacts, and conservation (Figure 3.8). Considering the extent of African women's vulnerability to climate change stressors (Awiti, 2022), the named areas represent key targets for relevant measures to be applied. The obtained results are not only expected to significantly reduce vulnerability and improve livelihood, but also to have positive impacts on other climate change and gender-related areas.



Figure 3.8 Areas where women's empowerment has most increased. Source: author.

During the past five years, the advances in women's engagement in climate change related processes have been associated with their larger involvement in gender-based analysis, decision-making, solutions and policy development and research (Figure 3.9). The predominance of the first two processes reflects the greater attention given to, and effectiveness of, measures to include the gender variable in climate change. The gender-based analysis helps to identify inequalities, provides

information on women's and men's different roles, and considers policies and legislation in terms of gender outcomes or potential differential impacts (International Training Centre of the International Labour Organisation, 2009). Respondents listed adaptation, mitigation, emission and disaster risk reduction, as well as financial resources and education, as areas that require further intervention due to the still-significant unequal gender representation in leadership positions and poor levels of women's empowerment.



Figure 3.9 The extent of women's engagement in climate-change-related processes. Source: author.

However, the increase in women's participation in climate change related processes does not always directly correlate with their influence on decision-making. Often, women's engagement signifies only their presence, without their opinions being taken into account by male colleagues. The respondents assessed the progress in women's influence, and rating it from no degree at all to a very significant degree, with the majority of responses referring to a small or certain degree (Figure 3.10). They also noted the change in women's ability to overcome the impacts of climate change, which is unequivocally interconnected with their growing influence (Figure 3.10). To ensure that women's inclusion in decision-making attains its objectives, it is necessary to assign the smallest weight to quantitative indicators of their presence when assessing the effectiveness of the respective measures. Undoubtedly, this will significantly increase the complexity of the process, but diminish the formal nature of the 'gender' requirement.



Figure 3.10 Progress in women's influence on climate-change-related decisions and improvements in their ability to overcome the impacts of climate change. Source: author.

Indicating potential hindrances to women's representation in climate-related leadership positions, the majority of respondents agreed on the lack of factors such as women's confidence in their ability to participate, social and professional experience, accumulated career capital, networks and contacts, as well as referencing women's low socio-economic status (Table 3.3). These factors are not specific to the climate change domain. Some of these barriers are also indicated in studies on women's potential and abilities to attain leadership positions in other areas, for instance, academia, business sectors, health care, and public sectors (Kalaitzi et al., 2017; Newman et al., 2017; Begashaw Abate and Terefe Woldie, 2022). However, the respondents do not consider the women's lack of leadership ambitions or a negative attitude towards this type of activity as being among these barriers (Table 3.3).

	Strongly Agree	Agree	Neutral	Disagree	Strongly Disagree
Lack of women's leadership ambitions					
Lack of women's confidence in their ability to participate					
Lack of knowledge and expertise					

Table 3.3 Barriers to women's representation in climate-related leadership positions. Source: author.



Assessing the overall outcomes of changes in women's and men's roles in climate action, the majority of respondents indicated their positive rather than negative character (Table 3.4). More than half of respondents also suggested a positive correlation between the number of women in climate-related leadership positions and the number of national gender-responsive climate policies, plans and programmes. These results follow earlier studies on the link between the number of women in leadership positions and the introduction of climate-related measures (Mavisakalyan and Tarverdi, 2019; Norgaard and York, 2005), supported by the works that indicate women's essential contribution to climate action (Mekonnen, 2022; Habib et al., 2022; Loarne-Lemaire et al., 2021). However, there are no works focusing solely on the change in men's role in climate action and its outcomes. Therefore, the specific characteristics of the 'new' roles of both women and men, and scale and effectiveness assessments of their positive and negative outcomes, require further, more rigorous investigation that lies beyond the scope of this work.

	Strongly Agree	Agree	Neutral	Disagree	Strongly Disagree
A change in women's role in climate action over the last five years, if any, has led to positive outcomes					
A change in men's role in climate action over the last five years, if any, has led to positive outcomes					
A change in women's role in climate action over the last five years, if any, has led to negative outcomes					
A change in men's role in climate action over the last five years, if any, has led to negative outcomes					

Table 3.4 Outcomes of the change in women's and men's roles in climate action. Source: author.

To consolidate and strengthen the positive results that have already been achieved in the reduction of gender imbalances in climate-related leadership positions, the factors perceived as hindrances should continue to be addressed. Among the possible measures, the respondents suggested improving climate change knowledge-sharing, additional training and the development of capacity-building, the introduction of a gender quota system and a reduction in corruption in recruitment processes. However, they expressed concern that the prioritisation of gender-based over meritbased opportunities might lead to negative consequences, including the amplification of stereotypes in case of women's failure.

Other issues that were raised included the need to guide particularly male leaders on ways to include women in climate action and demonstrate how their behaviour could affect women's engagement. Often, climate change–gender studies are women-centric (Bunce and Ford, 2015), discussing their vulnerabilities, empowerment and leadership, adaptation and mitigation. There is a very little investigation on men's attitudes, perceptions and opinions with respect to female leadership, empowerment and their roles as potential agents of change for climate change solutions.

It is worth mentioning that, even after overcoming the barriers to leadership positions, women continue to face a range of challenges. In this respect, the respondents outlined factors including insufficient education, training, skills, expertise, or experience, lack of political will and governmental support, and weak institutional linkages. It is important to note that some of these challenges are similar to the aforementioned causes of women's under-representation in leadership positions. Other factors include cultural and societal barriers and stereotypes, lack of or insufficient financial resources, and the corruption and non-transparency of selection processes.

The development, introduction and implementation of measures to overcome these challenges require the active involvement of different actors. The respondents outlined the role of national governments and experienced civil servants in encouraging women's advancement to leadership positions. However, the international, national and community efforts that have been taken to advance gender-equal participation in climate action and representation in climate-related positions in recent years were mainly assessed as being moderate (Figure 3.11). These results are in line with the global concern over the slow pace of progress towards reductions of gender imbalances in leadership and empowerment in climate change (UNFCCC, 2021).



Figure 3.11 Efforts to advance gender-equal participation and representation in climate action and leadership positions. Source: author.

Assessing the potential timeframe to achieve gender-equal participation and representation in climate leadership in their countries, the respondents' opinions varied between less than 5 years and more than 10 years, with the latter slightly prevailing. Several respondents were even more pessimistic, indicating that gender-equal representation will never be achieved (Figure 3.12). At international level, the UNFCCC estimated that the gender-balanced representation of delegates will be achieved in 24 years and of COP Heads of Delegation in 46 years following the current rate of change (UNFCCC WEDO, 2020). At national level, considering the recent global emergencies and the state of the global economy, it is very difficult to project the amount of time that will be required by the countries to achieve gender-balanced participation and leadership in their climate action.





One such global emergency is COVID-19 and its impacts. The pandemic unveiled and further amplified existing climate change vulnerabilities, including those related to gender inequalities (Nyahunda et al., 2021; Ingutia, 2021; Patrick et al., 2021). Furthermore, there has been a concern that the pandemic might undermine actions taken towards climate change mitigation and reduction of gender imbalances (OECD, 2020a; Women for Climate-Resilient Societies, 2020; Reilly et al., 2021). However, some works claim that the recovery plans could be seen as a strategic opportunity to use the lessons learned and additionally stimulate the climate agenda and its objectives (Newell and Dale, 2020; Nguyen et al., 2021). The gender-related measures taken in response to COVID-19 could also be applied to address the gender-related impacts of climate change (Akrofi et al., 2021; Women for Climate-Resilient Societies, 2020). The respondents did not come to an agreement on whether the pandemic's impacts on advancing gender-equal representation in climate-related leadership positions and women's empowerment in climate action were positive or negative. Their opinion ranged from a minor negative to a very substantial positive impact (Figure 3.13).



Figure 3.13 Impact of the COVID-19 pandemic. Source: author.

Considering the time required to develop, introduce and implement COVIDadjusted measures regarding gender-balanced leadership and women's empowerment in climate action, it is challenging to predict their actual outcome, particularly at this stage. In addition to time, the outcome will be shaped by the gender sensitivity of the planned recovery schemes, established interconnections between the pandemic and climate change, defined priority areas, and implementation strategies.

3.6.3 Final Remarks

Although the results of the bibliometric analysis and the online questionnaire are not fully comparable, the following can be inferred. In addition to the issues that received attention from researchers and respondents, some points were indicated by the latter that were not prominent on the co-occurrence maps. This might demonstrate that these topics have received limited attention from scholars or development organizations to date. Although the raised points were based on the respondents' experience and often subjective opinion, they might reflect another perspective on the climate change-gender domain in leadership/empowerment, from those facing the outcomes of these measures. In addition, there were 'not included' topics, such as, for instance, 'COVID-19 pandemic', 'role models' or 'quota systems', which have already been discussed with reference to climate change, gender, and leadership as standalone topics (OECD, 2020a; Women for Climate-Resilient Societies, 2020; Reilly et al., 2021; Dimitrova-Grajzl and Obasanjo, 2019; Bosha, 2014). Their lack of prominence on the maps demonstrates the need for more rigorous investigation with connection to climate change-gender in leadership/empowerment in African countries.

Even though the obtained findings are not directly connected to enterprises, several assumptions can be made. It could be implied that companies of any background will have to comply with stronger climate-change-related binding policies and regulations due to a positive trend in advances in women's leadership and empowerment regarding climate change. In addition, the gender gap will decrease on more executive boards and the number of women executives will reach the minimum number indicated by some studies as being necessary to integrate more climate-friendly activities and improve climate/environmental performance (Nuber and Velte, 2021; Caby et al., 2022).

Overall, the results of this study outline the thematic structure of the climate change–gender research domain, provide insights into the outcomes of the advances made towards reduction of gender imbalances in climate change leadership in African countries, and elucidate the areas and directions that might need to be addressed in more depth.

3.7 Conclusions

Gender-balanced representation is perceived as one of the integral characteristics of effective leadership and empowerment, which are required to overcome and tackle complex challenges such as climate change and its impacts. Being recognized as one of the most vulnerable global areas (IPCC, 2023), the African continent will continue to face climate change impacts that affect its socio-economic development, increase inequalities, and exacerbate threats to human health, food and water security. On the other hand, African countries are increasingly contributing to the global climate change agenda, for example by taking on commitments in the frame of the Paris Agreement. In this context, it is essential to ensure that the countries will also continue to amplify their efforts to reduce gender imbalances in leadership and empowerment. Therefore, the visibility of the already-attained outcomes, in terms of both gender and climate change experts, demonstrates their effectiveness, the potential benefits and risk reductions, which are key factors supporting the importance of, and need for, further measures.

The paper investigates the need and potential for gender-balanced leadership/empowerment in climate change adaptation and mitigation based on the views and perspectives of climate change experts regarding the advances made in African countries over the last five years. The study is complemented by a bibliometric analysis (i.e., co-occurrence analysis) of the literature published on the topic between the years 2015 and 2022, which demonstrates the considerable prevalence of grey literature compared to peer-reviewed works.

Although the results from the online questionnaire and the bibliometric analysis are not fully comparable, some of the points raised by the respondents are reflected on the maps. For instance, the questionnaire findings revealed agriculture, the topic with the high number of co-occurrences on the maps, as one of the sectors where women are better supported. This type of support is also accompanied by a high number of studies (publications), development projects and initiatives on vulnerability, adaptation, climate-smart agriculture, etc. However the co-occurrence maps do not demonstrate the prominence of such topics as 'COVID-19 pandemic', 'role models' or 'quota systems', although the respondents saw the latter as being among the potential solutions, and perceived that the pandemic has both positive and negative impacts on reducing gender imbalances in leadership/empowerment.

Additionally, the questionnaire findings indicated renewable energy, transportation, oil and gas, and mining as sectors where men are better supported. The study suggests that women's influence on climate change related decisions, as well as their ability to overcome its impacts, have changed, but mainly to a small or certain degree. The lack of knowledge, expertise, skills and qualities, political will and female leader role models were indicated as being among the hindrances to women's representation in climate-related leadership positions. Overcoming these barriers was linked to better climate change knowledge-sharing, additional training

and capacity building, skills development, the introduction of a gender quota system and a reduction in corruption in recruitment processes. The importance of providing merit-based and not only gender-based opportunities was highlighted, as the latter might amplify some stereotypes regarding women if they failed to succeed. Assessing the potential timeframe needed to achieve gender-equal representation in climate change leadership in their home countries, the respondents' opinions ranged from very optimistic (less than five years) to very pessimistic (never).

The current study has several limitations. The Google Scholar (GS) dataset used for the bibliometric analysis was limited to a maximum of one thousand records due to software features. Only English-language publications were retrieved and analysed, which might alter some of the obtained results. The structure of the questionnaire and the types of questions affected the number of respondents and countries' representation.

Nevertheless, despite the aforementioned limitations, the findings demonstrate the visibility of the measures that have already been taken to those who are not directly engaged in the gender/women field but undoubtedly experience its outcomes. Their opinion provides a better understanding of the actual progress and its implications, particularly those that are not quantifiable. This study can be used as a premise when identifying possible directions of further actions towards genderbalanced leadership/empowerment in climate change, supporting their development, types and areas to be applied in the adaptation and mitigation context in African countries.

4. Central Asia: Exploring Insights on Gender Considerations in Climate Change

This chapter was published as Kovaleva, M. et al. (2023) Sustainability, DOI:10.3390/su151612667

4.1 Abstract

More than three decades of independence of Central Asia (CA) countries have been marked by socio-economic, political, and legal reforms. Growing climate change impacts threaten the wellbeing and livelihood of the already vulnerable local population, more than half of which comprises women. In this context, it is essential to adequately include both women's and men's needs in multiple efforts taken by national governments to overcome climate change challenges. This paper explores how gender/women considerations have been already addressed in climate change in CA using bibliometric analysis, an expert-driven assessment approach, and a comprehensive analysis of thematically relevant development projects. The findings demonstrated a significant prevalence of grey literature implying much lower interest from academia. The experts outlined the importance of women's participation and consideration of their experiences, which were different from men's, in climate change decision making. The comprehensive analysis of the selected development projects revealed the leadership/empowerment domain of climate change-gender interconnections as being mostly addressed. The implications of this paper regarding the current knowledge on the topic related to the CA region are threefold. Firstly, it highlights a strong need for further scientific research that could be implemented through international research initiatives and national institutional programs. Secondly, it calls for increased input from women representation in climate action at all levels. Thirdly, it outlines areas that require stronger cooperation with international donors to mainstream gender/women considerations among a wide range of stakeholders engaged in climate change and its related fields. Overall, the paper lays a basis for further steps towards advancing gender-sensitive and responsive approaches in CA, particularly in those climate-change-related areas that are often perceived as being gender-neutral.

4.2 Introduction: Gender in the Climate Change Context

The importance of considering gender in all facets of climate action has been continuously growing, along with the acknowledgment of disproportional vulnerability, unequal empowerment and leadership, and potential benefits from a reduction of these types of imbalances. Vulnerability, leadership/empowerment, and benefits represent key domains of climate change–gender interconnections (Kovaleva et al., 2022). The vulnerability to climate change of different gender groups is shaped by numerous socio-economic, political, and cultural factors existing in every community. Very often, women, compared to men, are less resilient, with less capacity to overcome challenges of food, energy, and water insecurity, as well as to cope with disaster events and their aftermath. Empirical findings from various geographical locations provide evidence of differential gender vulnerabilities due to unequal or lack of access to natural, financial, and information resources, decision-making power, and limited property (e.g., land) ownership rights and paid employment opportunities (Collantes et al., 2018; Yadav and Lal, 2018; Jerin et al., 2023; Botreau and Cohen, 2020; Assaduzzaman et al., 2023). On the other hand, men

experience stronger climate change impacts on their mental health and physical health, and on their abilities to perform "traditional" roles such as providing income and securing the livelihood of their family (Alston and Kent, 2008; Babugura, 2010; WHO, 2014). Studies report that men farmers, for instance, are found to be more vulnerable to higher temperatures and changes in rainfall patterns (Bessah et al., 2021). Therefore, neglecting men's and women's differences in adaptation needs can not only exacerbate inequalities but also diminish women's potential as agents of change in the economic sectors that are sensitive to climate change, such as agriculture, energy, and water, as well as disaster risk reduction (Botreau and Cohen, 2020; Lau et al., 2021; Ampaire et al., 2020; Tobi et al., 2023).

Unbalanced empowerment between men and women, their representation in leadership positions, and participation in decision-making processes reduce the effectiveness of adaptation and mitigation measures and amplify existing vulnerabilities (Asongu et al., 2021; Glazebrook et al., 2020). Recent estimates demonstrate that men still dominate in various sectors despite wider acknowledgment of women's positive role in contributing to climate action and calls to broaden their integration in climate policy and decision making (UNFCCC Secretariat, 2020; 2022; Mavisakalyan and Tarverdi, 2019; Huyer et al., 2020). For instance, in 2022, in the EU-28 region, women, on average, occupied 44% of the senior administration positions in national ministries and 27% as members of the upper decision-making bodies in European agencies dealing with the environment and climate change (European Institute for Gender Equality, 2023). In the same year, the overall men/women ratio in UNFCCC boards and bodies varied between 0.2 and 9. The number of women members was greater in only 3 out of 17 divisions. The share of men in the national Party delegations at the Conference of Parties COP 27 exceeded that of women by almost two times, implying the lack of significant progress in achieving gender-balanced representation (WEDO, 2023). Globally, climate-sensitive economic sectors have been also characterized by underrepresentation of women in leadership positions, particularly in energy and agriculture, where they occupy 20% and 23%, respectively (World Economic Forum, 2022).

However, numerous steps have been taken to promote a wider inclusion of gender considerations in the climate change context worldwide across different sectors (e.g., Ampaire et al., 2020; Howland et al., 2019; UNDP, 2022), consequently creating additional benefits for adaptation and mitigation measures, poverty reduction, resilience, and capacity building (Patel et al., 2020; UN Women, 2016). For instance, highly qualified women, members of board committees, who are actively involved in companies' governance, undoubtedly affect their voluntary climate change disclosure (Ararat and Sayedy, 2019). There is also a positive correlation between the percentage of women occupying managerial positions and the reduction of carbon emissions in a company (Altunbas et al., 2022). Furthermore, more genderbalanced access to agricultural land and ownership rights decreases land degradation, increases land productivity, and triggers sustainable land use practices and efficient water technologies (Collantes et al., 2018; UNDP, 2018).

This study examines how gender/women considerations in the climate change context in Central Asia are reflected in the published literature, viewed by experts engaged in climate action, and supported by international donors through the implementation of thematically relevant development projects. The subject is of particular importance in the light of (i) continuously growing pressure from a changing climate on the livelihoods and wellbeing of the CA population; (ii) national efforts towards broader inclusion of gender considerations in adaptation and mitigation processes; and (iii) strengthening of the role of women as agents of change. Therefore, it is necessary to understand developments that have been already attained to lay a basis for further steps towards advancing gender-sensitive and - responsive approaches, particularly in those climate-change-related areas that are often perceived as being gender-neutral.

The paper is organized in the following way. The next section discusses the narrative of climate change and gender/women in the Central Asian countries, as well as international donors' assistance. It is followed by the Methods section, which describes instruments used to collect and analyze the data. The Results and Discussion section describes and discusses the obtained findings. The paper concludes by presenting some implications, recommendations, and limitations of the study.

4.3 Central Asia: Climate Change, Gender/Women, and International Donors

4.3.1 Climate change

Kazakhstan, Kyrgyzstan, Tajikistan, Turkmenistan, and Uzbekistan, five countries that comprise the Central Asia region, have been facing a large variety of challenges, reforms, and transformations in political and socio-economic life during their more than three decades of independence. Furthermore, the region, where between 42% and 72% of the population resides in rural areas (The World Bank, 2021a), heavily relies on natural resources and is mainly engaged in agricultural activities (Linn, 2012; Pradhan, 2022), and constantly experiences growing climate change impacts, including alterations in precipitation patterns, increasing aridity, and seasonal climatic shifts (Lioubimtseva, 2015; Stucker et al., 2012; IPCC, 2022; Hu et al., 2014; Reyer et al., 2017b; Hu and Han, 2022). Recent studies report increasing temperature trends across the region that lead to a rapid rate of shrinking of glacier areas and temporary alterations in groundwater and lake water levels (Haag et al., 2019; De Beurs et al., 2018). These changes amplify the magnitude and frequency of most types of natural disasters that CA is exposed to, including floods, landslides, extreme temperatures, and droughts (IPCC, 2022; De Beurs et al., 2018; GFDRR, 2020).

Additionally, the vulnerability of the local communities is aggravated by poverty, intensive outmigration, and relatively low levels of coping capacity (Xenarios et al., 2019; The World Bank Group and The Asian Development Bank, 2021a; 2021b; 2021c; 2021d; 2021e). The countries rank between 39th (Kazakhstan) and 132nd (Turkmenistan) among the 182 countries on the 2022 ND-GAIN (Notre Dame Global Adaptation Initiative) Index, which assesses their vulnerability to climate change and other global threats, together with their ability to increase resilience (University of Notre Dame, 2022). A changing climate also affects key economic sectors that already face consequences of ineffective environmental management implemented during the Soviet Union period (e.g., Howland et al., 2019; Qushimov et al., 2007; The World Bank, 2019). For instance, the agriculture sector is projected to absorb between USD 1.6 million (Tajikistan) and USD 50 million (Kazakhstan) of climate-change-induced economic losses by 2040 (Kunwar, 2020). Currently, it contributes between 5% and 25% to the national GDPs (Agency for Strategic Planning and Reforms of the Republic of Kazakhstan Bureau of National Statistics, 2023; The State Committee of the Republic of Uzbekistan on Statistics, 2023) and employs between 15% and 45% of the total labor force, of which between 13% and 60% are women (The World Bank, 2021a).

Overall, the national governments' efforts regarding climate change include the implementation of obligations and commitments in the frame of the ratified international climate initiatives and agreements, including the United Nations Framework Convention on Climate Change (UNFCCC), the Kyoto Protocol, and the Paris Agreement (United Nations Treaty Collection, 2021a; 2021b; 2021c), and the introduction of numerous short- and long-term development and climate strategies, e.g., the Strategy Kazakhstan-2050, the National Climate Change Strategy of Turkmenistan, the National Strategy for Climate Adaptation until 2030 of Tajikistan, the National development strategy of the Kyrgyz Republic for 2018–2040, and the "Concept of Environmental Protection of the Republic of Uzbekistan until 2030" ("Strategy Kazakhstan-2050", 2014; National Climate Change Strategy of Turkmenistan, 2012; Government of Tajikistan, 2019; Government of the Kyrgyz Republic, 2018; Decree by the President of the Republic of Uzbekistan on Approval of the "Concept of Environmental Protection of the Republic of Uzbekistan until 2030", 2019). However, the effective implementation of climate-change-related measures is hampered by various factors, including countries' economic situations, weak governance, insufficient regional integration, and external stressors (e.g., COVID-19, military conflicts). Furthermore, it remains challenging to include adaptation and mitigation steps in long-term sectoral development programs and strategies (Liu W. et al., 2020).

4.3.2 Gender/Women

In Central Asia, women comprise a larger share of the population (Table 4.1). Between 36% (Kazakhstan) and 70% (Tajikistan) of women reside in rural areas (Agency for Strategic Planning and Reforms of the Republic of Kazakhstan Bureau of National Statistics, 2023; Agency on Statistics under the President of the Republic of Tajikistan, 2021). Women are mainly employed in health care, service and accommodation, education, and real estate sectors, which often are considered as women-dominated sectors, while men are overrepresented in mining and quarrying, transportation, construction, and public administration (National Statistical Committee of the Kyrgyz Republic, 2021; Agency for Strategic Planning and Reforms of the Republic of Kazakhstan Bureau of National Statistics, 2022a; The State Committee of the Republic of Uzbekistan on Statistics, 2022; Agency on Statistics under President of the Republic of Tajikistan, 2020). The agriculture sector, previously the main employer in most of the CA countries, currently employs between 13% (Kazakhstan) and 60% (Tajikistan) of women out of the total labor force (The World Bank, 2021a; Mogilevskii, 2020; The World Bank, 2021b).

Table 4.1 Central Asia countries' characteristics. Source: Authors compilation based on data from the World Bank, UNDP, national official websites, and statistical offices.

Countries	Share of Women Population	Labor Force Participation Rate, Women out of the Total Employed Population	Share of Seats in Parliament Held by Women	Gender Gap Index Ranking out of 146 Countries, 2022 *
Kazakhstan	51.4% (2021)	55.2%	27.4% (Mazhilis) (2021)	65
Kyrgyzstan	50.4% (2022)	38%	17% (2021)	86

Tajikistan	50.8% (2021)	46.1%	23.8% (2021)	114
Turkmenistan	50.8% (2021)	45%	25% (2019)	
Uzbekistan	49.7% (2022)	41.3%	32% (Oliy Majlis) 25% (Senate) (2019)	

* Turkmenistan and Uzbekistan were not included in the ranking.

At the institutional level, women's and men's equal rights, freedoms, and opportunities are guaranteed and protected by national legislations, including the constitutions, specific laws, and countries' international obligations on the elimination of discrimination (The Constitution of the Kyrgyz Republic as Last Amended of 5 May 2021. Article 24; The Constitution of the Republic of Kazakhstan as Last Amended of 08.06.2022. Article 13; The Constitution of the Republic of Uzbekistan as Amended of 9 February 2021. Article 46; The Constitution of the Republic of Tajikistan as Last Amended of 22.05.2016. Article 17: The Constitution of Turkmenistan as Last Amended of 25 September 2020; The Law of the Republic of Kazakhstan on State Guarantees of Equal Rights and Equal Opportunities of Men and Women Dated 8 December 2009; The Law of Turkmenistan about the State Guarantees of Providing the Equal Rights and Equal Opportunities of Women and Men as Amended of the Law of Turkmenistan of 25 November 2017; The Law of the Republic of Uzbekistan "On Guarantees of Equal Rights and Opportunities for Women and Men" No. ZRU-562 as of 2 September 2019; The Law of the Republic of Tajikistan "On State Guarantees of Equality between Men and Women and Equal Opportunities for Their Implementation", Adopted on 11 February 2005; The Law of the Kyrgyz Republic on State Guarantees of Equal Rights and Opportunities for Men and Women as Amended of 14 July 2011; OHCHR, 2023). Furthermore, to support women's political empowerment and leadership, the governments introduced a national legislated quota system that regulates a minimal share of women candidates to be included in political party lists and as members of national parliaments. The current quota is set at 30% (Constitutional Law of the Kyrgyz Republic "On Elections of the President of the Kyrgyz Republic and Deputies of the Jogorku Kenesh of the Kyrgyz Republic". No 68 of 2 July 2011; Constitutional Law of the Republic of Kazakhstan "On Elections in the Republic of Kazakhstan". No 2464 of 28 September 1995; Election Code of the Republic of Uzbekistan of 26 June 2019). However, today, women occupy between 17% and 32% of the seats in national parliaments (Table 4.1). Studies also outline other socio-economic and cultural factors that shape gender gaps in various spheres. For instance, men are more often registered as agricultural landowners despite the lack of apparent genderrelated differences during the privatization period (The World Bank, 2009; Asian Development Bank, 2016a; Lastarria-Cornhiel and Garcia-Frias, 2005a). Differences in employment of women and men and their wage levels across sectors are linked to the outmigration of men or their leaving to better-paying sectors (Mogilevskii, 2020). The latest assessment of gender gaps, which passive in areas such as economic educational attainment, health and survival, and political participation, empowerment, ranked Kazakhstan 65th, Kyrgyzstan 86th, and Tajikistan 114th out of 146 countries, whereas Turkmenistan and Uzbekistan were not included in the list (World Economic Forum, 2022).

In the climate change context, women have been acknowledged as one of the most vulnerable groups whose needs are required to be given specific attention along

with the importance of their empowerment and inclusion in decision-making processes (e.g., Government of Tajikistan, 2019; Ministry of Water Resources of the Republic of Uzbekistan, 2020; Government of the Kyrgyz Republic, 2013). Furthermore, by being responsible for most food choices and related decisions in their homes, women could contribute to a reduction of carbon emissions and communities' resilience to climate change impacts (CAREC, 2020). They also could significantly contribute to the management of water resources, which are highly sensitive to a changing climate, by participating in the water user associations (WUAs). However, currently, these organizations are characterized by overrepresentation of men (Asian Development Bank, 2020). At the international level, women participate at UNFCCC meetings, where their share in national Party delegations has varied between 0% and 100% across different years and across the CA countries. For instance, in 2009, women accounted for, on average, 56% of delegates representing Kazakhstan, compared to 72% from Kyrgyzstan, 26% from Tajikistan, 77% from Turkmenistan, and 90% from Uzbekistan. On the other hand, in 2016, these values were 65%, 84%, 0%, 0%, and 75%, respectively (WEDO, 2023).

4.3.3 Donors

More than three decades of independence of the CA countries have been also characterized by the establishment of multilateral relations with numerous principal international organizations and institutions, including the United Nations (UN), Organization of Security and Co-operation in Europe (OSCE), European Union (EU), World Bank Group, and Asian Development Bank (ADB) etc. In the frame of this cooperation, the countries receive financial and expert assistance, in a wide spectrum of areas including but not limited to water resource management, agriculture, energy, environment protection, climate change, gender, and women (Dukhovny et al., 2015; Sehring et al., 2019; Rakhimov, 2010). Table 4.2 summarizes the official development assistance (ODA) received by the five countries between the years 2016 and 2020 to support environment protection, energy, agriculture, water supply and sanitation, women's rights organizations and movements, and government institutions. It is worth noting that the size of the latter is considerably smaller compared to others.

Table 4.2 Total net ODA disbursements received by the CA countries over the period						
2016–2020, USD million. Source: The authors' calculations based on the Query						
Wizard for International Development Statistics (QWIDS).						
					Women's Rights	

Countries	Environment Protection	Energy	Agriculture	Water supply and Sanitation	Women's Rights Organizations and Movements, and Government Institutions
Kazakhstan	30.7	38.6	11.9	2.1	0.23
Kyrgyzstan	24.4	159.9	38.6	75.2	3.7
Tajikistan	28.9	405.8	131.4	123.5	2.4
Turkmenistan	5.5	7.3	0.4		0.1
Uzbekistan	191.3	1129.5	659.3	386.2	1.1

To date, the ADB and the World Bank have committed to over a thousand commitments and projects having a total value of over USD 46 billion and focusing on, among other aspects, climate change, energy, water resources, agriculture, rural development, and environmental policies (Table 4.3) (Asian Development Bank, 2023; The World Bank, 2023). It is worth mentioning that the subsequent initiatives implemented within the frame of this financial assistance are subject to multiple factors, including national strategic priorities, socio-economic and political situations, and donors' requirements stipulated in respective bilateral/multilateral agreements.

Table 4.3 Number and size of the ADB and World Bank commitments by country; USD billions. Source: The authors, based on data from the Asian Development Bank and the World Bank 2023.

		Kazakhstan	Kyrgyzstan	Tajikistan	Turkmenistan	Uzbekistan
ld k	No. of projects*	60	131	118	7	75
Wor] Ban	Total project cost	10.76	3.87	2.24	0.12	8.67
DB	No. of Commitments ^{**}	196	127	146	13	228
A	Total size	2.4	6.0	2.2	0.632	10.4

* Includes active and closed projects only.

** Commitments include public sector loans, grants, and technical assistance.

4.4 Methods

The study deployed three methods to explore how gender/women considerations in the climate change context in the Central Asia countries are reflected in published peer-reviewed and grey literature, viewed by experts, and supported by international donors. The methods were (i) bibliometric analysis, (ii) an expert-driven assessment approach, and (iii) comprehensive analysis of climate-change-related development projects with reference to gender and/or women issues implemented in the region.

4.4.1 Bibliometric Analysis

In recent years, bibliometric analysis has been used by researchers as a tool for the evaluation of published scientific works and their trends. In this study, a term cooccurrence analysis was conducted to identify subjects with a connection to gender or women considerations in the climate change context in Central Asia, that are discussed in peer-reviewed and grey literature available in Scopus and Google Scholar (GS). The benefit of this method is in identifying relevant patterns in text data as well as getting insights on the interconnections between various concepts and terms (Radhakrishnan, 2017). Both Scopus and GS databases represent the largest collection of scientific and scholarly works published globally. Table 4.4 lists the selection criteria used to form the datasets. The search data range was limited to five CA countries: Kazakhstan, Kyrgyzstan, Tajikistan, Turkmenistan, and Uzbekistan, and the period between the years 2000 and 2023. Only works in English were selected. The full search strings included the following elements:

Scopus: TITLE-ABS-KEY (("clim* chang*" OR "clim* vulnerab*") AND ("gender" OR "wom*n") AND ("central asia*" OR "kyrgyz*" OR "kazakh*" OR "tajik*" OR "uzbek" OR "turkmen*")) AND (EXCLUDE (PUBYEAR, 1990)).

Google Scholar: Keywords: (gender OR women) AND ("climate change") AND ("central asia" OR kyrgyzstan OR kazakhstan OR uzbekistan OR tajikistan OR Turkmenistan); Publication year 2000–2022.

Criteria	Description		
Data range	2000–2023		
Language	English		
Type of publication	All types available in the collections		
Databases	Scopus, Google Scholar		
Geographical focus	Kazakhstan, Kyrgyzstan, Tajikistan, Turkmenistan, Uzbekistan		

Table 4.4 Selection criteria. Source: author.

The search and selection of literature from GS were undertaken using the Publish or Perish software program (Harzing, 2007). Both datasets were retrieved in February 2023 and included 13 (Scopus) and 982 (GS) publications characterized by such variables as "title" and "keywords" that effectively describe every publication (Springer, 2020).

The exclusion of duplicated and irrelevant works reduced the merged (Scopus/GS) dataset to 892 items, which served as input data for the term cooccurrence analysis. The results were visualized as a network map using the VOSviewer software (Van Eck and Waltman, 2010). Nodes on the map that represented terms or keywords were split into different thematic clusters, differentiated by color. The larger size of a node correlated with a higher cooccurrence of a term or keyword.

4.4.2 Expert-Driven Assessment Approach

The expert-driven assessment approach was used to reflect the opinions of those who are engaged in climate action and, consequently, in the measures taken towards the inclusion of gender/women considerations in climate change. The chosen approach is beneficial for this study because it provides an opportunity to gather and synthesize diverse views, particularly in the fields where data are limited or even not available, which may lead to severe challenges (Hemming et al., 2018). The approach is among the key elements used across various fields, including women roles and climate change adaptation and projections (Grainger et al., 2022; Oppenheimer et al., 2016; Agu et al., 2021).

In this work, an on-line survey instrument was used to elicit experts' opinion. The instrument comprised 45 open- and close-ended questions that were grouped into two categories: general and thematic, where the latter focused on three domains of climate change–gender interconnections: (i) vulnerability, (ii) benefits, and (iii) leadership/empowerment. The survey design, which was characterized by the prevalence of open-ended questions, was chosen due to a low number of similar studies investigating gender/women considerations in the climate change context in Central Asia. Open-ended questions provided respondents with more opportunities to share their own experiences and comments. However, it is worth mentioning that this type of survey design, as expected, resulted in a relatively low response rate. The initial set of questions was reviewed by a group of international experts working in thematically related fields. The final version of the survey was modified based on reviewers' comments and recommendations. The instrument was disseminated via

professional networks and emails. Participation was voluntary, and responses were collected in an anonymous mode.

4.4.3 Development Projects Analysis

The study was complemented by a comprehensive analysis of climate-change-related development projects with reference to gender/women. This method was one of the ways used to synthesize available information to uncover and demonstrate evidence on a meta- (Central Asia) level. The projects were and have been implemented in the CA countries between the years 2000 and 2023 in the frame of international assistance. These types of initiatives reflect, among other factors, donors' interest and willingness to support the integration of gender/women issues in the region. The projects were selected based on several criteria, including availability and accessibility of the relevant information to the general public, participation of a donor (e.g., international organization, governmental agency, or institution), and a reference of project objective(s) or outcome(s) to gender/women in climate change. The information was collected by means of extensive desktop research and summarized against a set of the following variables:

- project title;
- donor organization;
- country(-ies) of implementation;
- duration;
- short description;
- source of information.

Results of the comprehensive analysis of project characteristics including objectives, target audience, and outcomes were categorized according to three domains of climate change–gender interconnections: vulnerability, leadership/empowerment, and benefits (Kovaleva et al., 2022).

4.5 Results

4.5.1 Bibliometric Analysis

The analysis showed that 98% of the retrieved publications that comprised the merged dataset could be classified as grey literature. This mainly included reports, briefs, project findings, and guidelines. Thirteen peer-reviewed publications focused on a specific topic such as health, social factors, education, and energy and water resources with reference to climate change and gender. For instance, several studies investigated the impacts of weather shocks on child health (Freudenreich et al., 2022), extreme heat on the risk of preterm birth and stillbirth (McElroy, 2022), seasonality on anemia and eclampsia (Hlimi, 2015), and desertification in the Aral Sea Basin on the local communities' health (Orlovsky et al., 2001; Hill et al., 2013), and explored sexual and reproductive health and rights in the light of the post-2015 development goals, and the impact of climate change on the microelement status of the adult population (Batyrova, 2021). Another two publications discussed the relationship between social movements and the global carbon budget (Thiri et al., 2022) and the effects of the socioecological transformations such as climate change on local populations (Kassam, 2009). Kumar et al., 2021 focused on gender diversity in enrolment in Geo-Spatial Technology and Applications programs (Kumar et al., 2021). In the energy context, the authors investigated various factors, including the gender-shaped perception that affected energy security (Knox-Hayes et al., 2013; Sovacool et al., 2012). Water-related studies discussed the impacts of climate change on water resources and their effects on women and girls (Sen Roy, 2018b), and adaptation to climate-change-exacerbated water scarcity, droughts, and flash floods (Stucker et al., 2014). Only four publications focused solely on one or more CA countries, while the rest investigated within a much broader geographical scope.

The results of the term co-occurrence analysis are presented in the network map consisting of 62 terms divided into four clusters (Figure 4.1). The clusters describe energy, sustainability, and adaptive capacity (red cluster); biodiversity, global climate change, and each CA country (green cluster); empowerment, gender equality, disaster risk reduction, science, and knowledge (blue cluster); and health, women, children, and conflict (yellow cluster). It is worth mentioning the relatively high co-occurrence rate of geographical locations other than CA, e.g., China, the Middle East, and Europe. The domains of climate change–gender interconnections (Kovaleva et al., 2022) are not clearly distinct on the map. Furthermore, the terms "vulnerability", "benefit", "leadership" are not included in any of the clusters. Overall, all terms could be explored under every domain with a reference to the aforementioned geographical locations.





4.5.2 On-Line Survey

4.5.2.1 Respondents' Characteristics

Experts from all five CA countries participated in the on-line survey and shared their comments and opinions on gender, women, and men considerations in the climate

change context. Forty respondents, more than 50% of which were women, were representatives of international, governmental, and civil organizations, and the private sector working in such fields as climate change, disaster risk reduction (DRR), gender, environment protection, water, agriculture, and energy. It should be also noted that some of the respondents have expertise in more than one field due to the nexus character of respective issues. Table 4.5 summarizes the respondents' characteristics.

Characteristic	Percentage of Respondents	Characteristic	Percentage of Respondents	
Country		Type of organization:		
Kazakhstan	39.13%	international	32.61%	
Kyrgyzstan	23.91%	governmental	8.70%	
Tajikistan	21.74%	private	4.35%	
Turkmenistan	4.35%	civil society	54.35%	
Uzbekistan	10.87%			
Gender:		Worki	ng Experience:	
female	58.70%	1–5 years	64%	
male	41.30%	6–11 years	20%	
		15–30 years	16%	
Area of your expertise:		Degree of engagement with local communities:		
water	38.33%	not at all	10.87%	
energy	10.00%	only a little	15.22%	
climate/climate change	20.00%	to some extent	30.43%	
food/agriculture	11.67%	rather much	15.22%	
gender	15.00%	very much	28.26%	
environment protection	3.33%			
DRR	1.67%			

Table 4.5 Socio-demographic characteristics of the on-line survey respondents. Source: author.

4.5.2.2 Gender/Women in the Climate Change Context

Climate change affects CA local populations to various extents. More than 55% of the survey participants assessed its impacts as high, 25% as very high, and 20% as moderate. The respondents also indicated that rural men and women are more vulnerable to climate change compared to those who are residing in urban areas. The rural population faces impacts of climate change such as, among others, insect outbreaks, low water levels, disruption in irrigation services, and higher health risks (Figure 4.2).





The majority of the respondents were inclined to agree that in the agriculture, water, and energy sectors, men and women have been affected by the same problems and to a similar extent, regardless of their socio-economic status (Figure 4.3).



Figure 4.3 How men and women are affected in the key sectors. Source: author.

Specifying the main causes of rural men's and women's vulnerability to climate change, the respondents were inclined to categorize limited access to loans and financial resources as men-related, whereas unemployment and less capacity with "water providers" and household activities were categorized as women-related. Water scarcity and its low quality, interruptions in electricity supply, low level of education and medical services, and economic and financial issues were named as of both men's and women's. The comprehensive list of the main causes of rural men's and women's vulnerability to climate change indicated by the respondents is presented in Appendix II Table 12.2. The majority of the respondents agreed that women's experiences differ from those of men's and, therefore, should be also considered in discussions on climate change and its related issues. Furthermore, they acknowledged the need to address women's and men's differences in governments'

initiatives and to formulate and develop gender-sensitive climate change adaptation and mitigation measures (Figure 4.4).



Figure 4.4 Inclusion of women's and men's differences. Source: author.

Additionally, the survey participants outlined the importance of increasing rural women's participation in decision-making processes, particularly at the town/village level (Figure 4.5), despite them having a still small input into decisions on water distribution, use of renewables, and water efficient and adaptation practices (Figure 4.6).



Figure 4.5 Importance of including more rural women in decision-making processes on climate-change-related issues across all levels. Source: author.





The respondents also agreed that the role of women in climate change mitigation and adaptation over the last five years has stayed the same or improved insignificantly (Figure 4.7).



Figure 4.7 Change in women's role in climate change adaptation and mitigation (over the last five years). Source: author.

4.5.2 Development Projects Analysis

The study analyzed 17 climate-change-related development projects implemented in the region between the years 2003 and 2023. The list and their short description are presented in Appendix II Table 12.3. The projects, which vary in their scope, scale, geographical coverage, and duration, include gender/women considerations as one of the (sub-) components, objectives, or outcomes. Nine can be classified as standalone gender/women projects, i.e., gender or women considerations are the core focus. In the context of climate change–gender interconnections, most of the projects refer to more than one domain. It is worth mentioning that leadership/empowerment is addressed more often than other domains. For instance, the projects "Land Rights and Economic Security of Rural Women and Improved Food Security" and "Enhanced Livelihoods through Institutional and Gender Sensitive Land Reform" in Tajikistan were implemented as part of the national government's efforts to mainstream gender into climate change policy between the years 2003 and 2008. Led by the United Nations Development Fund for Women (UNIFEM), the projects contributed to the development of more gender-sensitive land and policy legislation, the introduction of gender statistics, and the improvement in women's land rights and sustainable livelihoods (Mirzoeva, 2009). The project "Women and Water in South and Central Asia" by the Central Asia Program (CAP) (George Washington University) and Women4Climate Mentorship Program (Nur-Sultan) by the C40 Cities focused solely on leadership/empowerment supporting young women leaders to exchange the knowledge and experience of innovative conflict resolution, water management, and climate change issues, and to increase their networking opportunities (Central Asia Program, 2022; Women4Climate, 2021). In 2019, the International Organization for Migration (IOM) launched the "Tajikistan: Understanding the Nexus of Migration, Gender, Climate Change and Agriculture" initiative, which can be assigned to the vulnerability domain. It focuses on the migration, gender, climate change, and agriculture nexus to address women's needs in climate change adaptation in one of the regions of Tajikistan. It also aims to mainstream migration in climate change policy and programming (IOM, 2019). The benefit domain is reflected by the CLIMADAPT project through, for instance, better access to climate technologies and practices that support efficient use of energy, and more effective collaboration of financial institutions with men and women separately (Climate Investment Funds, 2018).

The large-scale projects, where gender/women considerations are one of the (sub-) components, objectives, or outcomes, can also be classified under more than one domain. The Water Resource Management Project in Uzbekistan, which aimed at rehabilitation and upgrading of irrigation systems and inter-farm canals, created favorable conditions for women to become farmers and provided additional opportunities for seasonal jobs. Furthermore, the project activities contributed towards vulnerability reduction by improving women's economic situation, increasing their participation in water consumer associations, enhancing the environment, and reducing allergies (Asian Development Bank, 2017). There are projects that included a gender action plan (GAP) with specific targets and reporting requirements, instead of indicating gender/women considerations as an activity, objective, or pillar. For instance, the GAP of the Water Resource Management Project in Uzbekistan addressed equal participation of men and women. It also defined a set of indicators and goals, including identification of women's roles through information campaigns, gender sensitization of key stakeholders, and collection of gender-disaggregated information (Asian Development Bank, 2017).

4.6 Discussion

4.6.1 Bibliometric Analysis

Output in the form of scientific publications reflects research interest in an investigated field among the academic community, experts, and practitioners. The prevalence of grey literature in the retrieved and analyzed dataset is consistent with the findings of (Vakulchuk et al., 2022), and showed an overall low number of peer-reviewed publications on climate change or related topics in the region, particularly compared with the scope of the problem. It also implies significantly greater attention to gender/women considerations in the climate change context by international donor institutions. Taking into account a very small number of the connected peer-reviewed publications, their main benefit can be seen as an overall contribution to the

knowledge of climate change–gender interconnections in Central Asia. Each of these publications is novel, at least with reference to the geographical coverage, namely, to one or more CA countries. The current very small set of publications makes the comparison with other studies in this thematic field inconsequent.

The result of the co-occurrence analysis, that is, the collection of the terms, reflects areas that have been earlier acknowledged as having high relevance and importance in CA due to the region's vulnerability to climate change impacts and natural disasters that result in the growing risks of food, water, and energy insecurity (IPCC, 2022, 2014b). Furthermore, the transboundary character and unequal distribution of natural resources, and the reduction in their availability, create tension and conflicts within and between local communities (Pradhan, 2022; Peña-Ramos et al., 2021). One such example is the allocation of water resources for agricultural purposes, which is often made on a temporary basis due to the specificity of local irrigation systems. The respective decision-making processes are men-dominated, which in turn places women in more disadvantageous positions. Overall, all subject areas presented in Figure 28 have a direct or indirect linkage to the livelihood and wellbeing of the CA population, and hence require consideration of gender/women issues and therefore. They can therefore be investigated under vulnerability, benefits, and leadership/empowerment domains.

4.6.2 On-Line Survey

The susceptibility of agriculture, water, and energy sectors to climate change shocks compromises the livelihood and wellbeing of the CA local population (The World Bank Group, 2021). Rural men and women are more vulnerable and experience the impacts to a larger extent compared to those who are residing in urban areas. This expert opinion is consistent with the studies investigating the effects of climate change on rural populations in other world regions (Dumenu and Obeng, 2016; Mekonen and Berlie, 2021; Reyer et al., 2017a). Comparing the problems faced by men and women in the agriculture, energy, and water sectors, the respondents followed a more gender-neutral tone in their replies. One of the reasons might be a low number of studies on climate change adaptation and mitigation in the region, particularly of those with reference to gender and women considerations (Vakulchuk et al., 2022). Furthermore, there is an insufficient quantity of gender-/sexdisaggregated data due to gender-blind or gender-neutral institutional reporting requirements, and exclusion of gender analysis in respective activities. Additionally, gender/women considerations might not yet be perceived as an integral issue to be tackled in the climate change context. Nevertheless, the experts acknowledged the importance of women representation in managerial and leadership positions in mendominated water, energy, and agriculture sectors, despite their little contribution to decision-making processes. For instance, in Tajikistan, in 2015, women formally managed about 13% of the registered dekhkan (farms based on private, individual, and inheritable land shares) (Mukhamedova and Wegerich, 2018). In the same year, in Uzbekistan, women comprised only 7.6% of members of water consumer associations, and 18.6% occupied managerial positions in water supply and sanitation bodies in 2016 (Asian Development Bank, 2018). In Kazakhstan, threefourths of the farm households were headed by men, while, in the energy sector, only 12% of senior management positions were held by women (Mynbayeva, 2020; Agency for Strategic Planning and Reforms of the Republic of Kazakhstan Bureau of National Statistics, 2022b). These representation patterns are similar to those in other world regions (World Bank, 2019l; Khandker et al., 2020; Pilgrim et al., 2021).

The exclusion of women from decision-making processes is often attributed to societal stereotypes about the lower significance of their roles compared to men, patriarchal traditions, institutional barriers, and disproportional access to quality education, training, and resources (Ilesanmi, 2018; Mosso et al., 2022). On the other hand, numerous studies demonstrate the effectiveness of more balanced gender representation in leadership and management. Women in these types of positions are considered to be key agents of change, contributing to the adaptation of stringent climate change policies, and renewable energy transition and consumption (Allen et al., 2019; Atif et al., 2021), as well as improving companies' environmental performance (Birindelli et al., 2019). However, the processes of transforming women's role are characterized by their complexity, high costs, limited availability of resources, and prevalence of gender-neutral legislation and policies. This was also confirmed by the survey responses, which showed an insignificant improvement in the role of women in climate change mitigation and adaptation over the years, despite the national governments' efforts to reduce gender imbalances and better include women in socio-economic, environmental, and political areas. Therefore, the engagement of international agencies (donors) in supporting further developments of gender considerations in the climate change context can be considered as an additional stimulus in these processes.

4.6.3 Development Projects Analysis

Over the years, donors have assisted the CA countries in developing and introducing climate change adaptation and mitigation measures and building resilience. A large share of these initiatives has supported water resources, agriculture, and energy, sectors that are highly sensitive to climate change impacts. Types of gender/women considerations addressed in the projects reflect donors' interest in a specific domain of climate change-gender interconnection. The majority of the actions have been taken to improve women's leadership status and empowerment. This has been implemented through the improvement in the access to information and financial resources, provision of opportunities for climate change education, and training and workshops, including those on water resources, agricultural practices, disaster risk, and renewable energy resources. Projects outcomes, such as increased participants' income, better livelihoods, and environmental and health conditions, could be assigned to the benefits domain. At the same time, vulnerability has been rarely specified among objectives, pillars, or outcomes, but addressed indirectly. Its reduction has been often achieved as the result of broader objectives of the largescale projects, such as rehabilitation of irrigation infrastructure or reduction in (agricultural) land degradation. Donors' support can be perceived as an additional driver to encourage nation- and region-wide inclusion of gender/women consideration in the climate change context in CA. In addition to the financial aid, the countries receive experts' assistance, as well as non-financial resources, to advance the transition to gender-sensitive and -responsive approaches. Furthermore, strengthening gender-related requirements for project implementations will foster a more active collaboration of all stakeholders involved.

4.6.4 Implications and Recommendations

The implications of this study to the overall knowledge of gender considerations in climate change in CA are threefold. Firstly, the work demonstrates a significant prevalence of grey literature compared to peer-reviewed publications, highlighting a

need to stimulate scientific research in the field. Taking into account socio-economic and political settings in the region, this could be implemented through the introduction of specific international research initiatives and national institutional programs.

Secondly, the study emphasizes the importance of considering women's and men's experiences in discussions on climate change, addressing their differences in governments' initiatives, and developing gender-sensitive and -responsive adaptation and mitigation measures. This can be fostered by introducing measures to increase input from women representation, particularly at town and oblast levels. Additionally, government agencies need more evidence-based materials and information demonstrating these differences and the potential benefits of their consideration. On the other hand, the work draws attention to a largely gender-neutral perception of rural men's and women's vulnerability to climate change. Therefore, identifying recommendation areas for action requires various sets of studies focusing on the estimation and evaluation of the impacts of climate change on men and women across key sectors in each CA country. Furthermore, there is a need for the broader integration of gender analysis and collection of gender-disaggregated data to support the effective development and introduction of climate change adaptation and mitigation measures.

Thirdly, the study outlines donors' role in supporting and implementing activities to reduce gender imbalances in climate change in CA. It also shows that donors' standalone gender/women initiatives focus mainly on the leadership/empowerment domain. Consequently, the study highlights areas that require stronger cooperation to mainstream the importance of gender/women considerations among a wide range of stakeholders engaged in climate change and its related fields. Additionally, there is a need to explore the effectiveness of addressing gender inequalities as a (sub-) component in large-scale projects compared to standalone gender/women initiatives. The findings will provide a better overview of the intervention areas.

The study contributes to daily life by providing an overview of climate change impacts on local men and women, and helps in the making of informed and more targeted decisions and interventions in specific areas. Furthermore, the findings could be used to identify challenges and barriers that impede the achievement of national goals towards the reduction of climate change vulnerabilities of the local population.

4.7 Conclusions

More than three decades of independence of CA countries have been marked by continuous socio-economic, legal, and political reforms, along with the growing threats of a changing climate, particularly to agriculture, food, water, and energy security. To reinforce their response to these challenges, the CA national governments have developed and introduced adaptation strategies and instruments, and actively participated in international climate agreements. In this context, it is essential that gender/women considerations are not excluded from adaptation and mitigation measures, to reduce vulnerabilities and enhance the resilience of the local population, where women represent more than 50%.

The current paper examines how gender/women considerations in the climate change context in Central Asia are reflected in the published literature, viewed by experts engaged in climate action, and supported by international donors through the implementation of development projects. The subject is of particular importance in the light of increasing threats of a changing climate to the local population's livelihood and wellbeing, and efforts to broaden the inclusion of gender/women in adaptation and mitigation and to strengthen the role of women as agents of change.

The results of the bibliometric analysis demonstrated a significant prevalence of grey literature compared to peer-reviewed publications, implying smaller attention has been paid to the topic by the CA academic community. The analyzed publications mainly discussed issues concerning the region's vulnerability to climate change impacts and natural disasters, adaptation and mitigation, sustainable development, empowerment, and biodiversity. The experts, who were participants in the on-line survey, indicated a higher vulnerability to climate change of rural men and women compared to those who reside in urban areas. They also acknowledged the importance of considering women's and men's experiences in discussions on climate change and addressing their differences in governments' initiatives. Furthermore, they emphasized the need to develop gender-sensitive and -responsive climate change adaptation and mitigation measures. The analysis of the development projects showed a still low number of standalone climate-change-related projects focused on gender/women considerations. In the large-scale projects, the issues are addressed as a (sub-) component, objective, or in a gender action plan. Most of the projects focused on more than one domain of climate change-gender interconnections. Leadership/empowerment issues were addressed most often.

The study has several limitations, including several methodological drawbacks. The bibliometric analysis was based on a dataset that only included literature published in English, which may have led to language bias. Additionally, the voluntary participation mode of the on-line survey implies the omission of opinions of those experts who are engaged in climate action but do not consider the women/gender topic to be relevant to their working area. Consequently, the obtained survey results may not fully reflect the current situation. Furthermore, not all information about development projects may be available in open access sources due to organizations'/donors' disclosure requirements.

Nevertheless, this paper represents one of the first studies exploring gender/women considerations in climate change in Central Asia in terms of three areas: published literature, experts' opinions, and international development projects. The work outlines the issues and domains of climate change–gender interconnections that have gained more prominent attention from academia, government, civil society, and international institutions. Furthermore, the obtained findings provide valuable references for developing further steps toward the broader inclusion of gendersensitive and -responsive approaches, particularly in climate-change-related areas that are often perceived as being gender-neutral. The study demonstrates the need to support research in the field, introduce measures to increase women's participation and input in decision-making processes, and mainstream the importance of gender/women considerations among a wide range of stakeholders engaged in climate change and its related fields.

5. Feminization of Agriculture and Its Impact on Sustainable Development

This chapter was published as Kovaleva, M. (2021) *Gender Equality. Encyclopedia of the UN Sustainable Development Goals*, DOI: 10.1007/978-3-319-95687-9_92

5.1 Abstract

Definition

Feminization of agriculture refers to a significant increase in the female share in the agricultural labour force (Katz, 2003; Deere, 2005), compared to past years, regardless of whether women form the majority of those employed in agriculture (Slavchevska, 2016).

5.2 Introduction

In the process of rural development and transformation, globally, the share of agricultural employment has declined from 44% in 1991 to 28% in 2018, but still accounting to 63% in low-income countries, in 2018, only 8% less since 1991 (World Bank, 2019e). Women comprise about 66% and 2% of the agricultural labour force in low-income and high-income countries, respectively (World Bank, 2019h). Among the factors significantly affecting quantity of women and men employed in agriculture, as well as types of their roles are a socioeconomic and political situation in a country, country's geographical location, and culture and traditions.

6. Promoting gender equality across the sustainable development goals

This chapter was published as Leal Filho, W., Kovaleva, M. et al. (2022) *Environment, Development and Sustainability*, DOI: 10.1007/s10668-022-02656-1

6.1 Abstract

Gender issues, and gender equality in particular, can be regarded as cross-cutting issues in the implementation of the Sustainable Development Goals (SDGs), even though it is unclear how they are taken into account. This study addresses this information gap by performing an assessment of the emphasis on gender issues across all the other 16 SDGs, in addition to SDG5, through a literature review and case study analysis, the basis for the newly developed framework, highlighting specific actions associated to each SDG. The 13 countries addressed in the 16 case studies include China, India, or Australia and illustrate the inclusion of SDG5 into the SDGs. Using an SDG matrix, the SDG targets are analysed. Those where an emphasis on gender equality is important in allowing them to be achieved are listed. The novelty of our approach resides in offering an in-depth analysis of how gender issues interact with the other SDGs, proposing a new analysis framework clearly identifying SDGs 1, 4, 11, 12, 14 and 16 demanding further attention for successful SD gender implementation and illustrating specific areas where further actions may be necessary, which may be used by policy-makers, raising further awareness on gender equality contribution to achieve the SDGs. A set of recommendations aimed at placing gender matters more centrally in the SDGs delivery are presented as a final contribution. These focus on the need for greater awareness and attention to good practices, to achieve successful implementation initiatives.

6.2 Introducing SDG5-gender equality

In an unprecedented global effort, the heads of state and government and high representatives in the United Nations (UN) meeting of September 2015 put forward the '2030 Agenda', a global plan for human and environmental prosperity, structured in 17 Sustainable Development Goals (SDGs) and 169 targets, indicative of the scale and of the ambition of the global action to be pursued. The 2030 Agenda recognises that the achievements of the 17 SDGs are linked to human and planetary prosperity, strengthening universal peace, greater freedom and promoting the eradication of poverty, discrimination and inequalities in all forms (United Nations, 2015c). In the collective journey of meeting the SDGs and the UN 2030 Agenda targets, countries and stakeholders will act in partnership (Leal Filho et al., 2022e) to take a transformative and inclusive path towards a resilient and sustainable future in economic, social and environmental terms. The 2030 Agenda plans for the SDGs and the related targets trigger action in critical areas for human and planetary welfare. These include (United Nations, 2015c): (i) human existence in prosperity, equality and a healthy environment, (ii) planet conservation through timely climate action, sustainable production, consumption and management of natural resources, (iii) economic, social and technological prosperity in a harmonious symbiosis with nature, (iv) peaceful, just and inclusive societies and (v) revived global partnership of countries, stakeholders and people.

SDG5, 'Achieve gender equality and empower all women and girls', reflects the ever-increasing efforts of the UN towards gender equality, earmarked with the
establishment of the Commission on the Status of Women in 1946 (UN Women, 2020a) and the adoption of landmark agreements such as the Convention on the Elimination of All Forms of Discrimination against Women in 1979 (OHCHR, 2020), the Beijing Declaration and Platform for Action in 1995 (United Nations, 1995), and the establishment of UN Women in 2010 (United Nations, 2012). The important role of gender equality for socio-economic development is well highlighted in the UN publication "We the Peoples" (Annan, 2000), emphasising the untapped development potential due to social, economic and political inequalities arising from gender discrimination, deeply rooted and persistent in many developing and developed economies, related to access to decent work and equal pay, education, healthcare, resources, decision-making, among others (Brixiová et al., 2020; Tsige et al., 2020; Connor et al., 2020; Maheshwari and Nayak, 2020). Women are still more vulnerable to violence, discrimination, and underrepresentation in the political, economic, and business spheres (Milazzo and Goldstein, 2019; European Commission, 2019a). The recognition of the important role of women in global, social, economic and environmental prosperity is clearly stated in paragraphs 236-243 of the 'Future We Want' (United Nations, 2012) and in the Open Working Group Proposal for Sustainable Development Goals (2014).

SDG5 brings forward issues of gender-based discrimination such as unpaid work, sexual and reproductive rights, and gender-based violence (Hirsu et al., 2019). Achieving SDG5 is a priority that contributes to the increase in global well-being. SDG5 includes nine targets that aim at ending all forms of discrimination, as described in Table 6.1. These targets set the sustainable development (SD) goals to be achieved. The indicators provide the monitoring approaches for status, progress, and assessment, chosen according to the respective objectives and measured globally, or at regional and country levels.

Target	Indicators	
5.1 End discrimination against women and girls	5.1.1 Legal frameworks for gender equality and non-discrimination	
5.2 End all violence against and exploitation of women and girls	5.2.1 Violence against women from an intimate partner	
5.3 Eliminate forced marriages and genital mutilation	5.3.1 Women married before age 15 or 185.3.2 Female genital mutilation/cutting	
5.4 Value unpaid care and promote shared domestic responsibilities	5.4.1 Time spent on unpaid domestic and care work	
5.5 Ensure full participation in leadership and decision-making	5.5.1 Women in political positions5.5.2 Women in managerial positions	
5.6 Universal access to reproductive rights and health	5.6.1 Women's decision-making on contraceptive use and healthcare5.6.2 Guarantee of equal access to sexual and reproductive health care	
5.A Equal rights to economic resources, property ownership, and financial services	5.A.1 Female land rights or ownership 5.A.2 Equal rights to land ownership	

Table 6.1 SDG5 Targets and Indicators from United Nations (2021). Source: team of authors.

5.B Promote empowerment of women through technology	5.B.1 Mobile telephone ownership
5.C Adopt and strengthen policies and enforceable legislation for gender equality	5.C.1 Systems to track gender equality

But despite the relevance of the SDGs as a whole and the importance of handling gender issues, in particular, there is a research gap when it comes to looking at both topics in a combined way. In order to address this research need, this paper reports on a study aimed at fostering a thorough assessment of the emphasis that gender issues should be given in order to achieve all the SDGs. The research question pursued by the paper is the following: *to what extent are gender issues being considered in the overall implementation of the SDGs*?

Through a literature analysis and 16 case studies discussion in a sample of 13 developed and developing countries, e.g., China, India, Spain, and Morocco, this study sheds some light on the topic. The novelty behind this study consists in not only offering a sound analysis of how gender is considered across all other SDGs, but also indicating areas where further actions may be required. The innovation of this work is also based on the fact that it offers specific insights into gender equality and the SDGs. Also, this study may offer further guidance to policy-makers, thus prioritising women's empowerment in developing collaborative initiatives in the area of gender equality. Finally, this paper also serves the purpose of raising awareness about the need for capacity building and sensitisation around gender-related issues and their crucial contribution to the SDGs.

6.3 Research on gender equality and the SDGs: assessing the relations

SDGs have clear, often measurable and very straightforward targets aiming to improve the quality of life and living conditions for all. The interactions between these goals and the larger policy frameworks aiming to ensure economic growth from the country level to the regional level turns out to be more complex and challenging due to numerous types of constraints, from financial to cultural, when considering gender equality and ways of promoting it.

The global agenda for change, intent, purpose and overall goals were generally defined with the publishing of the Brundtland Report (1987), and the progress since then entered a new phase when the SDGs were adopted by the UN as the 2030 Agenda, while SD has been adopted across several economic policy fields in order to define specific objectives and goals. While highlighting SD challenges and opportunities, studies have included the gender dimension to a lesser extent, as illustrated in the existing literature that concerns the SDGs (Magendane and Kapazoglou, 2021; Scharlemann et al., 2020).

Gender inequality is pervasive across the world and women experience a series of disadvantages, in comparison to men. Yet, SD requires that we should all enjoy equal rights and be able to appreciate lives, free from violence and discrimination (UN Women, 2020a). There has been progress in some areas of discrimination, e.g., more girls in education, fewer girls forced into marriage, and more women in leadership roles, but policy decisions related to education, health and other sectors continue to take place in gendered contexts (Morgan et al., 2020). A situation where approximately half of the population is denied equal opportunities, equal participation in decision-making, and equal access to resources, education and employment will contribute to severely inhibiting SD and global prosperity (Dugarova, 2018).

Thus, and through SDG5, gender equality is rightfully at the heart of the 2030 Agenda for SD (United Nations, 2015c), recognised as an essential human right and important enough to be a goal in its own right, among other 16 SDGs. Its significance is such that it constitutes a cross-cutting theme spanning all the other 16 SDGs, with a total of 45 targets and 54 indicators gender-related. It is suggested that not only is SDG5 critical to all the other SDGs, with gender inequality being an obstacle to progress, but that it has the potential to serve as an SD accelerator, with a positive multiplier effect, to speed up the progress of the 2030 Agenda (UNSDG, 2018). Gender equality and women's empowerment should have a catalytic effect on human development (Odera and Mulusa, 2020) if gender is in fact actively addressed across all SDGs.

There are a number of reasons why gender equality has to be considered in relation to all of the SDGs. If under-utilising part of the world's talent, we fall short of reducing poverty (SDG1) and encouraging economic growth (SDG8). Gender equality in education and the labour market contributes to enhancing the gross domestic product and should help to reduce extreme poverty by 2030 (Dugarova, 2018). Compared to 1998, the gender gaps in the labour force, measured as the difference between the labour force participation rates of women and men, have decreased in most regions of the world in 2018, particularly in Latin America and the Caribbean, and Northern, Southern and Western Europe, but the gap has widened in Eastern Asia and Eastern Europe (Klasen, 2018). According to world regions, the Gender Inequality Index of 2020, can be seen in Figure 6.1. Also, the Life-course Gender Gap in 2019, translating into a deviation from gender parity, reveals the gaps in the adult population (Figure 6.2). The studies on gender equality reveal that women worldwide are more fragile in aspects such as poverty, representativeness in public employment positions, insecurity, or physical and sexual violence, thus emphasising the need to ensure a redesigned gender-responsive approach towards implementing the 2030 Agenda (Hirsu et al., 2019; Liu, 2019; Bourgault et al., 2021). Dugarova (2018) also demonstrates the multiple benefits of gender equality in relation to SD beyond SDG5, including food security, agricultural production, climate change (Caridade et al., 2022) and natural resource management. Similarly, Morgan et al. (2020) raise similar points but focusing on showing the importance of gender in relation to health and well-being (SDG3) and the less obvious connection between water and sanitation (SDG6) and energy (SDG7), illustrating the interconnected nature of SD and meaning that gender equality plays, in fact, an integral role to achieving all of the SDGs. Women are more likely to be impacted by unsafe water and poor sanitation (SDG6) and to die from unclean fuel (SDG7) (James et al., 2020), than men.



Figure 6.1 Gender Inequality Index, by developing region, 2020, modified from UNDP (2020) Source: team of authors.





The World Employment and Social Outlook suggests that women are underpaid and under-employed (ILO, 2018), although playing a central role in the household economy and being important influencers in peaceful societies. The study from Manandhar et al. (2018) suggests that the concept of gender in SDG5, seeking the achievement of gender equality and empowerment of all women and girls, is narrow, focusing on women-specific limited roles. When considered in terms of social context impact, gender inequality affects justice in opportunities, leading to economic inefficiency and thus inhibiting growth and global SD (de Jong and Vijge, 2021).

According to Agarwal (2018), a bold interpretation of SDG5 and the establishment of synergies with the other SDGs could allow ways for women to contribute to progress in different aspects concerning SD. Asadikia et al. (2020) show the lack of influence that SDG5 alone has on an SDG index based on all observations, clearly highlighting the need to interact with other SDGs to increase SDG5 influence. Accordingly, it is important that other SDGs should refocus on the interactions of gender equality to achieve specific global sustainability objectives by 2030. Fariña García et al. (2020) used a semantic network analysis, including computational linguistics and text processing of SDGs in official documents, to measure interactions in specific countries (Nigeria and Spain), to be used to planners in every country. The results revealed that each SDG is connected with all the other remaining 16 SDGs, despite the language used to search for information. SDGs 2, 8, 11 and 12, known as the driving forces, were found to be always connected to all the others, and SDG5 was not among them, being translated into a difficulty in terms of transitioning from current to sustainable systems of governance and management, and failing to address the gender agenda (Rai et al., 2019).

SDG5 is clearly dependent on how governments interpret targets in order to allow women to access resources and have effective participation in all levels of societal decision-making, by involving various stakeholders in order to implement and reinforce legal and institutional arrangements on gender equality (Obura, 2020). The identification of interlinkage between the SDGs (Bali Swain and Ranganathan, 2021; Del Río Castro et al., 2021) is critical in allowing policy-makers to prioritise SDG5 targets and strategies for SD and achieving the 2030 Agenda indivisibility (Bennich et al., 2020). Biggeri et al. (2019) highlight the importance of adjusting the targets and indicators with specific goals, aiming to increase gender awareness and consciousness in the selection of parameters and to allow different strategic options to be involved in the implementation of the 2030 Agenda (Nilsson et al., 2018; Obura, 2020; Parkes et al., 2020). When assessing the sustainability performance of the Organisation for Economic Co-operation and Development (OECD) countries, Lamichhane et al. (2020) found that only 35% of OECD countries had identified a key national system to monitor all SDGs, a significant gap.

Most studies suggest that progress in achieving gender equality continues to be slow. The Global Gender Gap 2020 (World Economic Forum, 2020) report highlights the urgency of achieving gender equality, while reporting gaps between men and women in health, education and policy areas, and across all forms of economic participation, reinforcing that there is a long way to go with a 31.4% distance to parity. Women are closer than men in indicators related to health (SDG3), but further away from them in terms of employment targets. There are undoubtedly a number of local projects addressing gender equality, but it is predicted that it will take almost 100 years to close the gap in relation to political empowerment. Even in Western Europe, the same report suggests that gender equality will not be achieved for another 54 years.

Many countries are not on track to achieving the SDGs, and the COVID-19 pandemic has and continues to exacerbating widespread gender inequity (Shulla et al., 2021). Lockdowns have further increased the burden placed on women in the

home and putting them at increased risk from domestic violence (Huiskes et al., 2022), with women also accounting for 70% of healthcare workers fighting the virus (UN Department of Economic and Social Affairs, 2020). In this context, and considering that the SDGs are not effectively considering gender in their implementation, the gender gap may widen, rather than narrow.

6.4 Methodology

The work performed in the scope of this study was undertaken in three different phases:

Phase 1:

Documenting the targets of all the 17 SDGs that would require gender issues to be accommodated before the respective SDGs can be implemented

For achieving phase 1, which also attempts to cover an information gap regarding the integration and interaction of the 17 SDGs, an effort was made to identify the main strands dominating the literature concerned with policies, aims, interactions and analytical approaches regarding SDG5 integration in the SDGs. The first step consisted in analysing how SDGs interact in the complex framework generated by the current world's economic and social context, and therefore the methodology was based on reviewing how literature integrates gender equality leading to the UN 2030 Agenda. This resulted in a set of questions for which answers still need to be provided by considering that all SDGs need to be and are in fact interacting, guided by indivisibility, thus requiring inclusiveness as the *sine qua non* condition. Literature review allows to obtain a road overview of the existing scientific research, as well providing the context for new research (Hempel, 2020), forming the basis of all scientific research (Block and Fisch, 2020), while allowing the researcher to establish the key constructs of a future research agenda based on the identified gaps (Paul and Criado, 2020).

Phase 2:

Presenting 16 international case studies in 13 countries that specifically reflect how gender issues are being considered when implementing the 17 SDGs

The case studies in phase 2 were selected using an open international call for collaboration, in the context of which different experts were invited to provide inputs. After a detailed and critical examination of the published research, this study allows to document the cross-cutting gender issues that should be included in the targets of each of the 17 SDGs to achieve SD, while considering SDG5. A case study was associated to each SDG, demonstrating how gender issues have been successfully infused into the actions driving the achievement of all the SDGs. Thus, by setting up the main interactions/relations and policies dominating the policy-making that addresses SDG5, and identifying current vulnerabilities, gaps and delays in this respect, the 16 international case studies reflect how gender issues are taken into account when implementing the SDGs, a necessary step in developing a judicious framework and recommendations for facilitating the achievement of SD across all SDGs, by integrating the SDG5 targets and indicators.

Phase 3:

Develop a framework that is able to consider how gender issues across all the

SDGs can be implemented to facilitate the achievement of SD at global level In phase 3, data was first collected by documenting targets related to gender for each SDG (data from phase 1). Then, a set of case studies reflecting how gender issues have been successfully infused into the achievement of each SDG was used (data from phase 2). The combined results of both phases 1 and 2 formed the basis for the framework developed in phase 3, analysing the impact of gender issues on all the SDGs. The impact indicator showed the percentage of particular goal targets impacted by gender inequality. It was calculated for each SDG by using the following equation:

$$\frac{IT}{TQ} * 100\% = PI \tag{1}$$

where IT represents the Impacted targets quantity, TQ, the total targets quantity of each goal and PI, the percentage impact.

The percentage values fall under one of the four categories:

- Low impact: 0%—39.9%
- Average impact: 40%—60%
- Highly impacted: 60.1%—99.9%
- Extremely impacted: 100%

The combined results from the three phases are presented and discussed in the next section.

6.5 Results and discussion

This section reports on the literature search information and data collected. The evidence collected using the case studies allowed the development of a proposed framework that can be helpful to practitioners in promoting a cross-cutting approach to gender issues in the context of all other SDGs.

6.5.1 Gender equality and the SDGs

In the attempt to identify the gender issues predominant trends, the findings based on reviewing specialised literature have shown that contributions to gender equality and SDGs are mostly theoretical, focusing on trade-offs and synergies, followed by studies concerned with policy implications, and possible methodological and empirical approaches about the interactions of all the SDGs, while suggesting a wide number of indicators that are currently used or that need further refinement for properly measuring progress in achieving the SDGs. These frameworks of analyses assume particular relevance in developing countries, but also developed ones alike, as inequalities are still deeply rooted, irrespective of the SD degree.

Studies have referred to interactions among the 17 goals, while neglecting the specifics of interactions with SDG5 on gender equity studies (Abualtaher et al., 2021; Miola et al., 2019), the focus of this study. Moreover, most studies propose models and approaches often contradictory, thus delivering inconsistent outcomes regarding costs and effectiveness of policies or measures and actions for achieving the SDGs. Most of the studies are in an increasing trend of building up on the findings of other studies, while failing the novelty dimension (Magendane and Kapazoglou, 2021).

Faced with the vast volume of recent research and studies in approaching the dimension of the interaction between the SDGs, and by assessing the outcomes of relevant studies at this regard, it may be stated that most studies seek to bring improvements for three main processes: policy development, impact assessment, and how synergies are achieved or not (Alcamo et al., 2020; Biggeri et al., 2019; Scharlemann et al., 2020), while this study aims to cover both the theoretical and practical issues related to gender equity, as included in the 17 SDGs. Based on the

analysed literature review, it is important to be careful about forming a generalised perspective by including general insights and gained knowledge about one SDG in relation to all other SDGs, because the context from the economic, social and environmental perspective is of paramount relevance (Nilsson et al., 2018). Integrated perspectives provide the best opportunities in assessing the relations and interactions with all other SDGs, while allowing for the identification of the main weaknesses, in particular regarding SDG5.

By affirming the overarching relevance of gender equality and its developments in the short time framework between 2015 and 2021 (Dugarova, 2018; Klasen, 2018; Odera and Mulusa, 2020), it was then possible to develop a general theoretic-empirical framework for underpinning the relevance of a gender-responsive approach to implementing the 2030 Agenda (Hirsu et al., 2019; Liu, 2019; Bourgault et al., 2021).

The above information does reveal the need to focus on specific practical implementation at local level, though benchmarking. The case studies presented below aim to illustrate successful implementations.

6.5.2 Case Studies

Gender issues extend beyond SDG5 and needs to be addressed within all the other SDGs. The international case studies included in this section have thus been chosen as illustrative examples of gender equality, considered in relation to each SDG, other than SDG5. Further detail on how a focus on gender has brought a positive benefit in relation to each SDG, as the full list of case studies, is given in Appendix III.

Non-governmental and governmental organisations are working together to help rural women improve the quality of their life by expanding access to sexual and reproductive health care in Tanzania (Engender Health, 2021). The Trans-Boundary Rivers of South Asia programme in Nepal promotes and supports women's leadership in water governance to increase their social accountability (Crawford, 2020). A case study from China demonstrates that the implementation of sustainable consumption and production (SCP) may significantly benefit from the integration of gender analysis into the design of SCP policies, strengthening women's participation in natural resource management and decision-making processes (Fan and Jaffre, 2020). In the frame of the educational programme Soochnapreneur (Information-Preneur) in India, rural women received necessary information and technology training to become change agents and assist in disseminating information regarding government schemes and benefits in communities. Participation in the programme not only develops their entrepreneurial abilities as Digital Information entrepreneurs but also allows them to charge a nominal amount for their services to sustain their livelihood (Soochnapreneur, 2021). In South Africa, the skills-driven project that supports the creation of rural, womenonly entrepreneur craft groups contributes towards improving quality of life and developing a more sustainable community (Pretorius and Nicolau, 2020). The Samoa's Ministry of Women, Community and Social Development and the Disaster Management Office are working towards increasing women's engagement and participation in climate change and Disaster risk reduction community discussions and development projects (Aipira et al., 2017). The 'Blue Economy Aquaculture Challenge' initiative supports projects for transforming sustainable aquaculture practises with solutions linked to gender equality, among others (Australian Government, 2018).

The addressed case studies illustrate useful approaches for tackling a variety of local problems in a cross-cutting way, as a support for governments as they focus on gender equality issues, showing that there is room for further similar initiatives in different geographical and socio-economic contexts. The case studies presented clearly indicate that various initiatives related to gender across the globe have been successfully addressed at local levels, and these initiatives have directly and indirectly affected the achievement of the particular SDG under analysis, thus affirming the need to infuse gender issues within all the targets of the 17 SDGs to ensure more productive outcomes and achievements in the drive to SD. It has been shown that governmental and non-governmental organisations cooperate in improving the overall quality of life for women, either in rural or urban areas and in regards to health, education and access to leadership/management positions. Still, it was found that much is still to be done, as shown found below, analysing the interaction with all of the SDGs.

6.5.3 The proposed framework for assessing gender equality impact across the SDGs

Achieving gender equality is a matter of human rights and is crucial to progress across all the goals and targets (Dhar, 2018), as highlighted before. Gender inequalities intersect other inequalities, power imbalance and discriminatory practices, and as such, they unequivocal serve as routes to addressing the causes preventing SD globally (Hepp et al., 2019). We have pointed out that while being a goal in its own right, gender equality cuts across all other SDGs and is reflected in 86 targets for the SDGs.

Through the use of the data collected by documenting targets related to gender for each SDG (see the SDG Matrix–Appendix IV) and the identification of fruitful case studies reflecting how gender issues have been successfully infused into the achievement of each SDG, both based on a detailed analysis and synthesis of the literature, the authors have used the lessons learnt to develop a framework aimed at analysing the impact of gender issues on all the SDGs, illustrated in Figure 6.3. This framework allows to establish which SDGs need the most attention for successful SD implementation and can serve as a guide for all practitioners in accommodating and promoting a cross-cutting approach of contemplating gender issues within the target of all the SDGs.

SDG	Impacted Targets	Percentage impact
1 ND 用: 		85.7% Highly impacted
2 ITANIE NAMBER		50% Average impacted
3 GOOD HEALTH AND WELL BEING 		30.7% Low impact
4 country Education		70% Highly impacted
6 CLAAN MATER AND SAMPTATION		37.5% Low impact
7 ATOBIAGIE AND CLAMINERITY		40% Average impact
8 BEEST WURK AND ECONOMIC GROWTH		25% Low impact
9 MOLTIP: INFORMER MEDIFICATION INFORMATION		12.5% Low impact
		40% Average impact
		100% Extremely impacted
12 RESPONDED CONCEMPTION AND PRODUCTION		63.6% Highly impacted
13 CLIMATE		60% Average impact
14 LEE BELOW WATER		80% Highly impacted
		33.3% Low impact
16 PEACE AUSTICE AND STRONG INSTITUTIONS		100% Extremely impacted
17 PARTNERSHIPS		43.8% Average impact

Figure 6.3 Proposed framework for considering gender impact across all the SDGs. Source: team of authors.

According to the results of calculations, the following SDGs are extremely or highly impacted by gender inequality and should be prioritised: SDG1 (No Poverty), SDG4 (Quality Education), SDG11 (Sustainable Cities and Communities), SDG12 (Responsible Consumption and Production), SDG14 (Life below Water) and SDG16 (Peace, Justice and Strong Institutions) (Figure 6.4). If government and nongovernmental organisations strive to achieve SD, as proposed by the 2030 Agenda, they would have to ensure that gender equality is prioritised in their endeavours, particularly in the context of the six aforementioned SDGs (1, 4, 11, 12, 14 and 16).





A fundamental part of achieving SD is the reduction of poverty, and this needs greater priority in policy decisions. The literature makes it clear that high poverty is interlinked with high gender disparities (Warchold et al., 2021), particularly in developing countries (Workneh, 2020). More women are affected by poverty due to their larger share of unpaid work, limited access to resources and social protection, and lack of control over spending decisions when compared to men (United Nations, 2015c). Countries that reflect statistics of more women in remunerated positions have lower poverty rates (Nieuwenhuis et al., 2018), though this might not be the case when the income size is below the poverty line (European Institute for Gender Equality, 2016). The COVID-19 pandemic is expected to have deepen gender poverty gaps, affecting women more strongly than men (Leal Filho et al., 2022a, 2022d). According to the report released by the United Nations Development Programme (UNDP) and UN Women, 232 million women will be living in extreme poverty in 2030, compared to 221 million men (Azcona et al., 2020).

Gender gaps in education negatively affect economic growth (Klasen and Lamanna, 2009). Globally, approximately 17% of women, compared to 10% of men, are illiterate. In developing countries, this gap is much larger. As example, only 26% of women are literate, compared to 46% of men in Mali, 27%, compared to 60% in 70%. compared to 45% in Afghanistan (World South Sudan, and Bank, 2020a, 2020b). Every additional year of primary school increases the future earnings of girls, decreasing their vulnerability to violence and motivating them to marry later (UN Women, 2012). Addressing gender imbalance in land ownership rights and access to natural, social and economic resources is essential for responsible consumption and production (Franco et al., 2018). Women demonstrate a higher tendency towards product reuse, waste reduction, and purchase of organic

and eco-labelled products (Bulut et al., 2017; OECD, 2018). The promotion of peaceful and inclusive societies for SD and access to justice for all are impossible without targeting gender inequalities. In 2020, the United Nations High Commissioner for Refugees (UNHCR) recorded more than 82 million people fleeing war, violation of human rights, persecution or conflict, of which 48% are women and girls (UNHCR, 2021). The COVID-19 pandemic and subsequent lockdowns have intensified domestic violence (Azcona et al., 2020; UN Women, 2020b; Akel et al., 2021; Bourgault et al., 2021). The preliminary data indicate a 25%-100% increase in reported cases globally (UN Women, 2020c), one of the consequences of the inability of institutions to provide equal gender access to justice and essential services, and of gender representation imbalance in global, regional or national governance (UN Women, 2018a). Particularly in developing countries, the achievement of the social inclusion of vulnerable groups such as women can be ensured by local government policies, especially related to well-being gender budgeting (Gunluk-Senesen, 2021). More equal gender participation is one of the key factors to sustainable peace.

6.6 Conclusions

A recent major challenge impairing the proper achievement of gender equality is the COVID-19 pandemic, which is causing an expansion of inequalities in topics related to education, employment and well-being, healthcare, consumption and production, or climate change, being imperative that all stakeholders involved in SD thus prioritise and infuse gender equality in all their endeavours, while policy-makers need to critically reflect on whether their strategies for particular individual goals would be enhanced by a broader consideration of gender equality issues. While most of the previous studies investigated the potential interactions of gender equality with other SDGs (Barbier and Burgess, 2019; Dawes, 2022; Pham-Truffert et al., 2020; Tremblay et al., 2020; van Zanten and van Tulder, 2021; Warchold et al., 2021), this study contributes to a better understanding of gender equality as a cross-cutting issue among all the SDGs, underscoring the need to prioritise gender issues at all scopes of SD.

This study aimed to assess and define the relations and interactions regarding gender inequality, based on specific literature related to main gender inequality concerns, access to education, employment and implicitly to equal pay, along with all other related issues, from legal aspects to metrics of violence. An extensive body of literature was explored in this study, also documenting 16 relevant international case studies in 13 countries to emphasise the significance of positive interventions in terms of gender equality, considered as a cross-cutting issue among all the other SDGs, as reflected in 86 targets. As a result, the study proposes an innovative qualitative assessment framework, according to which targets can be impacted negatively by gender inequality, an important factor that can impair the achievement of a particular SDG. Among the most-impacted SDGs that should more attentively consider the promotion of gender equality as an important condition for their achievement are SDGs 1, 4, 11, 12, 14 and 16, being possible to notice a strong diversity of approaches involved, covering issues of concern that are equally of future interest. Understanding the strong interconnectedness of the SDGs in terms of addressing the issues related to gender equality needs to become a trend. If widely spread, this trend may serve as an accelerator for the achievement of global SD, through the 17 SDGs, and can offer further guidance to policy-makers for prioritising the achievements of the targets, by empowering women worldwide. The literature review outlines that the progress in achieving gender equality continues to be slow, as many gaps still exist between men and women in health, education, politics, and across all forms of economic participation. However, as demonstrated by the successful case studies implemented worldwide, there is a growing interest among different stakeholders to develop collaborative initiatives that give particular attention to promoting gender equality, and the trend is likely to increase in the future. However, while the presented case studies illustrate positive interventions in terms of SDG5 contribution to SD, they are clearly still insufficient.

One all-encompassing finding is that in spite of a wide range of studies and academic papers related to SDGs and SD, there continues to be divisiveness in assessing the challenges and opportunities of the 2030 Agenda, associated with the need for developing sound frameworks for drafting and assessing ex-ante policies, measures and actions for ensuring the integrated interaction among the 17 SDGs, by considering necessary trade-offs and integrating other environmental, social and economic policy objectives. All these, while not explicitly mentioned in this study, have been implicitly considered, along with policy paradigms that consider the lifestyle, technological and even healthcare/educational changes. The 17 SDGs of the 2030 Agenda imply by their formulation a principle of indivisibility, as SDGs address the shared concerns of all humanity. In fact, it is precisely this governing principle which is the foundation for the approach used in this study, guided by the interest in analysing how SDG5 can be assessed and further implemented when associated to the other 16 SDGs, substantiated by the fact that the 2030 Agenda has an implied target-integrated approach regarding the SDGs. Investigating SDG5 relationship with the other 16 SDGs proved to be challenging and promising, as it provided for new insights about the relationships and interactions between all the SDGs. Thus, a key implication of this study is that it illustrates the fact that more attention should be given to mainstreaming the gender equality theme within all development initiatives of every country. Also, considerations to gender issues should be included in the design of targeted policies and programmes, data collection on indicators, and also in the defining of priorities in every region. The study has limitations. The first one is the fact that, being a qualitative study, it was not possible to cover all the works published in the field. Also, the selection of the case studies was not exhaustive or intended to cover all geographical regions, and it should be only regarded as an illustration of gender equality as a cross-cutting issue. Furthermore, the sample of 13 countries does not cater for a worldwide representation. However, despite these limitations, this study represents a significant knowledge addition to the existing literature on the connections between SDG5 and overall efforts to implement global SD and successfully advancing the SDGs.

Based on the evidence collected, the following **recommendations** may help in efforts aimed at placing matters related to gender more centrally in the delivery of the SDGs:

- Inclusion of gender issues as a cross-cutting topic in the implementation of the SDGs.
- A greater emphasis on gender equality in SDGs-related projects across all themes.
- An increased attention should be paid to the opinion, views and voices of women on SDGs-associated policies, a procedure often overlooked.
- More attention should be given to poverty alleviation, a trend often unnoticed in gender discussions.

• A more detailed and continued review of novel case studies across the globe should be undertaken to establish how existing good practices on mainstreaming gender are integrated into the targets of all the SDGs, and then to infuse these local initiatives into policy and development initiatives.

Finally, there is a perceived need to build more capacity among professionals involved in the implementation of the SDGs, so as to better sensitise them about the need to always consider gender issues, raising global awareness about gender-related matters.

7. Climate change, extreme events and mental health in the Pacific region

This chapter was published as Leal Filho, W., ... Kovaleva, M. (2022) *International Journal of Climate Change Strategies and Management*, DOI: 10.1108/IJCCSM-03-2022-0032

7.1 Abstract

Purpose

This paper aims to address a gap in investigating specific impacts of climate change on mental health in the Pacific region, a region prone to extreme events. This paper reports on a study on the connections between climate change, public health, extreme weather and climate events (EWEs), livelihoods and mental health, focusing on the Pacific region Islands countries.

Design/methodology/approach

This paper deploys two main methods. The first is a bibliometric analysis to understand the state of the literature. For example, the input data for term cooccurrence analysis using VOSviewer is bibliometric data of publications downloaded from Scopus. The second method describes case studies, which outline some of the EWEs the region has faced, which have also impacted mental health.

Findings

The results suggest that the increased frequency of EWEs in the region contributes to a greater incidence of mental health problems. These, in turn, are associated with a relatively low level of resilience and greater vulnerability. The findings illustrate the need for improvements in the public health systems of Pacific nations so that they are in a better position to cope with the pressures posed by a changing environment.

Originality/value

This paper contributes to the current literature by identifying the links between climate change, extreme events, environmental health and mental health consequences in the Pacific Region. It calls for greater awareness of the subject matter of mental health among public health professionals so that they may be better able to recognise the symptoms and relate them to their climate-related causes and co-determinant factors.

7.2 Introduction

7.2.1 Introducing climate change, extreme weather events and health

The influence of humans on the climate system is evident. The recent global anthropogenic greenhouse gas emissions due to population growth and massive industrial processes are the highest in history; therefore, they are the leading cause of climate change and global warming (IPCC, 2014a). While the evidence explicitly links anthropogenic climate change with extreme weather and climate events (EWEs) frequency and intensity, indicators suggest human-mediated global warming has likely increased compound EWEs since the 1950s (IPCC, 2012, 2018, 2019, 2021; Mycoo et al., 2022).

Besides its substantial environmental and economic risks, climate change is one of the world's leading health risks (Berry et al., 2018). It is a primary culprit for the

increased rates of many communicable and non-communicable diseases, including zoonoses (Leal Filho et al., 2022c). Of particular concern are the impacts of climate change on mental health and the stress related to it (Searle and Gow, 2010). There is a strong association between EWEs and mental illness (Berry et al., 2018). These impacts may range from significant short-term to long-term impacts. The short-term impacts on mental health occur during or in the few days following a EWE and are likely to subside with time. Heatwaves, for instance, were found to negatively impact mental well-being equivalent to unemployment (Ding et al., 2015). Severe environmental events such as floods result in injuries, loss of properties and businesses and loss of loved ones, aggravating or leading to mental illnesses, such as anxiety, depression, distress and trauma (Berry et al., 2018). Low-income and developing countries are naturally more prone to such effects due to their lack of preparedness and limited adaptation capacity (Alderman et al., 2012). Overall, the most prevalent psychological illness in people affected by EWEs is post-traumatic distress syndrome (PTSD), followed by depression and anxiety (Liu et al., 2006). The magnitude of stress experienced by the victims depends on the extent of the damage, losses and inconvenience caused by the event (Tapsell et al., 2002).

Although there is no solid evidence about how long psychological impacts could last after a flood event, some studies suggest that psychological distress caused by floods affects the quality of life in the long term, too (Berry et al., 2018). For example, the levels of psychological distress in the community affected by the floods in 2000 in Lewes, Southern England, had doubled, and those psychological problems were still recognisable four years after the flood (Reacher et al., 2004). Figure 7.1 presents an overview of some of the impacts of climate change on mental health.

In addition, there is growing evidence linking climate change and extreme environmental events to human migration that together with its accompanying stressors become a significant cause of increased mental illness rates (Bhugra, 2004; Black et al., 2013).



Figure 7.1 Some of the impacts of climate change on mental health. Source: team of authors.

7.2.2 Impacts of climate change and extreme events in the Pacific Island countries

7.2.2.1 Health impacts of climate change and extreme events as a whole

Current projections indicate that climate change and environmental disasters will become among the leading causes of migration in the 21st century (IPCC, 2014a). One of the world's most affected regions by climate change and EWEs is the Pacific Islands Countries (PICs), which comprise hundreds of scattered islands that are small in size and poor in natural resources. PICs have weak economies, poor infrastructure and a combined population of about 6.6 million. Most of the population work in agriculture (WHO, 2013), and 50% live within 1.5 km of the ocean (Tiatia-Seath et al., 2018).

Five PICs are ranked among the top 20 countries in the World Risk Index for countries at-risk of extreme natural events in 2019, having Vanuatu and Tonga islands as first and third on the list, respectively (The World Bank, 2020). Intense cyclones, floods, sea-level rise (SLR), freshwater shortage and changes in seasonal weather are expected consequences of climate change in the Pacific region. Due to the concentration of most people in the coastal areas, only a few people could be safe from these climatic phenomena (Tiatia-Seath et al., 2018). These consequences may have disastrous economic implications such as land and coastal infrastructure losses and severe health impacts such as malaria and dengue fever. Nevertheless, the impacts of EWEs on mental well-being, in particular, have recently gained much recognition by public health experts in the Pacific region. The damaging climaterelated impacts on mental health may be direct or indirect. The direct impacts include, for instance, anxiety, depression, distress and trauma occurring following devastating natural events. On the other hand, the indirect impacts include the movement of people due to the destruction of the infrastructure and the subsequent economic and social collapse of the affected cities. Furthermore, climate change in the Asia Pacific region might force up to 75 million people to migrate to other destinations by 2050 (Tiatia-Seath et al., 2018); this kind of forced migration represents a real stressor for the affected people and negatively affects their mental well-being.

The specific impacts of climate change and EWEs on mental health in the Pacific region and the different PICs are not well studied, and only little is known about it up to date. For instance, Gibson et al. (2020) stated,

"Evidence is emerging of the mental health impacts of climate change. Tuvaluans are experiencing distress because of the local environmental impacts caused or exacerbated by climate change and hearing about the potential consequences of climate change."

This paper tries to fill this gap by studying the connections between climate change, extreme weather events and mental health, focusing on the PICs in general, particularly the Federated States of Micronesia (FSM), Fiji, Kiribati, Marshall Islands, Nauru, Pal au, Samoa, Solomon Islands, Tuvalu and Vanuatu. Besides reviewing case studies of EWEs, the paper describes how they disrupt life-support systems and population livelihoods and well-being by affecting determinant factors such as food security, malnutrition, water security, vector- and water-borne diseases and displacement affect the population's mental health/psychosocial condition.

Pacific island countries (PICs) are among those most vulnerable to the health impacts of climate change (Hanna and McIver, 2014) due to their exposure to changing weather patterns (McIver et al., 2016). Consequently, extreme events'

frequency, timing, intensity and duration vary (Karl et al., 1995). Increased precipitation, drought, cyclone, hurricane, windstorm and SLR result in the rise of climate-sensitive health risks and limits the capacity of PICs to manage and adapt in the face of such risks.

The extent to which the impacts of EWEs could affect future susceptibility is determined by whether affected communities can prepare for and cope with the exposure and if the health systems can fully recover from an event before the next occurs (Ebi and Bowen, 2016). Further, EWEs can have prolonged effects on communities and health-care services; their impacts increase the vulnerability to successive events. For example, a hurricane followed by a flooding event in low-lying coastal areas does not allow adequate time for recovery, which can take decades (UNES-CAP, 2015).

Climate change has substantial and diverse impacts on human health (Kotcher et al., 2021). Furthermore, the pathways by which climate change affects health differ according to their modes of action (McIver et al., 2016). The potential health impacts of EWEs include direct impacts, such as traumatic deaths and injuries; indirect impacts, such as illnesses associated with water-borne, food-borne, vector-borne diseases and zoonoses; diffuse or delayed impacts such as mental/psychosocial health disorders, non-communicable diseases (NCDs), health system deficiencies through ecologic or social disruption (McIver et al., 2016; Butler and Harley, 2010).

7.2.2.2 Delayed health impacts of climate change and extreme events on mental health

Several climate-related health risks are of concern in the Pacific not documented elsewhere globally, for example, NCDs, mental/psychosocial health disorders and ciguatera (McIver et al., 2016; Mannava et al., 2015). The countries prone to climate-related mental or psychosocial health disorders (e.g., anxiety, depression and post-traumatic disorder) include FSM, Fiji, Marshal Islands, Nauru, Palau, Solomon Islands, Tuvalu and Vanuatu (McIver et al., 2016). In addition, EWEs affect an individual's emotional or mental health (Ebi and Bowen, 2016; Lawrance et al., 2021) as the aftereffect or recovery phase from catastrophic damages is full of social and financial challenges.

Figure 7.2 schematises the climatic changes and EWEs' direct and indirect adverse impacts on health (e.g., infectious diseases and malnutrition) and mental relocation/resettlement), highlighting health (e.g., the importance of windstorms/hurricanes and SLR on mental health. The most prevalent psychological illness in people affected by EWEs is PTSD, followed by depression and anxiety (Liu et al., 2006). The prevalence of mental illnesses in the first two years following flood events ranges from 8.6% to 53% (Jackson and Devadason, 2019). There is, however, no solid evidence about how long psychological impacts could last after flood events, but some surveys suggested that they may last for as long as six years (Jackson and Devadason, 2019).



Figure 7.2 A conceptualisation of the pathways by which climate change will affect mental health in the Pacific Islands Countries (PICs) and the significant anticipated impacts throughout the region. Source: team of authors.

7.2.2.3 Zooming in the pacific region: the effects of extreme events

The EWEs cause many deaths and injuries annually in the Pacific region and disrupt the local environment and communities (Lei and Zhou, 2012; NDMO, 2014; OCHA, 2015; UFCOP, 2017; Terry and Lau, 2018). Over the years, the severity of EWEs has forced the people to relocate, for example, the climate-induced relocation of the Vunisavisavi community in Fiji (Singh et al., 2020). For Pacific Islanders, the land (e.g., Vanua in Fiji) provides a sense of place and identity. Therefore, the relocation or resettlement from the roots of their cultural, psychological and spiritual well-being has emotional and psychological implications (McAdam, 2014; McNamara et al., 2021).

In the recent past, PICs have been a hotspot for a series of EWEs such as flooding, tropical cyclones and hurricanes (Chand and Walsh, 2009; Magee et al., 2016), causing injuries, loss of lives, massive destruction to infrastructure (Aquino et al., 2019) and economic loss (Benson, 1997). Cyclones and floods are the most frequent climate-related disaster and the leading cause of death and injuries from EWEs in PICs (Lei and Zhou, 2012; NDMO, 2014). Table 7.1 presents some of these events.

Future climate scenarios project an increase in the frequency of EWEs, SLR and, consequently, an increased risk of floods (Meehl et al., 2000; Vitousek et al., 2017), especially in low-lying coastal areas. The severity of a flood is determined mainly by topography, the surrounding infrastructure of the flooded area, various human-generated factors and the potential of floodwater to spread over a wide area.

Cyclones can be massive, cutting broadband of destruction as they traverse the PICs. For instance, Vanuatu experienced drastic effects from the windstorm in 1999, causing 32 deaths (Lei and Zhou, 2012). Severe drought conditions have been associated with widespread crop failure and food shortages, resulting in malnutrition and starvation in PICs (Barr, 1999; Hoot et al., 2012). Fiji has experienced six prolonged periods of drought events since 1970, affecting over 900,000 Fijians (Government of Fiji, 2017). A historical extreme rainfall deficit and drought occurred in Fiji during the El Niño event of 1997/1998 (Lightfoot, 1999). Thus, the factors that could reduce longer-term resilience from EWEs in PICs would likely include poor food and water security, mental health issues and displacement.

Name of Event and Year	Country Affected	No of People Affected	References
Drought—2011	Tuvalu	>5200	IFRC (2011), Kuleshov et al. (2014)
Floods—2012	Fiji	±150,000	Kuleshov et al. (2014) ReliefWeb (2012a)
Tropical cyclone (Evan)—2012	Samoa Fiji	>10,000	Kuleshov et al. (2014), ReliefWeb, 2012b)
Floods—2013	Solomon Islands	>10,000	BBC (2014), Noy (2016)
Tropical cyclone (Ian)— 2014	Tonga	±5500	United Nations (2015b), World Bank (2014a)
Tropical cyclone (Pam)—2015	Fiji, Vanuatu, Tuvalu, Solomon Islands, Tonga	±166,600	Le Dé et al. (2018), United Nations (2016), ReliefWeb (2015a)
Tropical cyclone (Mitag)—2002	Federated States of Micronesia	>175	Guha-Sapir (2018)
Tropical cyclone (Chata'an)— 2002	Federated States of Micronesia	±1448	Guha-Sapir (2018)
Tropical cyclone (Lupit)—2003	Federated States of Micronesia	>1000	Guha-Sapir (2018)
Tropical cyclone (Sudal)—2004	Federated States of Micronesia	±6008	Guha-Sapir (2018)
Coastal flooding—2008	Federated States of Micronesia	>1200	Guha-Sapir (2018)

Table 7.1 Examples of extreme weather events in the Pacific islands (2003–2020) Source: team of authors supported by the literature

Tropical cyclone (Maysak)— 2015	Federated States of Micronesia	>35,000	United Nations (2016), Guha-Sapir (2018), ReliefWeb, (2015b), Tiwari et al. (2019)
Drought—2016	Federated States of Micronesia	>100,000	Guha-Sapir (2018)
Tropical cyclone (Winston)— 2016	Fiji, Vanuatu	±150,000	European Commission (2019b), ReliefWeb (2016), Thomas et al. (2019)
Earthquake— 2018	Papua New Guinea	>544,000	OCHA (2019)
Tropical cyclone (Gita)—2018	Tonga	$\pm 80,000$	OCHA (2019), Foley (2020)
Tropical cyclone (Yasa)—2020	Fiji, Vanuatu	±93,000	Fiji Meteorological Service (2021), ReliefWeb (2020)
Tropical cyclone (Harold)—2020	Vanuatu, Fiji, Solomon Islands	±170,000	Ahmed and McDonnell (2020), FAO (2020c)

7.3 Methods

This paper deployed two main methods and consisted of two steps.

In Step 1, a bibliometric analysis was performed based on the need to review the existing literature. Bibliometric analysis has become an increasingly popular data analysis and visualisation method due to its ability to provide an overview of the structure and trends in a field or sub-field. Specifically, the text mining ability offered by bibliometric analysis tools allows for analysing relationships between specific sets of terms. As such, the authors perform a bibliometric analysis to understand the state of the literature concerning psychiatric conditions and climate change events. The input data for analysis using VOSviewer is bibliographic data of academic publications downloaded from Scopus. Scopus was selected given its broad coverage of quality peer-reviewed academic journals and also the compatibility of its bibliographic outputs with VOSviewer. Web of Science is another frequently used database. However, it was not used in this analysis since its coverage of journals is limited compared to Scopus.

To find relevant documents, the authors designed an inclusive search string of terms associated with mental health and climate change that was not limited by date (Appendix V). The search string was developed iteratively to ensure its comprehensiveness. For example, the authors first searched for "climate change" and "mental health". The authors then checked the returned articles and added other relevant terms to the string. This process was repeated until adding new terms did not result in retrieving new articles. Although some relevant articles may have been overlooked, the authors believe this approach has helped us retrieve as many articles as possible.

Overall, our initial search on 23 October 2021 returned 501 articles written in English. Titles and abstracts of these articles were screened, and 355 articles, book chapters, reviews, conference papers and letters related to this study's scope were selected for bibliometric analysis using VOSviewer. The excluded articles did not include issues associated with mental health. For the selected articles, "full record and cited references" were downloaded from the Web of Science to be used as input data for bibliometric analysis. After inputting these data into VOSviewer, the authors used the term co-occurrence analysis to understand the knowledge structure of the field (Figure 7.3).

Figure 7.3 shows that the output is presented as a graph, where the node size is proportional to the terms' frequency, and the connecting lines' width indicates their strength. Words that frequently co-occur form clusters that show major thematic research areas in a field.



Figure 7.3 Co-occurrence map of mental health and climate change-related terms. Source: team of authors.

In a second step, the authors deployed a follow-up method, namely, a description of case studies. This outline some of the region's EWEs. The case studies, which were compiled based on available evidence and first-hand information gathered by some of the authors in their home countries, provide concrete examples of the impacts of extreme events on the life-support systems, population livelihood, wellbeing, health (especially malnutrition and infectious diseases), focusing on mental health.

The results of the data collection approaches are presented in the next section.

7.4 Results and discussion

7.4.1 Bibliometric Analysis

Results of the bibliometric analysis are evidenced in Figure 7.3. It can be seen that climate change impacts and extreme events have been linked to various types of mental health issues in the literature. Some key features are as follows:

- Four major thematic clusters link climate change, EWEs and mental health.
- The green cluster mainly focuses on general and mental health impacts.
- Mental health has a central position in the term map and frequently cooccurred with climate change.
- Various climate-related stressors and EWEs indicate that the potential adverse climate-related impacts on mental health and public health are well-recognised in the literature.
- The red cluster mainly focuses on the mental health impacts of slow-onset and long-term climatic impacts such as temperature changes or exposure to environmental pollution exacerbated by climate change. Based on the interlinkages between the terms, it can be seen that in addition to increasing mortality rate, temperature and environmental impacts can be significant risk factors contributing to mental health issues such as mental disorders and suicide.
- The yellow cluster focuses on acute and rapid-onset disasters such as floods and hurricanes, closely linked to mental health issues such as anxiety and post-traumatic stress disorder.
- The blue cluster includes various demographic groups such as males, females, children, adolescents, young adults, middle-aged and aged people. These demographic groups are linked to different mental health issues from other clusters, indicating that climate change will likely impact all demographic groups negatively.

Overall, this term co-occurrence analysis suggests significant documented effects of climate change on mental health, with analyses into the effects of heat and temperature and natural disasters predominating. However, secondary climate change impacts such as loss of culture or community due to forced migration have received relatively less attention in the literature regarding mental health impacts (Hayes et al., 2018). Additionally, many psychological disorders, such as schizophrenia, bipolar disorder or obsessive-compulsive disorder, are not highlighted in this term map but could be significantly impacted by individuals' experiences of their environments that could be affected by climatic changes (Hayes et al., 2018). Additionally, the links between mental and physical health in the face of climate change are unclear in this analysis but remain salient to the discourse. While further research on such issues is needed, it is clear from this analysis that mental health impacts of climate change are essential and need to be appropriately integrated into vulnerability assessment and climate adaptation efforts and processes (Hayes and Poland, 2018). More details about the four clusters highlighted by the term co-occurrence analysis can be found in Section 7.4.3.1.

7.4.2 Case studies: reviewing the impacts of climate change and extreme weather and climate events on the life-support systems, livelihoods, well-being and mental health in the Pacific Islands Countries

7.4.2.1 The Federated States of Micronesia (FSM)

EWEs that disrupt and impact the FSM's lives include tropical cyclones, increased floods and droughts, landslides and coastal flooding events related to SLR (Table 7.1). While direct health effects of EWEs are traumatic injuries and deaths, it also includes psychosocial impacts. Diffuse effects include unspecified detrimental effects of social disruption, e.g., loss of life, land or livelihoods due to climate-change-related phenomena, including anxiety, depression and post-traumatic stress disorder.

Following typhoon Sudal in Yap in 2004, depression, anxiety and substance abuse (especially among youth) as the aftermath of disasters, especially in women and children, were reported in addition to the traumatic injuries. For socially disadvantaged atoll communities, post-disaster permanent displacement is stressful, and it unsettles and harms mental health through the loss of life-supporting traditional food production systems. The stress of migration is further aggravated by a lack of social support, inadequate health systems, economic hardship and lack of access to housing (FSM, 2015).

7.4.2.2 Tropical cyclone Winston 2016-Fiji island

TC Winston affected approximately 40% of Fiji's population, with a death toll of 44 people. Following TC Winston in February 2016, a typhoid outbreak of 35 cases was observed on the Northeast coast of Viti Levu, Fiji's main island, but no fatalities were reported. Also, vector-borne diseases were reported, e.g., Dengue and Zika (Fiji Health and Nutrition Cluster, 2016a, 2016b).

After TC Winston, malnutrition was a severe problem in some areas, especially among children under five (Fiji Health and Nutrition Cluster, 2016c). The cyclone significantly affected local fisheries, with about 96% of fishing boats sunk, affecting household food supplies (92% of which depend on local fisheries for subsistence), reducing the amount of fish delivered to the local school to 0% (World Conservation Society, 2016). On Koro Island, there were some cases of malnutrition. The terrestrial food supplies were disrupted or lost by severe winds, rain and storm surge due to the cyclone's damage to this island, resulting in saltwater pollution of the soil (Government of Fiji, 2016).

The cyclone also prolonged its effect on Koro's livelihoods, as Kava farming provided half of their revenue, which was utterly destroyed after the cyclone (World Conservation Society, 2016). In addition, the food situation was precarious (Government of Fiji, 2016).

Following TC Winston, 8,466 people received psychological first aid. Therefore, the Ministry of Health and Medical Services coordinated training trainers workshops to facilitate the deployment of workers in all areas of Fiji (Fiji Health and Nutrition Cluster, 2016a, 2016b). The cyclone affected people with disabilities as well. UNICEF and the Pacific Disability Forum surveyed 963 people. According to the results, 13.6% of disabled children needed psychological assistance and basic first aid and integrity kits to maintain a standard of hygiene that prevents the spread of diseases (Pacific Disability Forum and UNICEF, 2016).

7.4.2.3 Tropical cyclone Evan 2012-Samoa and Fiji

In December 2012, both Samoa and Fiji were hit by tropical cyclone Evan with a death toll of 14, and 10 sailors went missing. In addition, damage to plants and manufactured structures was substantial due to the storm moving near the Fijian islands, where over 8,000 people had to seek temporary shelter. Cyclone-related economic losses also remained quite large, inflicting considerable negative consequences on the economies of both countries. Fiji's total losses from Evan were about 2.6% of the country's gross domestic product (GDP), plus another 1% due to short to medium-term losses. Economic losses in Samoa were even higher, in the order of US \$203.9m (Kuleshov et al., 2014), about 25% of the country's GDP.

7.4.2.4 Floods 2012-Fiji

The March 2012 flood was the worst flood to hit Fiji. The flooding was most severe in the western half of Viti Levu's main island. Thousands of people were displaced, about 3,500 people were placed in temporary shelters, and 8 likely perished in the floods; the majority were on the main island. Power and water supplies were disrupted in several regions, and roads were washed away. Crops and other infrastructure were damaged. The entire cost of the flood in March 2012 was over FJ \$70 million (Kuleshov et al., 2014), about 0.9% of the country's GDP.

7.4.2.5 Tc Harold 2020-Vanuatu, Fiji and the Solomon Islands

Tropical Cyclone Harold wreaked havoc on the Solomon Islands, Vanuatu, Fiji and Tonga in early April 2020. Heavy rain and strong winds wreaked havoc on homes, schools and gardens across four provinces in the Solomon Islands. Around 27 individuals were reported missing after being washed away on a ship from Honiara to Malaita. The cyclone caused the heaviest damage in Vanuatu when it landed on 6 April as a category five cyclone with sustained winds of more than 200 km/h. Roads, hospitals, schools, residences and churches were severely damaged on several northern and central islands. The worst-affected areas were home to 92,300 people, accounting for 30% of the country's population (UNICEF Pacific, 2020).

Many more families were displaced, without access to food crops or sanitation services. This circumstance was highly harmful to young children and severely threatened their survival and well-being. In addition, cases of dengue fever and malaria had been reported in Sanma Province. On 8 April, the storm impacted Fiji, causing significant flooding due to heavy rain and strong winds. In its wake, 2,494 homes were damaged. According to preliminary data, 116 schools were destroyed, affecting 11,524 children, with schools in the eastern and central divisions being the most brutal hit. The storm then impacted Tonga, causing damage or destruction to an estimated 428 homes (UNICEF Pacific, 2020).

7.4.2.6 Drought 2011-Tuvalu

Samoa, Tokelau, Tonga and Tuvalu were affected by the La Niña-induced rainfall deficit in 2011. Due to severely low water supplies, Tuvalu's Government announced a state of emergency on 28 September 2011. Households were rationed to roughly 40 litres of fresh water daily since some areas of Tuvalu had just a two-day water supply (Kuleshov et al., 2014).

7.4.2.7 Trends from Kiribati

Kiribati is an island situated in the central Pacific Ocean. The entire population, about 115,840, lives only one kilometre from the sea (WHO, 2018). Due to frequent EWEs, Kiribati might become the first country to lose its national identity (WHO, 2018). The risk of climate-related transmission of vector and water-borne diseases is also high. However, the effects on Mental Health among the Kiribati population are unclear. Although per the WHO mental health report, mental disorders in the Republic of Kiribati have become a significant concern. Limited mental disease prevalence data is available, relying on a small national survey. Based on the disability national survey, 17% or 653 people were found to have a mental illness, including intellectual disability, epilepsy, or psychiatric illness. Moreover, the World Mental Health Survey 2004 reported that approximately 13% of the Kiribati young population (over 15) would experience mental disorders.

7.4.3 Overall Discussion

7.4.3.1 Bibliometric analysis

The green cluster mainly focuses on climate change's general health and mental health impacts. The term "mental health" has a central position in the term map and has co-occurred frequently with the term "climate change" and various climaterelated stressors and adverse events such as drought, natural disaster and extreme weather. This indicates that the potential adverse impacts of climate change on mental health and public health are well-recognised in the literature. It is now wellrecognised that climate change can have both direct and indirect negative impacts on mental health. The direct pathway occurs when rapid onset climate-induced disasters such as hurricanes, floods and wildfires expose people to trauma. Long-term climatic changes such as extreme weather events, droughts and SLR can also indirectly affect mental health through eroding the physical health and livelihood options of humans and threatening social processes essential for community well-being (Berry et al., 2010; Palinkas and Wong, 2020). It is argued that marginalised, poor and vulnerable groups are disproportionately affected by climate change impacts, including impacts on mental health (Berry et al., 2010). Despite this, the term map shows that literature on the nexus of climate change and mental health mainly focuses on developed countries such as Australia, Canada and the USA. More research on the mental health impacts in other contexts is, therefore, needed.

The red cluster mainly focuses on the mental health impacts of slow-onset and long-term climatic impacts such as temperature changes or exposure to environmental pollution that can be exacerbated by climate change. Based on the interlinkages between the terms, it can be seen that in addition to increasing mortality rate, temperature and environmental impacts can be significant risk factors contributing to mental health issues such as mental disorders and suicide. For example, evidence shows that environmental factors such as extreme heat and humidity have increased hospital admissions of patients with mood and mental disorders such as mania and schizophrenia. Such environmental stressors have particularly affected people with pre-existing mental health diseases and those with a history of drug and alcohol abuse (Hayes et al., 2018). Several studies have also examined associations between climatic changes such as increased heat risk and suicide rates in countries such as Italy, New Zealand and the USA (Dumont et al., 2020; Preti et al., 2007; Williams et al., 2015). Existing research indicates that, while

other confounding factors are also essential and one should be cautious when associating suicide rates with climatic changes, the risk of suicide is higher during hotter and more polluted days (Dumont et al., 2020; Preti et al., 2007; Williams et al., 2015).

The yellow cluster mainly focuses on acute and rapid-onset disasters such as floods and hurricanes. These are closely linked to mental health issues such as anxiety and PTSD. The association between mental health issues and climateinduced disasters such as floods, hurricanes and wildfires has been extensively studied in the literature (Crabtree, 2012; Upward et al., 2021). It has been reported that climate-induced disasters can trigger mental issues such as "post-traumatic stress disorder (PTSD), major depressive disorder (MDD), anxiety, depression, complicated grief, and survivor guilt" (Hayes et al., 2018). Further, major disasters could also trigger substance abuse and suicide ideation (Hayes et al., 2018). Suicide and natural disasters are not closely linked in the term map, and it has relatively been more studied in the context of heat-related extreme events, as discussed earlier.

Finally, the blue cluster includes terms related to various demographic groups such as males, females, children, adolescents, young adults, middle-aged and aged people. These demographic groups are linked to different types of mental health issues from other clusters, indicating that climate change is likely to impact all demographic groups negatively. For example, different studies have discussed the mental health impacts of climate change on groups such as children and adolescents (Clemens et al., 2020), women and older adults (Gifford and Gifford, 2016; Padhy et al., 2015). It is worth mentioning that while all groups are exposed to mental health issues of climate change, evidence shows that some groups, such as women and older adults, experience more issues such as anxiety, distress and mental disorders (Gifford and Gifford, 2016). These groups should, therefore, receive special attention in the resilience-building efforts.

7.4.3.2 Case studies

The tiny atoll ecosystems in the FSM are becoming uninhabitable, forcing communities to abandon their life-supporting systems and become refugees on elevated grounds or urban centres. Climate models show that climate will impact all aspects of life in the FSM yet in varying intensity. As the consequences of climate change are not inevitable, the island communities handle all the other significant challenges and opportunities on their terms and with external support.

These conditions have far-reaching environmental, social and livelihood effects that ultimately affect island communities' health and well-being. However, psychological aspects resulting from the climate-related impacts are largely unexplored in the FSM. While the climate-related mental health implications can affect anyone, the impacts amplify in the marginalised atoll populations. In addition, interference with livelihood opportunities, property and land damage and post-disaster displacement affect atoll communities' mental health and well-being.

Island communities in the FSM have an inseparable connection to and derive their sense of identity from the lands and resources of their islands. Climate change threatens this familial relationship with ancestral resources. It disrupts the continuity required for the health and well-being of these communities (Keener et al., 2018; CDC, 2020). Women are more vulnerable to climate risks due to their economic activities, safety, health and livelihoods. Coastal flooding associated with SLR imperils atoll communities' livelihood opportunities. SLR in the western Pacific is three times higher than the global average, impacting traditional agriculture, coastal infrastructure, food security and livelihoods. Separation from traditional lands harms the island communities' spiritual and mental health in the FSM. Given the mounting evidence of the link between climate change and mental health and the particular vulnerability to many impacts of climate change, EWEs and climate disasters, island communities, especially the atoll communities, are likely to be at high risk for climate-related mental illnesses. Cultivating resilience and engaging the displaced atoll communities in site-specific adaptation strategies empowered them to face hardships and overcome distress situations followed by disasters (Krishnapillai, 2017, 2018).

EWEs cause disasters in Fiji due to climate instability and transition. These disasters hit the health sector with a rise in hospital admissions and treatments for accidents and infectious diseases, including diarrhoea, typhoid, dengue and leptospirosis. Malnutrition and stress-related illnesses are also on the rise. Dengue fever is one of Fiji's four major climate-sensitive diseases (Guillemot, 2011). Diarrheal, typhoid and leptospirosis are the other three major climate-sensitive diseases.

Mental health is vulnerable to climate-related catastrophes such as droughts and floods (Berry et al., 2010; Cunsolo et al., 2013; Reynolds et al., 2010; Sharpe and Davison, 2021). Anxiety, sadness, excessive concern, PTSD, survivor guilt and "solastalgia" (distress experienced by those affected by the environmental change) may have previously been documented in many places, and the burden of these mental illnesses is considerable (McNamara and Westoby, 2011; Willox et al., 2012; Owusu et al., 2022).

Changes in local settings in the Pacific, where community identity and culture are tied to local settings, contribute to changes in cultural practices and identity, with potentially severe implications for mental health. Also, people in PICs are at high risk for mental diseases due to the link between climate-related disasters and mental health.

As the conditions seem alarming in Kiribati, the Kiribati Government has looked to relocate its population. Meanwhile, the Kiribati Ministry of Health is preparing a National Adaptation Programs of Action to help curb the impacts of climate change. In addition, the Kiribati Government has also drafted the Strategic Health Plan (2008–2011) figured six primary objectives, one of which is to improve, revise the mental health policies, implement the mental health plan and enhance the mental health service delivery (Oten et al., 2013).

On the population level, the residents of Kiribati have begun building walls from coral rocks. They have also intensified planting their lands with mangroves to protect the soil from erosion. Nevertheless, several communities have already moved to other islands (WHO, 2018).

A total of 10% of the Kiribati population will experience mild to moderate mental illness, and nearly 3% will experience a severe form of mental illness. The service utilisation data of 2011 estimates that only 364 people were treated, and those who were treated might have a severe mental disorder. The country holds only one psychiatric unit; there are no community mental health services, so most support and care are given to individual family members (Oten et al., 2013). A significant concern related to extreme weather events in the studied PICs is the decreased access to health services, which often occurs in the setting of disasters.

7.5 Conclusions

Climate change impacts, especially those from extreme events, have significant effects on mental health, though underexplored. The authors first conducted a bibliometric analysis of the relevant literature to explore the relationship between climate change and mental health impacts. The bibliometric analysis results highlighted that effects of heat and temperature predominate the literature, with secondary impacts such as loss of culture or community remaining largely absent in connection with mental health. In addition, anxiety, suicide and depression came up most often, with many other psychological disorders significantly impacted by individuals' experiences of their environments. To better understand the relationship between climate change, extreme weather events and mental health, the authors analysed several case studies in the Pacific region. A common thread amongst the case studies was the importance of place and local communities to small island nations, predominantly indigenous communities. The compound effects of climaterelated disasters on critical infrastructure such as housing, food delivery, education and essential service provision can significantly impact Pacific Island communities' physical and mental health. Additionally, the lack of health infrastructure to treat people suffering from mental health issues, especially after a disaster, presents a barrier to receiving care.

The findings illustrate the need for improvements in the public health systems of Pacific nations so that they are in a better position to cope with the pressures posed by a changing environment. Apart from increased funding for handling more patients, a greater awareness about the connections between climate change and mental health is needed among public health professionals, so that they may be better able to recognise the symptoms and relate them to their climate-related causes.

8. Assessing causes and implications of climate-induced migration in Kenya and Ethiopia

This chapter was published as Leal Filho, W., ... Kovaleva, M. et al. (2023) *Environmental Science and Policy*, DOI: 10.1016/j.envsci.2023.103577

8.1 Abstract

Climate-induced migration is an increasingly pressing issue in many African regions, as rising temperatures and extreme weather events have caused the displacement of vulnerable populations. This is especially so in Eastern Africa and the Horn of Africa, particularly Kenya and Ethiopia, where extreme weather events have led to rangeland degradation, crop failures, water shortages, and food insecurity. Hence, there is a perceived need to understand these processes better. Against this background, this paper reports on a study investigating the processes associated with climate-induced migration in Kenya and Ethiopia. The research method used consisted of an expertdriven assessment approach, which assesses the causes of climate-induced migration in Kenya and Ethiopia and its human and social implications on local communities. Data were collected from 110 experts residing and working on climate and migration-related issues in Ethiopia and Kenya via e-mail, whose knowledge of the current situation has enabled the identification of some important trends. The results show that climate change is a primary trigger of migration both internally and externally. The high number of migrants, many of whom living within levels of poverty in their home areas is straining resources and services in the receiving regions. Their presence is also leading to increased competition for jobs and resources. Additionally, it has increased urban poverty, as many migrants have little access to living space and health care. This paper provides a welcome addition to the literature in that it lists the causes and implications of climate-induced migration and, by doing so, fosters a better understanding of the current crisis and its implications. The implications of this paper to the overall knowledge of climate change and migration are twofold. First, it highlights the need for governments, international organisations, and other stakeholders better to understand the complex linkages between climate change and migration. Secondly, it shows the usefulness of better recognising how climate change can drive migration and the other factors shaping the decision to migrate. The paper concludes by stating the urgent need for policies and programmes that support climate change-induced migrants. Also, it draws attention to the usefulness of promoting sustainable development in their origin countries and destinations, so that migration is not necessarily perceived as the only response to climate change. A further conclusion is that there is a perceived need for providing access to resources such as education, health care, and livelihood opportunities and establishing mechanisms to ensure a safe and dignified return for those who choose to do so.

9. Synthesis of results

Addressing the interconnections between climate change and gender is often presented as one of the integral elements for successful development and implementation of mitigation and adaptation strategies and measures. An analysis of scientific and scholarly contributions that can be referred to one or more of the domains of climate change-gender interconnections shows an increasing attention of experts and research to the subject, particularly concerning vulnerability and associated issues. However, the number of publications remains significantly lower compared to other research areas related to climate change. On the other hand, the obtained findings reflect a notable role played by international organizations in supporting practical activities with specific narrowed-down targeted outcomes. Their focus is more often on empowering women and reducing their vulnerability to climate change impacts at local levels. Overall, the results of the analysis offer insights to develop scientifically-based studies on the domains of climate changegender interconnections which contribute to effectiveness of inclusive adaptation and mitigation solutions.

Leadership/empowerment, particularly of women, is linked to successfully combating climate change and improving resilience capacity of communities. This is often realized by reducing women underrepresentation in decision-making processes and positions through a wide range of measures including quota systems, eligibility and legal requirements. However, experts participated in the study noted that the number of leadership positions held by women remains low, as does their influence on climate change-related decisions. Such slow progress raises questions about the barriers hindering the process and effectiveness of already taken steps. Economic sectors vary in their what could be called gender-related support and efforts. Agriculture, for example, is undergoing the transition of domination from one gender group to another. The feminization of agriculture has created settings where women's underrepresentation. Given the role played by the sector in climate change, the phenomenon of feminization opens discussion and raises a wide range of questions including:

- To what extent a higher share of women in agriculture reduces its contribution to climate change?
- To what extent a higher share of women in agriculture improves the sector's resilience to climate change impacts?
- Is the consideration of climate change-gender interconnections in agriculture among the agenda top priorities in the context of the sector's feminization?
- What are the necessary characteristics of women's empowerment and leadership in agriculture to ensure a significant contribution of the sector, particularly to global climate change mitigation efforts? The characteristics might include the number and type of the leadership positions to be held, their level (local, national or regional), specific areas of empowerment, etc.

Geographical locations also play a role in advancement of the gender/women subject. Despite the overall growing trend, in some regions, for example, in CA, the scientific discussion on gender/women issues in the climate change context, including adaptation and mitigation remains at an initial stage. Consequently, there is a significant lack of scientifically-based evidence from the CA region to be used to foster the development of efforts towards consideration of climate change-gender

interconnections. Those activities that can be linked to this context have mainly been carried out in the frame of development projects funded by international organizations. This type of support implies, among others, the applied nature of narrow-focused interventions to reduce women's vulnerability and increase their empowerment and leadership opportunities. Additionally, the engagement in these projects familiarizes local experts with the widely disseminated notion of women's higher vulnerability to climate change and the importance of their participation in decision-making processes. However, it is worth mentioning that there is still a significant gap between the notion and actual research studies that can provide numerous evidences from each CA country or the region as a whole to support these statements. Another specificity of the CA region is national gender-oriented political and legal systems that ensure equal rights and opportunities for men and women, including quota systems to ensure a minimum share of women in power.

Nevertheless, it remains unclear at what stage 'equality' in climate change-gender interconnections may become a hindrance in the mitigation and adaptation contexts. This, as well as the aforementioned questions are also applied to the CA agricultural sector, considering its role in the national economies, the feminization trend experienced by the countries and their international obligations towards achieving the SDGs. Similar to agriculture, the UN SDGs represent another thematic area where climate change and gender are interconnected. The obtained findings highlight the inclusion of gender equality issues in strategies as one of the key factors to successfully achieve the SDGs. The addressed issues can be referred to one or more domains of the climate change-gender interconnections, depending on the goal, scope and implementation steps. Additionally, national and international SDGs efforts create a narrative with higher visibility of the interconnections and a set of evidences to demonstrate potential gains of their consideration.

Mental health represents another thematic area where the climate change-gender interconnections framework could be taken into account, particularly focusing on the vulnerability domain. The results in Pacific nations demonstrate that the consideration of gender differences, while identifying climate-related symptoms adds value to the improvements needed in the public health systems in the region.

Climate change-induced migration represents different settings for climate changegender interconnections. Unlike in many other thematic areas (e.g., agriculture), where the discussion is mainly linked to women and associated inequalities, climate change-induced migration is a men-dominated phenomenon. Therefore, it requires a redefined set of imbalances regarding knowledge, access to resources, opportunities, power distribution, and participation decision-making processes. These imbalances are often women-related and are listed among the main causes of their vulnerability to climate change. This type of settings also entails to explore the role and prerequisites of the inclusion approach as a tool for development of effective solutions for climate change mitigation and adaptation.

The here presented research work has several limitations. Some of these limitations are associated with the complexity and novelty of the subject. The novelty has been associated with the considerably low amount of peer-reviewed literature in English on gender/women in climate change adaptation and mitigation, particularly from the climate change perspective compared to other research areas in other climate-related contexts. This created a challenge in identifying a sufficient number of studies to support the findings of this research work. The complexity of the subject limited the number of variables shaping the climate change-gender interconnections that could be considered and investigated to keep this study feasible.

Another limitation concerns the level of bias in formation of survey respondents' opinions on the subject. Their responses may have been shaped by insufficient gender-disaggregated data, limited related information and local evidences, as well as a societal perception of the role of gender and women in a specific context. Bias also reduced the number of experts engaged in climate change adaptation and mitigation or related fields who were expected to participate in the surveys. Their willingness (not) to participate in the surveys was based on (not) acknowledging the relevance and importance of the consideration of gender differences in their expert area.

Nevertheless, despite the aforementioned limitations, this research contributes to the body of work on the linkages between climate change and gender/women in both adaptation and mitigation contexts. The findings reflect a new perspective under a unified umbrella of the climate change-gender interconnections framework. This research work provides insights into the progress made towards considering men's and women's differences in adaptation and mitigation, including potential added value it might bring from a climate change perspective at global level. It also outlines narrower issues of the three domains across the climate change impacted sectors which require more thorough attention and a broader range of action to better integrate the gender/women variable in adaptation and mitigation solutions. In addition to providing more evidences of disproportional effects of climate change on men and women and collecting sex-disaggregated data, among the other issues are

- transition in power and decision-making;
- impact assessment of gendered measures;
- women agents of change as consumers and users compared to decisionmakers; and
- specific role(s) of gender/women in efficient and long-lasting climate policies.

The climate change-gender interconnections framework can be also used to link the outcomes of applied measures that are often detached and scattered across sectors and geographical locations to support their scaling up and replication in a structured and consolidated manner.

10. Outlook for future research

The dissertation lays down the basis for broader multidisciplinary research on the use of climate change-gender interconnections to design new and/or more effective integrated scalable adaptation and mitigation solutions at a global level. Considering the already worldwide recognition of such interconnections, per se, the *following directions* in these efforts may be pursued based on the results obtained here.

The studies have shown the need to *introduce settings and instruments to track the till today progress* in addressing or considering these interconnections and particularly, their *impact* on climate change adaptation and mitigation in a quantitative manner. This will enable experts to establish a baseline to set specific national and international quantitative goals to be attained, followed by the development of quantifiable indicators for reporting purposes.

Along with that, a greater emphasis should be given to *investigate the optimal ways of integrating these interconnections* to achieve maximum outcomes from a climate change perspective. For instance, what might be the optimal men-women ratio of empowered persons and those on leadership positions to design and implement the most effective mitigation and adaptation activities. On the other hand, the indicators will provide an opportunity to continuously assess the national performance with reference to climate change-gender interconnections in adaptation and mitigation. Given the different priorities that gender/women issues have in the climate change context across the nations, more attention should be paid to the development of more detailed implementation steps of global initiatives taking into account countries' environmental, political and socio-economic specificities.

Another point to be further explored is the *minimum combination of additional factors required to maximize the effect* of using climate change-gender interconnections. Among these factors experts often name additional knowledge, statistical gender-disaggregated data, access to a specific type of resources, participation in decision-making processes at various levels, political will, and stronger climate-change-related binding policies and regulations, etc. However, not all of them might have the same value and impact when including the interconnections in mitigation and adaptation measures. Therefore, a more detailed assessment of each factor in this context should be undertaken to establish their ranking to make an 'optimal' choice for different types of interconnections.

The studies conducted in the frame of this dissertation also illustrate the *need to increase the visibility of the actual benefits* of considering women-men differences that represent the 'gender' side of the interconnections to the wider audience. More detailed works should be undertaken to isolate gains that have 'purely' resulted from gender sensitized adaptation and mitigation action.

Another topic that requires increased attention is an *analysis of barriers that impede the implementation* of the already introduced measures that can be referred to one or more domains of the climate change-gender interconnections in the way to achieve planned outcomes. It is worth mentioning that any type of research conducted with reference to the climate change-gender interconnections, whether theoretical or applied, needs to provide a possibility to aggregate and extrapolate obtained findings at a global level.

11. References

- Abbas, K., Li, S., Xu, D., Baz, K., Rakhmetova, A. (2020) Do socioeconomic factors determine household multidimensional energy poverty? Empirical evidence from South Asia. Energy Policy 146, 111754. https://doi.org/10.1016/j.enpol.2020.111754
- Abbass, K., Qasim, M.Z., Song, H., Murshed, M., Mahmood, H., Younis, I. (2022) A review of the global climate change impacts, adaptation, and sustainable mitigation measures. Environmental Science and Pollution Research 29, 42539– 42559. https://doi.org/10.1007/s11356-022-19718-6
- Abdulkadir, I., Kumar, J.S., Noon, M. (2019) Ratio of Land Consumption Rate to the Population Growth Rate- A Case of Metropolitan Gombe. 2019120047. https://doi.org/10.20944/preprints201912.0047.v1
- Abebaw, A.W. (2022) Climate change and water resource management in Ethiopia. Journal of Geology & Geophysics 11(7), 10001043.
- Abebe, A.M. (2014) Climate change, gender inequality and migration in East Africa. Wash. Journal of Environmental Law and Policy 4(1), 105–137.
- Abebe, O.J. (2016) An Analysis of Women and Sustainable Development Goals. UN Women. http://dx.doi.org/10.2139/ssrn.2797855
- Abualtaher, M., Rustad, T., Bar, E.S. (2021) Systemic Insights on the Integration of UN Sustainable Development Goals within the Norwegian Salmon Value Chain. Applied Sciences 11, 12042. https://doi.org/10.3390/app112412042
- Adams, J. (2013) The fourth age of research. Nature 497(7451), 557–560. https://doi.org/10.1038/497557a
- Adamu, M. (2012) Climate Change, Gender Inequality, and Migration: an Ethiopian Case Study. Lambert Academic Publishing.
- Adzawla, W., Baanni Azumah, S., Yao Anani, P., Donkoh, S.A. (2019) Gender perspectives of climate change adaptation in two selected districts of Ghana. Heliyon 5(11), e02854. https://doi.org/10.1016/j.heliyon.2019.e02854
- Adzawla, W., Baumüller, H., Donkoh, S.A., Serra, R. (2020) Effects of climate change and livelihood diversification on the gendered productivity gap in Northern Ghana. Climate and Development 12, 743–755.
- Agarwal, B. (2018) Gender equality, food security and the sustainable development goals. Current Opinion in Environmental Sustainability 34, 26–32. https://doi.org/10.1016/j.cosust.2018.07.002
- Agency on Statistics under President of the Republic of Tajikistan (2019a) Time series of gender indicators to the Strategy of enhancing the role of women in Tajikistan. https://www.stat.tj/en/gender-database Accessed 7 Nov 2019
- Agency on Statistics under President of the Republic of Tajikistan (2019b) Macroeconomic indicators. https://www.stat.tj/en/macroeconomic-indicators Accessed 7 Nov 2019
- Agency on Statistics under President of the Republic of Tajikistan (2019c) Real sector, Employment by economic of activity, 2011–2019. https://www.stat.tj/en/database-real-sector Accessed 7 Nov 2019
- Aguilar, L., Granat, M., Owren, C. (2015) Roots for the Future: The Landscape and Way Forward on Gender and Climate Change; IUCN and GGCA: Washington, DC, USA.
- Afriyie, K., Ganle, J.K., Santos, E. (2018) The floods came and we lost everything: Weather extremes and households' asset vulnerability and adaptation in rural Ghana. Climate and Development 10, 259–274.
- Agency for Strategic Planning and Reforms of the Republic of Kazakhstan Bureau of National Statistics. (2023) Statistics of National Accounts. https://stat.gov.kz/en/industries/economy/national-accounts/dynamictables/ Accessed 17 Feb 2023
- Agency for Strategic Planning and Reforms of the Republic of Kazakhstan Bureau of National Statistics. Share of Employed People by Groups of Types of Economic Activities, by Sex. (2022a) https://gender.stat.gov.kz/page/frontend/detail?id=21andslug=-16andcat_id=7andlang=en Accessed 25 Feb 2023
- Agency for Strategic Planning and Reforms of the Republic of Kazakhstan Bureau of National Statistics (2022b) Proportion of Women, Headed Peasant or Farm Households (PFH).

https://gender.stat.gov.kz/page/frontend/detail?id=5andslug=-

5andcat_id=1andlang=en Accessed 5 March 2023

- Agency on Statistics under President of the Republic of Tajikistan (2020) Women and Men of the Republic of Tajikistan. https://stat.tj/storage/posts/May2021/Tajikistan_Statistical_Publication.pdf Acce ssed on 25 Feb 2023
- Agency on Statistics under the President of the Republic of Tajikistan (2021) Population of the Republic of Tajikistan as of 1 January 2021. 30 Years of the State Independence.

https://stat.tj/storage/posts/August2021/macmuai_sumorai_aholi_01.01.2021.pdf Accessed on 25 Feb 2023

- Agu, H.U., Andrew, C., Gore, M.L. (2021) Mapping Terra Incognita: An Expert Elicitation of Women's Roles in Wildlife Trafficking. Frontiers in Environmental Science 2, 683979.
- Ahmed, I., McDonnell, T. (2020) Prospects and constraints of post-cyclone housing reconstruction in Vanuatu drawing from the experience of tropical cyclone Harold. Progress in Disaster Science 8, 100126.
- Aipira, C., Kidd, A., Morioka, K. (2017) Climate change adaptation in pacific countries: fostering resilience through gender equality. In Leal Filho, W. (Ed.) Climate change adaptation in Pacific countries. Climate change management. Cham: Springer. https://doi.org/10.1007/978-3-319-50094-2_13
- Akel, M., Berro, J., Rahme, C., Haddad, C., Obeid, S., Hallit, S. (2021) Violence Against Women During COVID-19 Pandemic. Journal of Interpersonal Violence. https://doi.org/10.1177/0886260521997953
- Akrofi, M.M., Mahama, M., Nevo, C.M. (2021) Nexus between the gendered socioeconomic impacts of COVID-19 and climate change: Implications for pandemic recovery. SN Social Sciences 1, 198.
- Akter, S., Rutsaert, P., Luis, J., Htwe, N.M., San, S.S., Raharjo, B., Pustika, A. (2017)
 Women's empowerment and gender equity in agriculture: A different perspective from Southeast Asia. Food Policy 69, 270–279. https://doi.org/10.1016/j.foodpol.2017.05.003
- Alcamo, J., Thompson, J., Alexander, A., Antoniades, A., Delabre, I., Dolley, J., Marshall, F., Menton, M., Middleton, J., Scharlemann, J.P.W. (2020) Analysing interactions among the sustainable development goals: Findings and emerging issues from local and global studies. Sustainability Science 15(6), 1561–1572. https://doi.org/10.1007/s11625-020-00875-x

- Alderman, K., Turner, L.R., Tong, S. (2012) Floods and human health: a systematic review. Environment International 47, 37–47. https://doi.org/10.1016/j.envint.2012.06.003
- Aleixandre-Tudo, J.L., Bolaños-Pizarro, M., Aleixandre, J.L., Aleixandre-Benavent, R. (2019) Current trends in scientific research on global warming: A bibliometric analysis. International Journal of Global Warming 17(2), 142–169. https://doi.org/10.1504/IJGW.2019. 097858
- Alexander, L.V., Allen, S.K., Bindoff, N.L., Br´eon, F.-M., Church, J.A., Cubasch, U., et al. (2013) IPCC 2013: summary for policymakers. In Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change.
- Alffram, H. (2011) Equal Access to Justice A Mapping of Experiences. Sida. https://publikationer.sida.se/contentassets/8d1d0ea3d9464589af9259c07937ce35 /equalaccess-to-justice-a-mapping-of-experiences_3124.pdf
- Agu, H., Gore, M.L. (2020) Women in wildlife trafficking in Africa: A synthesis of literature. Global ecology and Conservation 23. https://www.iied.org/role-influence-impactwomen-biodiversity-conservation
- Agyemang, F.S.K., Morrison, N. (2017) Recognising the barriers to securing affordable housing through the land use planning system in Sub-Saharan Africa: A perspective from Ghana. Urban Studies 55(12), 2640–2659. https://doi.org/10.1177/0042098017724092
- Alarcón, D.M., Cole, S. (2019) No sustainability for tourism without gender equality. Journal of Sustainable Tourism 27(7), 903–919.
- Alhamshry, A., Fenta, A.A., Yasuda, H., Kimura, R., Shimizu, K. (2020) Seasonal rainfall variability in Ethiopia and its long-term link to global sea surface temperatures. Water 12(1). https://doi.org/10.3390/w12010055
- Allen, E., Lyons, H., Stephens, J.C. (2019) Women's leadership in renewable transformation, energy justice and energy democracy: Redistributing power. Energy Research & Social Science 57, 101233.
- Alston, M. (2013) Women and adaptation. Wiley Interdisciplinary Reviews: Climate Change 4, 351–358.
- Alston, M., Kent, J. (2008) The Big Dry: The link between rural masculinities and poor health outcomes for farming men. Journal of Sociology 44, 133–147.
- Altunbas, Y., Gambacorta, L., Reghezza, A., Velliscig, G. (2022) Does gender diversity in the workplace mitigate climate change? Journal of Corporate Finance 77, 102303.
- Ampaire, E.L., Acosta, M., Huyer, S., Kigonya, R., Muchunguzi, P., Muna, R., Laurence, J. (2020) Gender in climate change, agriculture, and natural resource policies: Insights from east Africa. Climatic Change 158(1), 43–60. https://doi.org/10.1007/s10584-019-02447-0
- Andrijevic, M., Cuaresma, J.C., Lissner, T., Thomas, A., Schleussner, C.-F. (2020) Overcoming gender inequality for climate resilient development. Nature Communications 11, 6261.
- Angula, M.N., Mogotsi, I., Lendelvo, S., Aribeb, K.M., Iteta, A.-M., Thorn, J.P.R. (2021) Strengthening Gender Responsiveness of the Green Climate Fund Ecosystem-Based Adaptation Programme in Namibia. Sustainability 13, 10162.
 Annan, K. (2000) We the peoples. United Nations.
- Aon plc (2023) Global Catastrophe Recap 2023. https://www.aon.com/getmedia/7107985e-43d8-412b-a674-7722112cc2b0/20231018-q3-2023-catastrophe-recap.pdf

- Aquino, D.H., Wilkinson, S., Raftery, G.M., Potangaroa, R. (2019) Building back towards storm-resilient housing: lessons from Fiji's cyclone Winston experience. International Journal of Disaster Risk Reduction 33, 355–364.
- Ararat, M., Sayedy, B. (2019) Gender and Climate Change Disclosure: An Interdimensional Policy Approach. Sustainability 11, 7217.
- Arib, E. (2017) Policy, SDGs and fighting corruption for the people. Transparency International.

https://images.transparencycdn.org/images/2018_Report_PolicySDGsandFightin gCorruption_ EN.pdf

- Arimah, B.C. (2001) Slums as expressions of social exclusion: explaining the prevalence of slums in African countries. https://www.oecd.org/dev/pgd/46837274.pdf
- Arora-Jonsson, S. (2011) Virtue and vulnerability: Discourses on women, gender and climate change. Global Environmental Change 21(2), 744–751. https://doi.org/10.1016/j.gloenvcha.2011.01.005
- Asadikia, A., Rajabifard, A., Kalantari, M. (2020) Systematic prioritisation of SDGs: Machine learning approach. World Development. https://doi.org/10.1016/j.worlddev.2020.105269
- Asian Development Bank (2016a) Tajikistan Country Gender Assessment; Asian Development Bank: Mandaluyong City, Philippines.
- Asian Development Bank (2016b) Uzbekistan: Surkhandarya Water Supply and Sanitation Project; Project Number: 40007-013, Loan Number: 2466, Grant Number: 0131 July 2016; Asian Development Bank: Mandaluyong City, Philippines.
- Asian Development Bank (2017) Uzbekistan: Water Resource Management Sector Project. Completion Report August. Project Number: 40086-013 Loan Numbers: 2492 and 2493. https://www.adb.org/sites/default/files/projectdocuments/40086/40086-013-pcr-en.pdf Accessed 25 February 2023
- Asian Development Bank (2018) Uzbekistan Country Gender Assessment Update; Asian Development Bank: Manila, Philippines.
- Asian Development Bank (2020) A Study of Women's Role in Irrigated Agriculture in the Lower Vaksh River Basin, Tajikistan. https://www.adb.org/sites/default/files/publication/663141/womens-roleirrigated-agriculture-tajikistan.pdf Accessed 25 Feb 2023
- Asian Development Bank (2023) Members Fact Sheets. https://www.adb.org/publications/series/fact-sheets Accessed 25 April 2023
- Asongu, S.A., Messono, O.O., Guttemberg, K.T.J. (2021) Women Political Empowerment and Vulnerability to Climate Change: Evidence from Developing Countries. MPRA Working Paper No. 109849. https://mpra.ub.unimuenchen.de/109849/ Accessed 5 March 2023
- Assaduzzaman, M., Filatova, T., Lovett, J.C., Coenen, F.H.J.M. (2023) Gender-Ethnicity Intersectionality in Climate Change Adaptation in the Coastal Areas of Bangladesh. Sustainability 15, 3744.

Atif, M., Hossain, M., Alam, M.S., Goergen, M. (2021) Does board gender diversity affect renewable energy consumption? Journal of Corporate Finance 66, 101665.

Australian Government (2018) Report on the implementation of the sustainable Development goals.

https://www.sdgdata.gov.au/sites/default/files/voluntary_national_review.pdf

AWGGCC (2017) Gender Analysis of the Paris Agreement and Implications for Africa. International Development Research Centre: Ottawa, ON, Canada.

- Awiti, A.O. (2022) Climate Change and Gender in Africa: A Review of Impact and Gender-Responsive Solutions. Frontiers in Climate 4, 895950.
- Ayal, D.Y. (2020) Disaster risk management strategies: building the resilient human settlements. In Leal Filho, W., Marisa Azul, A., Brandli, L., Gokçin Ozuyar, P., Wall, T. (Eds.) Sustainable Cities and Communities. Encyclopedia of the UN Sustainable Development Goals. Springer, Cham.
- Ayal, D., Leal Filho, W. (2017) Farmers' perceptions of climate variability and its adverse impacts on crop and livestock production in Ethiopia. Journal of Arid Environments 140, 20-28.
- Ayanlade, A., Oluwaranti, A., Ayanlade, O.S., Borderon, M., Sterly, H., Sakdapolrak, P., Jegede, M.O., Weldemariam, L.F., Ayinde, A.F.O. (2022) Extreme climate events in sub-Saharan Africa: a call for improving agricultural technology transfer to enhance adaptive capacity. ClimateSERV 27, 100311.
- Azcona, G., Bhatt, A., Encarnacion, J., Plazaola-Castaño, J., Seck, P., Staab, S., Turquet, L. (2020) From insights to action: Gender equality in the wake of COVID-19. United nations entity for gender equality and the empowerment of women (UN Women).
- Babugura, A. (2010) Gender and Climate Change: South Africa Case Study. Heinrich Böll Foundation Southern Africa: Cape Town, South Africa.
- Bahoo, S.B., Ilan, A., Paltrinieri, A. (2020) Corruption in international business: A review and research agenda. International Business Review 29(4), 101660. https://doi.org/10.1016/j.ibusrev.2019.101660
- Bali Swain, R., Ranganathan, S. (2021) Modeling interlinkages between sustainable development goals using network analysis. World Development 138, 105136. https://doi.org/10.1016/j.worlddev. 2020.105136
- Bamber, P., Fernandez-Stark, K. (2013) Global value chains, economic upgrading, and gender in the horticulture industry. In Staritz, C., Reis, J.G. (Eds.) Global value chains, economic upgrading and gender. Case studies of the horticulture, tourism, and call center industries. World Bank. https://openknowledge.worldbank.org/bitstream/handle/10986/16976/832330W P0GVC0G0Box0382076B00PUBLIC0.pdf?sequence=1andisAllowed=y
- Barbier, B., Burgess, J.C. (2019) Sustainable development goal indicators: Analyzing trade-offs and complementarities. World Development 122, 295–305. https://doi.org/10.1016/j.worlddev.2019.05.026
- Barr, J. (1999) Drought assessment: the 1997-98 El Nino drought in Papua new guinea and the Solomon islands. Australian Journal of Emergency Management 14(2), p. 31.
- Barsted, L. (2005) The legal status of women in the context of agrarian reform. FAO. http://www.fao.org/3/a0297e/a0297e06.htm#bm6
- Batsaikhan, U., Dabrowski, M. (2017) Central Asia twenty-five years after the breakup of the USSR. Russian Journal of Economics 3, 296–320. https://doi.org/10.1016/j.ruje.2017.09.005
- Batyrova, G., Tlegenova, Z., Umarova, G., Kononets, V., Umarov, Y., Kudabayeva, K., Aitmaganbet, P., Amanzholkyzy, A. (2021) Microelement status of the adult population in western Kazakhstan. Ekologiya cheloveka (Human Ecology) 11, 42–49.
- BBC (2014) Deadly flash floods hit Solomon Islands' capital Honiara, Asia. www.bbc.com/news/world-asia-26880142 Accessed 15 April 2021

- Begashaw Abate, G., Terefe Woldie, A. (2022) Breaking Barriers to Women's Advancement in the Public Sector in Sub-Saharan Africa. Canadian Bureau for International Education: Ottawa, ON, Canada.
- Belsey-Priebe, M., Lyons, D., Buonocore, J.J. (2021) COVID-19s Impact on American Women's Food Insecurity Foreshadows Vulnerabilities to Climate Change. International Journal of Environmental Research and Public Health 18, 6867.
- Belter, C.W., Seidel, D.C. (2013) A bibliometric analysis of climate engineering research. Wiley Interdisciplinary Reviews: Climate Change 4(5), 417–427. https://doi.org/10.1002/wcc.229
- Bennich, T., Weitz, N., Carlsen, H. (2020) Deciphering the scientific literature on SDG interactions: A review and reading guide. Science of the Total Environment 728, 138405. https://doi.org/10. 1016/j.scitotenv.2020.138405
- Benson, C. (1997) The Economic Impact of Natural Disasters in Fiji. Overseas Development Institute (ODI), London.
- Beqiraj, J., McNamara, L. (2016) Children and Access to Justice in the Agenda for Sustainable Development. Briefing Paper by the Bingham Centre for the Rule of Law.
- Bereket, T.H., Kassahun, T.B., Tadesse, T.Z., Desalegn, Y.A., Gudina, L.F., Fikiru, A.A. (2022) Drought analysis using standardised evapotranspiration and aridity index at bilate watershed: sub-basins of Ethiopian rift valley. Scientific World Journal 1181198 1–4, 181198.
- Berger, G. (2020) New Opportunities in Monitoring Safety of Journalists through the UN's 2030 Sustainable Development Agenda. Media and Communication 8(1). http://dx.doi.org/10.17645/mac.v8i1.2660
- Berrang-Ford, L., Lesnikowski, A., Fischer, A., Siders, A.R., Mach, K., Thomas, A., et al. (2020) The Global Adaptation Mapping Initiative (GAMI): Part 1 Introduction and Overview of Methods. https://doi.org/10.21203/rs.3.pex-1240/v1
- Berry, H.L., Bowen, K., Kjellstrom, T. (2010) Climate change and mental health: a causal pathways framework. International Journal of Public Health 55(2), 123–132.
- Berry, H.L., Waite, T.D., Dear, K.B.G., Capon, A.G., Murray, V. (2018) The case for systems thinking about climate change and mental health. Nature Climate Change 8(4), 282-290.
- Bessah, E., Raji, A.O., Taiwo, O.J., Agodzo, S.K., Ololade, O.O., Strapasson, A., Donkor, E. (2021) Gender-based variations in the perception of climate change impact, vulnerability and adaptation strategies in the Pra River Basin of Ghana. Journal of Climate Change 13, 435–462.
- Biggeri, M., Clark, D.A., Ferrannini, A., Mauro, V. (2019) Tracking the SDGs in an 'integrated' manner: A proposal for a new index to capture synergies and tradeoffs between and within goals. World Development 122, 628–647. https://doi.org/10.1016/j.worlddev.2019.05.022
- Bhandari, M.P., Shvindina, H. (2019) Reducing Inequalities Towards Sustainable Development Goals: Multilevel Approach. River Publishers, Denmark/the Netherlands.
- Bhugra, D. (2004) Migration and mental health. Acta Psychiatrica Scandinavica 109(4), 243–258.
- Birindelli, G., Iannuzzi, A.P., Savioli, M. (2019) The impact of women leaders on environmental performance: Evidence on gender diversity in banks. Corporate Social Responsibility and Environmental Management 1485–1499.

- Birkmann, J., Liwenga, E., Pandey, R., Boyd, E., Djalante, R., Gemenne, F., Leal Filho, W. et al. (2022) Poverty, livelihoods and sustainable development. In Portner, H.-O., Roberts, D.C., Tignor, M., Poloczanska, E.S., Mintenbeck, K. et al. (Eds.) Climate Change 2022: Impacts, Adaptation and Vulnerability. Contribution of Working Group II to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge, UK and New York, NY, USA, pp. 1171–1274.
- Björnberg, K.E., Hansson, S.O. (2013) Gendering local climate adaptation. Local Environment 18, 217–232.
- Black, R., Arnell, N.W., Adger, W.N., Thomas, D., Geddes, A. (2013) Migration, immobility and displacement outcomes following extreme events. Environmental Science and Policy 27, S32-S43.
- Blei, A.M., Angel, S., Civco, D.L., Liu, Y., Zhang, X. (2018) Accuracy Assessment and Map Comparisons for Monitoring Urban Expansion: The Atlas of Urban Expansion and the Global Human Settlement. Lincoln Institute of Land Policy.
- Blind, P.K. (2019) How relevant is governance to financing for development and partnerships? Working Paper No. 162 ST/ESA/2019/DWP/162. United Nations Department of Economic and Social Affairs. https://www.un.org/esa/desa/papers/2019/wp162_2019.pdf
- Block, J.H., Fisch, C. (2020) Eight tips and questions for your bibliographic study in business and management research. Management Review Quarterly 70, 307–312.
- Bob, U., Babugura, A. (2014) Contextualising and conceptualising gender and climate change in Africa. Agenda 28, 3–15.
- Bolaji-Adio, A. (2015) The Challenge of Measuring SDG 16: What Role for African Regional Frameworks? Discussion Paper No. 175. https://ecdpm.org/wp-content/uploads/DP175- Challenge-Measuring-SDG16-May-2015.pdf
- Boserup, E. (1970) Woman's role in economic development. Earthscan, London/New York.
- Botreau, H., Cohen, M.J. (2020) Gender inequality and food insecurity: A dozen years after the food price crisis, rural women still bear the brunt of poverty and hunger. Advances in Food Security and Sustainability 5, 53–117.
- Bosha, S.L. (2014) Quota Systems and Women Political Leadership Development in Africa. Journal of African Union Studies 3, 103–114.
- Bourgault, S., Peterman, A., O'Donnell, M. (2021) Violence against women and children during COVID-19—One Year On and 100 Papers. A Fourth Research Round Up. CGD Notes. Center for Global Development.
- Bradshaw, S., Chant, S., Linneker, B. (2017) Gender and poverty: what we know, don't now, and need to know for Agenda 2030. Gender, Place & Culture 24(12), 1667-1688. https://doi.org/10.1080/0966369X.2017.1395821
- British Council (2018) A cultural relations contribution to peace, justice and strong institutions.

https://www.britishcouncil.org/sites/default/files/j082_sustainable_development _goals_report _final_web.pdf

- Brixiová, Z., Kangoye, T., Tregenna, F. (2020) Enterprising women in Southern Africa: When does land ownership matter? Journal of Family and Economic Issues 41, 37–51.
- Broeckhoven, N. (2014) Biodiversity Loss and Climate Change: Gender Issues in International Law and Policy. DiGeSt. Journal of Diversity and Gender Studies 1(2), 23–38.

- Bromley, M., Caparini, M., Malaret, A. (2019) Measuring illicit arms and financial flows: improving the assessment of sustainable development goal 16. SIPRI Background Paper. https://www.sipri.org/sites/default/files/2019-07/bp_1907_sdg_16.pdf
- Bryan, E., Deressa, T.T., Gbetibouo, G.A., Ringler, C. (2009) Adaptation to climate change in Ethiopia and South Africa: options and constraints. Environmental Science & Policy 12(4), 413–426.
- Brody, A., Demetriades, J., Esplen, E. (2008) Gender and Climate Change: Mapping the Linkages a Scoping Study on Knowledge and Gaps; BRIDGE, Institute of Development Studies (IDS), University of Sussex: Brighton, UK.
- Buhmann, K., Jonsson, J., Fisker, M. (2019) Do no harm and do more good too: connecting the SDGs with business and human rights and political CSR theory. Corporate Governance 19(3), 389–403.
- Bulut, Z. A., Çımrin, F. K., Doğan, O. (2017) Gender, generation and sustainable consumption: Exploring the behaviour of consumers from Izmir Turkey. International Journal of Consumer Studies 41(6), 597–604. https://doi.org/10.1111/ijcs.12371
- Bunce, A., Ford, J. (2015) How is adaptation, resilience, and vulnerability research engaging with gender? Environmental Research Letters 10(12), 123003. https://doi.org/10.1088/1748-9326/10/12/123003
- Butler, C.D., Harley, D. (2010) Primary, secondary and tertiary effects of ecoclimatic change: the medical response. Postgraduate Medical Journal 86(1014), 230–234.
- Bwire, G., Ari, A.R., Eyu, P., Ocom, F., Wamala, J.F., Kusi, K.A., Ndeketa, L. et al. (2022) The COVID-19 pandemic in the African continent. BMC Med 20(1), 167.
- Byravan, S., Rajan, S.C. (2022) Cross-border migration on a warming planet: a policy framework. Wiley Interdisciplinary Reviews 13(2), e763.
- Caby, J., Coron, C., Ziane, Y. (2022) The Effect of Top Management Team Gender Diversity on Climate Change Management: An International Study. Sustainability 14, 1032.
- Callister, L.C. (2018) Reducing Hunger Among Women and Children in India. MCN American Journal of Maternal/Child Nursing 43(4), 234. https://doi.org/10.1097/NMC.00000000000445
- Campbell, B.M., Corner-Dolloff, C., Girvetz, E., Loboguerrero, A.M., Ramirez-Villegas, J. (2016) Reducing risks to food security from climate change. Global Food Security 11, 34–43.
- Campbell, B.M., Hansen, J., Rioux, J., Stirling, C.M., Twomlow, S., Wollenberg, E.L. (2018) Urgent action to combat climate change and its impacts (SDG 13): transforming agriculture and food systems. Current Opinion in Environmental Sustainability 34, 13-20. https://doi.org/10.1016/j.cosust.2018.06.005
- CARE (2020) Evicted by Climate Change. In Confronting the Gendered Impacts of Climate-Induced Displacement; CARE Climate Change and Resilience Platform (CCRP); CARE: Atlanta, GA, USA. https://careclimatechange.org/wpcontent/uploads/2020/07/CARE-Climate-Migration-Report-v0.4.pdf Accessed 17 Feb 2022
- CAREC (2020) Women, Food and Climate Change in Central Asia. The Regional Environmental Centre for Central Asia https://zoinet.org/product/women-food-climate-ca/ Accessed 17 Feb 2022
- Caridade, S.M.M., Vidal, D.G., Dinis, M.A.P. (2022) Climate Change and Gender-Based Violence: Outcomes, Challenges and Future Perspectives. In Leal Filho,

W., Vidal, D.G., Dinis, M.A.P., Dias, R. C. (Eds.) Sustainable Policies and Practices in Energy, Environment and Health Research, Springer, pp. 167–176. https://doi.org/10.1007/978-3-030-86304-3_10

- Carothers, T., Brechenmacher, S. (2014) Accountability, transparency, participation, and inclusion. A New Development Consensus? Carnegie Endowment for International Peace.
- Carvajal-Escobar, Y., Quintero-Angel, M., Garcia-Vargas, M. (2008) Women's Role in Adapting to Climate Change and Variability. Advances in Geosciences 14, 277–280.
- Catalyst (2016) Women in Energy: Gas, Mining, and Oil. https://www.catalyst.org/knowledge/women-energygas-mining-oil Accessed 29 April 2022
- Catalyst (2022) Women CEOs of the SandP 500. https://www.catalyst.org/research/women-ceos-of-the-sp-500/ Accessed 6 June 2022
- CDC (2020) Preparing for the regional health impacts of climate change in the United States, Climate and Health Program, Centers for Disease Control and Prevention/National Center Environmental Health.
- CEDAW (2015) Consideration of reports submitted by States parties under article 18 of the Convention, Fifth periodic report of States parties due in 2014: Uzbekistan. https://www.refworld.org/docid/5653091e4.html Accessed 22 Nov 2019
- Central Asia Program (2022) Women and Water in South and Central Asia. https://centralasiaprogram.org/initiatives/women-and-water-in-central-andsouth-asia Accessed 17 Feb 2023
- CEPALSTAT (2019) Millennium development goals in Latin America and the Caribbean Target 3.A: gender equality. https://cepalstatprod.cepal.org/cepalstat/tabulador/ConsultaIntegradaProc_HTML.asp Accessed 28 Mar 2020
- Chand, S.S., Walsh, K.J. (2009) Tropical cyclone activity in the Fiji region: spatial patterns and relationship to large-scale circulation. Journal of Climate 22(14), 3877-3893.
- Chapman, S., Mustin, K., Renwick, A.R., Segan, D.B., Hole, D.G., Pearson, R. G., Watson, J.E.M. (2014) Publishing trends on climate change vulnerability in the conservation literature reveal a predominant focus on direct impacts and long time-scales. Diversity and Distributions 20(10), 1221–1228. https://doi.org/10.1111/ddi.12234
- Chattopadhyay, A., Mukherjee, A., Sudha, G. (2016) Prevailing Basic Facilities in Slums of Greater Mumbai. https://www.iipsindia.ac.in/sites/default/files/IIPS_Working_Paper_No13.pdf
- Chen, Y.Z., Tanaka, H. (2014) Women's Empowerment. In Michalos, A.C. (Ed.) Encyclopedia of Quality of Life and Well-Being Research; Springer: Dordrecht, The Netherlands.
- Chen, Y., Takeuchi, K., Xu, C., Chen, Y., Xu, Z. (2006) Regional climate change and its effects on river runoff in the Tarim Basin, China. Hydrological Processes 20 (10), 2207–2216.
- Cherotich, V.K., Saidu, O., Bebe, B.O. (2012) Access to climate change information and support services by the vulnerable groups in semi-arid Kenya for adaptive capacity development. African Crop Science Journal 20, 169–180.

- Cho, Y., Park, J., Park, H.Y. (2018) Women Leaders in the Corporate Sector. In Cho, Y., McLean, G.N. (Eds.) Korean Women in Leadership, Current Perspectives on Asian Women in Leadership. Palgrave Macmillan, Cham, pp. 121–139.
- Christoff, P., Saucedo Dávila, A., Kaur, J., Sommer, J.M. (2019) Cultivating leadership among Indian women in climate change adaptation. In Chao, C.C., Ha, L., (Eds.) Asian Women Leadership A Cross-National and Cross-Sector Comparison, 1st ed, Routledge: London, UK.
- Clemens, V., von Hirschhausen, E., Fegert, J.M. (2020) Report of the intergovernmental panel on climate change: implications for the mental health policy of children and adolescents in Europe-a scoping review. European Child and Adolescent Psychiatry 31(5), 701-713.
- Climate Investment Funds (2018) CLIMADAPT Gender-Sensitive Climate Resilience Investments in Tajikistan. https://d2qx68gt0006nn.cloudfront.net/sites/cif_enc/files/knowledge
 - documents/1091_gender_daycop24_case_study_final.pdf Accessed 7 Feb 2022
- Clos, J. (2016) A New Urban Agenda for the 21st century: The role of urbanization in sustainable development. OECD Regional Outlook 2016.
- Cohen, B. (2006) Urbanization in Developing Countries: Current Trends, Future Projections, and Key Challenges for Sustainability. Technology in Society 28(1-2), 63-80. https://doi.org/10.1016/j.techsoc.2005.10.005
- Collantes, V., Kloos, K., Henry, P., Mboya, A., Mor, T. Metternicht, G. (2018) Moving towards a twin-agenda: Gender equality and land degradation neutrality. Environmental Science & Policy 89, 247–253.
- Connell, J. (2016) Last days in the Carteret Islands? Climate change, livelihoods and migration on coral atolls. Asia Pacific Viewpoint 57(1), 3–15.
- Connor, J., Madhavan, S., Mokashi, M., Amanuel, H., Johnson, N.R., Pace, L.E., Bartz, D. (2020) Health risks and outcomes that disproportionately affect women during the Covid-19 pandemic: A review. Social Science and Medicine 266, 113364. https://doi.org/10.1016/j.socscimed.2020.113364
- Constitutional Law of the Kyrgyz Republic "On Elections of the President of the Kyrgyz Republic and Deputies of the Jogorku Kenesh of the Kyrgyz Republic". No 68 of 2 July 2011 as Last Amended by Constitutional Law No 81 of 06 April 2023. http://cbd.minjust.gov.kg/act/view/ru-ru/203244?cl=ru-ru#st_21-1 Accessed 27 April 2023
- Constitutional Law of the Republic of Kazakhstan "On Elections in the Republic of Kazakhstan". No 2464 of 28 September 1995, Article 104 (4) as Amended by Constitutional Law 335-VI 3PK of 25 May 2020, to be Enforce on 1 January 2023. https://adilet.zan.kz/eng/docs/Z950002464_ Accessed 27 April 2023
- Conway, D., Mould, C., Bewket, W. (2004) Over one century of rainfall and temperature observations in Addis Ababa, Ethiopia. International Journal of Climatology 24(1), 77–91.
- Cook, N.J., Grillos, T., Andersson, K.P. (2019) Gender quotas increase the equality and effectiveness of climate policy interventions. Nature Climate Change 9, 330– 334.
- Costa, B. (2010) Her mile. Women's rights and access to land. The last stretch of road to eradicate hunger. https://www.landcoalition.org/sites/default/files/documents/resources/HerMile_AAItaly.pdf Accessed 3 Dec 2019
- Council of Europe (2017) Ending all forms of violence against children by 2030: The Council of Europe's contribution to the 2030 Agenda and the Sustainable

Development

Goals.

https://violenceagainstchildren.un.org/sites/violenceagainstchildren.un.org/files/2030_agenda/sdg_leaflet.pdf.pdf

- Crabtree, A. (2012) Climate change and mental health following flood disasters in developing countries, a review of the epidemiological literature: what do we know, what is being recommended. Australasian Journal of Disaster and Trauma Studies 1, 21-30.
- Crawford, E. (2020) Achieving Sustainable Development Goas 5 and 6: The case for gender-transformative water programmes. Oxford, UK: Oxfam. https://doi.org/10.21201/2020.5884
- Cunsolo, W.A., Harper, S.L., Ford, J.D., Edge, V.L., Landman, K., Houle, K., Blake, S. Wolfrey, C. (2013) Climate change and mental health: an expository case study from rigolet, nunatsiavut, Canada. Climatic Change 121(2), 255–270.
- Dagnachew, A.G., Hof, A.F., van Soest, H., van Vuuren, D.P. (2021) Climate change measures and sustainable development goals. PBL Netherlands Environmental Assessment Agency, The Hague.
- Dahan, M., Gelb, A. (2015) The role of identification in the post-2015 development agenda. World Bank Working Paper. http://pubdocs.worldbank.org/en/149911436913670164/WorldBank-Working-Paper-Center-for-Global-Development-Dahan-Gelb-July2015.pdf
- Damodaran, A., Jörgensen, K., Schreurs, M., Beermann, J., Ollier, L. (2015) Sustainable cities – inclusive, green and competitive. https://smartnet.niua.org/sites/default/files/resources/giz20162d0389en2dindo2d german2dsust ainable2dcities.pdf
- Daniel, K. (2016) Public Spaces. A key tool to achieve the sustainable development goals. https://healthbridge.ca/images/uploads/library/Final_Electronic.pdf
- Dankelman, I. (2010) Introduction: Exploring Gender, Environment and Climate change. In Gender and Climate Change: An Introduction; Dankelman, I., Ed.; Routledge: London, UK.
- Dar, M.H., Waza, S.A., Nayak, S., Chakravorty, R., Zaidi, N.W., Hossain, M. (2020) Gender focused training and knowledge enhances the adoption of climate resilient seeds. Technology in Society 63, 101388.
- David, C. C., Ramon, J. Albert, G., Vizmanos, J. (2018) Sustainable Development Goal 5: How Does the Philippines Fare on Gender Equality? Research Paper Series No. 2018- 04. Philippine Institute for Development Studies. Quezon City.
- Davidson, D.J., Freudenburg, W.R. (1996) Gender and environmental risk concerns.EnvironmentandBehavior28(3),302–339.https://doi.org/10.1177/0013916596283003
- Davis, A., Roper, L., Miniszewski, U. (2015) Climate Justice and Women's Rights:
 A Guide to Supporting Grassroots Women's Action; Global Greengrants Fund (GGF): Boulder, CO, USA; International Network of Women's Funds (INWF):
 México City, Mexico, Alliance of Funds: São Paulo, Brazil.
- Dawes, J.H.P. (2022) SDG interlinkage networks: Analysis, robustness, sensitivities, and hierarchies. World Development 149, 105693. https://doi.org/10.1016/j.worlddev.2021.105693
- Dazé, A., Dekens, J. (2018) Towards Gender-Responsive NAP Processes Progress and Recommendations for the Way Forward. NAP Global Network Synthesis Report, 2017–2018, International Institute for Sustainable Development: Winnipeg, MB, Canada.

- Debebe, B., Senbeta, F., Teferi, E., Diriba, D., Teketay, D. (2023) Analysis of forest cover change and its drivers in biodiversity hotspot areas of the Semien Mountains National Park, Northwest Ethiopia. Sustainability 15(4), 3001.
- De Beurs, K.M., Henebry, G.M., Owsley, B.C., Sokolik, I.N. (2018) Large scale climate oscillation impacts on temperature, precipitation and land surface phenology in Central Asia. Environmental Research Letters 13, 065018.
- Decree by the President of the Republic of Uzbekistan on Approval of the "Concept of Environmental Protection of the Republic of Uzbekistan until 2030". No. UP-5863. (2019) https://lex.uz/ru/docs/4574010 Accessed 20 Feb 2023
- Deere, C.D. (2005) The feminization of agriculture? Economic restructuring in rural Latin America. http://www.unrisd.org/80256B3C005BCCF9/(httpPublications)/20024EBC6AB

9DA45C1256FE10045B101?OpenDocument Accessed 12 Nov 2019

- Deere, C.D. (2018) Sustainable Development Goals, Gender Equality and the Distribution of Land in Latin America. Cadernus Pagu 52. https://doi.org/10.1590/18094449201800520006
- de Jong, E., Vijge, M.J. (2021) From millennium to sustainable development goals: Evolving discourses and their reflection in policy coherence for development. Earth System Governance 7, 100087. https://doi.org/10.1016/j.esg.2020.100087
- Dekens, J., Dazé, A. (2019) Conducting Gender Analysis to Inform National Adaptation Plan (NAP) Processes: Reflections from Six African Countries. NAP Global Network; IISD: Winnipeg, MA, Canada.
- Dell, M., Jones, B.F., Olken, B.A. (2012) Temperature shocks and economic growth: evidence from the last half-century. American Economic Journal: Macroeconomics. 4(3), 66–95.
- Del Río Castro, G., González Fernández, M.C., Uruburu Colsa, Á. (2021) Unleashing the convergence amid digitalization and sustainability towards pursuing the sustainable development goals (SDGs): A holistic review. Journal of Cleaner Production 280, 122204. https://doi.org/10.1016/j. jclepro.2020.122204
- Department for International Development (2015) Why corruption matters: understanding causes, effects and how to address them Evidence paper on corruption. Department for International Development, UK Government.
- Desai, D. (2020) Urban Densities and the Covid-19 Pandemic: Upending the Sustainability Myth of Global Megacities. Observer research foundation. https://www.orfonline.org/wpcontent/uploads/2020/05/ORF_OccasionalPaper_2 44_PandemicUrbanDensities.pdf
- Desai, B.H., Mandal, M. (2021) Role of climate change in exacerbating sexual and gender-based violence against women: A new challenge for international law. Environmental Policy and Law 51(3), 137-157.
- de Schutter, O. (2013) The agrarian transition and the "feminization" of agriculture. Food sovereignty: a critical dialogue food sovereignty, 1–43. http://www.iss.nl/fileadmin/ASSETS/iss/Research_and_projects/Research_netw orks/ICAS/37_deSchutter_2013.pdf
- Dessalegn, M., Debevec, L., Nicol, A., Ludi, E. (2023) A critical examination of rural outmigration studies in ethiopia: considering impacts on agriculture in the sending communities. Land 12(1), 176. https://doi.org/10.3390/land12010176
- Development Initiatives Kenya (2019) Tracking Subnational Government Investments in Disaster Risk Reduction in Kenya. https://reliefweb.int/sites/reliefweb.int/files/resources/Tracking-subnationalgovernment-investments-in-disaster-risk-reductionin-Kenya.pdf

- Devisscher, T., Konijnendijk, C., Nesbitt, L., Lenhart, J., Salbitano, F., Cheng, Z.C., Lwasa, S., van den Bosch, M. (2019) SDG 11: Sustainable Cities and Communities – Impacts on Forests and Forest-Based Livelihoods. In Katila, P., Pierce Colfer, C.J., de Jong, W., Pacheco, P., Winkel, G. (Eds) Sustainable Development Goals: Their Impacts on Forests and People. Cambridge University Press, pp 349-385.
- Dhar, S. (2018) Gender and sustainable development goals (SDGs). Indian Journal of Gender Studies 25(1), 47–78. https://doi.org/10.1177/0971521517738451
- DIAUD/ CBM (2016) The Inclusion Imperative: Towards Disability-inclusive and Accessible Urban Development Key Recommendations for an Inclusive Urban Agenda. https://www.cbm.org/fileadmin/user_upload/Publications/The-Inclusion-Imperative-TowardsDisability-Inclusive-and-Accessible-Urb....pdf
- Di Matteo, G., Nardi, P., Grego, S., Guidim, C. (2018) Bibliometric analysis of climate change vulnerability Assessment research. Environmental Systems and Decisions 38(4), 508–516. https://doi.org/10.1007/s10669-018-9687-4
- Dimitrova-Grajzl, V., Obasanjo, I. (2019) Do parliamentary gender quotas decrease gender inequality? The case of African countries. Constitutional Political Economy 30, 149–176.
- Ding, N., Berry, H., O'Brien, L. (2015) The effect of extreme heat on mental health evidence from Australia. International Journal of Epidemiology 44(suppl_1), i64.
- Dixon, J., Gulliver, A., Gibbon, D., Hall, M. (2001) Farming Systems and Poverty: Improving Farmers' Livelihoods in a Changing World (English). Washington, DC World Bank Group. http://documents.worldbank.org/curated/en/126251468331211716/Farming-

systems-and-poverty-improving-farmers-livelihoods-in-a-changing-world Donatti, C.I., Harvey, C.A., Martinez-Rodriguez, M.R., Vignola, R., Rodriguez,

- C.M. (2019) Vulnerability of smallholder farmers to climate change in Central America and Mexico: current knowledge and research gaps. Climate and Development 11(3), 264–286. https://doi.org/10.1080/17565529.2018.1442796
- Doria, M. de F., Boyd, E., Tompkins, E.L., Adger, W.N. (2009) Using expert elicitation to define successful adaptation to climate change. Environmental Science & Policy 12(7), 810–819. https://doi.org/10.1016/j.envsci.2009.04.001
- Doss, C. (2014) If Women Hold up Half the Sky, How Much of the World's Food Do They Produce? In Quisumbing, A., Meinzen-Dick, R., Raney, T., Croppenstedt, A., Behrman, J., Peterman, A. (Eds.) Gender in Agriculture. Springer: Dordrecht, The Netherlands.
- Doss, C.R. (2018) Women and agricultural productivity: Reframing the issues. Development Policy Review 36, 35–50.
- Duc, N. M., Hiep, D. V., Thong, P. M., Zunic, L., Zildzic, M., Donev, D., Jankovic, S. M., Hozo, I., Masic, I. (2020) Predatory Open access journals are indexed in reputable databases: A revisiting issue or an unsolved problem. Medical Archives 74(4), 318–322. https://doi.org/10.5455/medarh.2020.74.318-322
- Dugarova, E. (2018) Gender equality as an accelerator for achieving the Sustainable Development Goals. Discussion Paper. United Nations Entity for Gender Equality and the Empowerment of Women, New York, USA.
- Dukhovny, V.A., Sokolov, V.I., Ziganshina, D.R. (2015) The Role of Donors in Addressing Water Problems in Central Asia. Irrigation and Drainage 65, 79–85.

- Dumenu, W.K., Obeng, E.A. (2016) Climate change and rural communities in Ghana: Social vulnerability, impacts, adaptations and policy implications. Environmental Science & Policy 55, 208–217.
- Dumont, C., Haase, E., Dolber, T., Lewis, J., Coverdale, J. (2020) Climate change and risk of completed suicide. The Journal of Nervous and Mental Disease 208(7).
- Dzebo, A., Shawoo, Z. (2023) Sustainable Development Goal interactions through a climate lens: a global analysis. Stockholm Environment Institute (SEI). https://doi.org/10.51414/sei2023.010
- Ebi, K.L., Bowen, K. (2016) Extreme events as sources of health vulnerability: drought as an example. Weather and Climate Extremes 11, 95–102.
- Ebi, K.L., Vanos, J., Baldwin, J.W., Bell, J.E., Hondula, D.M., Errett, N.A., Hayes, K., Reid, C.E., Shubhayu Saha, Spector, J., Berry, P. (2021) Extreme Weather and Climate Change: Population Health and Health System Implications. Annual Review of Public Health 42, 293-315. https://doi.org/10.1146/annurev-publhealth-012420-105026
- Eckstein, D., Künzel, V., Schäfer, L. (2021) Global climate risk index 2021 Who suffers Most from Extreme Weather Events? Weather-related Loss Events in 2019 and 2000 to 2019. Briefing paper, Germanwatch, Bonn, Germany.
- ECLAC, FAO, IICA (2017) The outlook for agriculture and rural development in the Americas: a perspective on Latin America and the Caribbean 2017–2018. https://repositorio.cepal.org/bitstream/handle/11362/42282/1/OutlookAgricultur e2017-2018.pdf
- Election Code of the Republic of Uzbekistan of 26 June 2019. https://lex.uz/docs/4386846 Accessed 25 Febr 2023
- EM-DAT, CRED / UCLouvain, Brussels, Belgium (2023) https://doc.emdat.be/docs/data-structure-and-content/emdat-public-table/ Accessed 26 Sept 2023
- Enarson, E., Morrow, B. (1998) Why gender? Why women? An Introduction to Women and Disaster. In Enarson, E., Morrow, B. (Ed.) The Gendered Terrain of Disaster: Through Women's Eyes, Praeger, Westport, Connecticut, 1–8.
- Engender Health (2021) Tanzania. https://www.engenderhealth.org/ourcountries/africa/tanzania/
- Equal Measures 2030 (2021) Why SDG 13 matters for gender equality. https://data.em2030.org/goals/sdg13/
- ESCAP (2017) Transport and Communications Bulletin for Asia and the Pacific No. 87 Transport and Sustainable Development Goals. https://www.unescap.org/sites/default/files/publications/bulletin87_Fulltext.pdf
- Eskenazi, B., Etzel, R.A., Sripada, K., Cairns, M.R., Hertz-Picciotto, I., Kordas, K., Machado Torres, J.P., Mielke, H.W., Oulhote, Y., Quirós-Alcalá, L. et al. (2020) The International Society for Children's Health and the Environment Commits to Reduce Its Carbon Footprint to Safeguard Children's Health. Environmental Health Perspectives 128, 014501.
- European Commission (2019a) 2019 Report on equality between women and men in the EU. Luxembourg: Publications Office of the European Union. https://doi.org/10.2838/395144
- European Commission (2019b) European Civil Protection and Humanitarian Aid Operations: Pacific Region, European Commission, Brussels. CHAPTER 7.
- European Environment Agency (2015) Urban sustainability issues What is a resourceefficient city? EEA Technical report No 23/2015. https://www.eea.europa.eu/publications/resource-efficient-cities/file

- European Institute for Gender Equality (2016) Poverty, gender and intersecting inequalities in the EU. Review of the implementation of Area A: Women and Poverty of the Beijing Platform for Action.
- European Institute for Gender Equality (2020) Care. Gender sensitive infrastructure. https://op.europa.eu/en/publication-detail/-/publication/4bec6067-cfd4-11ea-adf7- 01aa75ed71a1
- European Institute for Gender Equality (2023) Gender Statistics Database. https://eige.europa.eu/gender
 - statistics/dgs/browse/wmidm/wmidm_env/wmidm_env_nat Accessed 25 April 2023
- European Union Agency for Fundamental Rights (2020) Strong and effective national human rights institutions challenges, promising practices and opportunities. https://fra.europa.eu/sites/default/files/fra_uploads/fra-2020-strong-effective-nhris_en.pdf
- Fabbri, C., Bhatia, A., Petzold, M., Jugder, M., Guedes, A., Cappa, C., Devries, K. (2020) The right to protection ending violence against children. Child Abuse & Neglect 104897. https://doi.org/10.1016/j.chiabu.2020.104897
- Family Code of the Kyrgyz Republic amended on 16 January 2014 (2003) http://cbd.minjust.gov.kg/act/view/ru-ru/1327 Accessed 15 Nov 2019
- Fan, L., Jaffre, V.N. (2020) The gender dimension of sustainable consumption and production: A micro survey-based analysis of gender differences in awareness, attitudes, behaviours in the People Republic of China. Asian Development Bank, Hong Kong.
- Fanzo, J. (2019) Healthy and Sustainable Diets and Food Systems: the Key to Achieving Sustainable Development Goal 2? Food ethics 4, 159–174. https://doi.org/10.1007/s41055-019-00052-6
- FAO (1997) Gender: the Key to Sustainability and Food security. SD Dimensions. http://www.fao.org/sd/WPdirect/WPdoe001.htm
- FAO (2010) Gender dimensions of agricultural and rural employment: differentiated pathways out of poverty. Status, trends and gaps. Rome, Italy. http://www.fao.org/3/i1638e/i1638e.pdf
- FAO (2011) The state of food and agriculture. Women in Agriculture; Closing the Gender Gap for Development; Food and Agriculture Organization of the United Nations: Rome, Italy. http://www.fao.org/3/a-i2050e.pdf
- FAO (2013) Irrigation in Central Asia in figures. AQUASTAT Survey-2012. Rome. http://doi.wiley.com/10.1111/j.1467-6346.2007.00685.x Accessed 10 Dec 2019
- FAO (2014a) Rural Women in Eastern Europe and Central Asia. http://www.fao.org/3/a-i3840e.pdf
- FAO (2014b) FAO statistical yearbook 2014 Latin America and the Caribbean Food and Agriculture. http://www.fao.org/3/i3592e/i3592e.pdf Accessed 28 Mar 2020
- FAO (2016a) National gender profile of agricultural and rural livelihoods Kyrgyz Republic. Country gender assessment series. Ankara. http://www.fao.org/3/a-i5766e.pdf
- FAO (2016b) National gender profile of agricultural and rural livelihoods Tajikistan. Country gender assessment series. Ankara. http://www.fao.org/3/a-i6192e.pdf
- FAO (2017a) Enabling Frameworks. The Role of Gender in Climate-Smart Agriculture; Climate Smart Agriculture Sourcebook; FAO: Rome, Italy.

- FAO (2017b) Women in Latin America and the Caribbean face greater poverty and obesity compared to men. http://www.fao.org/americas/noticias/ver/en/c/473028/ Accessed 28 March 2020
- FAO (2018) The state of food and agriculture 2018. Migration, agriculture and rural development. Rome, Italy. http://www.fao.org/3/I9549EN/i9549en.pdf
- FAO (2019a) Sex-disaggregated data in agriculture and sustainable resource management. New approaches for data collection and analysis. Rome, Italy. http://www.fao.org/3/i8930en/i8930en.pdf
- FAO (2019b) Gender and land rights database. http://www.fao.org/genderlandrights-database/data-map/statistics/en/?sta_id=1168 Accessed 7 Nov 2019
- FAO (2019c) Gender, agriculture and rural development in Uzbekistan. Country gender assessment series Budapest. Budapest. http://www.fao.org/3/ca4628en/ca4628en.pdf Accessed 3 Dec 2019
- FAO (2019d) Regional gender equality and action plan for Europe and Central Asia 2019–2022. Budapest. http://www.fao.org/3/ca4521en/ca4521en.pdf
- FAO (2020a) Gender and land rights database. http://www.fao.org/genderlandrights-database/data-map/statistics/en/?sta_id=982 Accessed 28 Mar 2020
- FAO (2020b) Gender and land rights database. http://www.fao.org/genderlandrights-database/data-map/statistics/en/?sta_id=1162 Accessed 28 March 2020
- FAO (2020c) The pacific islands: Tropical cyclone harold situation report may 2020. Resilience. www.fao.org/resilience/resources/resourcesdetail/en/c/1274007/ Accessed 15 April 2021
- FAO (2020d) FAO Policy on Gender Equality 2020–2030. Food and Agriculture Organization of the United Nations: Rome, Italy.
- FAO, ARC (2021) Women's Leadership and Gender Equality in Climate Action and Disaster Risk Reduction in Africa—A Call for Action; FAO: Rome, Italy.
- FAO, CARE (2019) Good Practices for Integrating Gender Equality and Women's Empowerment in Climate-Smart Agriculture Programmes; FAO: Rome, Italy; CARE: Atlanta, GA, USA.
- Fariña García, M.C., de Nicolás, V.L., Yagüe Blanco, J.L., Fernández, J.L. (2020) Semantic network analysis of sustainable development goals to quantitatively measure their interactions. Environmental Development. https://doi.org/10.1016/j.envdev.2020.100589
- Fathallah, J., Pyakurel, P. (2020) Addressing gender in energy studies. Energy Research and Social Science 65, 101461. https://doi.org/10.1016/j. erss.2020.101461
- Feyen L., Ciscar J.C., Gosling S., Ibarreta D., Soria A. (Eds.) (2020) Climate change impacts and adaptation in Europe. JRC PESETA IV final report. EUR 30180EN, Publications Office of the European Union, Luxembourg. https://doi.org/10.2760/171121
- Fiji Health and Nutrition Cluster (2016a) Fiji Tropical Cyclone Bulletin, Fiji Health and Nutrition Cluster, Suva, Fiji.
- Fiji Health and Nutrition Cluster (2016b) Fiji Tropical Cyclone Bulletin 4, Fiji Health and Nutrition Cluster, Suva, Fiji.
- Fiji Health and Nutrition Cluster (2016c) Fiji Tropical Cyclone Bulletin 1, Fiji Health and Nutrition Cluster, Suva, Fiji.
- Fiji Meteorological Service (2021) Fiji climate summary January 2021. www.met.gov.fj/index.php?page=climateSummaries#January%202021cli mateSum2021.02.08%2009.51.17.pdf Accessed 15 April 2021

- Fikru, A.A., Kassahun, T.K., Tadesse, T.Z., Desalegn, Y.A., Gudina, L.F. (2021) Spatiotemporal hydro-climate variability in Omo-Gibe River Basin, Ethiopia. Climate Services 24, 100277.
- Fisher, P.G. (2020) Making the Financial System Sustainable. Cambridge University Press. https://doi.org/10.1017/9781108908269
- Fleurbaey M., Kartha, S. Bolwig, S., Chee, Y.L., Chen, Y. Corbera, E., Lecocq, F., Lutz, W. et al. (2014) Sustainable Development and Equity. In Edenhofer, O., Pichs-Madruga, R., Sokona, Y., Farahani, E. Kadner, S. et al (Eds.) Climate Change 2014: Mitigation of Climate Change. Contribution of Working Group III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.
- Flouris, A.D., Dinas, P.C., Ioannou, L.G., Nybo, L., Havenith, G. et al. (2018) Workers' health and productivity under occupational heat strain: a systematic review and meta-analysis. Lancet Planet Health 2(12), e521–31.
- Foley, A. (2020) The impact of connectivity on information channel use in Tonga during cyclone Gita: challenges and opportunities for disaster risk reduction in island peripheries. In Leal Filho, W. (Ed.) Managing Climate Change Adaptation in the Pacific Region, Springer, Berlin, pp. 255–271.
- Fontefrancesco M.F. (2019) Food Commodity Market: History and Impact of Food Trading Toward SDG2. In Leal Filho, W., Azul, A.M., Brandli, L., Özuyar, P.G., Wall, T. (Eds.) Zero Hunger. Encyclopedia of the UN Sustainable Development Goals. Springer, Cham. https://doi.org/10.1007/978-3-319-69626-3_13-1
- Ford, J.D., Berrang-Ford, L., Bunce, A., McKay, C., Irwin, M., Pearce, T. (2015) The status of climate change adaptation in Africa and Asia. Regional Environmental Change 15(5), 801–814. https://doi.org/10. 1007/s10113-014-0648-2
- Ford, J.D., King, D. (2015) A framework for examining adaptation readiness. Mitigation and Adaptation Strategies for Global Change 20, 505–526. https://doi.org/10.1007/s11027-013-9505-8
- Ford, J.D., Pearce, T. (2010) What we know, do not know, and need to know about climate change vulnerability in the western Canadian Arctic: A systematic literature review. Environmental Research Letters 5(1), 014008. https://doi.org/10.1088/1748-9326/5/1/014008
- Fortmann, L. (2010) The social dimensions of climate change. equity and vulnerability in a warming world. Experimental Agriculture, 46. The World Bank, Washington DC, p. 422.
- Forzieri, G., Bianchi, A., Batista e Silva, F., Marin Herrera, A., Antoine Leblois, A. et al. (2018) Escalating impacts of climate extremes on critical infrastructures in Europe. Global Environmental Change 48, 97–107. https://doi.org/10.1016/j.gloenvcha.2017.11.007
- Fothergill, A. (1999) Women's Roles in a Disaster. Applied Behavioral Science Review 7(2), 125–143.
- Franco, I. B., dos Muchangos, L., Okitasari, M., Mishra, R., Akhtar Mousumi, M., Nguyen, A., Kanie, N. (2018) Gender Mainstreaming in the 2030 Agenda: A Focus on Education and Responsible Consumption and Production. Policy brief No. 15. United Nations University Institute for the Advanced Study of Sustainability.
- Franco, I.B., Minnery, J. (2020) SDG 1 No Poverty. In Franco, I., Chatterji, T., Derbyshire, E., Tracey, J. (Eds.) Actioning the Global Goals for Local Impact.

Science for Sustainable Societies. Springer, Singapore. https://doi.org/10.1007/978-981-32-9927-6_2

- Freistein, K., Mahlert, B. (2015) The Role of Inequality in the Sustainable Development Goals, Conference Paper, University of Duisburg-Essen.
- Freudenreich, H., Aladysheva, A., Brück, T. (2022) Weather shocks across seasons and child health: Evidence from a panel study in the Kyrgyz Republic. World Development 155, 105801.
- FSM (2015) Second national communications to the United Nations framework convention on climate change. FSM.
- Fuso Nerini, F., Sovacool, B., Hughes, N., Cozzi, L., Cosgrave, E., Howells, M., Tavoni, M. et al. (2019) Connecting climate action with other Sustainable Development Goals. Nature Sustainability 2(8), 674–680. https://doi.org/10.1038/s41893-019-0334-y
- Gartaula, H., Sapkota, T.B., Khatri-Chhetri, A., Prasad, G., Badstue, L. (2020) Gendered impacts of greenhouse gas mitigation options for rice cultivation in India. Climatic Change 163, 1045–1063.
- GEF Independent Evaluation Office (2017) Evaluation on Gender Mainstreaming in the GEF. Proceedings of the GEF/ME/C.52/inf.09 and 52nd GEF Council Meeting, Washington, DC, USA, 23–25 May 2017.
- Gellers, J.C. (2016) Crowdsourcing global governance: sustainable development goals, civil society, and the pursuit of democratic legitimacy. International Environmental Agreements 16, 415–432. https://doi.org/10.1007/s10784-016-9322-0
- Gemenne, F., Blocher, J. (2017) How can migration serve adaptation to climate change? Challenges to fleshing out a policy ideal. The Geographical Journal 1–12.
- GFDRR (2020) Strengthening Financial Resilience and Accelerating Risk Reduction in Central Asia. Global Facility for Disaster Reduction and Recovery. https://www.gfdrr.org/en/program/SFRARR-Central-Asia Accessed on 27 Jan 2023
- Ghosh-Jerath, S., Kapoor, R., Singh, A., Downs, S., Barman, S., Fanzo, J. (2020) Leveraging Traditional Ecological Knowledge and Access to Nutrient-Rich Indigenous Foods to Help Achieve SDG 2: An Analysis of the Indigenous Foods of Sauria Paharias, a Vulnerable Tribal Community in Jharkhand, India. Frontiers in Nutrition 7(61). https://doi.org/10.3389/fnut.2020.00061
- Gibson, K.E., Barnett, J., Haslam, N. Kaplan, I. (2020) The mental health impacts of climate change: Findings from a 62 Pacific island Atoll nation. Journal of Anxiety Disorders 73.
- Gifford, E. Gifford, R. (2016) The largely unacknowledged impact of climate change on mental health. Bulletin of the Atomic Scientists 72(5), 292-297.
- Glänzel, W., Schubert, A. (2004) Chapter 11 Analyzing scientific networks through co-authorship. In Moed, H.F., Glänzel, W., Schmoch, U. (Eds.) Handbook of quantitative Science and Technology research, Springer Science and Business Media, pp. 257–276.
- Glass, L-M., Newig, J. (2019) Governance for achieving the Sustainable Development Goals: How important are participation, policy coherence, reflexivity, adaptation and democratic institutions? Earth System Governance 2, 100031. https://doi.org/10.1016/j.esg.2019.100031

- Glazebrook, T. (2011) Women and Climate Change: A Case-Study from Northeast Ghana. Hypatia 26, 762–782. http://www.jstor.org/stable/41328879 Accessed 17 Feb 2022
- Glazebrook, T., Noll, S., Opoku, E. (2020) Gender Matters: Climate Change, Gender Bias, and Women's Farming in the Global South and North. Agriculture 10, 267.
- Glemarec, Y., Qayum, S., Olshanskaya, M. (2016) Leveraging co-benefits between gender equality and climate action for sustainable development. Mainstreaming Gender Considerations in Climate Change Projects. UN Women: New York, NY, USA.
- Goh, A.H.X. (2012) A literature review of the gender-differentiated impacts of climate change on women's and men's assets and wellbeing in developing countries. CAPRi Working Paper No. 106. International Food Policy Research Institute, Washington, D.C. http://doi.org/10.2499/
- Gomez-Echeverri, L. (2018) Climate and development: enhancing impact through stronger linkages in the implementation of the Paris Agreement and the Sustainable Development Goals (SDGs). Philosophical Transactions of the Royal Society A 376(2119). https://doi.org/10.1098/rsta.2016.0444
- Goodman, S., Baudu, P. (2023) Climate change as a "threat multiplier": History, uses and future of the concept. Briefer, 38. Center for Climate and Security.
- Gopalan, K., Venkataraman, M. (2015) Affordable housing: Policy and practice in India. IIMB Management Review 27(2), 129-140.
- Gorettie, N.N., Namaalwa, J.J., Bomuhangi, A. (2019) Impacts of climate change on small holder households in Mt. Elgon region of Uganda: Does gender matter? In Bamutaze, Y., Kyamanywa, S., Singh, B. R., Nabanoga, G., Lal, R. (Eds.) Agriculture and ecosystem resilience in Sub Saharan Africa: Livelihood pathways under changing climate, Springer, pp. 673–690. https://doi.org/10.1007/978-3-030-12974-3_30
- Government of Andhra Pradesh (2017) Achieving sustainable development goals 2030 baseline, targets and strategy. http://4dj7dt2ychlw3310xlowzop2.wpengine.netdnacdn.com/wpcontent/uploads/2017/07/Andhra-Pradesh_Vision-2029.pdf
- Government of Canada (2020) An Open Justice Commitment for Canada -Discussion Paper. Government of Canada, Department of Justice. https://www.justice.gc.ca/eng/rp-pr/otherautre/trans/open-ouvert/ojcdpeejodt.html
- Government of Fiji (2016) Fiji Post-Disaster Needs Assessment, Tropical Cyclone Winston, Government of Fiji, Suva, Fiji.
- Government of Fiji (2017) Climate vulnerability assessment: making Fiji climate resilient. Government of Fiji, World Bank, and Global Facility for Disaster Reduction and Recovery.
- Government of Kenya (2010) National Climate Change Response Strategy, 2010. Government of Kenya. https://environmentalmigration.iom.int/resources/national-climate-changeresponse-strategy
- Government of Kenya (2013) National Climate Change Action Plan 2013 -2017.
- Government of Kenya (2018a) National Climate Change Action Plan (Kenya): 2018 –2022, Nairobi, Ministry of Environment and Forestry.
- Government of Kenya (2018b) Kenya: The 2018 Long Rains Season Assessment Report Africa Renewal, Report.

Government of Tajikistan (2019) National Strategy for Adaptation to Climate Change of the Republic of Tajikistan for the Period up to 2030. https://leap.unep.org/countries/tj/national-legislation/national-strategyadaptation alignets about a republic to illicitie to the Period

adaptation-climate-change-republic-tajikistan#:~:text=Policy-

,National%20Strategy%20for%20Adaptation%20to%20Climate%20Change%2 0of%20the%20Republic,land%20tenure%20and%20food%20security Accessed 25 April 2023

- Government of the Kyrgyz Republic (2013) Priorities for Adaptation to Climate Change in the Kyrgyz Republic Till 2017 (Updated to 2020); Resolution of the Government of the Kyrgyz Republic of 2 October 2013 No 549; Government of the Kyrgyz Republic: Bishkek, Kyrgyzstan.
- Government of the Kyrgyz Republic (2018) National Development Strategy of the Kyrgyz Republic for 2018–2040. Government of the Kyrgyz Republic: Bishkek, Kyrgyzstan. http://donors.kg/images/National_Development_Strategy_of_KR_2 018-2040_final_ENG.docx Accessed 20 Feb 2023
- Government of the Republic of Tajikistan (2012) Programme for Reforming the Agriculture Sector of the Republic of Tajikistan for 2012-2020. Decree of the Government of the Republic of Tajikistan №383 of August 1, 2012. http://base.spinform.ru/show_doc.fwx?rgn=54824
- Grainger, S., Dessai, S., Daron, J., Taylor, A., Ling Siu, Y. (2022) Using expert elicitation to strengthen future regional climate information for climate services. Climate Services 26, 100278.
- Grant Thornton (2021) Women in Business 2021 A Window of Opportunity; Grant Thornton International Ltd.: London, UK.
- Gray, C., Mueller, V. (2012) Drought and population mobility in rural Ethiopia. World Development 40(1), 134–145.
- Greenbaum, A. (1995) Taking stock of two decades of research on the social bases of environmental concern. In Mehta, D.M., Ouellet, E. (Eds.) Environmental sociology: Theory and Practice. North York, Ontario, Canada: Captus Press, pp. 125–152.
- Grillos, T. (2018) Women's participation in environmental decision-making: Quasiexperimental evidence from northern Kenya. World Development 108, 115–130.
- Grogan, J. (2020) America's legacy cities: building an equitable renaissance. Policy brief. Lincoln Institute of Land Policy.
- Guedes Vidal, D., Barros, N., Leandro Maia, R. (2019) Public and Green Spaces in the Context of Sustainable Development. In Leal Filho, W., Azul, A.M., Brandli, L., Özuyar, P.G., Wall, T. (Eds.) Sustainable Cities and Communities Living Edition. https://doi.org/10.1007/978-3-319-71061-7_79-1
- Guha-Sapir, D. (2018) The International disaster database; centre for research on the epidemiology of disasters (CRED), Louvain, Belgium. www.emdat.be/ Accessed 17 April 2021
- Guillemot, J. (2011) Summary report from Fiji's piloting climate change adaptation to protect human health (PCCAPHH) Project", Suva.
- Gumuci, T., Hansen, J., Huyer, S., van Huysen, T. (2019) Gender-responsive rural climate services: A review of the literature. Climate and Development 12(3), 241–254. https://doi.org/10.1080/17565529. 2019.1613216
- Gunluk-Senesen, G. (2021) Wellbeing gender budgeting to localize the UN SDGs: Examples from Turkey. Public Money and Management 41(7), 554–560. https://doi.org/10.1080/09540962.2021.1965402

- Gupta, J., Vegelin, C. (2016) Sustainable development goals and inclusive development. International Environmental Agreements: Politics, Law and Economics 16, 433–448. https://doi.org/10.1007/s10784-016-9323-z
- Haag, I., Jones, P.D., Samimi, C. (2019) Central Asia's Changing Climate: How Temperature and Precipitation Have Changed across Time, Space, and Altitude. Climate 7, 123.
- Habib, N., Alauddin, M., Cramb, R. (2022) What defines livelihood vulnerability to climate change in rain-fed, rural regions? A qualitative study of men's and women's vulnerability to climate change in Pakistan's Punjab. Cogent Social Sciences 8, 2054152.
- Habtezion, S. (2013) Overview of Linkages between Gender and Climate Change, Policy Brief; United Nations Development Programme: New York, NY, USA.
- Habtezion, S. (2016) Training Module 5 Gender and Climate Finance. United Nations Development Programme: New York, NY, USA.
- Hajžmanová, I. (2018) Drought-stricken Communities Hit by Destructive Floods in the Horn of Africa, International Displacement Monitoring Centre (IDMC).
- Hallegatte, S. (2016) Shock Waves: Managing the Impacts of Climate Change on Poverty. World Bank Publications.
- Hamidov, A., Helming, K., Balla, D. (2016) Impact of agricultural land use in Central Asia: a review. Agronomy for Sustainable Development 36, 1–23. https://doi.org/10.1007/s13593-015-0337-7
- Hanna, E.G. McIver, L. (2014) Small island states-canaries in the coal mine of climate change and health. In Butler, C.D. (Ed.) Climate Change and Global Health, 2nd ed., CABI, Wallingford, pp. 181–192.
- Hansen, N., Huis, M.A., Lensink, R. (2020) Microfinance Services and Women's Empowerment. In San-Josem, L., Retolaza, J., van Liedekerke, L. (Eds.) Handbook on Ethics in Finance. International Handbooks in Business Ethics, Springer, Cham. https://doi.org/10.1007/978-3-030-00001-1_4-1
- Harris, S., Abbott, K. (2018) The Business Case for Empowering Women through Climate-Resilient Supply Chains. https://www.bsr.org/en/our-insights/blogview/business-case-for-empowering-women-through-climate-resilientsupplychains Accessed 7 March 2022
- Hartmann, B. (2010) Rethinking climate refugees and climate conflict: rhetoric, reality and the politics of policy discourse. Journal of International Development 22(2), 233–246.
- Harvey, C.A., Rakotobe, Z.L., Rao, N.S., Dave, R., Razafimahatratra, H., Rabarijohn, R.H., Rajaofara, H., MacKinnon, J.L. (2014) Extreme vulnerability of smallholder farmers to agricultural risks and climate change in Madagascar. Philosophical Transactions of the Royal Society B: Biological Sciences 369(1639), 20130089. https://doi.org/10.1098/rstb.2013.0089
- Harzing, A.-W. (2007) Publish or Perish. https://harzing.com/resources/publish-orperish
- Hasson, F., Keeney, S., McKenna, H. (2000) Research guidelines for the Delphi survey technique. Journal of Advanced Nursing 32(4), 1008–1015.
- Haunschild, R., Bornmann, L., Werner, M. (2016) Climate change research in view of bibliometrics. PLoS One 11(7), e0160393. https://doi.org/10.1371/journal.pone.0160393
- Hayes, A.M., Bulat, J. (2017) Disabilities Inclusive Education Systems and Policies Guide for Low- and Middle-Income Countries. RTI Press Publication No. OP-

0043-1707. Research Triangle Park, NC: RTI Press. https://doi.org/10.3768/rtipress.2017.op.0043.1707

- Hayes, K., Blashki, G., Wiseman, J., Burke, S., Reifels, L. (2018) Climate change and mental health: risks, impacts and priority actions. International Journal of Mental Health Systems 12(1).
- Hayes, K., Poland, B. (2018) Addressing mental health in a changing climate: incorporating mental health indicators into climate change and health vulnerability and adaptation assessments. International Journal of Environmental Research and Public Health 15(9).
- Hemming, V., Burgman, M.A., Hanea, A.M., McBride, M.F., Wintle, B.C. (2018) A practical guide to structured expert elicitation using the IDEA protocol. Methods in Ecology and Evolution 9(1), 169–180. https://doi.org/10.1111/2041-210X.12857
- Hempel, S. (2020) Concise guides to conducting behavioral, health, and social science research series. American Psychological Association. https://doi.org/10.1037/0000155-000
- Hepp, P., Somerville, C., Borisch, B. (2019) Accelerating the United Nation's 2030 Global agenda: Why prioritization of the gender goal is essential. Global Policy 10(4), 677–685. https://doi.org/10.1111/1758-5899.12721
- Herbert, R., Falk-Kresinski, H.J., Plume, A. (2020) Sustainability through a gender lens: The extent to which researh on UN sustainable development goals (SDGs) includes sex and gender consideration. http://dx.doi.org/10.2139/ssrn.3689205
- Hilderbrand, M. (2015) Benefits and Costs of the Governance & Institutions Targets for the Post 2015 Development Agenda Post-2015 Consensus. Copenhagen Consensus Center. https://www.copenhagenconsensus.com/sites/default/files/governance assessme

nt_- _hilderbrand.pdf

- Hill, P.S., Huntington, D., Dodd, R., Buttsworth, M. (2013) From millennium development goals to post-2015 sustainable development: Sexual and reproductive health and rights in an evolving aid environment. Reprod Health Matters 21, 113–124.
- Hillis, S., Mercy, J., Amobi, A., Kress, H. (2016) Global Prevalence of Past-year Violence Against Children: A Systematic Review and Minimum Estimates. Pediatrics 2015-4079. https://doi.org/10.1542/peds.2015-4079
- Hirsu, L., Hashemi, L., Quezada-Rayes, Z. (2019) SDG 5: Achieve Gender Equality and Empower all Women and Girls. Jean Monnet Sustainable Development Goals Network Policy Brief Series. RMIT University. https://www.rmit.edu.au/content/dam/rmit/rmit-images/college-of-dscimages/eu-centre/sdg-5-policy-brief.pdf
- Hlimi, T. (2015) Association of anemia, pre-eclampsia and eclampsia with seasonality: A realist systematic review. Health Place 31, 180–192.
- Hoek-Smit, M., Kyung-Hwan, K., Wachter, S. (2020) Cities with Affordable Housing: Fulfilling the New Urban Agenda. https://realestate.wharton.upenn.edu/wpcontent/uploads/2020/06/Workingpaper-829.pdf
- Hoffiani, M. (2019) The Nexus between Corruption, Sustainable Development and Rule of Law. https://www.divaportal.org/smash/get/diva2:1352722/FULLTEXT01.pdf

- Hoffmann, R., Wiederkehr, C., Dimitrova, A., Hermans, K. (2022) Agricultural livelihoods, adaptation, and environmental migration in sub-Saharan drylands: a meta-analytical review. Environmental Research Letters 17, 083003.
- Holvoet, N., Inberg, L. (2014) Gender sensitivity of Sub-Saharan Africa National Adaptation Programmes of Action: Findings from a desk review of 31 countries. Climate and Development 6, 266–276.
- Hoornweg, D., Pope, K. (2017) Population predictions for the world's largest cities in the 21st century. Environment & Urbanization, International Institute for Environment and Development (IIED) 29(1), 195–216. http://dx.doi.org/10.1177/0956247816663557
- Hope, K.R. (2020) Corruption Reduction as a Target of the Sustainable Development Goals: Applying Indicators and Policy Frameworks. In Blaustein, J., Fitz-Gibbon, K., Pino, N.W., White, R. (Eds.) The Emerald Handbook of Crime, Justice and Sustainable Development, Emerald Publishing Limited, pp. 105-130.
- Hoot, W., Taborosi, D., Neth, Y. (2012) Timeline of natural disasters in the FSM, island research and education initiative. Pohnpei, FSM. https://fsmdata.sprep.org/system/files/FSM%20disaster%20timeline.pdf Accessed 17 April 2021
- Howland, F., Le Coq, J.F., Acosta, M. (2019) Gender Integration in Agriculture, Food Security and Climate Change Policy: A Framework Proposal Fanny Howland, Jean-Francois Le Coq, Mariola Acosta; Climate Change, Agriculture and Food Security (CCAFS): Palmira, Colombia.
- Hu, Q., Han, Z. (2022) Northward Expansion of Desert Climate in Central Asia in Recent Decades. Geophysical Research Letters 49, e2022GL098895.
- Hu, Z., Zhang, C., Hu, Q., Tian, H. (2014) Temperature Changes in Central Asia from 1979 to 2011 Based on Multiple Datasets. Journal of Climate 27, 1143–1167.
- Huang, L., Chen, K., Zhou, M. (2020) Climate change and carbon sink: A bibliometric analysis. Environmental Science and Pollution Research 27(8), 8740–8758. https://doi.org/10.1007/s11356-019-07489-6
- Huiskes, P., Dinis, M.A.P., Caridade, S. (2022) Technology-facilitated sexual violence victimization during the COVID-19 pandemic: Behaviours and attitudes. Journal of Aggression, Maltreatment and Trauma. https://doi.org/10.1080/10926771.2022.2089863
- Huyer, S. (2016) Closing the gender gap in agriculture. Gender, Technology and Development 20, 105-116.
- Huyer, S., Acosta, M., Gumucio, T., Ilham, J.I.J. (2020) Can we turn the tide? Confronting gender inequality in climate policy. Gender & Development 28, 571– 591.
- Hyder, A.A., Malik, A.M. (2007) Violence against Children: A Challenge for Public Health in Pakistan. Journal of Health, Population and Nutrition 25(2), 168–178. https://www.ncbi.nlm.nih.gov/pmc/articles/PMC2753994/pdf/jhpn0025-0168.pdf
- IAASTD (2009) Latin America and the Caribbean (LAC) report. https://www.weltagrarbericht.de/reports/LAC/LAC_full_report.pdf
- IDMC (2017) Ethiopia. Internal Displacement Monitoring Centre. http://www.internal-displacement.org/countries/ethiopia</underline
- Idowu, S.O., Schmidpeter, R., Zu, L. (Eds.) (2020) The Future of the UN Sustainable Development Goals: Business Perspectives for Global Development in 2030. Springer International Publishing. http://dx.doi.org/10.1007/978-3-030-21154-7

IFC (2016a) Investing in women along agribusiness value chains. International Finance Corporation. https://www.ifc.org/wps/wcm/connect/02c5b53e-420f-4bf4-82bb-

6f488ff75810/Women+in+Agri+VC_Report_FINAL.pdf?MOD=AJPERESand CVID=m0JfSbv

- IFC (2016b) The business case for women's employment in agribusiness. International Finance Corporation. https://www.ifc.org/wps/wcm/connect/187c236c-6efb-4a99-b5a9-1700da096437/Women_in_Agribusiness_Report+03.06.17cscovers.pdf?MOD=AJPERESandCVID=IGBMENs
- IFRC (2011) Red cross responds to the water crisis in drought-stricken Tuvalu. www.ifrc.org/en/news-and-media/news-stories/asia-pacific/tuvalu/red-crossresponds-to-water-crisis-in-drought-stricken-tuvalu/ Accessed 9 April 2021
- IFPRI, UNDP (2019) Building Resilience to Climate Shocks in Ethiopia. International Food Policy Research Institute, Washington, DC.
- IISD (2017) Achieve Gender Equality to Deliver the SDGs. http://sdg.iisd.org/commentary/policy-briefs/achieve-gender-equality-to-deliverthe-sdgs/
- Ilesanmi, O.O. (2018) Women's Visibility in Decision Making Processes in Africa— Progress, Challenges, and Way Forward. Frontiers in Sociology 3, 38.
- Ilkkaracan, I., Kim, K., Kayaaugust, T. (2015) The Impact of Public Investment in Social Care Services on Employment, Gender Equality, and Poverty: The Turkish Case. İstanbul Technical University Women's Studies Center in Science, Engineering and Technology and the Levy Economics Institute of Bard College.
- ILO (2018) World Employment and Social Outlook: Trends for women 2017. International Labour Organization.
- Ingutia, R. (2021) The impacts of COVID-19 and climate change on smallholders through the lens of SDGs; and ways to keep smallholders on 2030 agenda. International Journal of Sustainable Development & World Ecology 28, 693–708.
- Institute for Economics and Peace (2014) Measuring goal 16 identifying priority indicators based on key statistical and normative criteria. https://www.economicsandpeace.org/wpcontent/uploads/2015/06/Measuring-Goal-16.pdf
- Institute for Economics and Peace (2020) Ecological Threat Register 2020: Understanding Ecological Threats, Resilience and Peace, Sydney. https://www.economicsandpeace.org/wp-

content/uploads/2020/09/ETR_2020_web-1.pdf

- International Bank for Reconstruction and Development, World Bank (2020) Building effective, accountable, and inclusive institutions in Europe and Central Asia. https://www.pefa.org/sites/pefa/files/resources/downloads/Building-Effective-Accountableand-Inclusive-Institutions-in-Europe-and-Central-Asia-Lessons-from-the-Region.pdf
- International Council for Science (2011) Report of the ICSU Planning Group on Health and Wellbeing in the Changing Urban Environment: a Systems Analysis Approach. International Council for Science, Paris. https://sph.umd.edu/sites/default/files/files/health-and-wellbeingin-thechanging-urban-environment.pdf
- ILO (2012) Text of the Recommendation concerning National Floors of Social Protection. International Labour Convention. 14A ILO.

- International Training Centre of the International Labour Organisation (2009) Training Module—Introduction to Gender Analysis and Gender-Sensitive Indicators. Gender Campus: Turin, Italy.
- IOM (2014) IOM Outlook on Migration, Environment and Climate Change. International Organization for Migration. http://publications.iom.int/system/files/pdf/mecc_outlook.pdf
- IOM (2019) Tajikistan: Understanding the Nexus of Migration, Gender, Climate Change and Agriculture. International Organization for Migration. https://environmentalmigration.iom.int/tajikistan-understanding-nexusmigration-gender-climate-change-and-agriculture Accessed 17 Feb 2023
- IOM, Joint Migration and Development Initiative (2015) White Paper mainstreaming migration into local development planning and beyond. International Organization for Migration, United Nations Development Programme.

https://publications.iom.int/system/files/pdf/whitepaper_mainstreaming.pdf

- IPCC (2007) Climate Change 2007: Synthesis Report. Contribution of Working Groups I, II and III to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change, IPCC, Geneva, Switzerland, p. 104.
- IPCC (2012) Managing the risks of extreme events and disasters to advance climate change adaptation. Special Report of the Intergovernmental Panel on Climate Change, Cambridge University Press, Cambridge.
- IPCC (2014a) Climate change 2014: Synthesis Report. Contribution of Working Groups I, II and III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change, IPCC, Geneva, Switzerland.
- IPCC (2014b) Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press: Cambridge, UK; New York, NY, USA.
- IPCC (2018) Global warming of 1.5°C. An IPCC Special Report on the Impacts of Global Warming of 1.5°C above Pre-Industrial Levels and Related Global Greenhouse Gas Emission Pathways, in the Context of Strengthening the Global Response to the Threat of Climate Change, Sustainable Development, and Efforts to Eradicate Poverty, Cambridge University Press, Cambridge, and New York, NY.
- IPCC (2019) IPCC special report on the ocean and cryosphere in a changing climate.
- IPCC (2021) Climate change 2021: the physical science basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change, Cambridge University Press, Cambridge.
- IPCC (2022) Climate Change 2022: Impacts, Adaptation, and Vulnerability. Contribution of Working Group II to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press. Cambridge University Press, Cambridge, UK and New York, NY, USA.
- IPCC (2023) Climate Change 2023: Synthesis Report. Contribution of Working Groups I, II and III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. IPCC, Geneva, Switzerland. https://doi.org/10.59327/IPCC/AR6-9789291691647
- Iqbal, N., Gkiouleka, A., Milner, A., Montag, D., Gallo, V. (2018) Girls' hidden penalty: analysis of gender inequality in child mortality with data from 195 countries. BMJ Global Health 3(5), e001028. https://doi.org/10.1136/bmjgh-2018-001028

- IUCN (2015) Gender and climate change. Strengthening Climate Action by Promoting Gender Equality. Issues Brief; International Union for Conservation of Nature: Gland, Switzerland.
- Ivanovich, C.C., Sun, T., Gordon, D.R., Ocko I.B. (2023) Future warming from global food consumption. Nature Climate Change 13, 297–302. https://doi.org/10.1038/s41558-023-01605-8
- Jackson, L., Devadason, C.A. (2019) Climate change, flooding, and mental health. Secretariat of the Rockefeller Foundation Economic Council on Planetary Health.
- James, B.S., Shetty, R.S., Kamath, A., Shetty, A. (2020) Household cooking fuel use and its health effects among rural women in southern India—A cross-sectional study. PLoS ONE 15(4), e0231757. https://doi.org/10.1371/journal.pone.0231757
- Jankowska, M.M., Lopez-Carr, D., Funk, C., Husak, G.J., Chafe, Z.A. (2012) Climate change and human health: spatial modeling of water availability, malnutrition, and livelihoods in Mali, Africa. Applied Geography 33, 4–15.
- Jerin, T., Azad, M.A.K., Khan, M.N. (2023) Climate change-triggered vulnerability assessment of the flood-prone communities in Bangladesh: A gender perspective. International Journal of Disaster Risk Reduction 95, 103851.
- Jerneck, A. (2018) Taking gender seriously in climate change adaptation and sustainability science research: Views from feminist debates and sub-Saharan small-scale agriculture. Sustainability Science 13(2), 403–416. https://doi.org/10.1007/s11625-017-0464-y
- Jewell, S.T. (2018) 6 Providing meaningful information: Part D—current awareness. In DeRosa, A.P. (Ed.) A practical guide for informationists. Supporting research and clinical practice. Chandos Publishing, pp. 63–70. https://doi.org/10.1016/B978-0-08-102017-3. 00006-1
- Jizi, M., Nehme, R., Melhem, C. (2021) Board gender diversity and firms' social engagement in the gulf cooperation council (GCC) countries. Equality, Diversity and Inclusion 41, 186–206.
- JLL (2016) Affordable Housing in India. Key Initiatives for Inclusive Housing for All.
- Joshi, A., Kangave, J., Boogaard, V. (2020) Gender and Tax Policies in the Global South.

https://opendocs.ids.ac.uk/opendocs/bitstream/handle/20.500.12413/15450/817_ Gender_and_ Tax.pdf

- Kabir, R., Khan, H.T., Ball, E., Caldwell, K. (2016) Climate change impact: The experience of the coastal areas of Bangladesh affected by Cyclones Sidr and Aila. Journal of Environmental and Public Health 9654753.
- Kalaitzi, S., Czabanowska, K., Fowler-Davis, S., Brand, H. (2017) Women leadership barriers in healthcare, academia and business. Equality, Diversity and Inclusion 36, 457–474.
- Kaltenborn, M. (2017) Overcoming Extreme Poverty by Social Protection Floors Approaches to Closing the Right to Social Security Gap. Law and Development Review 10(2). https://doi.org/10.1515/ldr-2017-0014
- Kalungu, J.W., Leal Filho, W. (2016) Adoption of appropriate technologies among smallholder farmers in Kenya. Climate and Development 10(1), 84–96.
- Kandiyoti, D. (2002) Agrarian reform, gender and land rights in Uzbekistan. https://unece.org/fileadmin/DAM/hlm/prgm/cph/experts/uzbekistan/03_land_ad min_and_urban_devt/Agrarian_Reform.pdf

- Kanji, N., Tan, S.F., Toulmin, C. (2007) Introduction: Boserup revisited. In: Women's role in economic development. Earthscan, London/New York.
- Kansake, B.A., Sakyi-Addo, G.B., Dumakor-Dupey, N.K. (2021) Creating a genderinclusive mining industry: Uncovering the challenges of female mining stakeholders. Resources Policy 70, 101962.
- Karl, T.R., Knight, R.W., Plummer, N. (1995) Trends in high-frequency climate variability in the twentieth century. Nature 377 (6546), 217-220.
- Kassam, K. (2009) Viewing change through the prism of indigenous human ecology: Findings from the Afghan and Tajik Pamirs. Human Ecology 37, 677–690.
- Katila, P., McDermott, C., Larson, A., Aggarwal, S., Giessen, I. (2020) Forest tenure and the Sustainable Development Goals – A critical view. Forest Policy and Economics 120, 102294. https://doi.org/10.1016/j.forpol.2020.102294
- Katz, E. (2003) The changing role of women in the rural economies of Latin America. Current and emerging issues for economic analysis and policy research (CUREMIS II). FAO, pp 31–66. http://www.fao.org/3/a-y4940e.pdf
- Kaur, J. (2018) Impact Assessment of Access to Basic Services for Urban Poor in Chandigarh City, India. Asian Journal of Public Affairs 11(1), e2. Lee Kuan Yew School of Public Policy Research Paper No. 18–14. https://papers.ssrn.com/sol3/papers.cfm?abstract_id=3258884
- Keener, V., Helweg, D., Asam, S., Balwani, S., Burkett, M., Fletcher, C. et al. (2018) Hawai'i and U.S.-affiliated Pacific islands. In Maycock, T.K., Stewart, B.C. (Eds) Impacts, Risks, and Adaptation in the United States. In Fourth National Climate Assessment, Volume II; Fourth National Climate Assessment, Volume II, US Global Change Research Program, Washington, DC.
- Kefelegn, C. (2020) Impacts of climate change and variability on Rural Livelihoods and Community Responses: the Case of Merhabete Woreda, North Shewa Zone, Amhara National Regional State, Ethiopia.
- Keivani, R. (2010) A review of the main challenges to urban sustainability. International Journal of Urban Sustainable Development 1(1-2), 5–16. https://doi.org/10.1080/19463131003704213
- Kerras, H., Sanchez-Navarro, J., Lopez-Becerra, E. I., de-Miguel Gomez, M.D. (2020) The Impact of the Gender Digital Divide on Sustainable Development: Comparative Analysis between the European Union and the Maghreb Sustainability 12, 3347.
- Kessler, M.M. (1963) Bibliographic coupling between scientific papers. American Documentation 14(1), 10–25. https://doi.org/10.1002/asi. 5090140103
- Kett, M., Cole, E., Turner, J. (2020) Disability, Mobility and Transport in Low- and Middle-Income Countries: A Thematic Review. Sustainability 12(2), 589. https://doi.org/10.3390/su12020589
- Khandker, V., Gandhi, V.P., Johnson, N. (2020) Gender Perspective in Water Management: The Involvement of Women in Participatory Water Institutions of Eastern India. Water 12, 196.
- Khandekar, N., Gorti, G., Bhadwal, S., Rijhwani, V. (2019) Perceptions of climate shocks and gender vulnerabilities in the Upper Ganga Basin. Environmental Development 31, 97–109.
- Kironde, M.S., Durodola, O.S., Kanyunge, C.M. (2022) Integration of gender considerations into Tanzania's climate and water policies. Water Policy 24(1), 101. https://doi.org/110.2166/wp.2021.174

- Klarin, T. (2018) The Concept of Sustainable Development: From its Beginning to the Contemporary Issues. Zagreb International Review of Economics and Business 21(1), 67-94. https://doi.org/10.2478/zireb-2018-0005
- Klasen, S. (2018) The impact of gender inequality on economic performance in developing countries. Discussion Papers, No. 244. Georg-August-Universität Göttingen, Courant Research Centre Poverty, Equity and Growth (CRC-PEG), Göttingen.
- Klasen, S., Lamanna, F. (2009) The impact of gender inequality in education and employment on economic growth: new evidence for a panel of countries. Feminist Economics 15(3), 91–132. https://doi.org/10.1080/13545700902893106
- Knox-Hayes, J., Brown, M.A., Sovacool, B.K., Wang, Y. (2013) Understanding attitudes toward energy security: Results of a cross-national survey. Global Environmental Change 23, 609–622.
- Konte, M., Tirivayi, N. (Eds.) (2019) Women and sustainable human development: empowering women in Africa. Springer, Cham.
- Konte, M. (2020) Female Policymakers and Women's Well-Being in Africa. In Konte, M., Tirivayi, N. (Eds.) Women and Sustainable Human Development. Gender, Development and Social Change. Palgrave Macmillan, Cham. https://doi.org/10.1007/978-3-030-14935-2_18
- Kotcher, J., Maibach, E., Miller, J., Campbell, E., Alqodmani, L., Maiero, M., Wyns, A. (2021) Views of health professionals on climate change and health: a multinational survey study. The Lancet Planetary Health 5(5), e316-e323.
- Kovaleva, M., Leal Filho, W., Borgemeister, C., Kalungu, J.W. (2022) Understanding Needs and Potentials for Gender-Balanced Empowerment and Leadership in Climate Change Adaptation and Mitigation in Africa. Sustainability 14, 9410.
- Krall, S., (2015) What is sustainable agriculture? Deutsche Gesellschaft für Internationale Zusammenarbeit (GIZ) GmbH. https://www.giz.de/en/downloads/giz2015-en-what-is-sustain-agric.pdf
- Krishnapillai, M. (2017) Climate-friendly adaptation strategies for the displaced atoll population in Yap. In Leal Filho, W. (Ed.) Climate Adaptation in Pacific Countries: Fostering Resilience and Improving the Quality of Life, Springer, Berlin, pp. 101–117.
- Krishnapillai, M. (2018) Enhancing adaptive capacity and climate change resilience of coastal communities in Yap, in Leal Filho, W. (Ed.) Climate Change Impacts and Adaptation Strategies in Coastal Communities, Springer, Berlin, pp. 87–118.
- Kuleshov, Y., McGree, S., Jones, D., Charles, A., Cottrill, A., Prakash, B., Atalifo, T., Nihmei, S., Seuseu, F. (2014) Extreme weather and climate events and their impacts on island countries in the Western Pacific: cyclones, floods and droughts. Atmospheric and Climate Sciences 04(5), 803–818.
- Kumar, A.S., Aggarwal, S.P., Chauhan, P. (2021) Gender diversity in geo-spatial technology and applications disciplines uptake in developing asian countries—A survey-based competency analysis. Advances in Space Research 67, 1350–1364.
- Kunwar, S.B. (2020) Assessing the Economic Impact of Climate Change on Agriculture in Central Asia; CAREC Institute: Urumqi, China.
- Laborde, D., Martin, D., Swinnenand, J., Vos, R. (2020) COVID-19 risks to global food security. Economic fallout and food supply chain disruptions require attention from policymakers. Science 369(6503), 500–502.

- Lama, P., Hamza, M., Wester, M. (2021) Gendered dimensions of migration in relation to climate change. Climate and Development 13(4), 326–336. https://doi.org/10.1080/17565529.2020.1772708
- Lambrou, Y., Piana, G. (2006) Gender: The Missing Component of the Response to Climate Change; Food and Agriculture Organization of the United Nations: Rome, Italy.
- Lam Duyen, T.N., Rañola, R.F., Sander, B.O., Wassmann, R., Tien, N.D., Khanh Ngoc, N.N. (2020) A comparative analysis of gender and youth issues in rice production in north, central, and South Vietnam. Climate and Development 13(2), 115–127. https://doi.org/10.1080/17565529.2020.1734771
- Lamichhane, S., Eğilmez, G., Gedik, R., Bhutta, M.K.S., Erenay, B. (2020) Benchmarking OECD countries' sustainable development performance: A goalspecific principal component analysis approach. Journal of Cleaner Production. https://doi.org/10.1016/j.jclepro.2020.125040
- Lang, V.F., Lingnau, H. (2015) Defining and Measuring Poverty and Inequality Post-2015. https://doi.org/10.1002/jid.3084
- Larson, P.D., Larson, N.M. (2019) The Hunger of Nations: An Empirical Study of Interrelationships among the Sustainable Development Goals (SDGs). Journal of Sustainable Development 12(6). https://doi.org/10.5539/jsd.v12n6p39
- Lassi, M., Sonnenwald, D.H. (2010) Identifying factors that may impact the adoption and use of a social science collaboratory: A synthesis of previous research. Proceedings of the seventh international Conference on conceptions of library and Information science —"unity in diversity". IR Information Research 15(3). http://InformationR.net/ir/15-3/colis7/colis710.html
- Lastarria-Cornhiel, S. (2006) Feminization of agriculture: trends and driving forces. https://vtechworks.lib.vt.edu/bitstream/handle/10919/68838/4589_Lastarria_Cornhiel2006_Feminization_of_A.pdf?sequence=1 Accessed 7 Nov 2019
- Lastarria-Cornhiel, S., Garcia-Frias, Z. (2005a) Return to patriarchy in Uzbekistan. Gender and Land Compendium of Country Studies; Food and Agriculture Organization of the United Nations (FAO), Rome, Italy. https://www.fao.org/3/a0297e/a0297e08.htm#bm8.4.2 Accessed 20 Feb 2023
- Lastarria-Cornhiel, S., García-Frías, Z. (2005b) Gender and land rights: findings and lessons from country studies in Gender and land compendium of country studies. FAO, Rome. http://www.fao.org/3/a0297e/a0297e08.htm#bm8.6.1
- Lau, J.D., Kleiber, D., Lawless, S., Cohen, P.J. (2021) Gender equality in climate policy and practice hindered by assumptions. Nature Climate Change 11, 186– 192.
- Law of Kyrgyz Republic on State Guarantees for Ensuring Gender Equality Art. 13 (2011). http://cbd.minjust.gov.kg/act/view/ru-ru/202398 Accessed 21 Nov 2019
- Law on Agricultural Land Management (2001) Law on agricultural land management amended on 25 July 2006. http://cbd.minjust.gov.kg/act/view/ru-ru/386?cl=ru-ru Accessed 21 Nov 2019
- Lawrance, E., Thompson, R., Fontana, G., Jennings, N. (2021) The impact of climate change on mental health and emotional wellbeing: current evidence and implications for policy and practice. Briefing paper No 36, Grantham Institute, Institute of Global Health Innovation.
- Leach, M. (Ed.) (2015) Gender Equality and Sustainable Development. 1st edition, Routledge. https://www.routledge.com/Gender-Equality-and-SustainableDevelopment/Leach/p/book/9781138921313

- Leal Filho, W., Nzengya, D., Muasya, G., Chemuliti, J., Kalungu, J.W. (2017) Climate change responses among the Maasai Community in Kenya. Climatic Change 145(1), 71–83.
- Leal Filho, W., Taddese, H., Balehegn, M., Nzengya, D., Debela, N., Abayineh, A., et al. (2020a) Introducing experiences from African pastoralist communities to cope with climate change risks, hazards and extremes: fostering poverty reduction. International Journal of Disaster Risk Reduction 50, 1–11, 101738.
- Leal Filho, W., Nagy, G.J., Ayal, D.Y. (2020b) Viewpoint: climate change, health and pandemics – a wake-up call from COVID-19. International Journal of Climate Change Strategies and Management 12(4), 533–535. https://doi.org/10.1108/IJCCSM-08-2020-212
- Leal Filho, W., Azul, A.M., Brandli, L., Salvia, A.L., Özuyar, P.G., Wall, T. (Eds.) (2021) Peace, Justice and Strong Institutions. Springer, Cham https://doi.org/10.1007/978-3-319-71066-2
- Leal Filho, W., Henrique Paulino Pires Eustachio, J., Dinis, M.A.P., Sharifi, A., Venkatesan, M., Donkor, F.K., Doni, F., Abubakar, I.R., Cichos, K., Vargas-Hernandez, J. (2022a) Transient poverty in a sustainable development context. International Journal of Sustainable Development & World Ecology 29(5), 415– 428.
- Leal Filho, W., Olaniyan, O.F., Alverio, G.N. (2022b) Where to go? Migration and climate change response in West Africa. Geoforum 137, 83–87.
- Leal Filho, W., Ternova, L., Parasnis, S.A., Kovaleva, M., Nagy, G.J. (2022c) Climate change and zoonoses: a review of concepts, definitions, and bibliometrics. International Journal of Environmental Research and Public Health 19(2).
- Leal Filho, W., Vidal, D. G., Chen, C., Petrova, M., Dinis, M.A.P., Yang, P., Rogers, S. et al. (2022d) An assessment of requirements in investments, new technologies and infrastructures to achieve the SDGs. Environmental Sciences Europe 34, 1–17. https://doi.org/10.1186/s12302-022-00629-9
- Leal Filho, W., Wall, T., Barbir, J., Alverio, G.N., Dinis, M.A.P., Ramirez, J. (2022e) Relevance of International Partnerships in the Implementation of the UN Sustainable Development Goals. Nature Communications 13(1), 613. https://doi.org/10.1038/s41467-022-28230-x
- Least Developed Countries Expert Group (2015) Strengthening Gender Considerations in Adaptation Planning and Implementation in the Least Developed Countries; United Nations Framework Convention on Climate Change (UNFCCC): Bonn, Germany.
- Le Dé, L., Rey, T., Leone, F., Gilbert, D. (2018) Sustainable livelihoods and effectiveness of disaster responses: a case study of tropical cyclone pam in Vanuatu. Natural Hazards 91(3), 1203-1221.
- Lee, T.M., Markowitz, E.M., Howe, P.D., Ko, C-W., Leiserowitz, A.A. (2015) Predictors of public climate change awareness and risk perception around the world. Nature Climate Change 5, 1014–1020.
- Lei, X., Zhou, X. (2012) Summary of retired typhoons in the Western North pacific ocean. Tropical Cyclone Research and Review 1, 23-32.
- Leisher, C., Temsah, G., Booker, F., Day, M., Samberg, L., Prosnitz, D., et al. (2016) Does the gender composition of forest and fishery management groups affect resource governance and conservation outcomes? A systematic map. Environmental Evidence 5, 6.

- Lerman, Z., Sedik, D. (2018) Transition to smallholder agriculture in Central Asia. Journal of Agrarian Change 18. https://doi.org/10.1111/joac.12282 Accessed 18 Nov 2019
- Li, J., Wang, M.-H., Ho, Y.-S. (2011) Trends in research on global climate change: A Science citation index expanded-based analysis. Global and Planetary Change 77(1-2), 13–20. https://doi.org/10.1016/j.gloplacha.2011.02.005
- Lightfoot, C. (1999) Regional El Nino social and economic drought impact assessment and mitigation study. Disaster Management Unit, South Pacific Applied Geoscience Commission.
- Lima, V., Gomez, M. (2019) Access to Justice: Promoting the Legal System as a Human Right. In Leal Filho, W., Azul, A.M., Brandli, L., Salvia, A.L., Özuyar, P.G., Wall, T. (Eds.) Peace, Justice and Strong Institutions Living. Springer Publishing. https://doi.org/10.1007/978- 3-319-71066-2_1-1
- Lindsey, I., Chapman, T. (2017) Enhancing the Contribution of Sport to the Sustainable Development Goals. Commonwealth Secretariat. https://www.sportanddev.org/sites/default/files/downloads/enhancing_the_contribution_of_sp ort_to_the_sustainable_development_goals_.pdf
- Linn, J.F. (2012) Central Asian Regional Integration and Cooperation: Reality or Mirage? In Vinokurov, E. (Ed.) EDB Eurasian Integration Yearbook 2012. Eurasian Development Bank: Almaty, Kazakhstan, 96–117.
- Lioubimtseva, E.A. (2015) Multi-scale assessment of human vulnerability to climate change in the Aral Sea basin. Environmental Earth Sciences 73, 719–729.
- Liu, J. (2019) What does in-work poverty mean for women: comparing the gender employment segregation in Belgium and China. Sustainability 11, 5725. https://doi.org/10.3390/su11205725www.mdpi.com
- Liu, A., Tan, H., Zhou, J., L., S., Yang, T., Wang, J., Liu, J., Tang, X., Sun, Z., We n, S.W. (2006) An epidemiologic study of post-traumatic stress disorder in flood victims in Hunan, China. The Canadian Journal of Psychiatry 51(6), 350-354.
- Liu, R.-L., Hsu, C.-K. (2019) Improving bibliographic coupling with category-based Co-citation. Applied Sciences 9(23), 5176. https://doi.org/10.3390/app9235176
- Liu, W., Liu, L., Gao, J. (2020) Adapting to climate change: Gaps and strategies for Central Asia. Mitigation and Adaptation Strategies for Global Change 25, 1439– 1459.
- Liu, Z., de Jong, M., Hertogh, M., Dong, L. (2020) Towards inclusive urban accessibility: framework and methodology for urban transport inclusiveness assessment The Case of Xiong'an New Area. The 8th World Sustainability Forum.
- Loarne-Lemaire, S.L., Bertrand, G., Razgallah, M., Maalaoui, A., Kallmuenzer, A. (2021) Women in innovation processes as a solution to climate change: A systematic literature review and an agenda for future research. Technological Forecasting and Social Change 164, 120440.
- Lobell, D.B., Field, C.B. (2007) Global scale climate-crop yield relationships and the impacts of recent warming. Environmental Research Letters 2, 1–8. https://doi.org/10.1088/1748-9326/2/1/014002
- Loboguerrero, A.M., Campbell, B.M., Cooper, P.J.M., Hansen, J.W., Rosenstock, T., Wollenberg, E. (2019) Food and Earth Systems: Priorities for Climate Change Adaptation and Mitigation for Agriculture and Food Systems. Sustainability 11, 1372. https://doi.org/10.3390/su11051372
- Lopez-Ferrer, M. (2018) International Funding and Collaboration in Sea Level RISE Research. In: IGI Global, Climate Change and Environmental Concerns:

Breakthroughs in Research and Practice. Management Association, Information Resources. Science.

- Lu, K., Wolfram, D. (2012) Measuring author research relatedness: A comparison of word-based, topic-based, and author cocitation approaches. Journal of the American Society for Information Science and Technology 63(10), 1973–1986. https://doi.org/10.1002/asi.22628
- Lucci, P., Bhatkal, T., Khan, A., Berliner, T. (2015) What works in improving the living conditions of slum dwellers A review of the evidence across four programmes. https://www.odi.org/sites/odi.org.uk/files/odi-assets/publicationsopinion-files/10188.pdf
- Luetz, J.M. (2019) Climate refugees: why measuring the immeasurable makes sense beyond measure. In Leal Filho, W., Azul, A., Brandli, L., Ozuyar, P., Wall, T. (Eds.) Climate Action. Encyclopedia of the UN Sustainable Development Goals. Springer, Cham. https://doi.org/10.1007/978-3-319-71063-1_81-1
- Lukwale, S. R., Sife, A. S. (2017) Climate change research trends in Tanzania: A bibliometric analysis. International Journal of Biodiversity and Conservation 9(6), 224–231. https://doi.org/10.5897/IJBC2017.1099
- Mackey, T.K., Kohler, J.C., Savedoff, W.D., Vogl, F., Lewis, M., Sale, J., Michaud, J., Vian, T. (2016) The disease of corruption: views on how to fight corruption to advance 21st century global health goals. BMC Med 14, 149. https://doi.org/10.1186/s12916-016-0696-1
- Magazzino, C., Cerulli, G., Shahzad, U., Khan, U. (2023) The nexus between agricultural land use, urbanization, and greenhouse gas emissions: Novel implications from different stages of income levels. Atmospheric Pollution Research 14(9), 101846. https://doi.org/10.1016/j.apr.2023.101846
- Magee, A.D., Verdon-Kidd, D.C., Kiem, A., Royle, S.A. (2016) Tropical cyclone perceptions, impacts and adaptation in the southwest pacific: an urban perspective from Fiji, Vanuatu and Tonga. Natural Hazards and Earth System Sciences 16(5), 1091-1105.
- Magendane, K., Kapazoglou, M. (2021) Approaches to study SDG interactions: Literature review of relevant frameworks, University of Amsterdam, The Broker, and NOW.
- Magnusdottir, G.L., Kronsell, A. (2015) The (In)Visibility of Gender in Scandinavian Climate Policy-Making. International Feminist Journal of Politics 17, 308–326.
- Maheshwari, G., Nayak, R. (2020) Women leadership in Vietnamese higher education institutions: An exploratory study on barriers and enablers for career enhancement. Educational Management Administration and Leadership. https://doi.org/10.1177/1741143220945700
- Makinaa, A., Moyob, T. (2016) Mind the gap: Institutional considerations for gender-inclusive climate change policy in Sub-Saharan Africa. Local Environment 21, 1185–1197.
- Manandhar, M., Hawkes, S., Buse, K., Nosrati, E., Magar, V. (2018) Gender, health and the 2030 agenda for sustainable development. Bulletin of the World Health Organization 96, 644–653. https://doi.org/10.2471/BLT.18.211607
- Manby, B. (2017) Legal identity for all and childhood statelessness. Institute on Statelessness and Inclusion. http://children.worldsstateless.org/3/childhood-statelessness-and-thesustainable-development-agenda/legal-identity-for-all-and-childhood-statelessness.html

- Mannava, P., Abdullah, A., James, C., Dodd, R., Annear, P.L. (2015) Health systems and non-communicable diseases in the Asia-Pacific region: a review of the published literature. Asia Pacific Journal of Public Health 27(2), NP1-NP19.
- Mansell, P., Philbin, S. P., Broyd, T. (2020) Development of a New Business Model to Measure Organizational and Project-Level SDG Impact—Case Study of a Water Utility Company. Sustainability 12(16), 6413. https://doi.org/10.3390/su12166413
- Manuel, M., Manuel, C. (2018) Achieving equal access to justice for all by 2030 Lessons from global funds. Working paper 537. https://www.odi.org/sites/odi.org.uk/files/resourcedocuments/12307.pdf
- March, C., Smyth, I., Mukhopadhyay, M. (1999) A Guide to Gender Analysis Frameworks. Oxfam GB, UK.
- Marx, W., Haunschild, R., Bornmann, L. (2017a) Global warming and Tea production The bibliometric view on a newly emerging research topic. Climate 5(3), 46. https://doi.org/10.3390/cli5030046
- Marx, W., Haunschild, R., Thor, A., Bornmann, L. (2017b) Which early works are cited most frequently in climate change research literature? A bibliometric approach based on reference publication year spectroscopy. Scientometrics 110(1), 335–353. https://doi.org/10.1007/s11192-016-2177-x
- Mavisakalyan, A., Tarverdi, Y. (2019) Gender and climate change: Do female parliamentarians make difference? European Journal of Political Economy 56, 151–164.
- Mbow, C., Rosenzweig, C., Barioni, L.G., Benton, T.G., Herrero, M., Krishnapillai, M. et al. (2019) Food Security. In Shukla, P.R., Skea, J., Calvo Buendia, E., Masson-Delmotte, V., Pörtner, H.-O., Roberts, D.C., et al. (Eds.) Climate Change and Land: An IPCC Special Report on Climate Change, Desertification, Land Degradation, Sustainable Land Management, Food Security, and Greenhouse Gas Fluxes in Terrestrial Ecosystems. Intergovernmental Panel on Climate Change: Geneva, Switzerland.
- McAdam, J. (2014) Historical cross-border relocations in the pacific: lessons for planned relocations in the context of climate change. The Journal of Pacific History 49(3), 301-327.
- McAuley, L., Pham, B., Tugwell, P., Moher, D. (2000) Does the inclusion of grey literature influence estimates of intervention effectiveness reported in metaanalyses? The Lancet 356(9237), 1228–1231. https://doi.org/10.1016/S0140-6736(00)02786-0
- McElroy, S., Ilango, S., Dimitrova, A., Gershunov, A., Benmarhnia, T. (2022) Extreme heat, preterm birth, and stillbirth: A global analysis across 14 lowermiddle income countries. Environment International 158, 106902.
- McIver, L., Kim, R., Woodward, A., Hales, S., Spickett, J., Katscherian, D., Hashiz ume, et al. (2016) Health impacts of climate change in Pacific island countries: a regional assessment of vulnerabilities and adaptation priorities. Environmental Health Perspectives 124(11), 1707–1714.
- McKinsey (2015) A CEO's guide to gender equality: The case for gender equality is strong. Why Is Progress So Slow? Executive Briefing; McKinsey: New York, NY, USA.
- McLeman, R. (2017) Migration and displacement in a changing climate. In Werrell, C.E., Femia, F. (Eds.) Epicenters of Climate and Security: The New Geostrategic Landscape of the Anthropocene 100–109.

- McNamara, K., Westoby, R. (2011) Solastalgia and the gendered nature of climate change: an example from erub island, Torres strait. EcoHealth 8(2), 233-236.
- McNamara, K.E., Westoby, R., Chandra, A. (2021) Exploring climate-driven noneconomic loss and damage in the Pacific islands. Current Opinion in Environmental Sustainability 50, 1–11.
- McSweeney, C., New, M., Lizcano, G. (2012) UNDP Climate Change Country Profiles Kenya. http://country-profiles.geog.ox.ac.uk
- Meehl, G.A., Zwiers, F., Evans, J., Knutson, T., Mearns, L., Whetton, P. (2000) Trends in extreme weather and climate events: issues related to modeling extremes in projections of future climate change. Bulletin of the American Meteorological Society 81(3), 427–436.
- Meijer, K., Sullivan, C., Blaauw, J., Schasfoort, F., Ottow, B., Morales, D (2019) Social inclusiveness in floods and droughts How social variations in impacts and responses can be taken into account. Working paper. https://cms.deltares.nl/assets/common/downloads/Deltares-Working-Paper-Inclusiveness-2019.pdf
- Mekonnen, Z. (2022) Intra-household gender disparity: Effects on climate change adaptation in Arsi Negele district, Ethiopia. Heliyon 8, e08908.
- Mekonen, A.A., Berlie, A.B. (2021) Rural households' livelihood vulnerability to climate variability and extremes: A livelihood zone-based approach in the Northeastern Highlands of Ethiopia. Ecological Processes 10, 55.
- Mensah, M., Vlek, P.L.G., Fosu-Mensah, B.Y. (2022) Gender and climate change linkages in the semi-arid region of Ghana. GeoJournal 87, 363–376.
- Mercandalli, S., Losch, B., Belebema, M.N., B'eli`eres, J.-F., Bourgeois, R., Dinbabo, M.F., Freguin-Gresh, S., Mensah, C., Nshimbi, C.C. (2019) Rural Migration in sub-Saharan Africa: Patterns, Drivers, and Relation to Structural Transformation. FAO and CIRAD, Rome. https://doi.org/10.4060/ca7404en
- Midgley, A., Methner, N. (2016) Climate Adaptation Readiness for Agriculture: Drought Lessons from the Western Cape, South Africa/ SAIIA Policy Briefing, 154. Johannesburg: SAIIA.
- Mijatović, D. (2018) Paris Principles at 25: Strong National Human Rights Institutions Needed More Than Ever. https://www.coe.int/en/web/commissioner/-/paris-principles-at-25-strong-national-human-rights-institutions-needed-morethan-ever
- Milazzo, A., Goldstein, M. (2019) Governance and women's economic and political participation: Power inequalities formal constraints and norms. The World Bank Research Observer 34(1), 34–64. https://doi.org/10.1093/wbro/lky006
- Ministry of Environment and Natural Resources (2016) Kenya National Adaptation Plan, 2015–2030.

https://www4.unfccc.int/sites/NAPC/Documents%20NAP/Kenya_NAP_Final.p df

- Ministry of Foreign Affairs (2018) Climate Change Profile, Kenya. https://reliefweb.int/sites/reliefweb.int/files/resources/Kenya_2.pdf
- Ministry of National Economy of the Republic of Kazakhstan Statistics committee (2017a) Kazakhstan in figures. [in Russian]. https://gender.stat.gov.kz/file/Kazakhstaninfigures.pdf
- Ministry of National Economy of the Republic of Kazakhstan Statistics committee (2017b) Women and men in Kazakhstan 2017. [in Russian]. https://stat.gov.kz/edition/publication/collection

- Ministry of National Economy of the Republic of Kazakhstan Statistics committee (2019a) Main indicators. https://stat.gov.kz/official/industry/11/statistic/7 Accessed 7 Nov 2019
- Ministry of National Economy of the Republic of Kazakhstan Statistics committee (2019b) Statistics of labour and employment. http://stat.gov.kz/official/industry/25/statistic/7 Accessed 7 Nov 2019
- Ministry of National Economy of the Republic of Kazakhstan Statistics committee (2019c) Agriculture, forestry and fisheries in the Republic of Kazakhstan 2014–2018. Nur-Sultan city. https://stat.gov.kz/official/industry/14/publication Accessed 18 Nov 2019
- Ministry of National Economy of the Republic of Kazakhstan Statistics committee (2019d) Share of employed people by groups of types of economic activities, by sex. http://gender.stat.gov.kz/page/frontend/detail?id=21andslug=-16andcat_id=7andlang=ru
- Ministry of National Economy of the Republic of Kazakhstan Statistics committee (2019e) Proportion of adults who own the land. https://gender.stat.gov.kz/page/frontend/detail?id=4andslug=-4andcat id=1andlang=en Accessed 18 Nov 2019
- Ministry of Water Resources of the Republic of Uzbekistan (2020) On Approval of the Concept of Development of Water Management Sector of the Republic of Uzbekistan for 2020–2030. https://water.gov.ug/ep/posts/1545725855/206 Approval 25 March 2022

https://water.gov.uz/en/posts/1545735855/396 Accessed 25 March 2023

- Miola, A. et al. (2019) Interlinkages and policy coherence for the Sustainable Development Goals Implementation-An operational method to identify trade-offs and co-benefits in a systemic way, European Commission, JRC Technical Reports.
- Mirzoeva, V. (2009) Gender issues in land reform in Tajikistan. Economics and Rural Development 5, 23–29.
- Mnimbo, T.S., Mbwambo, J., Kahimba, F.C., Tumbo, S.D. (2016) A gendered analysis of perception and vulnerability to climate change among smallholder farmers: The case of same district, Tanzania. Climate Development 8, 95–104.
- Mogilevskii, R. (2020) Labour Market and Technological Development in Central Asia; Working Paper #58; University of Central Asia: Bishkek, Kyrgyzstan. https://ucentralasia.org/media/cx5p2yas/uca-

ippawp58labormarketeng.pdf Accessed 25 Feb 2023

- Mohai, P. (1997) Gender differences in the perceptions of most important environmental problems. Race, Gender and Class 5, 153–169.
- Mohommad, A., Pugacheva, E. (2022) Impact of COVID-19 on attitudes to climate change and support for climate policies. IMF Working Paper No. 2022/023. https://ssrn.com/abstract=4070768
- Mokku, J. (2023) Climate change destroys the livelihoods of Kenyan pastoralists. Sustainable Development Goals. Africa Renewal. https://www.un.org/africarenewal/magazine/january-2023/climate-changedestroys-livelihoods-kenyan-pastoralists
- Moreno, J., Van de Ven, D-J., Sampedro, J., Gambhir, A., Woods, J., Gonzalez-Eguino, M. (2023) Assessing synergies and trade-offs of diverging Pariscompliant mitigation strategies with long-term SDG objectives. Global Environmental Change 78. https://doi.org/10.1016/j.gloenvcha.2022.102624

- Morgan, R., Dhatt, R., Kharel, C., Muraya, K. (2020) A patchwork approach to gender equality weakens the SDGs: Time for cross-cutting action. Global Health Promotion 27(3), 3–5. https://doi.org/10.1177/1757975920949735
- Mosso, C., Pons, D., Beza-Beza, C. (2022) A Long Way toward Climate Smart Agriculture: The Importance of Addressing Gender Inequity in the Agricultural Sector of Guatemala. Land 11, 1268.
- Mueller, V., Sheriff, G., Dou, X., Gray, C. (2020) Temporary migration and climate variation in eastern Africa. World Development 126 https://doi.org/10.1016/j.worlddev.2019.104704
- Mugellini, G., Villeneuve, J-P. (2019) Monitoring the Risk of Corruption at International Level: The Case of the United Nations Sustainable Development Goals. European Journal of Risk Regulation 10(1), 201–207. https://doi.org/10.1017/err.2019.16
- Mukhamedova, N. (2018) What's the truth about the role of women in agriculture today? https://wle.cgiar.org/thrive/big-questions/what-truth/sweet-and-bitter-truths
- Mukhamedova, N., Wegerich, K. (2018) The feminization of agriculture in post-Soviet Tajikistan. Journal of Rural Studies 57, 128–139. https://doi.org/10.1016/j.jrurstud.2017.12.009
- Mycoo, M., Wairiu, M., Campbell, D., Duvat, V., Golbuu, Y., Maharaj, S., Nalau, J., Nunn, P., Pinnegar, J., Warrick, O. (2022) Chapter 15: small islands. IPCC WGII Sixth Assessment Report.
- Mynbayeva, J., Kelly, S., Kazembekova, L. (2020) Study on the Role of Women in Kazakhstan's Energy Sector; KAZENERGY Association: Astana, Kazakhstan; European Bank for Reconstruction and Development (EBRD): London, UK.
- Nabutola, W. (2004) Affordable Housing in Kenya: A Case Study of Policy on
Informal Settlements Kenya. 3rd FIG Regional Conference Jakarta, Indonesia,
October 3-7, 2004.

https://www.humanitarianlibrary.org/sites/default/files/2013/07/ts_01_2_nabutol a.pdf

- Nallathiga, R. (2019) Housing for the Urban Poor: The Case of Chandigarh Model. NAGARLOK VOL. LI, Part 2, 142-151.
- National Climate Change Strategy of Turkmenistan (2012) https://info.undp.org/docs/pdc/Documents/TKM/110712_Strategy_en.pdf Accessed 25 Feb 2023
- National Environment Management Authority (2015) Kenya- Second National Communication to the United National Framework Convention on Climate Change. https://unfccc.int/sites/default/files/resource/Kennc2.pdf
- National Statistical Committee of the Kyrgyz Republic (2021) Women and Men of the Kyrgyz Republic: 2016–2020. Bishkek. http://www.stat.kg/media/publicationarchive/8f7fc721-c04b-4376-b411-03777feef9a5.pdf Accessed 25 Feb 2023
- National Statistical Committee of the Kyrgyz Republic (2015) O polozhenii sel'skih zhenshhin v Kyrgyzskoj Respublike (On the situation of rural women in the Kyrgyz Republic) [in Russian]. http://www.stat.kg/media/files/af3cde42-5f4a-40f8-81fb-802d8fcafaf5.pdf Accessed 9 Dec 2019
- National Statistical Committee of the Kyrgyz Republic (2016) Women and men in the Kyrgyz Republic 2011–2015. http://stat.kg/en/publications/sbornikzhenshiny-i-muzhchiny-kyrgyzskoj-respubliki/ Accessed 9 Dec 2019

- National Statistical Committee of the Kyrgyz Republic (2019a) Population. http://stat.kg/en/statistics/naselenie/ Accessed 7 Nov 2019
- National Statistical Committee of the Kyrgyz Republic (2019b) National accounts. http://stat.kg/en/statistics/nacionalnye-scheta/ Accessed 7 Nov 2019
- National Statistical Committee of the Kyrgyz Republic (2019c) Gender statistics. http://stat.kg/en/statistics/gendernaya-statistika/ Accessed 7 Nov 2019
- National Statistical Committee of the Kyrgyz Republic (2019d) Employment. http://www.stat.kg/en/statistics/zanyatost/ Accessed 7 Nov 2019
- Nawrotzki, R.J., Hunter, L.M., Runfola, D.M., Riosmena, F. (2015) Climate change as a migration driver from rural and urban Mexico. Environmental Research Letters10(11), 114023.
- Ndiritu, S.W., Kassie, M., Shiferaw, B. (2014) Are there systematic gender differences in the adoption of sustainable agricultural intensification practices? Evidence from Kenya. Food Policy 49, 117–127.
- NDMO (2014) Mataniko river flash flood. National Disaster Management Office. https://www.ndmo.gov.sb/index.php Accessed 12 April 2021
- Neumayer, E., Plümper, T. (2007) The Gendered Nature of Natural Disasters: The Impact of Catastrophic Events on the Gender Gap in Life Expectancy, 1981–2002. Annals of the American Association of Geographers 97, 551–566.
- Newell, R., Dale, A. (2020) COVID-19 and climate change: An integrated perspective. Cities Health 5, S100–S104.
- Newman, C., Chama, P.K., Mugisha, M., Matsiko, C.W., Oketcho, V. (2017) Reasons behind current gender imbalances in senior global health roles and the practice and policy changes that can catalyze organizational change. Global Health, Epidemiology and Genomics 2, e19.
- Nguyen, X.P., Hoang, A.T., Ölçer, A.I., Huynh, T.T. (2021) Record decline in global CO2 emissions prompted by COVID-19 pandemic and its implications on future climate change policies. Energy Sources Part A Recovery Utilization, and Environmental Effects 1–4.
- Nhamo, G. (2014) Addressing women in climate change policies: A focus on selected east and southern African countries. Agenda 28, 156–167.
- Niang, I., Osman-Elasha, B., Githeko, A., Yanda, P.Z., Medany, M., Vogel, A., Boko, M., Tabo, R., Nyong, A. (2008) Africa Climate Change 2007: Impacts, adaptation and vulnerability: Contribution of working group II to the fourth assessment report of the intergovernmental panel on climate change. Cambridge University Press.
- Niang, I., Ruppel, O.C., Abdrabo, M.A., Essel, A., Lennard, C., Padgham, J., Urquhart, P. (2014) Africa. In Barros, V.R., Field, C.B., Dokken, D.J., Mastrandrea, M.D. et al. (Eds.) Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part B: Regional Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change; Cambridge University Press: Cambridge, UK; New York, NY, USA, pp. 1199– 1265.
- Nicolau, R., David, J., Caetano, M., Pereira, J.M.C. (2019) Ratio of Land Consumption Rate to Population Growth Rate—Analysis of Different Formulations Applied to Mainland Portugal. International Journal of Geo-Information 8(1), 10. https://doi.org/10.3390/ijgi8010010
- Nieuwenhuis, R., Munzi, T., Neugschwender, J., Omar, H., Palmisano, F. (2018) Gender equality and poverty are intrinsically linked. A contribution to the
continued monitoring of selected Sustainable Development Goals. Discussion Paper. UN Women.

- Nilsson, M., Chisholm, E., Griggs, D., Howden-Chapman, P., McCollum, D., Messerli, P., Neumann, B., Stevance, A.-S., Visbeck, M., Stafford-Smith, M. (2018) Mapping interactions between the sustainable development goals: Lessons learned and ways forward. Sustainability Science 13(6), 1489–1503. https://doi.org/10.1007/s11625-018-0604-z
- Norgaard, K., York, R. (2005) Gender Equality and State Environmentalism. Gender & Society 19, 506–522.
- Norichika, K., Biermann, F. (Eds.) (2017) Governing through Goals: Sustainable Development Goals as Governance Innovation. Cambridge, MA: MIT Press. https://www.earthsystemgovernance.org/publication/governing-through-goals-sustainabledevelopment-goals-as-governance-innovation/
- Noy, I. (2016) Natural disasters in the pacific islands countries: new measurements of impacts. Natural Hazards 84(S1), 7–18.
- Nuber, C., Velte, P. (2021) Board gender diversity and carbon emissions: European evidence on curvilinear relationships and critical mass. Business Strategy Environment 30, 1958-1992.
- Nucera, G.G., 2023. Addressing climate-induced migration through adaptation measures: an emerging human rights-based approach? Quarterly on Refugee Problems 62(1), 15-34. https://doi.org/10.57947/qrp.v62i1.21
- Nygård, H.M. (2017) Achieving the sustainable development agenda: The governance conflict nexus. International Area Studies Review 20(1), 3–18.
- Nyahunda, L. (2021) Social Work Empowerment Model for Mainstreaming the Participation of Rural Women in the Climate Change Discourse. Journal of Human Rights and Social Work 6, 120–129.
- Nyahunda, L., Chibvura, S., Tirivangasi, H.M. (2021) Social Work Practice: Accounting for Double Injustices Experienced by Women Under the Confluence of Covid-19 Pandemic and Climate Change Impacts in Nyanga, Zimbabwe. Journal of Human Rights and Social Work 6, 213–224.
- Nzau, B., Trillo, C. (2020) Affordable Housing Provision in Informal Settlements through Land Value Capture and Inclusionary Housing. Sustainability 12(15), 5975. https://doi.org/10.3390/su12155975
- Obura, D.O. (2020) Getting to 2030 Scaling effort to ambition through a narrative model of the SDGs. Marine Policy 117, 103973. https://doi.org/10.1016/j.marpol.2020.103973
- OCHA (2015) Situation report vanuatu: severe tropical cyclone pam. Bangkok: OCHA Regional Office for the Pacific. https://reliefweb.int/report/vanuatu/vanuatu-severe-tropical-cyclone-pamsituation-report-no-1-15-march-2015 Accessed 12 April 2021
- OCHA (2019) 2018 Year in review: responding to and preparing for disasters across Asia-Pacific, Year in review. Regional Office Asia-Pacific. https://ocharoap.exposure.co/2018-year-in-review Accessed 12 April 2021
- Odera, J.A., Mulusa, J. (2020) SDGs, Gender Equality and Women's Empowerment: What Prospects for Delivery? In Kaltenborn, M. Krajewski, M., Kuhn, H. (Eds.) Sustainable Development Goals and Human Rights. Interdisciplinary Studies in Human Rights, vol 5. Springer, Cham. https://doi.org/10. 1007/978-3-030-30469-0_6
- OECD (2018) Policy Coherence for Sustainable Development 2018 Towards Sustainable and Resilient Societies. https://www.oecd-

ilibrary.org/development/policy-coherence-forsustainable-development-2018_9789264301061-en

- OECD (2019a) Governance frameworks to ensure equal access to justice and citizens' legal empowerment. https://www.oecd-ilibrary.org/sites/cae781ceen/index.html?itemId=/content/component/cae781ce-en
- OECD (2019b) Governance as an SDG Accelerator Country Experiences and Tools. https://www.oecd.org/publications/governance-as-an-sdg-accelerator-0666b085en.htm
- OECD (2020a) COVID-19 and the Low-Carbon Transition: Impacts and Possible Policy Responses; Organisation for Economic Co-Operation and Development: Paris, France, 2020.
- OECD (2020b) Bridging the digital gender divide. http://www.oecd.org/digital/bridging-thedigital-gender-divide.pdf
- OECD (2021) Employment in Agriculture as a Share of Total Employment in Africa from 2010 to 2020 [Graph]. Statista. 2021. https://www.statista.com/statistics/1230868/employment-in-agriculture-asshare-of-total-in-africa/ Accessed 29 March 2022
- OECD/FAO (2019) OECD-FAO agricultural outlook 2019–2028, OECD Publishing, Paris/Food and Agriculture Organization of the United Nations, Rome. https://doi.org/10.1787/agr_outlook-2019-en Accessed 28 Mar 2020
- OHCHR (2020) Convention on the Elimination of All Forms of Discrimination against Women New York, 18 December 1979. https://www.ohchr.org/en/professionalinterest/pages/cedaw.aspx
- OHCHR (2023) Status of Ratification of Convention on the Elimination of All Forms of Discrimination against Women. Office of the United Nations High Commissioner for Human Rights. https://indicators.ohchr.org/ Accessed 27 April 2023
- Olinto, P., Beegle, K., Sobrado, C., Uematsu, H. (2013) The State of the Poor: Where Are The Poor, Where Is Extreme Poverty Harder to End, and What Is the Current Profile of the World's Poor? The World Bank. http://gesd.free.fr/wbpoor13.pdf on 12.01.21
- Onsrud, H., Paixao, S., Nichols, S. (2005) Women and land reform in Brazil. Department of Geodesy and Geomatics Engineering, Technical report no. 239, University of New Brunswick, Fredericton, New Brunswick, Canada. http://www2.unb.ca/gge/Pubs/TR239.pdf
- Open Government Partnership (2019) Access to Justice, Open Government Partnership Global Report Democracy beyond the ballot box. Justice Policy Series, Part I. https://www.opengovpartnership.org/wpcontent/uploads/2019/09/Justice-Policy-SeriesAccess-to-Justice.pdf
- Open Working Group (2014) Open Working Group Proposal for Sustainable Development Goals. https://sustainabledevelopment.un.org/index.php?page=viewandtype=400andnr= 1579andmenu=1300
- Oppenheimer, M., Little, C.M., Cooke, R.M. (2016) Expert judgement and uncertainty quantification for climate change. Nature Climate Change 6, 445–451.
- Organization for Security and Co-Operation in Europe (2022a) Women, WaterManagementandConflictPrevention—Phasehttps://www.osce.org/node/503986Accessed 17Feb 2023

- Organization for Security and Co-Operation in Europe (2022b) Empowering Central Asian Women in Renewable Energy Mentoring Program. https://www.osce.org/oceea/511819 Accessed 7 Feb 2023
- Orlovsky, N., Radzinsky, V., Orlovsky, L. (2001) Desertification and population health in the Turkmenistan part of the Aral Sea region. Environmental Health Risk Assessment 5, 267.
- Osman-Elasha, B. (2013) Women in the Shadow of Climate Change; UN Chronicle; United Nations: New York, NY, USA. https://www.un.org/en/chronicle/article/womenin-shadow-climate-change Accessed 17 Feb 2022
- Oten, B., Reiffer, A., Funk, M., Shields, L., Ruteru, K., Hughes, F. et al. (2013) WHO Pro-File on Mental Health in Development (WHO proMIND): Republic of Kiribati, World Health Organization, Geneva.
- Overseas Development Institute (2018) Migration and the 2030 Agenda for Sustainable Development. https://www.odi.org/sites/odi.org.uk/files/resourcedocuments/12422.pdf
- Oyawole, F.P., Shittu, A., Kehinde, M., Ogunnaike, G., Akinjobi, L.T. (2020) Women empowerment and adoption of climate-smart agricultural practices in Nigeria. African Journal of Economic and Management Studies 12, 105–119.
- Owain, E.L., Maslin, M.A. (2018) Assessing the relative contribution of economic, political and environmental factors on past conflict and the displacement of people in East Africa. Palgrave Communications 4(1) https://doi.org/10.1057/s41599-018-0096-6
- Owusu, E., Shalaby, R., Eboreime, E., Nkire, N., Lawal, M.A., Agyapong, B., Pazderka, H. et al. (2022) Prevalence and determinants of generalized anxiety disorder symptoms in residents of fort McMurray 12 months following the 2020 flooding. Frontiers in Psychiatry 13.
- Oxfam (2010) The Rain Doesn't Come on Time Anymore Poverty, Vulnerability, and Climate Variability in Ethiopia. Oxfam International Research Report. https://www.oxfam.org/sites/www.oxfam.org/files/file_attachments/rainpoverty-vulnerability-climate-ethiopia-2010–04-22_3.pdf
- Özatağan, G., Ayalp, E.K. (2021) Sustainable futures of agro-food? İzmir's sustainable agro-food transitions in the making. Environmental Innovation and Societal Transitions 40, 283-295. https://doi.org/10.1016/j.eist.2021.08.003
- Pacific Disability Forum, UNICEF (2016) TC Winston Disability Needs Assessment Fiji Islands, UNICEF, Suva.
- Padhy, S.K., Sarkar, S., Panigrahi, M., Paul, S. (2015) Mental health effects of climate change. Indian Journal of Occupational and Environmental Medicine 19(1), 3-7.
- Paez, A. (2017) Gray literature: An important resource in systematic reviews. Journal of Evidence-Based Medicine 10(3), 233–240. https://doi.org/10.1111/jebm.12266
- Palacios-Lopez, A., Christiaensen, L., Kilicc, T. (2017) How much of the labor in African agriculture is provided by women? Food Policy 67, 52–63.
- Palinkas, L.A., Wong, M. (2020) Global climate change and mental health. Current Opinion in Psychology 32, 12-16.
- Pandey, U.C., Kumar, C. (2019) The Relationship of SDG5 to Other Goals. SDG5 Gender Equality and Empowerment of Women and Girls (Concise Guides to the United Nations Sustainable Development Goals). Emerald Publishing Limited, Bingley, pp. 103-120. https://doi.org/10.1108/978-1-78973-521-520191008

- Parker, C.F., Karlsson, C., Hjerpe, M. (2015) Climate change leaders and followers: Leadership recognition and selection in the UNFCCC negotiations. International Relations 29, 434–454.
- Parkes, C., Kolb, M., Schlange, L., Gudić, M., Schmidpeter, R. (2020) Looking forward: Leadership Development and Responsible Management Education for advancing the implementation of the Sustainable Development Goals (SDGs). The International Journal of Management Education 18(2), 100387. https://doi.org/10.1016/j.ijme.2020.100387
- Patel, S.K., Agrawal, G., Mathew, B., Patel, S., Mohanty, B., Singh, A. (2020) Climate change and women in South Asia: A review and future policy implications. World Journal of Science, Technology and Sustainable Development 17, 145–166.
- Pathfinders for Peaceful, Just and Inclusive Societies (2017) The Roadmap for Peaceful, Just and Inclusive Societies – A Call to Action to Change our World. New York: Center on International Cooperation. https://cic.nyu.edu/sites/default/files/sdg16_roadmap_en_20sep17.pdf
- Patnaik, H. (2021) Gender and participation in community-based adaptation: Evidence from the decentralized climate funds project in Senegal. World Development 142, 105448.
- Patrick, H.O., Khalema, E.N., Abiolu, O.A., Ijatuyi, E.J., Abiolu, R.T. (2021) South Africa's multiple vulnerabilities, food security and livelihood options in the COVID-19 new order: An annotation. Journal for Transdisciplinary Research in Southern Africa 17, a1037.
- Pattnaik, I., Lahiri-Dutt, K., Lockie, S., Pritchard, B. (2017) The feminization of agriculture or the feminization of agrarian distress? Tracking the trajectory of women in agriculture in India. Journal of the Asia Pacific Economy 23, 138–155. https://doi.org/10.1080/13547860.2017.1394569
- Paul, J., Criado, A.R. (2020) The art of writing literature review: What do we know and what do we need to know? International Business Review 29(4), 101717.
- Pauna, V.H., Picone, F., Le Guyader, G., Buonocore, E., Franzese, P.P. (2018) The scientific research on ecosystem services: A bibliometric analysis. Ecological Questions 29(3), 53–62. https://doi.org/10.12775/EQ.2018.022
- Pearl-Martinez, R., Stephens, J.C. (2016) Toward a gender diverse workforce in the renewable energy transition. Sustainability: Science, Practice and Policy 12(1), 8–15. https://doi.org/10.1080/15487733. 2016.11908149
- Pearse, R. (2016) Gender and climate change. WIREs Climate Change 8, e451.
- Pelling, M., Garschagen, M. (2019) Put equity first in climate adaptation. Nature 569(7756), 327–329. https://doi.org/10.1038/d41586-019-01497-9
- Peña-Ramos, J.A., Bagus, P., Fursova, D. (2021) Water Conflicts in Central Asia: Some Recommendations on the Non-Conflictual Use of Water. Sustainability 13, 3479.
- Perrault, N., Arellano, B. (2011) A rights-based approach to birth registration in Latin America and the Caribbean. UNICEF – Regional Office for Latin America and the Caribbean Documentation Centre. https://repositorio.cepal.org/bitstream/handle/11362/35984/1/challenges-13-ECLACUNICEF_es.pdf
- Peterman, A., O'Donnell, M. (2020) COVID-19 and Violence against Women and Children A Second Research Round Up. https://resourcecentre.savethechildren.net/node/18275/pdf/covid-19-andviolence-againstwomen-and-children-second-research-round.pdf

- Petrick, M., Pomfret, R. (2016) Agricultural policies in Kazakhstan. Leibniz Institute of Agricultural Development in Transition Economies (IAMO), Halle (Saale). http://nbn-resolving.de/urn:nbn:de:gbv:3:2-54045
- Pham, T.T., Brockhaus, M. (2015) Gender mainstreaming in REDD+ and PES Lessons learned from Vietnam. Brief for GSDR 2016 Update. Center for International Forestry Research (CIFOR): Bogor, Indonesia.
- Pham, H., Saner, M.A. (2021) Systematic Literature Review of Inclusive Climate Change Adaption. Sustainability 13, 10617. https://doi.org/10.3390/su131910617
- Pham-Truffert, M., Metz, F., Fischer, M., Rueff, H., Messerli, P. (2020) Interactions among sustainable development goals: Knowledge for identifying multipliers and virtuous cycles. Sustainable Development 28, 1236–1250. https://doi.org/10.1002/sd.2073
- Phogat, V. (2015) Right to information in consonance with right to privacy. https://cic.gov.in/sites/default/files/Internship%20Research%20Paper-%20Vratika%20Phogat.pdf
- Pickson, R.B., Boateng, E. (2022) Climate change: A friend or foe to food security in Africa? Environment, Development and Sustainability 24, 4387-4412.
- Pilgrim, G., Nicholson, D.-J., Johnstone, N., Nghiem, A. (2021) Women in Senior Management Roles at Energy Firms Remains Stubbornly Low, But Efforts to Improve Gender Diversity Are Moving Apace; International Energy Agency (IEA): Paris, France. https://www.iea.org/commentaries/women-in-seniormanagement-roles-at-energy-firms-remains-stubbornly-low-but-efforts-toimprove-gender-diversity-are-moving-apace Accessed 25 Feb 2023
- Pisano, P., Lange, L., Berger, G., Hametner, M. (2015) The Sustainable Development Goals (SDGs) and their impact on the European SD governance framework Preparing for the post- 2015 agenda. European Sustainable Development Network.
- Poisson, M. (2016) Promoting public access to information is key to improving transparency and accountability. http://www.iiep.unesco.org/en/promoting-public-access-information-keyimproving-transparency-and-accountability
- Pradhan, R. (2022) Natural Resources and Violent Conflicts: Water and Energy in Kyrgyzstan. Journal of Asian and African Studies 57, 650–666.
- Pratiwi, N.A.H., Rahmawati, Y.D., Setiono, I. (2017) Gender equality in climate change adaptation: A case of Cirebon, Indonesia. The Indonesian Journal of Planning and Development 2(2), 74–86. https://doi.org/10.14710/ijpd.2.2.74-86 Preti, A., Lentini, G., Maugeri, M. (2007) Global warming possibly linked to an enhanced risk of suicide: Data from Italy, 1974–2003. Journal of Affective Disorders 102(1-3), 19-25.
- Pretorius, R.W., Nicolau, M.D. (2020) Empowering communities to drive sustainable development: Reflections on experiences from rural South Africa. In Leal Filho, W., Tortato, U., Frankenberger, F. (Eds.) Universities and Sustainable Communities: Meeting the goals of the Agenda 2030, Cham: Springer, pp. 529– 545. https://doi.org/10.1007/978-3-030-30306-8-32
- Pudaruth, S., Devi Juwaheer, T., Seewoo, Y.D. (2015) Gender-based differences in understanding the purchasing patterns of eco-friendly cosmetics and beauty care products in Mauritius: a study of female customers. Social Responsibility Journal 11(1), 179-198. https://doi.org/10.1108/SRJ-04-2013-0049
- Punnett, B.J., Clarke, L.N. (2017) Women and Leadership in Africa. In Lituchy, T.R., Galperin, B.L., Punnett, B.J. (Eds.) LEAD: Leadership Effectiveness in Africa and the African Diaspora, Palgrave Macmillan: London, UK, pp. 217–236.

- Qualcomm Wireless Reach (2018) Soochnapreneur. https://www.qualcomm.com/media/documents/files/indiasoochnapreneurprogram.pdf
- Qoraboyev, I. (2021) Global governance. In De Feyter, K., Turkeli, G.E., De Moerloose, S. (Eds.) Law and Development Encyclopedia. Edward Elgar, pp. 99-103.
- Queen Mary University of London (2018) Gender inequality could be driving the deaths of girls under five. Science Daily.
- Quinn, S., Sannerholm, S. (2019) Rule of law washing and the Sustainable Development Goals. ILAC Policy Brief 4. http://ilacnet.org/wpcontent/uploads/2019/12/Using-SDG16-forrule-of-lawwashing_20191217_V4_Final2.pdf
- Quisumbing, A.R. (2003) Household decisions, gender, and development: a synthesis of recent research. Washington, DC. http://ebrary.ifpri.org/cdm/ref/collection/p15738coll2/id/129647 Accessed 21 Nov 2019
- Qushimov, B., Ganiev, I., Rustamova, I., Haitov, B., Islam, K.R. (2007) Land degradation by agricultural activities in Central Asia. In Lal, R., Suleymenov, M., Stewart, B.A. et al. (Eds.) Climate change and terrestrial carbon sequestration in Central Asia. Taylor and Francis, London, pp. 137–147. https://doi.org/10.1201/9780203932698.ch10
- Rabinovych, M. (2020) Where Economic Development Meets the Rule of Law? Promoting Sustainable Development Goals Through the European Neighborhood Policy. Brill Open Law 2(1), 140–174. https://doi.org/10.1163/23527072-20191017
- Racioppi, L., Rajagopalan, S. (2016) Women and Disasters in South Asia: Survival, Security and Development; Routledge: London, UK.
- Radhakrishnan, S., Erbis, S., Isaacs, J.A., Kamarthi, S. (2017) Novel keyword cooccurrence network-based methods to foster systematic reviews of scientific literature. PLoS ONE 12, e0172778.
- Rahman, M.S., Khatun, M. (2019) Climate change and gender based vulnerability nexus: an evidence from cyclonic storm surges area of Bangladesh. Bangladesh Journal of Agricultural Economics XL 1&2, 69-86.
- Rai, S.M., Brown, B.D., Ruwanpura, K.N. (2019) SDG 8: Decent work and economic growth A gendered analysis. World Development 113, 368-380. https://doi.org/10.1016/j.worlddev.2018.09.006
- Rakhimov, M. (2010) Internal and external dynamics of regional cooperation in Central Asia. Journal of Eurasian Studies 1, 95-101.
- Raman, S., Muhammad, T., Goldhagen, J., Seth, R., Kadir, A., Bennett, S. et al. (2020) Ending violence against children: What can global agencies do in partnership? Child Abuse & Neglect 104733. https://doi.org/10.1016/j.chiabu.2020.104733
- Rao, N., Lawson, E.T., Raditloaneng, W.N., Solomon, D., Angula, M.N. (2019) Gendered vulnerabilities to climate change: Insights from the semi-arid regions of Africa and Asia. Climate and Development 11(1), 14–26. https://doi.org/10.1080/17565529.2017.1372266
- Reacher, M., McKenzie, K., Lane, C., Nichols, T., Kedge, I., Iversen, A., Hepple, P., Walter, T., Laxton, C., Simpson, J. (2004) Health impacts of flooding in Lewes: a comparison of reported gastrointestinal and other illness and mental health in

flooded and non-flooded households. Communicable Disease and Public Health/PHLS 7(1), 39-46.

- Reckova, D., Irsova, Z. (2015) Publication bias in measuring anthropogenic climate change. Energy and Environment 26(5), 853–862. https://doi.org/10.1260/0958-305X.26.5.853
- Reggers, A. (2019) Climate Change Is Not Gender Neutral: Gender Inequality, Rights and Vulnerabilities in Bangladesh. In Huq, S., Chow, J., Fenton, A., Stott, C., Taub, J., Wright, H. (Eds.) Confronting Climate Change in Bangladesh. The Anthropocene: Politik—Economics—Society—Science, Springer: Cham, Switzerland, Volume 28.
- Reilly, J.M., Chen, Y.H.H., Jacoby, H.D. (2021) The COVID-19 effect on the Paris agreement. Humanities and Social Sciences Communications 8, 16.
- ReliefWeb (2012a) Fiji: Floofds Mar 2012. https://reliefweb.int/disaster/tc-2012-000044-fji Accessed 12 April 2021
- ReliefWeb (2012b) Tropical cyclone Evan Dec 2012. https://reliefweb.int/disaster/tc-2012-000201-wsm Accessed 12 April 2021
- ReliefWeb (2015a) Tropical cyclone Pam Mar 2015. https://reliefweb.int/disaster/tc-2015-000020-vut Accessed 12 April 2021
- ReliefWeb (2015b) Typhoon Maysak Mar 2015. https://reliefweb.int/disaster/tc-2015-000028-fsm Accessed 14 April 2021
- ReliefWeb (2016)TropicalcycloneWinston-Feb2016. https://reliefweb.int/disaster/tc-2016-000014-fjiAccessed 14 April 2021ReliefWeb (2020)TropicalcycloneYasa-Dec
- 2020. https://reliefweb.int/disaster/tc-2020-000238-fji Accessed 12 March 2022
- Reliefweb (2022) Kenya: Impact of drought on the arid and semi-arid regions. Thematic Report. https://reliefweb.int/report/kenya/acaps-thematic-report-kenyaimpact-drought-arid-and-semi-arid-regions-29-march-2022
- Remteng, C., Nkem, J., Mofor, L., Murombedzi, J. (2021) Gender in the nationally determined contributions of African countries: A way forward for effective implementation of adaptation and mitigation strategies. Ecofeminism and Climate Change 3, 2633-4062.
- Resurrección, B.P., Bee, B.A., Dankelman, I., Park, C.M.Y., Halder, M., McMullen, C.P. (2019) Gender-transformative climate change adaptation: advancing social equity. Background paper to the 2019 report of the Global Commission on Adaptation. Stockholm Environment Institute, Rotterdam and Washington, DC.
- Reuter, P. (2017) Illicit Financial Flows and Governance: The Importance of Disaggregation. Background paper for World Development Report 2017.
- Reyer, C., Adams, S., Albrecht, T., Baarsch, F., Boit, A., Trujillo, N.C., Cartsburg, M., Coumou, D. Eden, A., Fernandes, E. et al. (2017a) Climate change impacts in Latin America and the Caribbean and their implications for development. Regional Environmetal Change 17, 1601–1621.
- Reyer, C.P.O., Otto, I.M., Adams, S., Albrecht, T., Baarsch, F., Cartsburg, M., Coumou, D. et al. (2017b) Climate Change Impacts in Central Asia and their Implications for Development. Regional Environmental Change 17, 1639-1650.
- Reynolds, T.W., Bostrom, A., Read, D., Morgan, M.G. (2010) Now what do people know about global climate change? Survey studies of educated laypeople. Risk Analysis 30(10), 1520-1538.
- Rigaud, K.K., Sherbinin, A.D., Jones, B., Bergmann, J., Clement, V., Ober, K., Schewe, J., Adamo, S., McCusker, B., Heuser, S. et al. (2018) Groundswell:

Preparing for Internal Climate Migration, (Groundswell: Preparing for Internal Climate Migration); World Bank: Washington, DC, USA.

- Ringler, C., Zhu, T., Cai, X., Koo, J., Wang, D. (2010) Climate change impacts on food security in sub-Saharan Africa. International Food Policy Research Institute discussion paper.
- Rocque, R.J., Beaudoin, C., Ndjaboue, R., Cameron, L., Poirier-Bergeron, L., Poulin-Rheault, R-A., Fallon, C. (2021) Health effects of climate change: an overview of systematic reviews. BMJ Open 11, e046333. https://doi.org/10.1136/bmjopen-2020-046333
- Rodríguez-Pose, A., Hardy, D. (2015) Addressing poverty and inequality in the rural economy from a global perspective. Applied Geography 61, 11–23. https://doi.org/10.1016/j.apgeog.2015.02.005
- Rohr, U. (2007) Gender, climate change and adaptation. Introduction to the Gender Dimensions. Background Paper for the Both ENDS Briefing Paper "Adapting to Climate Change: How Local Experiences Can Shape the Debate"; Both ENDS: Amsterdam, The Netherlands.
- Rosenberg, M., Armanios, D.E., Aklin, M., Jaramillo, P. (2020) Evidence of gender inequality in energy use from a mixed study in India. Nature Sustainability 3, 110–118.
- Ros-Tonen, M.A.F., Agergaard, J., Zoomers, A. (2016) SDG Policy Brief #3 Support positive links between urban, peri-urban and rural areas. https://doi.org/10.13140/RG.2.1.1869.9761
- Rotich, B., Ojwang, D. (2021) Trends and drivers of forest cover change in the Cherangany hills forest ecosystem, western Kenya. Global Ecology and Conservation e01755.
- Sams, I.S. (2019) Impacts of Climate Change Induced Migration on Gender: A Qualitative Study from the Southwest Coastal Region of Bangladesh. International Journal of Social Science Studies 7, 57–68.
- Santiago Pineda, V., Meyers, S., Cruz, J.P. (2017) The Inclusion Imperative. Forging an Inclusive New Urban Agenda. The Journal of Public Space 2(4), 1. https://doi.org/10.5204/jps.v2i4.138
- Santoro, P.F. (2015) Urban planning to provide affordable housing in infrastructured areas, with social cohesion, through the market: Real estate profitability or right to the city assurance? RC21 International Conference on "The Ideal City: between myth and reality. Representations, policies, contradictions and challenges for tomorrow's urban life" Urbino (Italy). http://www.rc21.org/en/conferences/urbino2015/
- Saroar, M., Routray, J.K. (2010) Why does climate change awareness differ? Lessons learned from Bangladesh. Proceedings of the Second International Conference on Climate Change, Sustainability and Development in Semi-Arid Regions, Fortaleza, 16–20 August 2010, 48-53.
- Sartor, M.A., Beamish, P.W. (2020) Private Sector Corruption, Public Sector Corruption and the Organizational Structure of Foreign Subsidiaries. Journal of Business Ethics 167, 725–744. https://doi.org/10.1007/s10551-019-04148-1
- Satterthwaite, D. (2017) Successful, safe and sustainable cities: towards a New Urban Agenda. Commonwealth Journal of Local Governance. https://doi.org/10.5130/cjlg.v0i19.5446
- Satterthwaite, M.L., Dhital, S. (2019) Measuring Access to Justice: Transformation and Technicality in SDG 16.3. Global policy 10(S1), 96-109. https://doi.org/10.1111/1758-5899.12597

- Save The Children International Asia (2020) Because we matter. Addressing COVID-19 and Violence Against Girls in Asia-Pacific. https://resourcecentre.savethechildren.net/node/17928/pdf/pi_stc_becausewemat terpolicybrief -final.pdf
- Scalise, E., Undeland, A. (2016) Kyrgyz Republic: women and community pasture management. https://www.landesa.org/wp-content/uploads/2016-Best-Practices-Case-Kyrgyzstan.pdf
- Schalatek, L. (2018) Gender and Climate Finance. Climate Finance Fundamentals 10. Heinrich Böll Stiftung North America: Washington, DC, USA.
- Scharlemann, J.P.W. et al. (2020) Towards understanding interactions between sustainable development goals: The role of environment-human linkages. Sustainability Science 15(6), 1573–1584. https://doi.org/10.1007/s11625-020-00799-6
- Scheffran, J., Battaglini, A. (2011) Climate and conflicts: the security risks of global warming. Regional Environmental Change 11(SUPPL. 1). https://doi.org/10.1007/s10113-010-0175-8
- Schipper, E.L.F. (2020) Maladaptation: When Adaptation to Climate Change Goes Very Wrong. One Earth 3(4), 409-414. https://doi.org/10.1016/j.oneear.2020.09.014
- SDG Compass (2021) SDG 10: Reduce inequality within and among countries. https://sdgcompass.org/sdgs/sdg-10/
- Searle, K., Gow, K. (2010) Do concerns about climate change lead to distress? International Journal of Climate Change Strategies and Management 2(4), 362-379.
- Sehring, J., Ziganshina, D.R., Krasznai, M., Stoffelen, T. (2019) International actors and initiatives for sustainable water management. In Xenarios, S., Schmidt-Vogt, D., Qadir, M., Janusz-Pawletta, B., Abdullaev, I. (Eds.) The Aral Sea Basin Water for Sustainable Development in Central Asia, 1st ed., Routledge: London, UK, pp. 155–175.
- Sellers, S. (2016) Gender and Climate Change: A Closer Look at Existing Evidence; Global Gender and Climate Alliance (GGCA): Bali, Indonesia.
- Sen Roy, S. (2018a) The Three "E" Approach to Gender Mainstreaming in Climate Change: Enumeration, Education, Empowerment. In Sen Roy, S. (Ed.) Linking Gender to Climate Change Impacts in the Global South, Springer Climate, Springer: Cham, Switzerland, pp. 139–148.
- Sen Roy, S. (2018b) Water. Springer: Cham, Switzerland, pp. 75–91.
- Serdeczny, O., Adams, S., Baarsch, F., Coumou, D., Robinson, A., Hare, W., Schaeffer, M., Perrette, M., Reinhardt, J. (2017) Climate change impacts in Sub-Saharan Africa: from physical changes to their social repercussions. Regional Environmental Change 17(6). https://doi.org/10.1007/s10113-015-0910-2
- Servaes, J. (Ed.) (2017) Sustainable Development Goals in the Asian Context. Springer Singapore. https://doi.org/10.1007/978-981-10-2815-1
- Seto, K C., Dhakal, S., Bigio, A., Blanco, H., Delgado, G.C., Dewar, D., Huang, L. et al. (2014) Human Settlements, Infrastructure and Spatial Planning. In Edenhofer, O., Pichs-Madruga, R., Sokona, Y., Farahani, E., Kadner, S. et al. (Eds.) Climate Change 2014: Mitigation of Climate Change. Contribution of Working Group III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.

- Shakti, B.S. (2017) Tackling Violence Against Women: A Study of State Intervention Measures. A comparative study of impact of new laws, crime rate and reporting rate, Change in awareness level. https://wcd.nic.in/sites/default/files/Final%20Draft%20report%20BSS_0.pdf
- Shamoo, A.E., Resnik, D.B. (2003) Responsible conduct of research. Oxford University Press Inc.
- Sharpe, I., Davison, C.M. (2021) Climate change, climate-related disasters and mental disorder in low- and middle-income countries: a scoping review. BMJ Open 11(10).
- Shulla, K., Voigt, B. F., Cibian, S., Scandone, G., Martinez, E., Nelkovski, F., Salehi, P. (2021) Effects of Covid-19 on the sustainable development goals (SDGs). Discover Sustainability. https://doi.org/10.1007/s43621-021-00026-x
- Sida (2015) Gender Analysis Principles & Elements. Swedish International Development Cooperation Agency. https://www.sida.se/contentassets/3a820dbd152f4fca98bacde8a8101e15/gendertool-analysis.pdf
- Sifa, S.F., Sultana, R., Bodrud-Doza, M. (2021) Climate Change and COVID-19: Crisis Within Crises for Eradication of Poverty in Bangladesh. In Muthu, S.S. (Ed.) COVID-19. Environmental Footprints and Eco-design of Products and Processes. Springer, Singapor.
- Singh, K., Bloom, S., Brodish, P. (2015) Gender equality as a means to improve maternal and child health in Africa. Health care for women international 36(1), 57–69. https://doi.org/10.1080/07399332.2013.824971
- Singh, P., Charan, D., Kaur, M., Railoa, K., Chand, R. (2020) Place attachment and cultural barriers to climate change-induced relocation: lessons from vunisavisavi village, vanua levu, Fiji. In Leal Filho, W. (Ed.) Managing Climate Change Adaptation in the Pacific Region. Climate Change Management, Springer Cham, pp. 27-43.
- Sinha, A., Sengupta, T., Alvarado, R. (2020) Interplay between technological innovation and environmental quality: Formulating the SDG policies for next 11 economies. Journal of Cleaner Production 242, 118549.
- Slavchevska, V. (2016) World feminization of agriculture in the context of rural transformations: what is the evidence? The World Bank, Washington, DC. https://doi.org/10.1596/25099
- Slutskiy, P. (2020) Freedom of Expression, Social Media Censorship, and Property Rights. Tripodos 48, 53-67. http://doi.org/10.51698/tripodos.2020.48p53-67
- Smas, L., Fredricsson, C., Claessen, H. (2013) Demographic changes, housing policies and urban planning Examples of situations and strategies in Nordic municipalities. NORDREGIOWORKING PAPER 2013, 4. https://www.divaportal.org/smash/get/diva2:700286/FULLTEXT01.pdf
- Smiciklas, J., Menon, M., Carriero, D., Wakhlu, V., Geray, O., Vartto, H., Stankus, A., Galal, H., Demaithan, H.B. (2017) Connecting cities and communities with the Sustainable Development Goals. CBD, ECLAC, FAO, ITU, UNDP, UNECA, UNECE, UNESCO, UN Environment, UNEP-FI, UNFCCC, UN-Habitat, UNIDO, UNU-EGOV, WMO. Switzerland, Geneva. https://unece.org/DAM/hlm/documents/Publications/U4SSC_Deliverable-Connecting-Citiesand-Communities.pdf
- Sogani, R., Viswanathan, K.R. (2020) Gender-Sensitive approaches and issues of Urban climate changes: Benefits and challenges. In Joshi, D., Brassard, C. (Eds.)

Urban spaces and gender in Asia. Sustainable Development Goals series, Springer Cham, pp. 177–196. https://doi.org/10.1007/978-3-030-36494-6_11

- Song, Y., Zhang, L., Sun, D. et al. (2009) Feminization of agriculture in rapid changing rural China: policy implication and alternatives for an equitable growth and sustainable development. FAO-IFAD-ILO workshop on Gaps, trends and current research in gender dimensions of agricultural and rural employment: differentiated pathways out of poverty. Rome. http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.521.8027andrep=rep1 andtype=pdf
- Soochnapreneur (2021) About the Programme. https://soochnapreneur.in/about-the-programme/
- Soroptimist International of the Americas (2011) White Paper: Reaching Out to Women When Disaster Strikes; Soroptimist International of the Americas: Philadelphia, PA, USA.
- Sovacool, B.K., Valentine, S.V., Jain Bambawale, M., Brown, M.A., de Fátima Cardoso, T., Nurbek, S., Suleimenova, G., Li, J., Xu, Y., Jain, A. et al. (2012) Exploring propositions about perceptions of energy security: An international survey. Environmental Science & Policy 16, 44–64.
- Sovacool, B.K., Rogge, J-C. Saleta, C., Masterson-Cox, E. (2019) Transformative versus conservative automotive innovation styles: contrasting the electric vehicle manufacturing strategies for the BMW i3 and Fiat 500e. Environmental Innovation and Societal Transitions 33, 45-60.
- Spiteri, J. (2018) Why we should start early with ESD for lifelong learning. In Leal Filho, W., Mifsud, M., Pace, P. (Eds.) Handbook of Lifelong Learning for Sustainable Development. Springer: World Sustainability Series.
- Spiteri, J. (2020) Early childhood education for sustainability. In Leal Filho, W., Azul, A.M., Brandli, L., Ozuyar, P.G., Wall, T. (Eds.) Quality Education. Encyclopedia of the UN Sustainable Development Goals. Switzerland, Cham: Springer.
- Springer (2020) Title, Abstract and Keywords https://www.springer.com/gp/authorseditors/authorandreviewertutorials/writing-ajournal-manuscript/title-abstractand-keywords/10285522 Accessed 22 Feb 2021
- Sterling, S. (2016) A Commentary on Education and Sustainable Development Goals. Journal of Education for Sustainable Development 10(2), 208-213. https://doi.org/10.1177/0973408216661886
- Stevanović, M., Popp, A., Lotze-Campen, H., Dietrich, J.P., Müller, C., Bonsch, M., Schmitz, C., Bodirsky, B.L., Humpenöder, F., Weindl, I. (2016) The impact of high-end climate change on agricultural welfare. Science Advances 2(8). https://doi.org/10.1126/sciadv.1501452
- "Strategy Kazakhstan-2050": A New Political Course of the Established State (2014) https://afmrk.gov.kz/en/activity/strategy-and-program/strategy-kazakhstan-2050/ Accessed 20 Feb 2023
- Stromquist, N.P. (2020) Girls and women in the educational system: The curriculuar challenge. Prospects 49, 47–50. https://doi.org/10.1007/s11125-020-09482-1
- Stucker, D., Kazbekov, J., Yakubov, M., Wegerich, K. (2012) Climate Change in a Small Transboundary Tributary of the Syr Darya Calls for Effective Cooperation and Adaptation. Mountain Research and Development 32, 275–285.
- Stucker, D., Kazbekov, J., Yakubov, M., Wegerich, K. (2014) Adaptation to climate change-exacerbated water scarcity, droughts and flashfloods: The khojabakirgansai, a small transboundary tributary of the Syr Darya in Kyrgyzstan

and Tajikistan. Adaptation to Climate Change through Water Resources Management: Capacity, Equity and Sustainability; Routledge: London, UK, pp. 43–66.

- Sweetman, C. (Ed.) (1999) Women, land and agriculture. Oxfam GB. https://oxfamilibrary.openrepository.com/bitstream/handle/10546/121139/bkwomen-land-agriculture-010199-en.pdf?sequence=1andisAllowed=y
- Swiss Re Institute (2021) The economics of climate change: no action not an option. https://www.swissre.com/dam/jcr:e73ee7c3-7f83-4c17-a2b8-8ef23a8d3312/swiss-re-institute-expertise-publication-economics-of-climatechange.pdf
- Tapsell, S., Penning-Rowsell, E., Tunstall, S.M., Wilson, T.L. (2002) Vulnerability to flooding: Health and social dimensions, Philosophical Transactions. Series A, Mathematical, Physical, and Engineering Sciences 360(1796), 1511-1525.
- Teotia, M.K. (2015) Housing for the Urban Poor in Chandigarh. Shelter 16(2), 56-63.
- Terry, J.P., Lau, A.A. (2018) Magnitudes of nearshore waves generated by tropical cyclone winston, the strongest landfalling cyclone in south pacific records. unprecedented or unremarkable? Sedimentary Geology 364, 276-285.
- Terry, K., Rai, A. (2023). Amid record drought and food insecurity, East Africa's Protracted Humanitarian Crisis Worsens, Migration Policy Institute.
- The Constitution of the Kyrgyz Republic as Last Amended of 5 May 2021. Article 24. http://cbd.minjust.gov.kg/act/view/ru-ru/112213?cl=ru-ru Accessed 20 Feb 2023
- The Constitution of the Republic of Kazakhstan as Last Amended of 08.06.2022. Article 13. https://www.akorda.kz/en/official_documents/constitution Accessed 25 Feb 2023
- The Constitution of the Republic of Tajikistan as Last Amended of 22.05.2016. Article 17. https://mfa.tj/en/main/view/70/constitution-of-the-republic-of-tajikistan Accessed 25 Feb 2023
- The Constitution of the Republic of Uzbekistan as Amended of 9 February 2021. Article 46. https://lex.uz/docs/35869 Accessed 20 Feb 2023
- The Constitution of Turkmenistan as Last Amended of 25 September 2020. https://online.zakon.kz/Document/?doc_id=31337929andpos=125;-44#pos=125;-44 Accessed 25 Feb 2023
- The Economist Intelligence Unit (2019) The critical role of Infrastructure for the Sustainable Development Goals. https://1lib.in/book/5418287/4ece50
- The State Committee of Republic Uzbekistan on Statistics (2019a) Demography. https://stat.uz/en/181-ofytsyalnaia-statystyka-en/6383-demography Accessed 7 Nov 2019
- The State Committee of Republic Uzbekistan on Statistics (2019b) National accounts. https://stat.uz/en/181-ofytsyalnaia-statystyka-en/6373-national-accounts Accessed 7 Nov 2019
- The State Committee of Republic Uzbekistan on Statistics (2019c) Labor market. https://stat.uz/ru/164-ofytsyalnaia-statystyka-ru/6580-rynok-truda2 Accessed 7 Nov 2019
- The State Committee of Republic Uzbekistan on Statistics (2019d) Indicators of population employment. Employed persons by sex and economic activity in 2018. https://gender.stat.uz/ru/osnovnye-pokazateli/trud/zanyatost-naseleniya/1025-raspredelenie-chislennosti-zanyatykh-po-vidam-ekonomicheskoj-deyatelnosti-po-polu-v-srednem-za-2018-god Accessed 7 Nov 2019

- TheGlobalEconomy.com (2022) Rural Population, Percent in Africa. https://www.theglobaleconomy.com/rankings/rural_population_percent/Africa/ Accessed 6 June 2022
- The Law of the Kyrgyz Republic on State Guarantees of Equal Rights and Opportunities for Men and Women as Amended of 14 July 2011 No. 97. http://cbd.minjust.gov.kg/act/view/ru-ru/202398 Accessed 20 Feb 2023
- The Law of the Republic of Kazakhstan on State Guarantees of Equal Rights and Equal Opportunities of Men and Women Dated 8 December 2009 No.223—IV. https://adilet.zan.kz/eng/docs/Z090000223 Accessed 25 Feb 2023
- The Law of the Republic of Tajikistan "On State Guarantees of Equality between Men and Women and Equal Opportunities for Their Implementation", Adopted on 11 February 2005. http://ncz.tj/system/files/Legislation/89_ru.pdf Accessed 25 Feb 2023
- The Law of the Republic of Uzbekistan "On Guarantees of Equal Rights and Opportunities for Women and Men" No. ZRU-562 as of 2 September 2019. https://lex.uz/docs/4494873 Accessed 25 Feb 2023
- The Law of Turkmenistan about the State Guarantees of Providing the Equal Rights and Equal Opportunities of Women and Men as Amended of the Law of Turkmenistan of 25 November 2017 No. 661-V. https://cislegislation.com/document.fwx?rgn=78372 Accessed 25 Feb 2023
- The State Committee of the Republic of Uzbekistan on Statistics (2022) Gender Statistics. Labour. https://gender.stat.uz/en/main-indicators/labor Accessed 25 Feb 2023
- The State Committee of the Republic of Uzbekistan on Statistics (2023) National Accounts. https://stat.uz/en/official-statistics/national-accounts Accessed 17 Feb 2023
- The World Bank (2009) Agricultural Activities, Water, and Gender in Tajikistan's Rural Sector: A Social Assessment of Konibodom, Bobojon Ghafurov, and Yovon. World Bank, Government of the Republic of Tajikistan Ferghana Valley, Water Resources Management Project (FVWRMP); The World Bank: Washington, DC, USA.
- The World Bank (2018) Environmental Land Management and Rural Livelihoods Project. https://projects.worldbank.org/en/projects-operations/projectdetail/P122694 Accessed 17 Feb 2023
- The World Bank (2019) Financial Inclusion. Europe and Central Asia Economic Update (Spring), World Bank: Washington, DC, USA.
- The World Bank (2020) The World Bank on Pacific islands, Pacific Islands Overview. www.worldbank.org/en/country/pacificislands/overview Accessed 9 April 2021
- The World Bank (2021a) World Development Indicators. https://databank.worldbank.org/source/world-development-indicators Accessed 17 Feb 2023
- The World Bank (2021b) Employment in Agriculture (% of Total Employment) (Modeled ILO Estimate). International Labour Organization. ILO Modelled Estimates Database. ILOSTAT. https://data.worldbank.org/indicator/SL.AGR.EMPL.ZS?locations=UZ Accesse d 25 March 2023
- The World Bank (2021c) Climate and Environment (CLIENT) Program in Central Asia. https://www.worldbank.org/en/topic/environment/brief/climate-and-environment-program-in-central-asia#Overview Accessed 7 Feb 2023

- The World Bank (2022) Central Asia Water and Energy Program. 2022. https://www.worldbank.org/en/region/eca/brief/cawep Accessed 17 Feb 2023
- The World Bank (2023) Countries and Economies. Projects and Operations. https://data.worldbank.org/country Accessed 25 April 2023
- The World Bank Group, The Asian Development Bank (2021a) Climate Risk
CountryProfile:Kazakhstan.https://climateknowledgeportal.worldbank.org/sites/default/files/2021-08/15834-
WB_Kazakhstan%20Country%20Profile-WEB.pdf Accessed 25 Feb 2023
- The World Bank Group, The Asian Development Bank (2021b) Climate Risk Profile: Kyrgyz Republic. https://www.adb.org/sites/default/files/publication/706596/climate-risk-countryprofile-kyrgyz-republic.pdf Accessed 25 Feb 2023
- The World Bank Group, The Asian Development Bank (2021c) Climate Risk Country Profile:

Tajikistan. https://climateknowledgeportal.worldbank.org/sites/default/files/202 1-09/15919-WB_Tajikistan%20Country%20Profile-WEB.pdf Accessed 25 Feb 2023

- The World Bank Group, The Asian Development Bank (2021d) Climate Risk Country Profile: Turkmenistan. https://climateknowledgeportal.worldbank.org/sites/default/files/2021-06/15837-Turkmenistan%20Country%20Profile-WEB_0.pdf Accessed 25 Feb 2023
- The World Bank Group, The Asian Development Bank (2021e) Climate Risk Country Profile: Uzbekistan. https://climateknowledgeportal.worldbank.org/sites/default/files/2021-09/15838-Uzbekistan%20Country%20Profile-WEB.pdf Accessed 25 April 2023
- Thiri, M.A., Villamayor-Tomás, S., Scheidel, A., Demaria, F. (2022) How social movements contribute to staying within the global carbon budget: Evidence from a qualitative meta-analysis of case studies. Ecological Economics 195, 107356.
- Thomas, A.S., Mangubhai, S., Vandervord, C., Fox, M., Nand, Y. (2019) Impact of tropical cyclone winston on women mud crab fishers in Fiji. Climate and Development 11(8), 699-709.
- Thorlakson, T., Neufeldt, H. (2012) Reducing subsistence farmers' vulnerability to climate change: evaluating the potential contributions of agroforestry in western Kenya. Agriculture & Food Security 1(1), 15. https://doi.org/10.1186/2048-7010-1-15
- Thornton, T.F., Comberti, C. (2017) Synergies and trade-offs between adaptation, mitigation and development. Climatic Change 140, 5–18.
- Tiatia-Seath, J., Underhill-Sem, Y., Woodward, A. (2018) The nexus between climate change, mental health, and wellbeing and pacific peoples. Pacific Health Dialogue 21(2) 47-49.
- Tiwari, G., Kumar, S., Routray, A., Panda, J., Jain, I. (2019) A high-resolution mesoscale model approach to reproduce super typhoon maysak (2015) over northwestern pacific ocean. Earth Systems and Environment 3(1), 101–112.
- Tobi, S.U.M., Razak, K.A., Siow, Y.M., Ramlee, L.H.S., Aris, N.A.M. (2023) Empowering women for disaster risk reduction: A case study of geologically based disaster at Yan, Kedah, Malaysia. IOP Conference Series: Earth and Environmental Science 1144, 012013.

- Tremblay, D., Fortier, F., Boucher, J., Riffon, O., Villeneuve, C. (2020) Sustainable development goal interactions: An analysis based on the five pillars of the 2030 agenda. Sustainable Development 28(6), 1584–1596.
- Trisos, C.H., Adelekan, I.O., Totin, E., Ayanlade, A., Efitre, J., Gemeda, A., et al. (2022) Africa. In: Portner, H.-O., Roberts, D.C., Tignor, M., Poloczanska, E.S., Mintenbeck, K., Alegría, A. et al. (Eds.) Climate change 2022: Impacts, adaptation and vulnerability. Contribution of working group II to the sixth assessment report of the intergovernmental panel on climate change. Cambridge University Press, Cambridge, UK and New York, NY, USA, pp. 1285–1455.
- Tsige, M., Synnevåg, G., Aune, J.B. (2020) Gendered constraints for adopting climate-smart agriculture amongst smallholder Ethiopian women farmers. Scientific African 7, e00250. https://doi.org/10. 1016/j.sciaf.2019.e00250
- Tully, C. (2015) The critical role of effective, accountable and inclusive institutions in implementing the Sustainable Development Goals. Foundation for Democracy and Sustainable Development, London. http://www.fdsd.org/wpcontent/uploads/2015/10/The-critical-role-ofeffective-accountable-andinclusive-institutions.pdf
- UFCOP (2017) Urban flood risk management in the pacific: tracking progress setting priorities. http://documents.worldbank.org/curated/en/421071516106649769/pdf /122545-REVISED Accessed 14 April 2021
- UFPA (2021) Multi-Country Study on Access to Justice for Women and Girls in East and Southern Africa. United Nations Entity for Gender Equality and the Empowerment of Women, UN Women East and Southern Africa Regional Office. https://africa.unwomen.org/en/digital-library/publications/2021/11/multicountry-study-on-access-to-justice-for-women-and-girls-in-east-and-southernafrica
- UN Commission on the Status of Women (2002) Report on the forty-sixth session. Economic and Social Council Official Records, 2002 supplement No. 7 (E/2002/27-E/CN.6/2002/13). United Nations, New York.
- UNCTAD, UNODC (2021) Promotion of international cooperation to combat illicit financial flows and strengthen good practices on asset return to foster sustainable development: Achievement, challenges and way forward. https://unctad.org/system/files/officialdocument/webosg2021d1_en.pdf
- UN Department of Economic and Social Affairs (2020) Goal 5. Achieve gender equality and empower all women and girls. https://sdgs.un.org/goals/goal5
- UN Department of Economic and Social Affairs (2022) World Population Prospects 2022: Summary of Results. UN DESA/POP/2022/TR/NO. 3.
- UNDP (2003) China's accession to WTO: challenges for women in the agricultural and industrial sectors. Overall report. https://www.undp.org/content/dam/china/docs/Publications/UNDP-CH-DG-Chinas%20Ascension%20to%20WTO%20Challenges%20for%20Women%20in %20the%20Agricultural%20and%20Industrial%20Sectors.pdf
- UNDP (2007) Gender Mainstreaming: A Key Driver of Development in Environment and Energy; United Nations Development Programme: New York, NY, USA.
- UNDP (2018) Climate Change Adaptation in Europe and Central Asia: Adapting to a Changing Climate for Resilient Development; Istanbul Regional Hub Regional Bureau for Europe and the CIS United Nations Development Programme; UNDP: New York, NY, USA.
- UNDP (2020) The 2020 Human Development Report. New York, USA.

- UNDP (2021) Gender Equality in Public Administration; UNDP University of Pittsburgh: Pittsburgh, PA, USA.
- UNDP (2022) Advancing Gender Equality in National Climate Plans: Progress and Higher Ambitions. https://www.undp.org/publications/advancing-genderequality-national-climate-plans-progress-and-higher-ambitions Accessed 15 Feb 2023
- UNDP-UNEP Poverty-Environment Initiative (2015a) Kyrgyzstan. https://www.unpei.org/kyrgyzstan-2/ Accessed 17 Feb 2023
- UNDP-UNEP Poverty-Environment Initiative (2015b) Tajikistan. https://www.unpei.org/tajikistan-2/ Accessed 17 Feb 2023
- UNES-CAP (2015) Overview of natural disasters and their impacts in Asia and the Pacific 1970–2014. www.unescap.org/resources/overview-natural-disasters-and-their-impacts-asia-and-pacific-1970-2014 Accessed 14 April 2021
- UNESCO (2014) UNESCO Priority Gender Equality Action Plan 2014-2021. Complementary strategic document to the 37 C/4 and 37 C/5.
- UNESCO (2018a) Situation analysis of SDG4 with a gender lens, Target 4.7. https://unesdoc.unesco.org/ark:/48223/pf0000371227
- UNESCO (2018b) World trends in freedom of expression and media development: global report https://unesdoc.unesco.org/ark:/48223/pf0000261065
- UNFCCC (2017) Gender and climate change. Draft conclusions proposed by the Chair. Recommendation of the Subsidiary Body for Implementation. Draft Decision-/CP.23 Establishment of a Gender Action Plan; FCCC/SBI/2017/L.29; United Nations Framework Convention on Climate Change; UNFCCC: Bonn, Germany.
- UNFCCC (2019) Differentiated impacts of climate change on women and men; the integration of gender considerations in climate policies, plans and actions; and progress in enhancing gender balance in national climate delegations. Subsidiary Body for Implementation, Fiftieth Session, FCCC/SBI/2019/INF.8; Synthesis report by the secretariat. UNFCCC: Bonn, Germany. https://unfccc.int/sites/default/files/resource/sbi2019_inf8.pdf
- UNFCCC (2021) Gender composition Report by the secretariat. Proceedings of the Conference of the Parties Twenty-Sixth Session Glasgow, Glasgow, UK, 31 October–12 November 2021; FCCC/CP/2021/4. UNFCCC: Bonn, Germany.
- UNFCCC Secretariat (2020) Enhanced Lima Work Programme on Gender and Its Gender Action Plan; Decision 3/CP.25; UNFCCC Secretariat: Geneva, Switzerland.
- UNFCCC Secretariat (2022) Dimensions and examples of the gender-differentiated impacts of climate change, the role of women as agents of change and opportunities for women. Synthesis report by the secretariat. Proceedings of the Bonn Climate Change–Conference–June 2022, Bonn, Germany, 6–16 June 2022.
- UNFCCC WEDO (2020) Report: Women's Participation in the UNFCCC. https://wedo.org/wp-content/
- UN General Assembly (2015) Resolution adopted by the General Assembly on 25 September 2015. Transforming our world: the 2030 Agenda for Sustainable Development.

https://www.un.org/ga/search/view_doc.asp?symbol=A/RES/70/1andLang=E

UN-Habitat (2005) Land tenure, housing rights and gender in Brazil. https://www.suelourbano.org/wp-content/uploads/2017/09/Law-Land-Tenureand-Gender-Review-Latin-America-Brazil-1.pdf Accessed 28 Mar 2020

- UN-Habitat (2021) Climate Change and Water 2021. https://unhabitat.org/events/climate-change-and-water-2021
- UNICEF (2009) A Study on Violence Against Girls Report on the International Girl Child Conference. https://www.unicef-

 $irc.org/publications/pdf/violence_girls_eng.pdf$

UNICEF (2017) Ethiopia: Initial Summary of Humanitarian Response Planning 2017.

https://www.unicef.org/ethiopia/ECO_Ethiopia_Initial_Summary_Humanitarian _Response_Planning_for_2017.pdf

- UNICEF Office of Research (2017) Building the Future: Children and the Sustainable Development Goals in Rich Countries. Innocenti Report Card 14, UNICEF Office of Research Innocenti, Florence. https://www.unicef-irc.org/publications/pdf/RC14_eng.pdf
- UNICEF Pacific (2020) Pacific islands tropical cyclone harold humanitarian situation report No. 4.
- UNICEF, The Regional Office for CEE/CIS (2015) Children's Equitable Access to Justice: Central and Eastern Europe and Central Asia. UNICEF, Geneva, https://www.unicef.org/media/50996/file/Equitable_access_to_justice_for_child ren_in_Central_and_Eastern_Europe_and_Central_Asia_-v2_1.pdf
- United Nations (1994) United Nations Convention to Combat Desertification. https://www.unccd.int/sites/default/files/relevantlinks/2017-

01/UNCCD_Convention_ENG_0.pdf

- United Nations (1995) Beijing Declaration and Platform for Action. https://www.un.org/womenwatch/daw/beijing/pdf/BDPfA%20E.pdf
- United Nations (1999) Economic and Social Council's Substantive Session of 1999, Item 14(a) of the provisional agenda, "Social and human rights questions: advancement of women" questions: advancement of women. https://www.un.org/esa/documents/ecosoc/docs/1999/e1999-66.htm
- United Nations (2002) Report of the World Summit on Sustainable Development Johannesburg, South Africa. A/CONF.199/20.
- United Nations (2008) An Introduction to Human Trafficking: Vulnerability, Impact and Action.

https://www.unodc.org/documents/humantrafficking/An_Introduction_to_Huma n_Trafficking_-_Background_Paper.pdf

United Nations (2012) Resolution adopted by the General Assembly on 27 July 2013. A/RES/66/288.

https://www.un.org/ga/search/view_doc.asp?symbol=A/RES/66/288andLang=E United Nations (2015a) Paris Agreement. United Nations: New York, NY, USA.

- United Nations (2015b) Disasters in Asia and the Pacific: 2014 year in a review, knowledge products. www.unescap.org/resources/disasters-asia-and-pacific-2014-year-review-0 Accessed 14 April 2021
- United Nations (2015c) Transforming our world: The 2030 agenda for sustainable development. New York: United Nations, Department of Economic and Social Affairs.
- United Nations (2015d) Eliminating corruption is crucial to sustainable development. Press release. United Nations Office on Drugs and Crime. https://www.unodc.org/unodc/en/press/releases/2015/November/eliminating-corruption-iscrucial-to-sustainable-development.html

United Nations (2016) Disasters in Asia and the Pacific: 2015 year in review. www.unescap.org/sites/default/d8files/knowledge-

products/2015_Year%20in%20Review_final_PDF.pdf Accessed 14 April 2021

- United Nations (2017) Draft decision -/CP.23 Establishment of a gender action plan. Subsidiary Body for Implementation Forty-seventh session FCCC/SBI/2017/L.29.
- United Nations (2018) Working Together: Integration, institutions and the Sustainable Development Goals, World Public Sector Report 2018. Division for Public Administration and Development Management, Department of Economic and Social Affairs, (DPADM), New York.
- United Nations (2019) Gender and climate change proposal by the president Draft decision -/CP.25 Enhanced Lima work programme on gender and its gender action plan. Conference of the Parties twenty-fifth session, FCCC/CP/2019/L.3.
- United Nations (2021) Department of Economic and Social Affairs Sustainable Development, 5 Achieve gender equality and empower all women and girls. https://sdgs.un.org/goals/goal5
- United Nations Environment Programme (2017) Resilience and Resource Efficiency in Cities.

https://wedocs.unep.org/bitstream/handle/20.500.11822/20629/Resilience_resou rce_efficienc%20y_cities.pdf?sequence=1&isAllowed=y

- United Nations Environment Programme (2023) Adaptation Gap Report 2023: Underfinanced. Underprepared. Inadequate investment and planning on climate adaptation leaves world exposed. Nairobi. https://doi.org/10.59117/20.500.11822/43796
- United Nations Human Rights (2020) Appeal 2021. https://www.ohchr.org/Documents/Publications/AnnualAppeal2021.pdf
- United Nations Statistics Division (2016) Goal 11 Make cities and human settlements inclusive, safe, resilient and sustainable. https://unstats.un.org/sdgs/files/metadatacompilation/Metadata-Goal-11.pdf
- United Nations System Chief Executives Board for Coordination (2017) Leaving No One Behind: Equality and Non-Discrimination at the Heart of Sustainable Development: A Shared United Nations System Framework for Action. New York: United Nations.
- United Nations Treaty Collection. Status as at: 27-12-2021 10:15:39 edt Chapter XXVII Environment 7. United Nations Framework Convention on Climate Change New York, 9 May 1992. (2021a) https://treaties.un.org/Pages/ViewDetailsIII.aspx?src=INDandmtdsg_no=XXVII -7andchapter=27andTemp=mtdsg3andclang=_en Accessed 25 Feb 2023
- United Nations Treaty Collection. Status as at: 27-12-2021 10:15:39 edt Chapter XXVII Environment 7. a Kyoto Protocol to the United Nations Framework Convention on Climate Change. Kyoto, 11 December 1997. (2021b) https://treaties.un.org/Pages/ViewDetails.aspx?src=TREATYandmtdsg_no=XX VII-7-aandchapter=27andclang=_en Accessed 25 Feb 2023
- United Nations Treaty Collection. Status as at: 27-12-2021 10:15:39 edt Chapter XXVII Environment 7. d Paris Agreement. Paris, 12 December 2015. (2021c) https://treaties.un.org/Pages/ViewDetails.aspx?src=TREATYandmtdsg_no=XX VII-7-dandchapter=27andclang=_en Accessed 15 Feb 2023
- University of Notre Dame (2022) Notre Dame Global Adaptation Initiative. Country Index. https://gain.nd.edu/our-work/country-index/ Accessed 25 Feb 2023

- UN Secretary-General (2008) Overview of United Nations activities in relation to climate change: report of the Secretary-General (A/62/644). United Nations: New York, NY, USA. https://digitallibrary.un.org/record/617513?ln=en
- UNSDG (2018) Gender equality: A key SGD accelerator. A case study from the Republic of Moldova.
- UN Sustainable Development Goals (2020) Gender equality grows as key aspect of sustainable ocean management. https://www.un.org/sustainabledevelopment/blog/2020/03/genderequality-grows-as-key-aspect-of-sustainable-ocean-management/
- UN Women (2001) Gender Mainstreaming. Concepts and definitions. https://www.un.org/womenwatch/osagi/conceptsandefinitions.htm
- UN Women (2012) Facts and Figures. https://www.unwomen.org/en/news/in-focus/commission-on-the-status-of-women-2012/facts-and-figures
- UN Women (2016) Implementation of Gender-Responsive Climate Action in the Context of Sustainable Development. https://unfccc.int/files/gender_and_climate_change/application/pdf/egmreport.p df Accessed 25 Feb 2023
- UN Women (2018a) Turning promises into action: Gender equality in the 2030 Agenda for Sustainable Development.
- UN Women (2018b) Gender Equality and the Sustainable Development Goals in Asia and the Pacific. Baseline and pathways for transformative change by 2030. https://asiapacific.unwomen.org/en/digital-library/publications/2018/10/apsdg
- UN Women (2020a) Commission on the Status of Women. https://www.unwomen.org/en/csw
- UN Women (2020b) COVID-19 and Violence against Women and Girls: Addressing the Shadow Pandemic. Policy Brief no. 17. New York: UN Women.
- UN Women (2020c) Policy Brief: The Impact of COVID-19 on Women. United Nations Entity for Gender Equality and the Empowerment of Women.
- UN Women (2020d) About UN Women. https://www.unwomen.org/en/about-us/about-un-women
- UN Women (2020e) Fund for Gender Equality. https://www.unwomen.org/en/trustfunds/fund-for-gender-equality
- UN Women (2020f) SDG12: Ensure sustainable consumption and production patterns. https://eca.unwomen.org/en/news/in-focus/women-and-the-sdgs/sdg-12-responsibleconsumption-production
- UN Women (2020g) Women and sustainable development goals. https://sustainabledevelopment.un.org/content/documents/2322UN%20Women %20Analysis%20on%20Women%20and%20SDGs.pdf
- UN Women, UNDP, UNEP (2015) Empowering women for sustainable energy solutions to address climate change Experiences from UN Women and UNDP-UNEP PEI Africa. Working Paper. UN Women Regional Office for Eastern and Southern Africa (ESARO): Nairobi, Kenya; United Nations Development Programme–United Nations Environment Programme Poverty-Environment Initiative (UNDP-UNEP PEI) Africa: New York, NY, USA.
- Upreti, B.R., Ghale, Y., Shivakoti, S., Acharya, S. (2018) Feminization of agriculture in the eastern hills of Nepal: a study of women in cardamom and ginger farming. SAGE Open 8, 1–12. https://doi.org/10.1177/2158244018817124
- Upward, K., Saunders, V., Maple, M., Usher, K. (2021) Mental health, climate change, and bushfires: what's colonisation got to do with it? International Journal of Mental Health Nursing 30(6), 1473–1475.

- USAID (2021) Women's Empowerment and Gender Equality. USAID Programs Empower Women and Girls in Central Asia. https://2017-2020.usaid.gov/centralasia-regional/fact-sheets/womens-empowerment-and-gender-equality Accessed 7 Aug 2023
- Vakulchuk, R., Daloz, A.S., Overland, I., Sagbakken, H.F., Standal, K. (2022) A void in Central Asia research: Climate change. Central Asian Survey 42, 1–20.
- Vandenabeele, C., Lao, C.V. (2007) Legal Identity for Inclusive Development. Asian Development Bank. https://thinkasia.org/bitstream/handle/11540/227/legalidentity.pdf?sequence=1
- van der Straaten, J. (2015) Legal Identity for All by 2030: How Will We Know? workshop of the Open Society Justice Initiative (OSJI) with support from Civil Registration Centre for Development (CRC4D).
- van Eck, N. J., and Waltman, L. (2010) Software survey: VOSviewer, a computer program for bibliometric mapping. Scientometrics 84(2), 523–538. https://doi.org/10.1007/s11192-009-0146-3
- van Eck, N.J., Waltman, L. (2014) Visualizing Bibliometric Networks. In Ding, Y., Rousseau, R., Wolfram, D. (Eds.) Measuring Scholarly Impact Springer International Publishing, Springer: Cham, Switzerland. https://doi.org/10.1007/978-3-319-10377-8 13
- Vanwey, L. (2005) Land ownership as a determinant of international and internal migration in Mexico and internal migration in Thailand. International Migration Review 39(1), 141–172.
- van Zanten, J.A., van Tulder, R. (2021) Improving companies' impacts on sustainable development: A nexus approach to the SDGS. Business Strategy and the Environment 30(8), 3703–3720. https://doi.org/10.1002/bse.2835
- V-Dem Institute (2015) Measuring Accountability in Sustainable Development Target 16.6 with V-Dem Data. Policy Brief No. 1.
- V-Dem Institute (2017) Measuring Corruption in Sustainable Development, Target 16.5 with V-Dem Data. Policy Brief No. #13. https://www.vdem.net/media/filer_public/a3/1b/a31b2dda-4b98-47fb-b1f4-58f8075e1da7/vdem_policybrief_13_2017.pdf
- Vepa, S.S. (2005) Feminisation of agriculture and marginalisation of their economic stake. Economic and Political Weekly 40, 2563–2568. https://www.jstor.org/stable/4416785 Accessed 22 Nov 2019
- Vitousek, S., Barnard, P.L., Fletcher, C.H., Frazer, N., Erikson, L., Storlazzi, C.D. (2017) Doubling of coastal flooding frequency within decades due to sea-level rise. Scientific Reports 7(1), 1-9.
- von Lander Svendsen, N., Weber, K., Factor, G., Winther Engelsbak, L., Fischer-Bogason, R. (2022) How Climate Policies Impact Gender and Vice Versa in the Nordic Countries; Nordic Council of Ministers: Copenhagen, Denmark.
- von Stechow, C., Minx, J.C., Riahi, K., Jewell, J., McCollum, D.L., Callaghan, M.W., Bertram, C., Luderer, G., Baiocchi, G. (2016) 2 °C and SDGs: united they stand, divided they fall? Environmetal Research Letters 11(3), 034022. https://doi.org/10.1088/1748-9326/11/3/034022
- Vyas-Doorgapersad, S. (2019) Gender equality for achieving sustainable development goal one (no poverty) in South African municipalities. International Journal of Social Sciences and Humanity Studies 11(1), 1150–1198. https://doi.org/10.3390/su11041150
- Wagner, R., Ward, N., Percy, F. (2015) ALP Adaptation Strategies Compendium; Adaptation Learning Programme CARE International: Nairobi, Kenya.

- Wang, B., Pan, S.-Y., Ke, R.-Y., Wang, K., Wie, Y.-M. (2014) An overview of climate change vulnerability: A bibliometric analysis based on Web of Science database. Natural Hazards 74(3), 1649–1666. https://doi.org/10.1007/s11069-014-1260-y
- Wang, X., Ren, H., Wang, P., Yang, R., Luo, L., Cheng, F. (2018a) A Preliminary Study on Target 11.4 for UN Sustainable Development Goals. International Journal of Geoheritage and Parks 6(2), 18–24. https://doi.org/10.17149/ijgp.j.issn.2577.4441.2018.02.002
- Wang, Z., Zhao, Y., Wang, B. (2018b) A bibliometric analysis of climate change adaptation based on massive research literature data. Journal of Cleaner Production 199(20), 1072–1082. https://doi.org/10.1016/j. jclepro.2018.06.183
- Warchold, A., Pradhan, P., Kropp, J.P. (2021) Variations in sustainable development goal interactions: Population, regional, and income disaggregation. Sustainable Development 29, 285–299.
- Warner, K., Hamza, M., Oliver-Smith, A., Renaud, F., Julca, A. (2010) Climate change, environmental degradation and migration. Natural Hazards 55(3), 689–715.
- Watkiss, P. Troeltzsch, J., Katriona McGlade, K. (Eds.) (2018) The Economic Cost of Climate Change in Europe: Synthesis Report on State of Knowledge and Key Research Gaps. Policy brief by the COACCH project.
- Watts, N., Amann, M., Arnell, N., Ayeb-Karlsson, S., Belesova, K., Boykoff, M. et al. (2019) The 2019 report of The Lancet Countdown on health and climate change: ensuring that the health of a child born today is not defined by a changing climate. Lancet 394(10211), 1836–78.
- WBG Climate Change Knowledge Portal (2020) Kenya Water Dashboard. Data Description.

https://climateknowledgeportal.worldbank.org/country/kenya/climate-sector-water

- WECF International (2021) Green Villages Central Asia. https://www.wecf.org/green-villages/ Accessed 17 Feb 2023
- WEDO (2008) Global gender and climate alliance. Women's Environment and Development Organization. https://wedo.org/global-genderand-climate-alliance/
- WEDO (2023) Women's Participation Statistics in Climate Diplomacy. Gender Climate Tracker. https://genderclimatetracker.org/womens-participation-party-delegations Accessed 25 April 2023
- Weiss, T.G., Wilkinso, R. (2018) The Globally Governed— Everyday Global Governance. Global Governance 24(2), 193–210. https://doi.org/10.1163/19426720-02402003
- Wen, N., Xiaoming, H., George, C. (2013) Gender and political participation: News consumption, political efficacy and interpersonal communication. Asian Journal of Women's Studies 19, 124–149.
- Whaites, A. (2016) Achieving the impossible: can we be SDG 16 believers? GovNet Background Paper No2, 2016. OECD.
- White, H.D., McCain, K.W. (1998) Visualizing a discipline: An author co-citation analysis of information science, 1972–1995. Journal of the American Society for Information Science 49(4), 327–355.
- WHO (2011) World report on disability 2011. https://disabilityinclusion.msf.org/assets/files/WorldReport_eng.pdf

- WHO (2013) Country page/ South pacific 2013. https://apps.who.int/iris/bitstream/handle/10665/136831/ccsbrief_pci_en.p df?sequence=1andisAllowed=y Accessed 9 April 2021
- WHO (2014) Gender, Climate Change and Health; World Health Organization: Geneva, Switzerland. https://apps.who.int/iris/handle/10665/144781
- WHO (2018) Climate and health country profile 2017 Kiribati climate and health country Profile 2017.
- WHO (2020) Global status report on preventing violence against children 2020. https://resourcecentre.savethechildren.net/node/17789/pdf/9789240004191eng.pdf
- WHO (2023) Climate change. https://www.who.int/news-room/fact-sheets/detail/climate-change-and-health
- Williams, M.N., Hill, S.R., Spicer, J. (2015) Will climate change increase or decrease suicide rates? The differing effects of geographical, seasonal, and irregular variation in temperature on suicide incidence. Climatic Change 130(4), 519-528.
- Willox, A., Harper, S., Ford, J., Landman, K., Houle, K., Edge, V., Rigolet Inuit Community Government (2012) From this place and of this place: climate change, sense of place and health in nunatsiavut Canada. Social Science and Medicine 75(3), 538-547.
- Wilson, C., Pettifor, H., Cassar, E., Kerr, L., Wilson, M. (2018) The potential contribution of disruptive low-carbon innovations to 1.5 °C climate mitigation. Energy Efficiency 12, 423–440. https://doi.org/10.1007/s12053-018-9679-8
- WMO (2021a) State of the Climate in Asia 2020. World Meteorological
OrganizationMeteorological
Switzerland.

https://library.wmo.int/doc_num.php?explnum_id=10867

- WMO (2021b) WMO Atlas of mortality and economic losses from weather, climate and water extremes (1970–2019). World Meteorological Organization. https://library.wmo.int/doc_num.php?explnum_id=10989
- WMO (2022) 2022 State of Climate Services: Energy. WMO-No. 1301. World Meteorological Organization, Geneva. https://library.wmo.int/idurl/4/58116
- Woetzel, J., Ram, S., Mischke, J., Garemo, N., Sankhe, S. (2014) A blueprint for addressing the global affordable housing challenge. McKinsey Global Institute.
- Woetzel, J., Madgavkar, A., Ellingrud, K., Labaye, E., Devillard, S., Kutcher, E., Manyika, J., Dobbs, R., Krishnan, M. (2015) The Power of Parity: How Advancing Women's Equality Can Add \$12 Trillion to Global Growth; McKinsey Global Institute, McKinsey and Company: New York, NY, USA.
- Women for Climate-Resilient Societies (2020) Think Piece: Gender and Climate Change in the Context of COVID-19. https://wedocs.unep.org/20.500.11822/32901
- Women4Climate(2021)Women4ClimateNur-Sultan.https://w4c.org/mentorship/women4climate-nur-sultan Accessed 17Feb 2023
- Women4ClimateAction (2019) Daring Circle Women Leading Climate Action: A World Within Reach; Women's Forum for the Economy and Society: Deauville, France.
- Working group of the Interdepartmental Commission on the implementation of the international obligations of Turkmenistan in the field of human rights and international humanitarian law (2016) National integrated review of implementation of the Beijing Declarations and Platform for Action 1995 (period

2014–2019 years) Turkmenistan. Ashgabat. https://www.unece.org/fileadmin/DAM/Gender/Beijing_20/Turkmenistan.pdf

- Workneh, M.A. (2020) Gender inequality governance, and poverty in Sub-Saharan Africa. Poverty and Public Policy 12(2), 150–174. https://doi.org/10.1002/pop4.278
- World Banana Forum (2017) Women's employment in the banana industry. FAO. http://www.fao.org/3/i8285en/i8285en.pdf
- World Bank (2014a) Building back better in Tonga after cyclone Ian. www.worldbank.org/en/results/2014/10/01/building-back-better-tonga-cyclone-ian Accessed 9 April 2021
- World Bank (2014b) Initiatives to promote gender equality in Latin America and the Caribbean. https://www.worldbank.org/en/events/2014/11/24/gender-equality-lac Accessed 28 Mar 2020
- World Bank (2015) International Bank for Reconstruction and Development, International Finance Corporation Multilateral investment guarantee agency joint country engagement note (CEN) for Turkmenistan for the period FY16-FY17. http://documents.worldbank.org/curated/en/371591467987825776/Turkmenistan -Country-engagement-note-for-the-period-FY20167 Accessed 22 Nov 2019
- World Bank (2016) Women in agriculture. The impact of male out-migration on women's agency, household welfare, and agricultural productivity. http://documents.worldbank.org/curated/en/162161468017454186/pdf/AUS914
 7-REVISED-WP-new-title-PUBLIC-women-agriculture-04-14-FINAL.pdf
 Accessed 28 Mar 2020
- World Bank (2018a) Male outmigration and women's work and empowerment in agriculture. The case of Nepal and Senegal. Washington, DC. http://documents.worldbank.org/curated/en/653481530195848293/pdf/Male-outmigration-and-womens-work-and-empowerment-in-agriculture-the-case-of-Nepal-and-Senegal.pdf
- World Bank (2018b) Rural population (% of total population) Latin America and Caribbean.

https://data.worldbank.org/indicator/SP.RUR.TOTL.ZS?locations=ZJ Accessed 28 Mar 2020

- World Bank (2018c) Disaster Risk Management Development Policy Financing with a Catastrophe Deferred Drawdown Option. http://documents.worldbank.org/curated/en/131661529811034069/pdf/KENYA-DDO-NEWPAD-2-05312018.pdf
- World Bank (2019a) World Bank Country and Lending Groups. https://datahelpdesk.worldbank.org/knowledgebase/articles/906519-world-bankcountry-and-lending-groups Accessed 7 Nov 2019
- World Bank (2019b) The World Bank in Turkmenistan. Overview. https://www.worldbank.org/en/country/turkmenistan/overview Accessed 7 Nov 2019
- World Bank (2019c) Personal remittances, received (% of GDP). https://data.worldbank.org/indicator/BX.TRF.PWKR.DT.GD.ZS Accessed 7 Nov 2019
- World Bank (2019d) Rural population (% of total population) Turkmenistan. https://data.worldbank.org/topic/agriculture-and-ruraldevelopment?locations=TMandview=chart Accessed 7 Nov 2019

- World Bank (2019e) World employment social outlook. Trends 2019. Geneva. http://www.ilo.org/wcmsp5/groups/public/%2D%2D-dgreports/%2D%2Ddcomm/%2D%2D-publ/documents/publication/wcms_615594.pdf
- World Bank (2019f) Employment in agriculture, male (% of male employment) (modeled ILO estimate). https://data.worldbank.org/indicator/SL.AGR.EMPL.MA.ZS Accessed 7 Nov 2019
- World Bank (2019g) Employment in agriculture (% of total employment) (modeled ILO estimate). https://data.worldbank.org/indicator/SL.AGR.EMPL.ZS Accessed 7 Nov 2019
- World Bank (2019h) Employment in agriculture, female (% of female employment) (modeled ILO estimate). https://data.worldbank.org/indicator/SL.AGR.EMPL.FE.ZS Accessed 7 Nov 2019
- World Bank (2019i) Population, total Latin America and Caribbean. https://data.worldbank.org/indicator/SP.POP.TOTL?locations=ZJ Accessed 28 Mar 2020
- World Bank (2019j) World Development Indicators (WDI). https://databank.worldbank.org/source/world-development
 - indicators/Type/TABLE/preview/on# Accessed 28 Mar 2020
- World Bank (2019k) Agriculture, forestry, and fishing, value added (% of GDP) Latin America and Caribbean. https://data.worldbank.org/indicator/NV.AGR.TOTL.ZS?locations=ZJ Accessed 28 Mar 2020
- World Bank (2019l) Women in Water Utilities: Breaking Barriers; World Bank: Washington, DC, USA.
- World Bank (2020a) Literacy rate, adult female (% of females ages 15 and above). https://data.worldbank.org/indicator/SE.ADT.LITR.FE.ZS
- World Bank (2020b) Literacy rate, adult male (% of males ages 15 and above). https://data.worldbank.org/indicator/SE.ADT.LITR.MA.ZS
- World Bank (2020c) Employment in agriculture (% of total employment) (modeled ILO estimate). https://data.worldbank.org/indicator/SL.AGR.EMPL.ZS
- World Conservation Society (2016) Impact of Tropical Cyclone Winston on Fisheries-Dependent Communities in Fiji, World Conservation Society, Suva, Fiji.
- World Economic Forum (2020) Global Gender Gap Report 2020. World Economic Forum, Switzerland. http://www3.weforum.org/docs/WEF_GGGR_2020.pdf
- World Economic Forum (2021) Global Gender Gap Report 2021. World Economic Forum: Geneva, Switzerland. https://www3.weforum.org/docs/WEF_GGGR_2021.pdf Accessed 7 March 2022
- World Economic Forum. Global Gender Gap Report (2022) https://www3.weforum.org/docs/WEF_GGGR_2022.pdf
- World Wide Fund for Nature (2006) Climate Change Impacts on East Africa: A Review of the Scientific Literature. WWF, Morges, Gland.
- Wrigley-Asante, C., Owusu, K., Egyir, I.S., Owiyo, T. M. (2019) Gender dimensions of climate change adaptation practices: The experiences of smallholder crop farmers in the transition zone of Ghana. African Geographical Review 38(2), 126– 139. https://doi.org/10.1080/19376812.2017.1340168

- Wu, F., Geng, Y., Tian, X., Zhong, S., Wu, W., Yu, S., Xiao, S. (2018) Responding climate change: A bibliometric review on urban environmental governance. Journal of Cleaner Production 204, 344–354. https://doi.org/10.1016/j.jclepro.2018.09.067
- Xenarios, S., Gafurov, A., Schmidt-Vogt, D., Sehring, J., Manandhar, S., Hergarten, C., Shigaeva, J., Foggin, M. (2019) Climate Change and Adaptation of Mountain Societies in Central Asia: Uncertainties, Knowledge Gaps, and Data Constraints. Regional Environmental Change 19, 1339–1352.
- Xi, H. (2016) Special report: Climate change poses grave threats to China's essential infrastructure. China dialogue. https://www.chinadialogue.net/article/show/single/en/8847-Special-report-Climate-change-posesgrave-threats-to-China-s-essential-infrastructure
- Yadav, S.S., Lal, R. (2018) Vulnerability of women to climate change in arid and semi-arid regions: The case of India and South Asia. Journal of Arid Environments 149, 4–17.
- Yared, T., Ayal, D.Y., Kassahun, T., Tadesse, T., (2022) Impact of climate variability and change on household's food security status; the case of Godere woreda, Gambella region, Ethiopia. Climate Services 27, 100307.
- Zakari, S., Ibro, G., Moussa, B., Abdoulaye, T. (2022) Adaptation strategies to climate change and impacts on household income and food security: Evidence from Sahelian region of Niger. Sustainability 14, 2847.
- Zamfir, I. (2020) Peace, justice and strong institutions EU support for implementing SDG 16 worldwide. European Parliamentary Research Service.
- Zapp, M. (2018) The scientization of the world polity: International organizations and the production of scientific knowledge, 1950–2015. International Sociology 33(1), 3–26. https://doi.org/10.1177/0268580917742003
- Zhang, L., Gong, J., Zhang, Y. (2016) A review of ecosystem services: A bibliometric analysis based on Web of science. Acta Ecologica Sinica 36(18), 5967–5977. https://doi.org/10.5846/stxb201504060688
- Zhang, X., Estoque, R.C., Xie, H., Murayama, Y., Ranagalage, M. (2019) Bibliometric analysis of highly cited articles on ecosystem services. PLoS One 14(2), e0210707. https://doi.org/10.1371/journal.pone.0210707
- Zhou, Y., Sun, X. (2020) Toward gender sensitivity: Women and climate change policies in China. International Feminist Journal of Politics 22(1), 127–149. https://doi.org/10.1080/14616742.2019.1687001
- Zupic, I., Čater, T. (2014) Bibliometric Methods in Management and organization. Organizational Research Methods 18(3), 429–472. https://doi.org/10.1177/1094428114562629
- Zyoud, S.H., Fuchs-Hanusch, D. (2020) Mapping of climate change research in the Arab world: A bibliometric analysis. Environmental Science and Pollution Research 27(3), 3523–3540. https://doi.org/10.1007/s11356-019-07100-y

12. Appendices

Appendix I

Table 12.1. List of Abbreviations of the UNFCCC bodies (based on Gender Climate Tracker). Source: author.

Abbreviation	Name
CTCN Advisory Board	Climate Technology Centre and Network Advisory Board
CC Facilitative Branch	Compliance Committee Facilitative Branch
CC Enforcement Branch	Compliance Committee Enforcement Branch
CGE	Consultative Group of Experts
CDM Executive Board	Clean Development Mechanism Executive Board
GCF Board	Green Climate Fund Board
JISC	Joint Implementation Supervisory Committee
LEG	Least Developed Countries Expert Group
РССВ	Paris Committee on Capacity-building
SCF	Standing Committee son Finance
TEC	Technology Executive Committee
FWG	Facilitative Working Group
KCI	Katowice Committee of Experts on the Impacts of the Implementation of Response Measures
PAICC	Paris Agreement Implementation and Compliance Committee
СОР	Conference of the Parties

Appendix II

Water	Economic Development	Education	Agriculture	Health	Information	Weather events	Other
Women's - Low v qualit - Water scarci - Droug - Insuff water irrigat purpo - Lack low le of wa supply house needs irrigat purpo - Addit to house and jo respon ties bu findin water irrigat	 vater y No economic robustness Decrease in economic status due to reduction in yields and land degradation Unemployment (men are hired more often because of the physical strength needed for heavy jobs) hold and ion sees ional hold bb nsibili urden g for ion ng ses No economic yrous and the physical strength is the physical	 Lack or low level of education including applied education Low level of knowledge Lack of technical knowledge of adaptation Scarcity of trainings, field trainings, publications and videos 	 Agriculture is the main source of income Reduction in crop yields 	 Low level of medical service Endocrine diseases High percentage of death from cardiovascular diseases Sanitation and hygiene 	- Lack of full access to information among the rural population as a whole regardless of gender	 Temperature increases Heat strokes and sunburns Emergency situations 	 Household activities Low trust in governmental authorities Many project activities on climate change are mainly at the national level, but a larger emphasis should be at the local level Lack of equal access to employment opportunities and decision making Lack of capacity Families of many children High physical load in household High physical load needed for

Table 12.2 Main causes of rural men's and women's vulnerability to climate change. Source: Authors' compilation from the survey responses.

	during drought events - Men farmers can easier and faster negotiate with "water providers" than women.							agricultural production - Traditions/Asian mentality - Low level of culture - Gender inequality in rural areas - Low level of participation
Men's	 Low water quality Water scarcity Drought Water distribution Lack or low level of water supply including for irrigation purposes 	 Low profitability and income Decrease in economic status due to reduction in yields and land degradation Access to loans Enforced inner migration Limited access to financial services 	 Lack or low level of education, including applied education Low level of knowledge Lack of technical knowledge of adaptation Lack of insufficient knowledge on intensive growth of agricultural crops and water savings (switch to drip irrigation) 	 Exposure to natural hazards Reduction in crop yields Low level of food security More problems related to taking care of cattle and harvest Low level of modernization and mechanization of agriculture There are no mechanisms (tractors and machines) that ease the workload of rural men; 	 Low level of medical services Deterioration of health High percentage of death from cardiovascular diseases 	 Lack of full access to information among the rural population as a whole regardless of gender Lack of more precise information on climate change and agronomy 	 Temperature increase Heat strokes and sunburns Emergency situations 	 Men are at risk to their lives in all cases Stress because of responsibility for family and household Too lazy to go out to the field Alcohol addiction

	therefore, the health of rural men decreases every year.		
	- Investments in agriculture are very costly, and the return on these investments is long-term		
	- Increased physical activity in agricultural production		
	- Dependency of crops cultivation and livestock rearing on weather		
	conditions and irrigation water		

#	Project Title	Donor Organization	Country*	Duration	Short Description	Climate Change– Gender Interconnections Domain	Source
1	Land Rights and Economic Security of Rural Women in Tajikistan	UNIFEM, Government of Tajikistan	Tajikistan	2003–2005	Both projects supported land rights and the sustainable livelihoods of women. Among the outcomes are the development of gender statistics and specific indicators, a creation of a	Leadership/ Empowerment Vulnerability	Mirzoeva (2009)
2	Improved Food Security and Enhanced Livelihoods through Institutional and Gender Sensitive Land Reform in Tajikistan	UNIFEM, Government of Tajikistan	Tajikistan	2007–2008	gender network, enhancement of livelihood conditions for women and poor rural families. The projects advanced changes in land policies and legislation, supported farmers' social mobilization.	Leadership/ Empowerment Vulnerability	
3	Surkhandarya Water Supply and Sanitation Project—Outcomes of the Gender Action Plan	ADB	Uzbekistan	2009–2015	The project improves living standards, environment, and public health in the Surkhandarya Province. Beneficiaries of the project are about 340,000 people, approximately 50% of which were women residing in rural and urban areas. Women represented 50% of the participants in all project activities and public meetings; 68% of 153 representatives of consumer and project support groups in all subprojects were women; women contributed to 40% of suggestions for effective project implementation. Women staff contributed to keeping more accurate records of water in branches of the water consumers associations (WCAs).	Vulnerability Leadership/ Empowerment Benefits	Asian Development Bank (2016b)

Table 12.3 Development projects supported by international organizations/donors. Source: author.

	1	1	1	1	1		1
4	Uzbekistan: Water	ADB	Uzbekistan	2009–2016	The project aimed at rehabilitating and	Benefits	Asian
	Resource Management				upgrading selected irrigation systems	Vulnerability	Development
	Project				and improving water management. In	Leadership/	Bank (2017)
					the Namangan, Samarkand, and	Empowerment	
					Fergana regions, the number of women		
					farmers increased by 30%–60% and		
					over 5000 seasonal jobs for women		
					were created by female-headed farms.		
					Profits of women farmers increased		
					37% per hectare making their average		
					profit margin higher in comparison to		
					that of men farmers. Most women in		
					the project area established		
					greenhouses and gardening plots as		
					part of their own business. Women's		
					participation in water consumers		
					associations increased from 3.5% to		
					7.6%. Rehabilitated pump stations		
					improved the environment and reduced		
					allergies among women and children.		
5	Central Asia Water &	The World Bank	Central Asia *	2012-2024	CAWEP program strengthens the	Benefits	The World Bank
	Energy Program				environment to promote energy and	Leadership/	(2022)
					water security at CA and country levels	Empowerment	· · ·
					by leveraging the benefits of improved	1	
					cooperation. The program promotes		
					gender aspects in IWRM, female		
					inclusion in all water/energy/climate		
					change activities, and knowledge-		
					sharing events.		
6	Women & Water in	CAP, George	South and	2013	The project supported a program of	Leadership/	Central Asia
	South & Central Asia	Washington	Central Asia		international leadership and exchange	Empowerment	Program (2022)
		University			of knowledge for innovative conflict	I COLORA	<u> </u>
					resolution with a sustainable and		
					multiplying effect among young		

					women social entrepreneurs and activists from Kyrgyzstan, Tajikistan, Afghanistan, India, and Pakistan. The activists receive an opportunity to share their experience in water management, improve their expertise and leadership skills.		
7	ELMARL (Environmental Land Management and Rural Livelihoods)	The World Bank PPCR Global Environment Facility (GEF)	Tajikistan	2013–2018	The project focused on the improvement of natural resource management to increase production and build resilience to climate change. Women represented 40% of the 320,000 project beneficiaries. The project used a community-led, participatory implementation approach with a main focus on women's participation. 15 locally based international agencies or non- governmental organizations focused on community mobilization to promote gender equality and inclusion of marginalized groups.	Vulnerability Benefits	The World Bank (2018)
8	CLIMADAPT	European Bank for Reconstruction and Development (EBRD) PPCR of the Climate Investment Funds (CIF) Government of the UK and the multi-donor	Tajikistan	2016	The project supported gender-sensitive climate resilience investments in Tajikistan. It assisted men and women farmers to cope with climate change impacts and supported the country's transition towards a green economy by promoting the efficient use of resources such as water, energy and land and increasing access to climate technologies.	Benefits Leadership/ Empowerment	Climate Investment Funds (2018)

		EBRD Early Transition Countries Fund					
9	Women, Water Management and Conflict Prevention— Phase II	OSCE	Central Asia	2017–2022	The project specifically supported better inclusion of gender aspects in water management and water diplomacy, as well as the targeted empowerment of women water professionals. The main beneficiaries are men and women working in the water sector, from government bodies, academic institutions, NGOs and technical agencies with particular attention to young professionals.	Leadership/ Empowerment	Organization for Security and Co- Operation in Europe (2022a)
1 0	Tajikistan: Understanding the Nexus of Migration, Gender, Climate Change and Agriculture	IOM	Tajikistan	2019–2021	The project addressed the migration, gender, climate change, and agriculture nexus in Tajikistan. It also aimed at evaluating capacity-building activities for women from remittance-recipient households to support their adaptation.	Vulnerability	IOM (2019)
1 1	Central Asia climate information products 2020	Zoï CAREC national partners	Central Asia	2020	Number of new climate information products including "Women, food and climate change in Central Asia"	Leadership/ Empowerment Benefits	CAREC (2020)
1 2	Women4Climate Mentorship Program (Nur-Sultan)	C40 Cities	Kazakhstan	2021	The program focuses on sharing knowledge and experiences with the next generation of women leaders on climate change issues. In the frame of the project, 10 women work with mentors, and women leaders from various areas of the private and public. It also supports women in broadening their knowledge of climate change and enhancing their leadership skills	Leadership/ Empowerment	Women4Climate (2021)

	ſ		Г			ر
				through distance learning and		
				increasing networking opportunities.		
1	UNDP-UNEP Poverty-	UNDP	Kyrgyzstan	The initiative collaborates with main	Leadership/	UNDP-UNEP
3	Environment Initiative	UNEP	Tajikistan	government partners to raise	Empowerment	Poverty-
				awareness, influence policy making	Benefits	Environment
				and strengthen the mainstreaming of	Vulnerability	Initiative (2015a;
				the poverty-environment into budget		2015b)
				processes, sector programs, and sub-		
				national planning.		
				Relevant actions:		
				- Kyrgyzstan:		
				Increasing knowledge and		
				understanding of the interconnections		
				between gender, women's		
				empowerment, and environmental		
				sustainability to reduce inequality.		
				Supporting local educational		
				institutions in mainstreaming gender-		
				specific activities, including research		
				on gender aspects of poverty,		
				biodiversity, and climate change		
				linkages.		
				- Tajikistan:		
				At least 50% of the green micro-loans		
				were targeted towards women-led		
				initiatives.		
				Collection of sector and local examples		
				on the poverty, environment and		
				gender nexus to inform planning and		
				budgeting for pro-poor environmental		
				sustainability.		

1 4	Green villages Central Asia	WECF international	Tajikistan, Kyrgyzstan		Collaboration with women farmer organizations, renewable energy cooperatives, water providers, and eco- tourism operators. Rural women's organizations work on women's economic empowerment and rights. Networking with local women and environmental partners, cooperation with local authorities, financial micro- credit cooperatives, and United Nations	Leadership/ Empowerment	WECF International (2021)
					agencies. The Network of rural women organizations in Kyrgyzstan created women's rural cooperatives, and advocates for women's rights and gender equality.		
1 5	Empowering Central Asian Women in Renewable Energy Mentoring Program	OSCE	Central Asia		The program aims at increasing women's representation in managerial and decision-making positions in the renewables sector in Central Asia. It supports mid-career women working in the energy transition to build self- confidence and position themselves for leadership roles.	Leadership/ Empowerment	Organization for Security and Co- Operation in Europe (2022b)
1 6	USAID programs empower women and girls in Central Asia	USAID	Central Asia		Programs promote gender equity in climate change mitigation, natural resource management and water usage; strengthen women and girls' capacity to support their full engagement as managers, partners, and entrepreneurs in water-related activities.	Leadership/ Empowerment	USAID (2021)
1 7	Climate and Environment	The World Bank PROGREEN	Central Asia	2021–2026	The program supports CA countries to attain sustainable, resilient, and inclusive economic growth with a	Leadership/Empower ment	The World Bank (2021c)

(CLIENT) Program in	focus on, among other aspects, climate	
Central Asia	resilience and resilient landscape	
	restoration.	
	Within pillar 3: Communication for	
	Climate and Awareness–C4CA; one of	
	the activities was to implement a	
	feasibility study to raise awareness of	
	climate change, pollution, and	
	landscape restoration practices among	
	Tajikistan's rural youth and women	
	using the AnchorEd Schools project	
	model.	

* if not specified, the project is implemented in all five CA countries
Appendix III

Table 12.4 International case studies illustrating worldwide initiatives that address gender equality among the SDGs. Source: team of authors.

Country	SDG	Case study description	References
International	SDG 1 (No poverty)	Study focusing on "gendered poverty." The study's main aim is on the UN Women and the progress of women in the years 2015-2016, acknowledging that gendered poverty is a consequence of gender inequality.	Bradshaw et al. (2017)
India	SDG 2 (Zero hunger)	Study documenting the relationship between gender inequality and hunger in India. 95 million (2.5% global hunger burden) undernourished people exist. The most vulnerable are women and children. Malnutrition has more effect on women's and girls' health, at a greater risk of giving birth to low-birth weight infants.	Callister (2018)
Tanzania	SDG 3 (Good health and well- being)	Engender Health Tanzania is helping women accessing to reproductive healthcare. With a population of nearly 56.32 million, many Tanzanian women, do not have access to contraceptives, HIV testing/counselling services and health facilities to deliver babies. Since 1982, EngenderHealth has been working with the government, contributing to improving healthcare.	Engender Health (2021)
India	SDG 4 (Quality Education)	Soochnapreneur (Information-Preneur) is a unique project focusing on education about the different government schemes and citizen's rights for people in rural and remote parts of India. Bridging the digital-divide, young rural women and youth, acting as Information-Preneurs, provide the required information to those who have no access to technology, also helping them to generate some small amount of money through information services. 100 Information-Preneurs have assisted more than 7,000 rural people. Many women participate in the project, benefitting directly by earning as Information-Preneurs and indirectly through access to the different government schemes.	Soochnapreneur (2021) Qualcomm Wireless Reach (2018)
Nepal	SDG 6 (Clean water and sanitation)	Study related to the Trans-Boundary Rivers of South Asia (TROSA) programme, an opportunity for women's leadership in water governance to increase social accountability. Women are sensitised on their rights regarding riverine water resource planning and decision-making, advocating for key water governance issues through service providers and local government.	Crawford (2020)

India	SDG 7 (Affordable and clean energy)	Using India as an illustrative case, a mixed-methods study was conducted. Women are neither the sole nor primary beneficiaries of electricity access, even when appliances that would particularly benefit them are affordable. While energy access could improve gender equity, intra-household power dynamics were highlighted as an important boundary condition in realising more equitable energy access.	Rosenberg et al. (2020)
Turkey	IrkeySDG 8 (Decent work and economic growth),Study indicating that a fiscal prioritisation on building a social infrastructure of care, g than investments in physical infrastructure/construction or cash transfers, presents an enormous potential for decent job creation, particularly in the female-dominated occup and sectors, promoting gender equality and better access to decent work.		Ilkkaracan et al. (2015)
Comparison France, Spain, Morocco and Algeria	SDG 9 (Industry innovation infrastructure)	Research focusing on the importance of women's participation as a key element in achieving the SDGs, by analysing the existing digital gender gap with the achievement of the SDGs, concluding that equal access to digital technologies is important to achieving all the SDGs.	Kerras et al. (2020)
Philippines	SDG 10 (Reducing inequality)	Research documenting gender-related indicators that can be used to monitor gender equality and women empowerment. An overview of equality of human capabilities, equality of economic opportunity, equality in political voice and leadership, and the safety of women and girls is provided. Priorities for public policy involving gender transformational issues to attain equality and women's empowerment are given.	David et al. (2018)
South Africa	SDG 11 (Sustainable cities and communities)	Skills-driven project enabling a rural community to drive changes that will improve quality of life. The creation of a female-only entrepreneur craft group is included. The craft group initiative gives the group of women the opportunity to contribute meaningfully towards a collective drive to a more sustainable community.	Pretorius and Nicolau (2020)
Peoples Republic of China	SDG 12 (Responsible consumption and production)	Study focusing on gender awareness in sustainable consumption and production (SCP). Research concludes that integrating gender analysis into the design of SCP policies, addressing remaining gender gaps, and strengthening women's participation in natural resource management and decision-making can have a positive effect and support the shift towards SCP.	Fan and Jaffre (2020)

Pacific Countries	SDG 13 (Climate action)	Study focusing on the findings of research by United Nations (UN) Women 2015 to investigate the connection between gender and climate change in the Pacific Region. The study provides evidence-based information about the gender impact of climate change, how the gender inequalities are directed by climate change adaptation and how to enhance women empowerment.	Aipira et al. (2017)
Australia	SDG 14 (Life below water)	For implementing SDGs 14, Australia funded the "Blue Economy Aquaculture Challenge" initiative, in order to support, projects for transforming sustainable aquaculture practices. Many of the solutions promised outcomes linked to other SDGs, such as gender equality (SDG5), health and nutrition (SDG3) employment, and SDP (SDG 12).	Australian Government (2018)
Latin American countries	SDG 15 (Life on land)	Research focusing on women's unequal access to land and the proposed indicators to measure progress. The study in Latin America demonstrates the current degree of inequality in the gender distribution of landholders and landowners, and why it is important that countries improve gender statistics, collecting gender disaggregated data on both land ownership and agricultural decision-making.	Deere (2018)
Australia	SDG 16 (peace, justice and strong institutions)	The Australian government supports global peace and justice, by acting as a global leader for advancing commitments to the Women, Peace and Security Agenda. Considerable work was done to integrate a gender perspective into international peace and security policy outcomes, training Australian personnel to safeguard the needs of women in conflict zones.	Australian Government (2018)
Australia	SDG 17 (Partnerships for the goal)	The Australian government, in partnership with the Australian National University and the International Women's Development Agency, has introduced the Individual Deprivation Measure (IDM), a gender-sensitive, multidimensional and below household level (individual level). It aligns with 25% of the 53 gender-related SDG Indicators and identifies gender sensitive deprivation, in order to address poverty.	Australian Government (2018)

Appendix IV

Table 12.5 SDG Matrix. Source: team of authors.

SDG Number and Target	Target description	Implications	References
SDG 1 End pov	verty in all its forms everywhere		
Target 1.1	By 2030, eradicate extreme poverty for all people everywhere, currently measured as people living on less than \$1.25 a day	Women and the most vulnerable in society are for the most part more impacted by poverty. As such, ensuring the eradication of extreme poverty will benefit more women and help end and/or reduce poverty. Extreme poverty and its impact on women is exacerbated by discriminatory social norms that alienate women when it comes to equal opportunities in education as well as employment and as such leads to those affected living on less than \$1.25 a day in mostly the developing world.	Olinto et al. (2013) Lang and Lingnau (2015) Franco and Minnery (2020)
Target 1.2	By 2030, reduce at least by half the proportion of men, women and children of all ages living in poverty in all its dimensions according to national definitions	Strong and robust policies can play a role in reducing poverty. For example, as poverty takes many forms including that of energy, a consideration of place of residence, house ownership status, family size, and the age of the primary breadwinner can mitigate against multidimensional energy poverty and play a significant role in national definitions of poverty. Such considerations could have an impact on poverty reduction in at least half of the proportion of men, women and children in the most deprived areas of society.	Abbas et al. (2020)
Target 1.3	Implement nationally appropriate social protection systems and measures for all, including floors, and by 2030 achieve substantial coverage of the poor and the vulnerable	Social security is an important human right which is central to the reduction of poverty, achieving equality and avoiding social exclusion. These aspects allow for the promotion of equal opportunities including gender equality, which if properly implemented can assist with the development of appropriate social security systems and measures that could have a positive impact for both men and women.	ILO (2012) Kaltenborn (2017)
Target 1.4	By 2030, ensure that all men and women, in particular the poor and the vulnerable, have equal rights to economic resources, as well as access to basic services, ownership and control over land	Land and resource tenure, including inheritance, access to and use of technology, financial services, and access to microfinance are important for meeting SDG targets such as the one in question. For example, when it comes to land ownership, because of unequal rights to land, there ought to	Katila et al. (2020) Hansen et al. (2020)

	and other forms of property, inheritance, natural resources, appropriate new technology and financial services, including microfinance	be explicit protection of land rights that are recognised by both customary and statutory law for women, which can allow them to fairly participate in land ownership, management and administration. Furthermore, microfinance gives women the opportunity to empower themselves and improve their lives. This has the ability to remove them from poverty.	
Target 1.5	By 2030, build the resilience of the poor and those in vulnerable situations and reduce their exposure and vulnerability to climate-related extreme events and other economic, social and environmental shocks and disasters	Technical, policy, capacity enhancement and finance that are among the elements of transformative approaches in agriculture and food security will be necessary to tackle climate change. In particular, because women are most impacted by poverty and hunger, climate change has a direct impact on them in this respect.	Campbell et al. (2018)
Target 1.b SDG 2 End hu	Create sound policy frameworks at the national, regional and international levels, based on pro- poor and gender-sensitive development strategies, to support accelerated investment in poverty eradication actions inger, achieve food security and improved nutrition	The participation of women in policy making and the holding of high office such as parliamentary seats has a positive impact on women's well- being, including girls' education, fertility, child and infant mortality, and early marriage	Konte (2020)
Target 2.1	By 2030, end hunger and ensure access by all people, in particular the poor and people in vulnerable situations, including infants, to safe, nutritious and sufficient food all year-round.	It is well understood that due to gender inequalities, women and the more vulnerable in society will be more affected by hunger. Due to evident societal gender inequalities resulting from social, political and economic marginalisation of women, such inequalities have an impact on food security and resulting hunger, not only for women but for infants as well. With hunger come issues around health and the inability, as well as a lack of capability, to be able to effectively contribute to sustainable development in society	Larson and Larson (2019)
Target 2.2	By 2030, end all forms of malnutrition, including achieving, by 2025, the internationally agreed targets on stunting and wasting in children under 5 years of age, and address the nutritional needs of adolescent girls, pregnant and lactating women and older persons	Women are more likely to be overweight and obese. In addition, one in three women of reproductive age suffer from anaemia. This has implications for children as well as pregnant and lactating women, due to the lack of nutrients that such ailments have. As such, it is imperative that nutrition takes into consideration healthy and sustainable dietary needs when looking to end all forms of malnutrition. Healthy, nutritional and sustainable diets will need to be considered into ethically thought-out national policies that cover issues around taxes, incentives, nudges, and subsidies, especially for those most affected.	Fanzo (2019) Ghosh-Jerath et al. (2020)

Target 2.3 Target 2.c	By 2030, double the agricultural productivity and incomes of small-scale food producers, in particular women, indigenous peoples, family farmers, pastoralists and fishers, including through secure and equal access to land, other productive resources and inputs, knowledge, financial services, markets and opportunities for value addition and non-farm employment Adopt measures to ensure the proper functioning of food commodity markets and their derivatives	Furthermore, encouraging indigenous food systems in traditional communities can play a role in sustainable nutritional diets that can help end malnutrition. When it comes to the developing of national policies around and/or encouraging good dietary consumption of indigenous food systems, mothers, who are usually responsible for family food and meals, should be a targeted priority. Equal participation of all relevant stakeholders, which should include women in agricultural productivity, will result in increased productivity. In particular, development policies should not only favour urban industrial areas but should be inclusive of agricultural and rural settings where there might be more women inhabiting such spaces in order to increase agricultural productivity for sustainable development that could help end hunger.	Rodríguez-Pose and Hardy (2015) Fontefrancesco (2019)
	of food commodity markets and their derivatives and facilitate timely access to market information, including on food reserves, in order to help limit extreme food price volatility	to recognising the need for information on food commodity markets and the fact that most small, medium-sized and family farms are run by women, it might prove a challenge to have proper functioning food commodity markets. Therefore, measures and policies that consider necessary policy measures that address and acknowledge this phenomenon will go a long way in achieving the target in question.	
SDG 3 Ensure	healthy lives and promote well-being for all at all a	ages	
Target 3.1	By 2030, reduce the global maternal mortality ratio to less than 70 per 100,000 live births	Equalising the social and economic status of women would improve the women's health outcomes	Abebe (2016)
Target 3.2	By 2030, end preventable deaths of new-borns and children under 5 years of age, with all countries aiming to reduce neonatal mortality to at least as low as 12 per 1,000 live births and under-5 mortality to at least as low as 25 per 1,000 live births	Giving indistinguishable preference to all genders will ensure child survival with the reduction in excess male over female child mortality	Iqbal et al. (2018)
Target 3.7	By 2030, ensure universal access to sexual and reproductive health-care services, including for family planning, information and education, and	Giving equal opportunity and easy accessibility to formal and quality education to females will impact by generating responsible attitudes towards health	Queen Mary University of London (2018)

	the integration of reproductive health into national		
	strategies and programmes.		
Target 3.8	Achieve universal health coverage, including	Inclusion of women in economic and decision-making activity will	Singh et al. (2015)
	financial risk protection, access to quality	increase the ability to access existing maternal and child health services	
	essential health-care services and access to safe,		
	effective, quality and affordable essential		
	medicines and vaccines for all.		
SDG 4 – Ensur	e access to inclusive and equitable quality education	on for boys and girls, from pre-primary through to secondary education, a	and promote lifelong
learning oppor	tunities for all		
Target 4.1	By 2030, ensure that all girls and boys complete	Education of girls is also important for the achievement of a sustainable	Herbert et al. (2020)
	free, equitable and quality primary and secondary	future.	Spiteri (2020)
	education leading to relevant and effective		Spiteri (2018)
	learning outcomes		UNESCO (2018a)
Target 4.2	By 2030, ensure that all girls and boys have	All children need good-quality early childhood education, which sets the	Stromquist (2020)
	access to quality early childhood development,	foundations for lifelong learning. It helps to improve their future outcomes	
	care and pre-primary education so that they are	and alleviate poverty.	
	ready for primary education		
Target 4.3	By 2030, ensure equal access for all women and	Lifelong education for women is important at every stage of life.	
	men to affordable and quality technical,		
	vocational and tertiary education, including		
	university		
Target 4.5	By 2030, eliminate gender disparities in education	Women should have equal access to education to benefit from future	
	and ensure equal access to all levels of education	employment prospects.	
	and vocational training for the vulnerable,		
	including persons with disabilities, indigenous		
	peoples and children in vulnerable situations		
Target 4.6	By 2030, ensure that all youth and a substantial	The education of women is important for the achievement of a sustainable	
	proportion of adults, both men and women,	society.	
	achieve literacy and numeracy		
Target 4.7	By 2030, ensure that all learners acquire the	Education, and education for sustainable development (ESD) in particular,	
	knowledge and skills needed to promote	help promote gender equality and human rights, thus encouraging children	
	sustainable development, including, among	and adults to become ecologically responsible citizens. Therefore, barriers	
	others, through education for sustainable	to access to education need to be overcome by governments to ensure	
	development and sustainable lifestyles, human	quality education for all and eliminate gender disparities.	

	rights, gender equality, promotion of a culture of peace and non-violence, global citizenship and the appreciation of cultural diversity and of		
Target 4.A	culture's contribution to sustainable developmentBuild and upgrade education facilities that arechild, disability and gender sensitive and providesafe, nonviolent, inclusive and effective learningenvironments for all	Girls and boys with disabilities have a right to good quality education that promotes sustainable behaviours in children.	
SDG 6 Ensure	e availability and sustainable management of water	and sanitation for all	
Target 6.1	By 2030, achieve universal and equitable access to safe and affordable drinking water for all.	Proportion of population using safely managed drinking water services	Freistein and Mahlert (2015)
Target 6.2	By 2030, achieve access to adequate and equitable sanitation and hygiene for all and end open defecation, paying special attention to the needs of women and girls and those in vulnerable situations.	Proportion of population using (a) safely managed sanitation services and (b) a hand-washing facility with soap and water	Nygård (2017)
Target 6.b	Support and strengthen the participation of local communities in improving water and sanitation management	Proportion of local administrative units with established and operational policies and procedures for participation of local communities in water and sanitation management	Klarin (2018)
SDG 7 Ensure	e access to affordable, reliable, sustainable and mod	ern energy for all	
Target 7.1	By 2030, ensure universal access to affordable, reliable and modern energy services.	Proportion of population with access to electricity	
Target 7.b	By 2030, expand infrastructure and upgrade technology for supplying modern and sustainable energy services for all in developing countries, in particular least developed countries, small island developing States and landlocked developing countries, in accordance with their respective programmes of support	International financial flows to developing countries in support of clean energy research and development and renewable energy production, including in hybrid systems. Installed renewable energy-generating capacity in developing countries (in watts per capita)	Bhandari and Shvindina (2019)
SDG 8 Promo	te sustained, inclusive and sustainable economic gr	owth, full and productive employment and decent work for all	

Target 8.3	Promote development-oriented policies that support productive activitiesaccess to financial services.	SDG 8 promotes "sustained, inclusive and sustainable economic growth, full and productive employment and decent work for all". Businesses play a key role for job creation and economic growth. Instituting non- discriminatory practices and embracing diversity and inclusion will also lead to greater access to skilled, productive talent.	Dugarova (2018) Pandey and Kumar (2019) Buhmann et al. (2019) Rai et al. (2019) Manandhar et al. (2018) Alarcón and Cole (2019)
Target 8.5	By 2030, achieve full and productive employment and decent work for all women and men, including for young people and persons with disabilities, and equal pay for work of equal value.	SDG 8 promotes 'sustained, inclusive and sustainable economic growth, full and productive employment and decent work for all'. Businesses play a key role for job creation and economic growth. Instituting non- discriminatory practices and embracing diversity and inclusion will also lead to greater access to skilled, productive talent.	
Target 8.8	Protect labour rightsand those in precarious employment	The main objective is to achieve an inclusive employment. Protection of labour rights by enhancing health and safety at work. Key themes: economic inclusion; non-discrimination	
SDG 9 Build	resilient infrastructure, promote inclusive and susta	inable industrialization and foster innovation	
Target 9.2.	Promote inclusive and sustainable industrialization and, by 2030, significantly raise industry's share of employment and gross domestic product, in line with national circumstances, and double its share in least developed countries.	Increasing investments in infrastructure has positively affected gender equality by reducing unpaid care and domestic work. Better infrastructure can improve health and education for women and children. Women's access to education and employment contributes to industrialisation. Investing in social infrastructure could increase employment both in emerging countries and in developed economies. Reducing gender gap and subject segregation in education (by increasing gender diversity in STEM) can boost innovations.	Sinha et al. (2020)
SDG 10 Redu	ce inequality within and among countries		
Target 10.2	By 2030, empower and promote the social, economic and political inclusion of all, irrespective of age, sex, disability, race, ethnicity, origin, religion or economic or other status	Improving the labour market conditions by promoting decent work and equal employment opportunities to all genders	Abebe (2016)

Target 10.3	Ensure equal opportunities and reduce inequalities of outcome, including by eliminating discriminatory laws, policies and practices and promoting appropriate legislation, policies and action in this regard	Promoting equal pay for equal work, irrespective of gender, will encourage the suppressive gender to join the workforce, enjoying equal status	Abebe (2016)
Target 10.4	Adopt policies, especially fiscal, wage and social protection policies, and progressively achieve greater equality	Reinforce labour laws, including minimum wages, prevention of sexual harassment of women at the workplace, decent work, freedom of association and collective bargaining	United Nations System Chief Executives Board for Coordination (2017)
Target 10.5	Improve the regulation and monitoring of global financial markets and institutions and strengthen the implementation of such regulations	Promoting transparency and accountability of private companies to reduce inequalities among different genders	SDG Compass (2021)
SDG 11 Make	cities and human settlements inclusive, safe, resilie	ent and sustainable	
Target 11.1	By 2030, ensure access for all to adequate, safe and affordable housing and basic services and upgrade slums	Proportion of urban population living in slums, informal settlements or inadequate housing	Lucci et al. (2015) Nzau and Trillo (2020) Nabutola (2004) Hoek-Smit et al. (2020) JLL (2016) Chattopadhyay et al. (2016) Kaur (2018) Teotia (2015) Nallathiga (2019) Arimah (2001) Santoro (2015) Gopalan and Venkataraman (2015) Woetzel et al. (2014)
Target 11.2	By 2030, provide access to safe, affordable, accessible and sustainable transport systems for all, improving road safety, notably by expanding public transport, with special attention to the needs of those in vulnerable situations, women, children, persons with disabilities and older persons	Proportion of population that has convenient access to public transport, by sex, age and persons with disabilities	Kett et al. (2020) European Institute for Gender Equality (2020) Leach (2015) UN Women (2018b) Liu Z. et al. (2020) ESCAP (2017)

Target 11.3	By 2030, enhance inclusive and sustainable urbanization and capacity for participatory, integrated and sustainable human settlement planning and management in all countries	Ratio of land consumption rate to population growth rate	Blei et al. (2018) Agyemang and Morrison (2017) United Nations Environment Programme (2017) European Environment Agency (2015) Desai (2020) Mansell et al. (2020) Abdulkadir et al. (2019) Nicolau et al. (2019)
Target 11.4	Strengthen efforts to protect and safeguard the world's cultural and natural heritage	Total expenditure (public and private) per capita spent on the preservation, protection and conservation of all cultural and natural heritage.	Wang et al. (2018a) Satterthwaite (2017) Smiciklas et al. (2017) Idowu et al. (2020) Sterling (2016) Seto et al. (2014) OECD (2018)
Target 11.5	By 2030, significantly reduce the number of deaths and the number of people affected and substantially decrease the direct economic losses relative to global gross domestic product caused by disasters, including water-related disasters, with a focus on protecting the poor and people in vulnerable situations	Number of deaths, missing persons and directly affected persons attributed to disasters per 100,000 population	International Council for Science (2011) Smas et al. (2013) Hoornweg and Pope (2017)
Target 11.6	By 2030, reduce the adverse per capita environmental impact of cities, including by paying special attention to air quality and municipal and other waste management	Proportion of urban solid waste regularly collected and with adequate final discharge out of total urban solid waste generated, by cities	United Nations (2018) Ros-Tonen et al. (2016) Keivani (2010) Overseas Development Institute (2018)
Target 11.7	By 2030, provide universal access to safe, inclusive and accessible, green and public spaces,	Provide universal access to safe, inclusive and accessible green and public spaces by 2030.	Damodaran et al. (2015) Devisscher et al. (2019)

			1
	in particular for women and children, older persons and persons with disabilities		Guedes Vidal et al. (2019) Santiago Pineda et al. (2017) DIAUD/ CBM (2016) Daniel (2016)
Target 11. 8	Support positive economic, social and environmental links between urban, peri-urban and rural areas by strengthening national and regional development planning	Proportion of population living in cities that implement urban and regional development plans integrating population projections and resource needs, by size of city.	Grogan (2020) Clos (2016) Cohen (2006)
Target 11.9	By 2020, substantially increase the number of cities and human settlements adopting and implementing integrated policies and plans towards inclusion, resource efficiency, mitigation and adaptation to climate change, resilience to disasters, and develop and implement, in line with the Sendai Framework for Disaster Risk Reduction 2015-2030, holistic disaster risk management at all levels	Substantially increase the number of cities and human settlements adopting and implementing holistic disaster risk management at all levels	United Nations Statistics Division (2016)
Target 11.10	Support least developed countries, including through financial and technical assistance, in building sustainable and resilient buildings utilizing local materials	Proportion of financial support to the least developed countries that is allocated to the construction and retrofitting of sustainable, resilient and resource-efficient buildings utilising local materials	The Economist Intelligence Unit (2019) Norichika and Biermann (2017) Servaes (2017)
SDG 12 Ensur	e sustainable consumption and production pattern	S	
Target 12.1	Implement the 10-year framework of programmes on sustainable consumption and production, all countries taking action, with developed countries taking the lead, taking into account the development and capabilities of developing countries	Gender issue is underestimated when it comes to sustainable production and consumption. In general SDG 12 has been considered 'gender blind'.	Herbert et al. (2020) UN Women (2020f) Pudaruth et al. (2015) Cho et al. (2018)
Target 12.2	By 2030, achieve the sustainable management and efficient use of natural resources	Women are under-represented in decision-making related to sustainable management and efficient use of natural resources, so it is important to	

		increase their participation in leadership and decision–making processes, for the sustainable management of production and consumption patterns	
Target 12.3	By 2030, halve per capita global food waste at the retail and consumer levels and reduce food losses along production and supply chains, including post-harvest losses	Women have an important role in achieving this goal, however the recognition and actions to address this are rather limited still.	
Target 12.4	By 2020, achieve the environmentally sound management of chemicals and all wastes throughout their life cycle, in accordance with agreed international frameworks, and significantly reduce their release to air, water and soil in order to minimize their adverse impacts on human health and the environment	Women are under-represented in decision-making related to management of chemicals in the life cycle.	
Target 12.5	By 2030, substantially reduce waste generation through prevention, reduction, recycling and reuse	Women are known to be more in contact with packaging, and their role in reduction, recycling and reuse of plastic waste in specific should be considered crucial.	
Target 12.6	Encourage companies, especially large and transnational companies, to adopt sustainable practices and to integrate sustainability information into their reporting cycle	Women are under-represented in leadership of large and transnational companies due to difficulties found in work-life imbalance in many countries.	
Target 12.8	By 2030, ensure that people everywhere have the relevant information and awareness for sustainable development and lifestyles in harmony with nature	Women in general are more interested in increasing awareness for sustainable development and lifestyles in harmony in nature, and therefore their role in achieving this target is crucial.	
SDG 13 Take u	rgent action to combat climate change and its imp	pacts	
Target 13.1	By 2030, strengthen resilience and adaptive capacity to climate-related hazards and natural disasters worldwide	Climate change is experienced differently by women and men because of the interplay between gender, poverty and participation in the political arena. When taking action, the interplay between gender relations and the vulnerability of women needs to be taken into consideration. An understanding of gender relations is important for the transformation in social behaviour with climate change.	Herbert et al. (2020) Pearse (2016) Equal Measures 2030 (2021)
Target 13.2	By 2030, integrate climate change measures into national policies, strategies and planning	Since women are underrepresented in government agencies and policy- making, they are negatively affected by this. This also limits their agency in climate advocacy and mitigation measures.	

Target 13.3	By 2030, improve education, awareness-raising and human and institutional capacity on climate change mitigation, adaptation, impact reduction and early warning	Gender inequalities are evident, in that women, especially those living in the global South, are more prone to resource scarcity, poverty and risks, such as environmental stress, posed by climate change. Also, socio- cultural and economic circumstances negatively affect women's economic status (poverty) and education, and women become more susceptible to climate risks.	
SDG 14 Conse	rve and sustainably use the oceans, seas and marin	e resources for sustainable development	
Target 14.1	By 2025, prevent and significantly reduce marine pollution of all kinds, in particular from land-based activities, including marine debris and nutrient pollution	Women empowerment and education related to marine pollution in sea- based livelihoods is important so that they can lead sustainable development initiatives for pollution prevention	UN Women (2018a) IISD (2017) UN Sustainable Development Goals
Target 14.2	By 2020, sustainably manage and protect marine and coastal ecosystems to avoid significant adverse impacts, including by strengthening their resilience, and take action for their restoration in order to achieve healthy and productive oceans	Women are under-represented in policy and management in the fishing industry, so it is important to increase their participation in leadership and decision-making processes for the sustainable management of coastal ecosystems	(2020)
Target 14.4	By 2020, effectively regulate harvesting and end overfishing, illegal, unreported and unregulated fishing and destructive fishing practices and implement science-based management plans, in order to restore fish stocks in the shortest time feasible, at least to levels that can produce maximum sustainable yield as determined by their biological characteristics	Women are mostly affected by unsustainable fishing because the majority of them deal with fishing activities near shores, not having the resources for sophisticated equipment. Usually they possess only simple instruments for fishing.	
Target 14.5	By 2020, conserve at least 10 per cent of coastal and marine areas, consistent with national and international law and based on the best available scientific information	Equal participation and co-management of resources is important in order for policies of conservation to be accepted by coastal female communities. Also, equal access to knowledge production and information related to coastal conservation.	
Target 14.7	By 2030, increase the economic benefits to Small Island developing States and least developed countries from the sustainable use of marine resources, including through sustainable management of fisheries, aquaculture and tourism	Providing women in small islands developing states and least developed countries with equal opportunities in training and new skills acquisition in the fisheries sector can ensure a more equal access to resources and improve their economic conditions, thus contributing to the GDP increase by sustainable fisheries.	

Targat 14 a Increase scientific knowledge develop research Women are under represented in marine science, so it is important to the science of the science o	portant to
anget 14.a Intrease selentine knowledge, develop research wonten ale under tepresented in mante a safe environment for	r them so
capacity and transfer manne technology, taking provide budget and support and to create a safe environment for	
into account the intergovernmental Oceanographic that their participation in science and the knowledge exchange in	
Commission Criteria and Guidelines on the of marine technology and ocean-related industries can be increase	d.
Transfer of Marine Technology, in order to	
improve ocean health and to enhance the	
contribution of marine biodiversity to the	
development of developing countries, in particular	
small island developing States and least developed	
countries	
Target 14.b Provide access for small-scale artisanal fishers to Promotion of female fishing leaders and protecting their rights is	important
marine resources and markets to increase their access to small-scale fisheries and marine	resources
markets, because often they are underpaid or without contracts or l	health and
safety insurance, even though, according to the Food and A	griculture
Organization, 'women represent nearly half of the estimated 18	30 million
people worldwide working in fisheries and aquaculture'.	
Target 14.c Enhance the conservation and sustainable use of Increased female representation in international advocacy can su	apport the
oceans and their resources by implementing processes of ratification by countries of institutional framework	s, ocean-
international law as reflected in UNCLOS, which related instruments, etc., for conservation and sustainable use of t	he oceans
provides the legal framework for the conservation and their resources.	
and sustainable use of oceans and their resources.	
as recalled in paragraph 158 of The Future We	
Want	
SDG 15 Protect, restore and promote sustainable use of terrestrial ecosystems, sustainably manage forests, combat desertificatio	on. and halt and reverse land
degradation and halt biodiversity loss	,
Target 15.1 By 2020, ensure the conservation, restoration and Gender differentiated responsibilities vary region to region, but in	many Agu and Gore (2020)
sustainable use of terrestrial and inland freshwater communities around the world, women act as primary caretakers	and Broeckhoven (2014)
ecosystems and their services, in particular natural resource managers – procuring water and firewood, managers	ging
forests, wetlands, mountains and drylands, in line waste and providing health care, often through plant-based medic	vines.
with obligations under international agreements Women's roles also mean they hold vast knowledge on sources of	f water,
storing and caring for seeds, and the diverse uses and benefits of t	plants.
including for food, medicine, art. and avoiding and mitigating lan	d
degradation. This knowledge is crucial to preserving biodiversity.	

Target 15.3	By 2030, combat desertification, restore degraded	The role of women in international law and policy is still low, especially in	
	land and soil, including land affected by	developing countries, and the decision-making about combating	
	desertification, drought and floods, and strive to	desertification and restoration of land and soil is limited.	
	achieve a land degradation-neutral world		
Target 15.5	Take urgent and significant action to reduce the	Especially in developing countries, the role of women in the reduction of	
	degradation of natural habitats, halt the loss of	natural habitat destruction and biodiversity loss has been considered	
	biodiversity and, by 2020, protect and prevent the	important. Women need to be equally and actively involved in processes to	
	extinction of threatened species	conserve and sustainably use biodiversity because they play critical roles	
		as primary land managers and resource users, and they face	
		disproportionate impacts both from biodiversity loss and gender-blind	
		conservation measures.	
Target 15.8	Take urgent action to end poaching and trafficking	Efforts to overcome challenges to combatting wildlife trafficking will likely	
	of protected species of flora and fauna and address	benefit from increased attention to, and mainstreaming of, the role of	
	both demand and supply of illegal wildlife	women in wildlife trafficking, especially in Africa. The role of women has	
	products	been considered low in Asia.	
SDG 16 Promo	ote peaceful and inclusive societies for sustainable o	development, provide access to justice for all and build effective, accounta	ble and inclusive
institutions at a	all levels	1	1
Target 16.1	Significantly reduce all forms of violence and	Significantly reduce all forms of violence and related death rates across all	Leal Filho et al. (2021)
	related death rates everywhere	countries by 2030	Institute for Economics
			and Peace (2014)
			Pathfinders for Peaceful,
			Just and Inclusive
			Societies (2017)
			UNICEF Office of
			Research (2017)
			Bolaji-Adio (2015)
			Shakti (2017)
Target 16.2	End abuse, exploitation, trafficking and all forms	Proportion of children aged 1–17 years who experienced any physical	Council of Europe
	of violence against and torture of children	punishment and/or psychological aggression by caregivers in the past	(2017)
		month.	Hyder and Malik (2007)
			Raman et al. (2020)
			UNICEF (2009)
			Peterman and O'Donnell
			(2020)

			WHO (2020) Save The Children International Asia (2020) Fabbri et al. (2020)
Target 16.3	Promote the rule of law at the national and international levels and ensure equal access to justice for all	Proportion of victims of violence in the previous 12 months who reported their victimisation to competent authorities	Hillis et al. (2016) Rabinovych (2020) Beqiraj and McNamara (2016) Quinn and Sannerholm (2019) Government of Canada (2020) Lima and Gomez (2019) Alffram (2011) Open Government Partnership (2019) Satterthwaite and Dhital (2019) UNICEF and Regional Office for CEE/CIS (2015) OECD (2019) Manuel and Manuel (2018)
Target 16.4	By 2030, significantly reduce illicit financial and arms flows, strengthen the recovery and return of stolen assets and combat all forms of organized crime	Total value of inward and outward illicit financial flows	Arib (2017) Reuter (2017) Bromley et al. (2019) UNCTAD and UNODC (2021) Fisher (2020)
Target 16.5	Substantially reduce corruption and bribery in all their forms	Substantially reduce corruption and bribery in all their forms across all countries by 2030.	Hoffiani (2019) Whaites (2016) Hope (2020)

			Mugellini and Villeneuve (2019) V-Dem Institute (2017) Department for International Development (2015) Bahoo et al. (2020) Mackey et al. (2016) Sartor and Beamish (2020)
Target 16.6	Develop effective, accountable and transparent institutions at all levels	Primary government expenditures as a proportion of original approved budget, by sector (or by budget codes or similar)	United Nations (2015d) Tully (2015) Poisson (2016) OECD (2019b) V-Dem Institute (2015) Carothers and Brechenmacher (2014) Blind (2019) International Bank for Reconstruction and Development and World Bank (2020)
Target 16.7	Ensure responsive, inclusive, participatory and representative decision-making at all levels	Proportions of positions (by sex, age, persons with disabilities and population groups) in public institutions (national and Local legislatures, public service, and judiciary) compared to national distributions	Hayes and Bulat (2017) WHO (2011) European Union Agency for Fundamental Rights (2020) Mijatović (2018)
Target 16.8	Broaden and strengthen the participation of developing countries in the institutions of global governance	Proportion of members and voting rights of developing countries in international organisations	Glass and Newig (2019) Qoraboyev (2021) Weiss and Wilkinso (2018) Gellers (2016) OECD (2019a)

	Manby (2017) Vandenabeele and Lao (2007) Dahan and Gelb (2015) Perrault and Arellano (2011)
Target 16.10 Ensure public access to information and protect fundamental freedoms, in accordance with national legislation and international agreements Number of verified cases of killing, kidnapping, enforced disappear arbitrary detention and torture of journalists, associated media personation and protect trade unionists and human rights advocates in the previous 12 monted in the previous 12 mo	rance, Slutskiy (2020) onnel, Berger (2020) ths UNESCO (2018b) Phogat (2015) United Nations (2008)
Target 16.11 Strengthen relevant national institutions, including through international cooperation, for building capacity at all levels, in particular in developing countries, to prevent violence and combat terrorism and crime Existence of independent national human rights institutions in comparison with the Paris Principles	pliance Manuel and Manuel (2018) Hilderbrand (2015) Zamfir (2020)
Target 16.12 Promote and enforce non-discriminatory laws and policies for sustainable development Proportion of population reporting to having personally felt discrimagiants or harassed in the previous 12 months on the basis of a grout discrimination prohibited under international human rights law Target 16.12 Promote and enforce non-discriminatory laws and policies for sustainable development Proportion of population reporting to having personally felt discrimagiants or harassed in the previous 12 months on the basis of a grout discrimination prohibited under international human rights law	inated Pisano et al. (2015) ind of British Council (2018) IOM and Joint Migration and Development Initiative (2015) Government of Andhra Pradesh (2017) United Nations Human Rights (2020) Lindsey and Chapman (2017) United Nations System Chief Executives Board for Coordination (2017)
SDG 17 Strengthen the means of implementation and revitalize the global partnership for sustainable development	

Target 17.1	Strengthen domestic resource mobilization, including through international support to developing countries, to improve domestic capacity for tax and other revenue collection	Creating equal working conditions and opportunities for women especially in the Global South, where the majority of women do not work formally, and thus do not contribute to the income taxes.	Joshi et al. (2020) UN Women (2020g) OECD (2020b)
Target 17.3	Mobilize additional financial resources for developing countries from multiple sources	It is important that foreign aid considers gender equity and provides funds for empowering woman. According to UN Women, 'Only 5 per cent of foreign aid funds had gender equality as a principle objective in 2012- 2013'.	
Target 17.7	Promote the development, transfer, dissemination and diffusion of environmentally sound technologies to developing countries on favourable terms, including on concessional and preferential terms, as mutually agreed	Clean technologies related to everyday work and activities of women (i.e., clean cooking technologies) can affect their health and improve their life conditions.	
Target 17.8	Fully operationalize the technology bank and science, technology and innovation capacity- building mechanism for least developed countries by 2017 and enhance the use of enabling technology, in particular information and communications technology	Women are under-represented in IT jobs, top management and academic careers, but it is important that woman are included in digital transformation and that there is no digital gender divide. In many countries, women have less access to technological devices. According to the OECD, approx. 327 million fewer women than men in the world can use a smartphone or can have access to mobile internet.	
Target 17.10	Promote a universal, rules-based, open, non- discriminatory and equitable multilateral trading system under the World Trade Organization, including through the conclusion of negotiations under its Doha Development Agenda	Women can be empowered and protected by fair trade practices, which consider labour conditions, prevent discrimination, promote equal access to employment, etc.	
Target 17.12	Realize timely implementation of duty-free and quota-free market access on a lasting basis for all least developed countries, consistent with World Trade Organization decisions, including by ensuring that preferential rules of origin applicable to imports from least developed countries are transparent and simple, and contribute to facilitating market access		

Target 17.17	Encourage and promote effective public, public-	It is important that female organised networks or civil society organisations
	private and civil society partnerships, building on	that deal with gender equality, women rights, etc. are part of the partnerships
	the experience and resourcing strategies of	and coalitions for goals, in order to increase their representation in national
	partnerships	and international processes
Target 17.19	By 2030, build on existing initiatives to develop	Many countries do not have enough data that identify women's issues or
	measurements of progress on sustainable	gender-based data, which is a critical aspect for supporting policies for
	development that complement gross domestic	gender equality. According to UN Women, around a third of countries have
	product, and support statistical capacity-building	an office for gender statistics.
	in developing countries	

Appendix V

The keyword search for the bibliometric analysis was as follows:

TITLE-ABS-KEY ("anxiet*" OR "schizophrenia" OR "mood disorder*" OR "depression" OR "suicide" OR "aggressive behavio*rs" OR "despair" OR "mental health") AND ("extreme weather event" OR "loss of landscape" OR "climate change*" OR "global warming" OR "climatic change*").

Acknowledgments

A journey of a thousand miles begins with a single step. Lao Tzu

Approaching the final step of this journey and remembering all the miles, endeavours, aspirations and challenges, I would like to convey my sincere appreciation and admiration for all the encounters and the role they have played along the way.

First and foremost, I would like to express my deepest and profound gratitude to my esteemed supervisors Prof. Dr. Walter Leal and Prof. Dr. Christian Borgemeister for their guidance throughout the course of my PhD work.

I am particularly grateful to Prof. Dr. Walter Leal for the privilege of joining the diverse cohort of his PhD students and being 'spoilt' by the ultimate working and research environment, a myriad of opportunities, constant action, ambitions and immersion in the multifaceted realm of academia.

I am equally grateful to Prof. Dr. Christian Borgemeister for his unique way and commitment of supervision, collaboration, providing feedback and posing questions as well as insightful discussions and the opportunity to join the interdisciplinary doctoral program at the Center for Development Research (ZEF) and the ZEF Gender Group.

I would like to express my appreciation to both Prof. Dr. Walter Leal and Prof. Dr. Christian Borgemeister for their immense support, flexibility, positivity and patience, and especially, for their exceptional talent as supervisors, complementing rather than contradicting each other in mentoring and guidance.

My sincere thanks go to the DAAD (German Academic Exchange Service) for giving me the opportunity to become a DAAD scholarship recipient under the Research Grants - Doctoral Programmes in Germany, to conduct my research, and join this multinational community of scholars and alumni. I would also like to thank the DAAD coordinators and persons in charge for their support, recommendations and advice throughout the scholarship period.

I extend my thanks to all those experts from different geographical regions who expressed their interest and generously shared their thoughts and opinions, and dedicated their time to participate in the surveys conducted during this research work. I would also like to thank all the co-authors for their efforts, motivation, contributions, great collaboration, tremendous experience, networking and knowledge sharing.

I would like to acknowledge the extraordinary team at the Research and Transfer Centre "Sustainable Development and Climate Change Management" (FTZ-NK) at the Hamburg University of Applied Sciences, which I have had the pleasure to join and work with.

I am deeply thankful to Julia Komagaeva, Sr. Country Operations Officer at the World Bank Group for her support and opportunities to collaborate with her.

I would like to express my special gratitude and appreciation to Gulnara Nurieva, Assistant Professor at the American University of Central Asia, Kyrgyzstan for her time, advice, rationality, comments and recommendations, and professional expertise.

My heartfelt gratefulness goes to the people in my life for their tremendous and unwavering understanding, dedication, respect and for giving me the freedom to follow my way throughout this journey of a thousand miles.